Development of a Detection Model for Curve Progression in Adolescents with Idiopathic Scoliosis

by

Mahdieh Khodaei Jalalabadi

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

Medical Sciences - Radiology and Diagnostic Imaging

University of Alberta

©Mahdieh Khodaei Jalalabadi, 2022

Abstract

Adolescent idiopathic scoliosis (AIS) is a three-dimensional (3D) structural spinal disorder recognized by lateral curvatures. The routine for monitoring of AIS is taking an X-ray every six months to check the curve severity and curve progression. Taking repetitive radiographs is not desirable for children since it can increase the risk of cancer. Based on the literature, demographic and radiographic parameters may help predict curve progression. However, there is no accurate prediction model available yet. The objectives of this thesis were to develop and validate a detection model by combining curve characteristics, reflection coefficient (RC) from ultrasound (US) signals, and clinical information to accurately detect the progressive cases thus avoiding unnecessary radiographs.

The specific objectives were

- a) To identify potential clinical predictors based on prediction models from the literature
- b) To determine if the US reflection coefficient correlates with curve severity and curve progression
- c) To determine whether other US parameters obtained from a single US scan associated with curve progression
- d) To develop a predictive model to detect the progressive curves of AIS
- e) To validate the model based on clinical data

To identify prognostic factors of curve progression, a systematic review was conducted. The results showed limited or conflicting evidence for most parameters, requiring further investigation. Among the most frequently found potential parameters, age, body mass index (BMI), menarche status, frontal X-ray Cobb, number of curve (NOC), and Risser sign were selected for model development. As new parameters, I suggested the US Cobb change, the US Cobb angle in plane of maximum curvature (PMC), the kyphotic angle (KA), and axial vertebral rotation (AVR) measured on US images. Also, the US RC, which can provide the bone quality information, was investigated.

Experimental and clinical studies were conducted to ensure the validity and reliability of this index. The results showed the larger the RC index value, the stiffer the bone. Also, a clinical study demonstrated that this index could be measured reliably and 68% of children with AIS without progression had a larger RC value than the progression group.

The kyphotic angle was also considered as a potential predictor in the literature. A new method was developed and a pilot study to investigate the reliability of the KA measurement on US images was conducted. The results showed KA could be reliably measured. The maximum difference of the KA measurements between the US images and radiographs was 4°. When a larger clinical study was conducted, it further confirmed the reliability of the KA measurements. The factor which might affect the accuracy of the KA measurements was the posture of the participants during data acquisitions.

Similarly, AVR was also reported as one of the potential predictors and was studied. Before using the AVR as a parameter in model development, the reliability of the AVR measurements was studied. A clinical study was conducted, and the results showed a poor to moderate reliability, ICC \geq 0.49, with the maximum average difference between X-ray and US being 4.6°. The factors that showed a significant influence on the differences of the measurements were AVR measurements at the apical region and larger AVR severity. Prior to conduct a large clinical study, a pilot study on development of detection model was conducted. In this study age, menarche status, X-ray Cobb angle, RC index, and US Cobb change were used. Of these parameters, only RC and US Cobb change were retained as predictors of curve progression. To develop the final model, a large clinical study including 162 girls was conducted. Among those, 100 participants including 25 progression cases were used for model development. The selected parameters included demographic information: age, BMI, and menarche status; radiographic parameters: X-ray Cobb angle, NOC, and Risser sign; and the US parameters: US Cobb change, RC index, KA, maximum AVR, and PMC Cobb angle. The risk of progression equation was Log (p/1-p) = -1.40+0.28 (US Cobb change)-39.45(RC)+1.36 (NOC). The model was then validated on 62 cases with 11 progressions. The sensitivity, specificity, and accuracy results were over 90%.

In conclusion, the developed model showed promising results and the accuracy was better than other studies in the literature. Further study on a larger sample size is needed to offer a stronger validation.

Preface

The research described in this thesis had received ethics approval from the Health Research Ethics Board from the University of Alberta. The project title is "Using Ultrasound to Assess Spinal Deformity for Adolescent Idiopathic Scoliosis", reference number: Pro00005707.

Some of the research in this thesis is formed from collaborative work with various coauthors.

Chapter 3 is based on a submitted paper:

 Khodaei M, Parent E, Hill D, Wong J, Chan A, Gergelé E, Ng K, Leung C, Le L, Lou E. Identifying Prognostic Factors of Curve Progression in Adolescents with Idiopathic Scoliosis: A Systematic Review. Submitted to *PLOS one*, 2021.

I conceived of the direction, performed screenings, extractions, quality appraisals, and compiled the data into the journal article. I collaborated with many reviewers to perform these tasks. Screening co-reviewers included: Dr. Eric Parent, Doug Hill, Dr. Andrew Chan, and Jason Wong. Extraction and quality appraisal co-reviewers were Eloi Gergelé, Jason Wong, Kenwick Ng, Carmen Leung, and Doug Hill. I also consulted with Liz Dennett, a medical librarian for constructing the search. Direction, guidance, and review were provided by Drs. Parent, Lou and Le.

Chapter 4 is based on one submitted and one published paper:

- Pham T, Le L, Khodaei M, Zheng R, Lou E. Investigation of Ultrasonic Soft Tissue-Bone Reflection Coefficients Correlating with Curve Severity in Children with Adolescent Idiopathic Scoliosis. Submitted to *Journal of Engineering in Medicine*, 2021.
- Khodaei M, Sayed T, Hill D, Parent E, Moreau M, Stampe K, Southon S, Le L, Lou E. Reliability of measurements of a reflection coefficient index to indicate spinal bone strength on adolescents with idiopathic scoliosis (AIS): a pilot study.

Eur Spine J. 2021 Jul;30(7):1888-1895. doi: 10.1007/s00586-021-06871-8. Epub 2021 May 25

For the first paper, I collected the clinical data, measured the data on US images, and helped for experimental set-up. Tu Pham was the first author and she performed experiments, collected the experimental data, analyzed the data, and prepared the manuscript. Dr. Zheng provided the fundamental theories for conducting the study. Advice, guidance, and review of the paper were provided by Drs. Le and Lou.

For the second paper, I collected the clinical data, measured the data on US images, analyzed the results and prepared the manuscript. Tehzeeb Sayed helped on data measurement and analysis. There was also consultation with the Edmonton scoliosis research group for clinical advice and guidance: Doug Hill, Dr. Moreau, Dr. Stampe, Sarah Southon. Direction, guidance, and review were provided by Drs. Lou, Parent, and Le.

Chapter 5 is based on two submitted papers:

- Sayed T, Khodaei M, Hill D, Lou E. Intra- and Inter-rater Reliabilities and Differences of Kyphotic Angle Measurements on Ultrasound Images versus Radiographs for Children with Adolescent Idiopathic Scoliosis – A Preliminary Study. Submitted to Spine Deformity, 2021.
- Khodaei M, Parent E, Le L, Lou E. Factors Influencing Kyphotic Angle Measurements on Ultrasound Spinal Images of Children with Adolescent Idiopathic Scoliosis (AIS). Submitted to *European Spine Journal*, 2021.

For the first paper, I collected clinical data, helped on development of the measurement method, measured the US images, and helped on the data analysis. Tehzeeb Sayed helped on data measurement, analyzed the data and prepared the manuscript for the publication. Dr. Lou also helped on data measurement. Direction, guidance, and review were provided by Dr. Lou and Doug Hill.

For the second paper, I collected clinical data measured the US images, performed data analysis, and prepared the manuscript. Dr. Lou also helped on data measurement. Direction, guidance, and review were provided by Drs. Lou, Parent, and Le.

Chapter 6 is based on one submitted paper:

 Khodaei M, Parent E, Le L, Lou E. Investigation of Factors Influencing the Reliability and Measurement Variation of Ultrasound Axial Vertebral Rotation on Adolescents with Idiopathic Scoliosis. Submitted to *The Spine Journal*, 2021.

I collected clinical data measured the US images, performed data analysis, and prepared the manuscript. Dr. Lou also helped on data measurement. Direction, guidance, and review were provided by Drs. Lou, Parent, and Le.

Chapter 7 is based on two submitted papers:

- Khodaei M, Parent E, Le L, Southon S, Hill D, Stampe K, Huang E, Lou E. Development and Validation of a Model to Predict X-ray Progression at a Followup Visit Based on Ultrasound and Clinical Parameters for Adolescent Idiopathic Scoliosis (AIS). Submitted to *Spine Deformity*, 2021.
- Khodaei M, Parent E, Le L, Southon S, Hill D, Stampe K, Huang E, Lou E. Identifying Prognostic Parameters to Develop a Detection Model of Curve Progression with Clinical Validation on Adolescents with Idiopathic Scoliosis (AIS). Submitted *European Spine Journal*, 2021.

For the first study, I collected clinical data measured the US images, performed data analysis, and prepared the manuscript. There was also consultation with the Edmonton scoliosis research group team for clinical advice and guidance: Doug Hill, Dr. Moreau, Dr. Stampe, Sarah Southon. Direction, guidance, and review were provided by Drs. Lou, Parent, and Le.

For the second study, I collected clinical data measured the US images, performed data analysis, and prepared a draft for the publication. Tehzeeb Sayed and Veena Logithasan also helped on data measurements. There was also consultation with the Edmonton

scoliosis research group team for clinical advice and guidance: Doug Hill, Dr. Moreau, Dr. Stampe, Sarah Southon. Direction, guidance, and review were provided by Drs. Lou, Parent, and Le.

Acknowledgements

I would like to thank my supervisor, Dr. Lou for providing this wonderful opportunity to participate in this interesting field of research and trusting my abilities. Thank you, Dr. Lou for your patience and understanding throughout these five years of my PhD research. Thank you, Dr. Lou, for supporting me to travel, learn and mentor many students and learn how to be an independent researcher. I would also like to thank my co-supervisor, Dr. Le for his guidance, advice, and support throughout the research work. Thank you Dr. Le for pushing me forward, and helping me to grow my research abilities, communication skills and improve my confidence. I would not be here without your enthusiasm and support. I would like to acknowledge Dr. Parent for his patience, support and understanding. Thank you, Dr. Parent, for all your constructive comments, I have learned a lot from you which helped me to improve my research abilities.

I am grateful for the friendships with my wonderful lab mates Dr. Chan, Kenwick Ng, Jason Wong, Tu Pham, Duc Nguyen, Tehzeeb Sayed, Carmen Leung and others for being supportive and helping me throughout my research. I enjoyed every moment of working in the lab with you all. I would like to truly appreciate Tu Pham and Duc Nguyen for their significant contributions to my PhD work helping me on the experiments, software development, and my research. Tu Pham also provided a significant contribution to the work on the ultrasound reflection coefficient which has been reported in Chapter 4. I would like to thank you all along with Susanna Trac, Eloi Gergelé for your huge help on my systematic reviews. Thank you, Dr. Zheng, Tehzeeb Sayed, Amber Sayed, Ayushi Patel, Carmen Leung and Veena Logithasan for helping me in clinic and data measurements. I would also like to thank the researchers and staffs from the Glenrose Rehabilitation Hospital including Doug Hill and others for their support. As well, many thanks to the Edmonton Scoliosis Research Group for supporting the research work. I would like to acknowledge Kathleen Shearer who helped me a lot for data recruitment and clinic, without her help I would not be able to progress in my research. I acknowledge and appreciate the fundings support from the Scoliosis Research Society (SRS), Women and Children's

Health Research Institute (WCHRI), Edmonton Orthopedic Research Committee, David Lawson Graduate Studentship, Graduate Students Association (GSA) Academic Travel Award, and Medical Sciences Graduate Program Scholarship Competition without which my research would not be possible.

I would like to thank my family and my friends who have been a constant encouragement for all the challenges, the ups and downs throughout my research. Finally, and most importantly, I give thanks to God to giving me all the power to believe in myself, never give up and go on.

Table of Contents

Abstractii
Prefacev
Acknowledgementsix
Table of Contentsxi
List of Tablesxvii
List of Figures xix
List of Equationsxxiii
List of Abbreviations xxiv
Chapter 1: Introduction 1
1.1. Motivation
1.2. Objectives
1.3. Scope of Work
1.4. Thesis Organization
Chapter 2: Background Information 5
2.1. Overview
2.2. Spine Anatomy
2.2.1. Vertebral Column Structure
2.3. Scoliosis
2.3.1. Adolescent Idiopathic Scoliosis (AIS)
2.3.2. Progression of the Curvature
2.3.3. Treatment Options 11
2.3.4. Observation
2.3.5. Orthotic Treatment
2.3.6. Exercise Therapy 12
2.3.7. Surgical Treatment 12
2.4. X-ray Based Imaging Modalities Used for AIS 13
2.4.1. Basics of X-ray Based Imaging Modalities
2.4.2. Conventional X-ray Systems 14
2.4.3. EOS Imaging System (2D/3D)14

2.4.4. Computed Tomography (CT) Scan	17
2.5. Use of X-ray Systems to Image Curve Characteristics of AIS	17
2.5.1. Cobb Angle	17
2.5.2. Spinal Flexibility Assessment	18
2.5.3. Axial Vertebral Rotation (AVR)	19
2.5.4. Sagittal Plane Parameters	22
2.5.5. Kyphotic Angle (KA)	22
2.5.6. Lordotic Angle (LA)	23
2.5.7. The Negative Effects of Ionizing Radiation	24
2.6. Non-ionizing Radiation Imaging Modalities for AIS	25
2.6.1. Magnetic Resonance Imaging (MRI)	26
2.6.2. Ultrasound (US) Imaging Method	27
2.6.3. Ultrasound for Imaging of AIS	27
2.7. Bone Property Assessment	29
2.7.1. Dual-energy X-ray absorptiometry (DXA)	30
2.7.2. Quantitative Ultrasound (QUS)	31
2.8. Prediction of Curve Progression	32
2.8.1. Methods for Prediction of Curve Progression	32
2.9. Summary	33
Chapter 3: A Systematic Review on Prognostic Factors of Curve Progression in AIS	34
3.1. Overview	34
3.2. Introduction	34
3.3. Methods	36
3.3.1. Protocol and Registration	36
3.3.2. Inclusion Criteria	36
3.3.3. Exclusion Criteria	36
3.3.4. Information Sources and Search Strategy	36
3.3.5. Study Selection	37
3.3.6. Data Collection Process and Data Items	37
3.3.7. Risk of Bias of Studies	38

3.3.8. Summary Measures	. 38
3.3.9. Synthesis of Results	. 39
3.3.10. Additional Analyses	. 39
3.4. Results	. 39
3.4.1. RoB Within Studies	. 43
3.4.2. Study Characteristics	. 43
3.4.3. Sample Characteristics	. 44
3.4.4. Reviewer Agreement Analyses	. 45
3.4.5. Synthesis of the Results	. 45
3.5. Discussion	. 52
3.6. Conclusions	. 54
3.7. Summary	. 54
References	. 54
Appendices	. 64
Appendix 3.1 The Search Strategy for the Selected Databases	. 64
Appendix 3.2 List of Excluded Articles with the Reason of Exclusion	. 75
Appendix 3.3 The Quality Appraisal Tailored for each Domain of the QUIPS Tool	. 91
Appendix 3.4 Summary of the Parameters for the Articles with Progression Criteria	
Appendix 3.5 Summary of Articles for the Progression Criteria to a Magnitude $\geq 30^{\circ}$ Requiring Surgery	
Appendix 3.6. Summary of Articles that Reported Change in Cobb Angle as °/year	124
Chapter 4: Ultrasound Reflection Coefficient (RC) Index	126
4.1. Overview	126
4.2. The Fundamentals of the Reflection Coefficient (RC) Index	126
4.2.1. Introduction	126
4.2.2. Materials and Methods	128
4.2.3. Theoretical Formulation	128
4.2.4. Phantom Studies	131
4.2.5. In-vivo Pilot Study	133
4.2.6. Signal Processing	135

4.2.7. Results	135
4.2.8. Phantom Studies	135
4.2.9. In-vivo Study	139
4.2.10. Discussion	140
4.2.11. Conclusion	142
References	142
4.3. Test-retest and Reliability Analyses of the Reflection Coefficient (RC) Inc	dex 148
4.3.1. Introduction	148
4.3.2. Methods	149
4.3.3. Clinical Participants	149
4.3.4. Equipment	150
4.3.5. Data Collection	150
4.3.6. Data Measurements	151
4.3.7. Reflection Coefficient (RC)	151
4.3.8. Measurement of the Reflection Coefficient (RC)	151
4.3.9. Data Analysis	153
4.3.10. Results	153
4.3.11. Test-retest Reliability Analysis	153
4.3.12. Intra-rater Reliabilities	154
4.3.13. Inter-rater Reliabilities	156
4.3.14. Discussion	157
4.3.15. Conclusions	159
4.4. Summary	159
References	160
Chapter 5: Ultrasound Kyphotic Angle (KA)	163
5.1. Overview	163
5.2. Reliabilities and Differences of Kyphotic Angle Measurements on U Images versus Radiographs	
5.2.1. Introduction	163
5.2.2. Methods	165
5.2.3. Clinical Participants	165

5.2.4. Data Acquisition 165
5.2.5. Raters
5.2.6. Data Measurements 166
5.2.7. Statistical Analysis168
5.2.8. Results
5.2.9. Discussion
5.2.10. Conclusions
References174
5.3. Factors Influencing Kyphotic Angle Measurements on Ultrasound images 178
5.3.1. Introduction
5.3.2. Methodology
5.3.3. Clinical Participants
5.3.4. Data Acquisition
5.3.5. Data Measurement
5.3.6. Data Analysis
5.3.7. Results
5.3.8. Reliability and Agreement Analysis
5.3.9. Factors Influencing the Accuracy and Reliability
5.3.10. Discussion
5.3.11. Conclusions
5.4. Summary
References
Chapter 6: Reliability and Factors Influencing the Measurement Variation of Ultrasound Axial Vertebral Rotation (AVR)
6.1. Overview
6.2. Introduction
6.3. Methods
6.3.1. Clinical Participants
6.3.2. Data Acquisition
6.3.3. Raters
6.3.4. AVR Measurements

6.3.5. Data Analysis)()
6.3.6. Results)1
6.3.1. Factors Influencing the AVR Measurement Difference)3
6.4. Discussion)4
6.5. Conclusion)6
6.6. Summary)6
References)7
Chapter 7: Development of Detection Models for Curve Progression	1
7.1. Overview	1
7.2. A Preliminary Analysis on Development and Validation of a Model to Detect Progression	
7.3. Identifying Prognostic Parameters to Develop a Detection Model of Curv Progression with Clinical Validation	
7.3.1. Introduction	.2
7.3.2. Materials and Methods	.4
7.3.3. Clinical Participants	4
7.3.4. US Data Acquisition	4
7.3.5. Investigated Parameters & Data Measurements	5
7.3.6. Statistical Analysis	7
7.3.7. Results	8
7.3.8. Discussion	21
7.3.9. Conclusion	24
7.4. Summary	24
References	25
Chapter 8: Conclusions and Recommendations	29
8.1. Overview	29
8.2. Summary of Thesis Work	29
8.3. Major Contributions	51
8.4. Clinical Significance	52
8.5. Limitations and Future Recommendations	3
References	\$5

List of Tables

Table 3.1 Rules for determining of the level of evidence when formulating summarystatements about whether a variable was predictive of progression or not
Table 3.2 Included articles and RoB assessments 41
Table 3.3 Study characteristic and baseline information summary
Table 3.4 Levels of evidence summary statements about whether each predictive factor investigated does or does not predict curve progression $\geq 5^{\circ}$ over short-term/long-term follow-up intervals
Table 3.5 Levels of evidence summary statements about whether each predictive factor investigated does or does not predict curve progression when attempting to predict progression $>30^{\circ}$ or surgery
Table 3.6 Levels of evidence summary statements about whether each predictive factor investigated does or does not predict curve progression defined as change in Cobb angle (°/year) over short-term and long-term follow-up intervals
Table 4.1 Properties of the materials used in the study. 132
Table 4.2 The calculated and measured reflection coefficients of the interfaces involved in this study. ϵ is given by the absolute value of the difference (calculated - measured) × 100% divided by the calculated value
Table 4.3a The average RC values for scan 1 and 2 from rater 1
Table 4.3b The test-retest reliability of RC values between scan 1 and 2 from rater 1154
Table 4.3c The MAD of soft tissue thickness h (cm) measured by the MRC, ARC andCARC methods on the first 24 images from rater 1.154
Table 4.4a The average RC measurements on the first and second measurements (M1 andM2) on the same 58 images using MRC and ARC methods from rater 1 (R1) and rater 2(R2)
Table 4.4b The intra-rater reliability on RC measurements from R1 and R2 on the same 58images using the MRC and ARC methods.155
Table 4.4c The mean±SD and intra-rater reliability for RC measurements from R1 and R2on the same 58 images using the CARC method
Table 4.5a The inter-rater reliability for RC measurements from R1 and R2 on the same 58images using the MRC and ARC methods.156
Table 4.5b The inter-rater reliability for RC measurements on the same 58 images using the CARC method, the soft tissue thickness (h) measurement difference and the frame number selection differences between R1 and R2

Table 5.1 KA measurement means, standard deviations, and range on radiographs and US images $(n = 20)$
Table 5.2 The intra-rater reliability of the US KA measurements $(n = 20)$ 169
Table 5.3 The inter-rater reliability for KA measurements on both US and radiographs (n $= 20$)
Table 5.4 The inter-method comparison and reliability of the KA measurements from both raters (n=20)
Table 5.5 The average, standard deviation, SEM and intra-rater reliability of the KA measurements for both vertebral levels measurement methods on the X-ray and US images. 185
Table 5.6 The comparison of inter-method differences and reliabilities at different occasions (M1, M2) and for different levels measurement methods as T1/T12 and T4/T12.
Table 5.7 Test of association between the magnitude of the absolute measurement difference and the KA severity, the X-ray system and the curve type for the T1/T12 measurements using a) a 5° MAD threshold, and b) for a 7° MAD threshold
Table 5.8 The association between the magnitude of the absolute errors of measurements (5° and 7° thresholds) and the apex location of the scoliotic curves
Table 6.1 The average, standard deviation, SEM and intra-rater results of the AVR measurements on both X-ray and US images
Table 6.2 The MAD, SEM and inter-method reliability results of the AVR measurementsbetween X-ray and US methods202
Table 6.3 Comparison of the AVR measurement difference (MAD) in terms of region ofmeasurement, curve severity and curve type203
Table 6.4 Comparison between the average of the inter-method difference in AVR severity 204
Table 7.1 The mean±SD for age, BMI, X-ray Cobb and US parameters for the cases with and without progression and the statistical t-test values on the described parameters between the two groups
Table 7.2 The association between categorical parameters with curve progression 219
Table 7.3 The final predictors and the statistics of the developed model
Table 7.4 The validation results from the test data set
Table 7.5 Predictors values (US Cobb change, RC and NOC) as well as the predicted probability of progression for the 5 cases misclassified as progression cases

List of Figures

Figure 4.3 (a) The ultrasound scan of children with AIS in standing position. (b) A B-mode Figure 4.4: (a) Envelopes of the recorded echo and two reverberations within the BP. The amplitude of the 2nd reverberation is small and the zoomed signal is shown in the inset. (b) The linear regression line of the three data points. (c) Comparison between the measured and predicted reflection coefficients. Error bars denote the standard deviations (see Table 4.2). (d) The simulated amplitude ratio with change of soft-tissue thickness for three αs -Figure 4.5 The ultrasonographs (Left) and the corresponding RF data (Right) of the following target: (a) acrylic phantom with the SFS, (b) acrylic phantom with the RCIS, (c) Figure 4.7 Acquisition of ultrasound data from a participant with AIS in a standardized standing position with feet shoulder-width apart, hands holding the wooden frame, looking Figure 4.8a-e The 3 steps process to measure the reflection coefficient (RC) from a Figure 4.9 Distribution of the participants with and without progression based on the Figure 5.1 (a) An ultrasound image showing a coronal PA view of spine with T1-T3 and T10-T12 identified on the image, (b) an US transverse view of T1 with the laminas identified and joined using a line and the midpoint of the line as the spinous process, (c) US sagittal view of spine and the kyphotic angle measured 27 ° using the slope of a line joining spinous process of T1 and T2 at the top to T11 and T12 at the bottom, d) the sagittal view X-ray of the spine correspond to the same patient and the kyphotic angle measured Figure 5.2 Inter-method comparison between the radiographic measurements and US measurements from both raters (R1 and R2) and the average of both raters with R2 Figure 5.3 The Bland-Altman plot of mean and ± 1.96 standard deviations of the mean of radiographic and US measurements versus the differences between the radiographic and US measurements for (a) Rater 1 and (b) Rater 2.....171 Figure 5.4 The process of measuring KA on US images. a) The center of the laminae of T1 to T5 and of T10 to T12 are first identified on the coronal US image. b) Then, the two centers of laminae and the tips of spinous process are identified in the axial view of the 3D US image at the level passing horizontally by both laminae. c) The spinous processes locations are then displayed in the sagittal view and used to measure the KA. The T1/T12 KA is obtained by measuring the angle between segments connecting T1T2 and T11/T12.

Figure 6.1 The process of AVR measurement starting from coronal to axial US images. a) The center of laminae are identified on the coronal view, and at each of the vertebra levels (T4 to L4) a line is automatically drawn and joined the COL. b) On the axial view, the selected vertebra (T6), the center of the laminae can be fine-tuned, and the AVR angle forms between the line which joins the COL and a horizontal reference line (dashed line).

Figure 6.3 Scatter plot of the inter-method measurement difference and curve severity 205

Figure 7.1The measurements of US parameters including. a) US Curve angle, b) AVR, c) KA and d) RC index on US images. a) US Cobb angle measured 28° on a coronal US image using the center of laminae method, b) the US AVR is measured on the transverse view US image using the center of laminae, c) to measure the KA, the angle between the two line that joins the spinous process of T1 and T2 and T11 and T12 is measured d) the RC index is measured based on the magnitude of the US refection from the laminae at L5 level by accounting the thickness of the soft tissue from the skin to laminae...... 217

Figure 8.1 Clinical application	of the detection model	to detect cases with progression i	n
clinical practice			3

List of Equations

Equation 2.1 Shows the logistic regression equation
Equation 4.1 Shows calcualtion of the ultrasound echo at soft tissue bone interface 129
Equation 4.2 Calculation of the reflection coefficient based on accoustic impedance 129
Equation 4.3 to 4.6 Show derviations of a formula for ultrasound multiple reflections. 129
Equation 4.7 to 4.8 Show the effect of absorption upon reflection coefficnet
Equation 4.9 Shows the calcualtion of the reflection coefficent (RC) index 151

List of Abbreviations

- 2D: Two Dimensional
- **3D**: Three Dimensional
- AIS: Adolescent Idiopathic Scoliosis
- ARC: Average Reflection Coefficient
- AUC: Area Under the Curve
- AVR: Axial Vertebral Rotation
- **BMD**: Bone Mineral Density
- **BMI**: Body Mass Index
- **BP**: Blue Phantom
- BUA: Broadband Ultrasound Attenuation
- CA: Cobb Angle
- CARC: Combined Average Reflection Coefficient
- CI: Confidence Interval
- COL: Center of Lamina
- **CT**: Computed Tomography
- CTLSO: Cervical-Thoracic-Lumbar-Sacral-Orthosis
- **DXA**: Dual X-ray Absorptiometry
- **FAI**: Frequency Amplitude Index
- **GPS**: Global Positioning System
- H: Soft Tissue Thickness
- ICC: Intra-class Correlation Coefficient

IS: Idiopathic Scoliosis

KA: Kyphotic Angle

L: Lumbar

LAT: Lateral

MAD: Mean Absolute Difference

MIAS: Medical Image Analysis Software

MRC: Maximum Reflection Coefficient

MRI: Magnetic Resonance Imaging

MT: Main Thoracic

NOC: Number of Curves

NPV: Negative Predictive Value

PA: Posterior-Anterior

PMC: Plane of Maximum Curvature

PPV: Positive Predictive Value

PT: Proximal thoracic

QUIPS: Quality in Prognostic Studies

QUS: Quantitative Ultrasound

RC: Reflection Coefficient

RCS: Rough Curved Surface

ROB: Risk of Bias

ROC: Receiver Operating Curve

SCS: Smooth Curved Surface

SD: Standard Deviation

SEM: Standard Error of Measurements

SFS: Smooth Flat Surface

SI: Stiffness Index

SOS: Speed of Sound

SOSORT: Society on Scoliosis Orthopedic and Rehabilitation Treatment

SP: Spinous Process

SRS: Scoliosis Research Society

TGC: Time Gain Compensation

TL: Thoracolumbar

TLSO: Thoracic-Lumbar-Sacral-Orthosis

TP: Transverse Process

US: Ultrasound

UT: Upper Thoracic

VOS: Velocity of Sound

Chapter 1: Introduction

Adolescent idiopathic scoliosis (AIS) is a three-dimensional (3D) spine condition that has an unknown cause [1]. This condition is recognized by lateral curvature of the spine but can affect the axial and sagittal planes as well. To manage this condition, observation, nonsurgical and surgical interventions are used. The major goal of the treatment is to improve the cosmetic appearance and preserve the curve progression.

1.1. Motivation

The current method to monitor the curve severity and check if the curve has progressed or not is taking consecutive radiographs every six months. These repetitive radiographs increase the level of radiation that children must be exposed to. This is an important issue because these adolescents are in the midst of their growth spurts. The ionizing radiation may not only have short-term negative effects such as increasing the cancer risk, but also have long-term effects for their future generations [2], [3]. Therefore, it is important to reduce the level of radiation for these children.

One way to lower the level of radiation is to reduce the number of radiographs. Based on the literature, only 15% of the children with AIS show curve progression [4]. This means that most of the radiographs acquired during follow-up visits are unnecessary and do not make any significant change in the management of their conditions. On the other hand, clinical parameters that have been associated with curve progression may help to predict the risk of curve progression [5]. Using these parameters, it is possible to develop prediction models for curve progression. However, there are still controversies on predictors to estimate curve progression among studies. Currently, there is no accurate prediction model available to use in everyday clinics [5]. Also, most of the reported models are lacking a comprehensive analysis and have not been validated.

Ultrasound (US) imaging has been proven to monitor scoliosis safely and effectively. A single ultrasound scan can provide images in coronal, axial, and sagittal planes. The spinal curvature information measured on US images is reliable and accurate [6], [7]. Since 38%

of children with AIS have been reported to have low bone mass [8], using US reflection signals from the spine to extract the properties of bone may have potential to help predict progression. Therefore, further investigation of the association of US parameters, as well as other clinical records with curve progression, to develop an accurate prediction model for children with AIS should be conducted.

The hypotheses of this research are:

a) a high accuracy curve progression detection model can be developed using ultrasound parameters and clinical information for AIS children at a follow-up visit.

b) using the developed model, children with AIS will not need to undergo unnecessary radiographs and this may save them from extra exposure.

1.2. Objectives

The specific objectives of this research are:

- 1. To identify potential prognostic predictors based on state-of-the-art prediction models from the literature
- 2. To extract a new US parameter called reflection coefficient (RC) index from US reflection signal, investigate the repeatability and reliability of this index in clinical practice, and correlate it with curve severity and progression
- 3. To investigate the reliability and accuracy of kyphotic angle (KA) measurements on ultrasound images and investigate their association with the progression of scoliosis
- 4. To investigate the factors influencing the reliability and variation of US axial vertebral rotation (AVR) measurements and investigate the association of AVR with curve progression
- 5. To develop and validate a detection model based on US parameters and clinical information to accurately detect cases with progression at a follow-up visit for children with AIS

6. To investigate if we can reduce the number of radiographs and save children from unnecessary radiation exposure

1.3. Scope of Work

The first scope of work included understanding the current literature on the predictors of curve progression. Therefore, a comprehensive systematic review was conducted to learn about the potential prognostic factors found in high-quality studies for different progression criteria and follow-up intervals.

A new US parameter called RC index—extracted from US reflection data—was analyzed based on the fundamental theory of physics and through experiments. Then, the repeatability and reliability of this index were investigated before applying it to the development of a detection model.

Following the RC extraction from the US images, US kyphotic angle and axial vertebral rotation parameters measured on the sagittal and axial planes, respectively, were studied. The goal of the KA studies was to determine a method that could be used to measure the KA reliably on US images. Then, the reliability and factors influencing variation in the measurements of the ultrasound KA and AVR needed to be investigated.

After all of the US parameters and potential clinical parameters associated with curve progression were identified. The developed model was validated with clinical data.

1.4. Thesis Organization

This thesis is organized into eight chapters.

Chapter 1 states the motivations, specific objectives, and scope of work of this thesis, along with an overview of the thesis chapters.

Chapter 2 provides background information on spine anatomy, scoliosis, X-ray-based imaging modalities and the scoliotic curve characteristics measured on these modalities, non-ionizing imaging modalities for scoliosis, bone property assessment methods, the

potential predictors of curve progression, and the method used to develop the prediction model.

Chapter 3 presents a comprehensive systematic review for identifying the predictors of curve progression for AIS.

Chapter 4 describes a new US index which is called the reflection coefficient (RC). The fundamental theory of this index as well as its correlation with curve severity is explained. Also, a clinical study describes the reflection coefficient as an indicator of bone strength and presents the repeatability and reliability of this index and its association with curve progression.

Chapter 5 describes how to measure the kyphotic angle (KA) on ultrasound images for children with AIS. A pilot study investigating the reliabilities and differences in KA measurements on US images compared with X-ray is described. Then, factors influencing KA measurements on US images are reported.

Chapter 6 presents a clinical study investigating the factors influencing the reliability and measurement variation of US axial vertebral rotation (AVR) versus X-ray.

Chapter 7 presents a pilot and a large clinical study on the development of a detection model for curve progression in AIS at the follow-up visit using US and clinical parameters.

Chapter 8 provides a summary of this PhD thesis work, the major contributions, the clinical significance, the limitations, and future recommendations for this research.

Chapter 2: Background Information

2.1. Overview

This chapter provides background information on spine anatomy, scoliosis, imaging of scoliosis, methods of bone properties assessment, and methods on prediction of progression. Section 2.2 provides the anatomy of the spine including structures and features of vertebrae. Section 2.3 presents an overview of scoliosis, curve progression, and available treatment options for scoliosis. Section 2.4 describes X-ray-based imaging modalities that are used for imaging scoliosis. In Section 2.5, the curve characteristics of scoliosis measured on X-ray based modalities are described. Section 2.6 describes the non-ionizing imaging modalities used for scoliosis. Section 2.7 presents the bone property assessment methods. In section 2.8, a summary of the predictors of curve progression (detailed description provided in Chapter 3) as well as the methods to predict curve progression are described.

2.2. Spine Anatomy

The human spine or vertebral column is a bony structure consisting of 33 individual vertebrae: 7 cervical vertebrae, 12 thoracic vertebrae, 5 lumbar vertebrae, 5 fused sacral, and 4 fused coccygeal vertebrae from the superior to inferior direction as shown in Figure 2.1 [9]. When viewed from the front, the normal spine is vertically straight, but from the side, it has four curvatures including cervical, thoracic, lumbar, and sacral curves. The cervical and lumbar curvatures are concave posteriorly which is known as cervical and lumbar lordosis, while the thoracic and sacral curvatures are concave anteriorly called thoracic and sacral kyphosis (Figure 2.1) [10]. These curvatures help to maintain spine alignment and to distribute the mechanical loading incurred when the body is either at rest or in movement.

2.2.1. Vertebral Column Structure

Every vertebra is an irregular bone consisting of a vertebral body, a vertebral arch, and vertebral processes (Figure 2.2) [11]. The vertebral body is the largest part of the vertebra and is located anteriorly. The vertebral arch is located posteriorly and is formed by a pair of pedicles and a pair of laminae. The pedicles are short, thick processes that extend from the side of the vertebral body and join the body to the arch. The laminae are broad plates extended from the pedicle processes backward and medially forming a posteriorly protruding process known as the spinous process. The other processes include two transverse processes and four articular processes. The transverse processes are formed laterally at the junctional point of the laminae with the pedicles from either side of the vertebra. There are superior and inferior articular processes that are on each side of the vertebra having facet joints that connect each vertebra with the upper and lower vertebrae. The size and shape of vertebrae are different depending on their location. Normally, the size of the vertebrae increases from the top to the bottom of the vertebral column. As shown in Figure 2.2, distinctive features that only exist in thoracic vertebrae are the costal facets on each side of the thoracic vertebral body and on the transverse processes that articulate with the head and tubercles of the ribs, respectively.



Figure 2.1 The anatomy of the spine: (a) The front view, (b) the back view, and (c) the sagittal view [10]



Figure 2.2 The landmarks of a thoracic vertebra (T6): (a) The superior view of a T6 vertebra, (b) the lateral view of a T6 vertebra [11]

2.3. Scoliosis

Scoliosis is a complicated structural spine condition that is generally characterized by lateral curvature of the spine which can also affect the axial and sagittal planes [12]. This leads to scoliosis being recognized as a three-dimensional (3D) spinal curvature condition. The causes of scoliosis are either non-idiopathic or idiopathic. Non-idiopathic scoliosis includes congenital, neuromuscular, and syndrome-related scoliosis. Congenital scoliosis is related to skeletal abnormalities including formation and segmentation failures appearing at the time of birth [13]. Neuromuscular and syndrome-related scolioses arise from a long list of neurological conditions, muscular abnormalities, and syndromes that are associated with scoliosis [14]. Some of these conditions include but are not limited to cerebral palsy, arthrogryposis, Marfan's syndrome, neurofibromatosis, Duchenne muscular dystrophy, and paralysis [15].

However, 80% of scoliosis has an unknown cause which is known as idiopathic [1].

2.3.1. Adolescent Idiopathic Scoliosis (AIS)

Adolescent idiopathic scoliosis (AIS) is the most common type of idiopathic scoliosis with a prevalence of 1-3% [16]. Patients with AIS normally present after 10 years of age corresponding with rapid adolescent growth [15].

Treatment choices for AIS are considered based on the severity of the curvature, the maturity status of the patients, the patients' self-image, risk of curve progression, and type of scoliosis [11, 12]. Under clinical circumstances, most treatment planning is based on the major curve progression (the largest curve of the spine). If the patients are left untreated, some future consequences may occur. These problems include worsening or progression of the curvature, back pain, increased risk of mortality, and negative psychological issues [18].

2.3.2. Progression of the Curvature

A scoliotic curve is comprised of three components: the apex (or the apical vertebra), and the superior and inferior end vertebrae along a curve. The apex is defined as the vertebra that has the largest lateral deviation from the axis passing through the center of the sacrum. The end vertebrae are those that define the end of the curve in a coronal or sagittal projection. The superior end vertebra is the first vertebra from the apex in the cephalad direction whose superior surface has the maximum tilt toward the concavity of the curve. The inferior endplate is the first vertebra in the caudal direction whose inferior surface has the maximum tilt toward the concavity of the curve. The severity or curve magnitude of AIS is measured using a standard clinical method known as the Cobb angle. As Figure 2.3 shows, the Cobb method measures the angle between the superior endplate of the proximal most-tilted vertebra and the inferior endplate of the distal most-tilted end vertebra [19]. A scoliotic curve is defined when the Cobb angle is greater than 10 degrees [20]. The Cobb angle is measured on a coronal x-ray. Curve progression is defined as an increase of curve magnitude >5° between two visits with a 6 to 12 month time interval [21]. The results of a study on untreated children with AIS showed that 68% of the untreated children showed curve progression after skeletal maturity [18]. Overall, curvature less than 30° at skeletal maturity did not progress while the curvatures between 50 to 75°, particularly those located at thoracic regions, progressed at skeletal maturity [11, 16]. The reason why curve progression occurs has been explained by the Heuter-Volkmann Principle [23]. This principle is related to the function of growth plates which are at each end of vertebral bodies and are responsible for longitudinal spinal growth. When these plates are loaded, their growth rate is delayed. When the plates are unloaded, their growth is stimulated. Therefore, when there is a curvature in the spine, the uneven mechanical pressure on the vertebrae within the curve leads to uneven growth of the plates. The uneven growth of the plates exacerbates the curvature. This phenomenon is a vicious cycle and is the fundamental treatment theory of conservative treatment of AIS [24].



Figure 2.3 The Cobb angle method measuring the severity of the curve on a coronal X-ray
2.3.3. Treatment Options

The management of AIS includes three treatment options: observation, non-surgical, and surgical interventions. The non-surgical management of AIS includes both bracing and specific exercises [25], but bracing has been validated as effective [26]. The details of these methods are described in the following sections.

2.3.4. Observation

Observation means children with AIS are monitored by radiographs every 4 to 12 months to check if the curve has progressed or not. Observation is recommended for immature patients with mild curves of Cobb angle $< 25^{\circ}$ that show no sign of curve progression [16].

2.3.5. Orthotic Treatment

Orthotic treatment (bracing) management is recommended for immature children with moderate curves, Cobb angle between 25° and 45°[16]. A brace (orthosis) is a hard plastic shell that a scoliotic child wears to stop curve progression until maturity occurs. There are different types of braces mainly grouped as Thoraco-Lumbo-Sacral Orthosis (TLSO), Cervico-Thoraco-Lumbo-Sacral Orthosis (CTLSO), and night-time orthoses [27]. The TLSO covers the front and back from the pelvis to the armpits. Normally, these types of braces are used for curvatures with an apex at or below T8. The CTLSO is used for the curvatures with an apex at or above T8 [27]. The night-time braces are used during sleep with the most common one being the Charleston brace. Overall, bracing has been reported to decrease risk of curve progression and incidence of surgical intervention by 75% for skeletally immature children with AIS versus untreated ones [26]. A study reported that when a brace, prescribed for full time wear, has been worn for over 12 hours a day, the success rate can reach up to 93% [26]. However, long-time brace wear is not desirable for most children because most braces are bulky, uncomfortable to wear, and restrict children's activities.

2.3.6. Exercise Therapy

The general aim of exercise therapy is to reduce the progression of scoliotic curvature and postpone or avoid the need to wear a brace [28]. Exercise therapies for AIS, which include physical exercises to strengthen and mobilize the spine, machine-assisted exercises, breathing, and postural correction exercises, along with less important ones such as yoga and tai chi, help to improve trunk flexibility and strength. The Schroth method is a scoliosis-specific exercise that includes sensorimotor, postural, and breathing exercises, and is used to recalibrate normal postural alignment, static/dynamic postural control, and spinal stability [29]. Studies showed that physical exercises may help to improve back strength and breathing function significantly [30]. Also, they help to reduce the rate of curve progression and curve severity in AIS [31]. In addition to the improvement in scoliotic characteristics, function, and quality of life, exercises may also increase self-esteem and provide positive psychological outcomes [25], [30], [32]. The results of a systematic review showed that Schroth exercises significantly reduce the Cobb angle and improve quality of life in AIS with level II evidence of support [25].

2.3.7. Surgical Treatment

Surgical interventions are considered for severe curves greater than 45° to correct the curve, preserve sagittal alignment and balance, as well as improve the quality of life [27, 28]. The two main approaches for surgery are anterior and posterior fusions. In the posterior fusion, surgeons access the posterior surface of the spine for instrumentation (such as longitudinal rods, hooks, sublaminar wires, and pedicle screws) [35]. Figure 2.4 illustrates a pre- and post-operative radiograph using posterior spine fusion on a case with AIS. The anterior approach is usually used for thoracolumbar or lumbar curves [36]. For the anterior approach, less instrumentation is attached to the spine; however, some complications such as poorer pulmonary function as well as higher risks of injury to critical organs and vessels may occur [27, 30]. The most used approach is posterior fusion. Although surgery significantly corrects the curve, some complications such as hook dislodgment, infections, and neurologic deficits are still of concern [37].



Figure 2.4 The posterior spinal fusion and instrumentation on a case with AIS: (a) Preoperative X-ray, (b) post-operative X-ray

2.4. X-ray Based Imaging Modalities Used for AIS

The X-ray based modalities used to image AIS include images acquired from:

- Conventional X-ray systems (screen-film and digital radiography)
- EOS imaging (2D/3D)
- Computed tomography (CT) scan

2.4.1. Basics of X-ray Based Imaging Modalities

Radiography is an imaging technique using X-rays to image the body. An X-ray system consists of an X-ray tube that generates X-ray photons which are exposed to the target of interest. The X-ray photons interact with the body and a portion of them are absorbed, some are scattered, and the rest is transmitted through the body and received by a detector. The higher density body tissues such as bone absorb more X-rays than soft tissue like muscle.

The detector is either a photographic film in analog radiography or a digital detector in digital radiography.

2.4.2. Conventional X-ray Systems

Generally, the most used method to image AIS is capturing an X-ray via conventional systems. Scoliosis radiography was generally performed by screen-film radiography. However, there are some drawbacks with screen-film radiography such as the image contrast cannot be changed after films are processed, the films are expensive, and the film processing procedures require the use of some hazardous chemicals and are labor-intensive [38]. In the last 2 decades, digital radiography has become more common. It includes two forms computed radiography (CR) and direct radiography (DR) [39]. The difference between CR and DR is related to the mechanism involved in the design of detectors to transfer X-ray energy to an electrical signal [39]. However, it is difficult to image the whole spine using digital radiography since the detector plates are not large enough to include the whole length [40]. One approach is to stitch together individual images acquired from different exposures into one image or image reconstruction which may cause image distortion and inferior image quality [41]. Nevertheless, changing from screen-film X-ray systems to digital radiography has reduced the X-ray retake from 5.5% to 1% which significantly prevents repeated exposure [42]. The absorbed dose for a imaging full spine significantly is reduced when DR is used versus CR and screen-film radiography [40]. A study reported that DR showed the lowest radiation to the whole body (effective dose), up to 43% lower than the other two methods (CR and screen-film radiography) [39].

2.4.3. EOS Imaging System (2D/3D)

The EOS X-ray machine is a Nobel prize-winning invention in physics capable of capturing biplanar (2D) X-ray images simultaneously by slot scanning the whole body in an upright and load-bearing position, while using low radiation doses [43]. The two source-detector pairs are positioned orthogonally so that two simultaneous coronal and sagittal images of

the spine are acquired. Figure 2.5 illustrates a coronal and sagittal view of a scoliotic spine on both 2D and 3D reconstructed EOS images.

The EOS imaging method is different from other X-ray based imaging in several ways including [44]:

- It allows for scanning the whole body in a standing position
- It reduces the radiation
- It creates 3D model reconstruction of upright images of the skeletal structure

Whole-body scan in standing position: Since the ideal position to image AIS is in a standing position, EOS has been popular for imaging of AIS for orthopedics applications. This ability to perform a whole-body scan in standing position is very useful to evaluate the spinopelvic relationship and to investigate secondary reasons leading to scoliosis such as leg length discrepancy [45]. However, the lengthy data acquisition process may cause artifacts on the image due to movement that can degrade the image quality. Also, it cannot be used for children who cannot stand in the device or infants with scoliosis. In addition, the standing posture is important. The degree of shoulder flexion may affect the sagittal alignment. While studies suggest different positioning during scan capture, the most widely used position is putting hands on cheeks [46].

Radiation dose: The radiation dose is reduced by a factor of 2.5 to 10 using EOS versus conventional radiography [43], [47]. The dose reduction even reaches 800-1000 when compared with 3D CT scan [43], [48]. To compare the dose from EOS versus DR radiography, the average dose area product (DAP) is used. The DAP considers the absorbed dose for a specific area, in this context, the spine. The DAP was 23.6 mGy·cm² /kg±4.32 for EOS versus 95.7 mGy·cm² /kg±30.39 for digital radiography for both AP and lateral spine exposures in children with idiopathic scoliosis [44].

EOS 3D images: The simultaneous capture of AP and sagittal images by EOS enables us to generate a 3D reconstruction of the skeletal system using a special software as shown in Figure 2.5 c and d. This feature is suitable for imaging of AIS. When the 3D reconstruction process is complete, the software automatically generates measurement files which reports

the curve characteristics in coronal, axial, and sagittal planes, as well as pelvic parameters. However, a few situations may make the EOS 3D reconstruction difficult. The first scenario is when the scoliotic curve is severe; the operator may have difficulty identifying landmarks and performing adjustments to overlay a vertebra to the database model. Also, in abnormal anatomical situations such as sacralization of L5 (the fifth lumbar vertebra is fused to the sacrum bone) or lumbarization of S1 (the first sacral vertebra is not completely attached to its fused sacral components and instead the first sacral vertebra looks like a lumbar vertebra). The operator may have difficulty identifying the vertebrae correctly. The 3D reconstruction process is time-consuming. The mean reconstruction time is about 20–30 minutes for 3D reconstruction of the spine and 35–45 minutes for 3D spine and lower limb rendering [44]. For severe idiopathic scoliosis, it may take longer than an hour to reconstruct.



Figure 2.5 The EOS image of a scoliotic spine: (a) The 2D coronal view of spine; (b) the 2D sagittal view; (c) the 3D reconstructed image of the spine projected in coronal view; (d) the 3D reconstructed image of the spine projected in sagittal view [49]

2.4.4. Computed Tomography (CT) Scan

Computed Tomography (CT) provides 3D images of the bony structures. The system includes an X-ray source and detectors that are located in opposite directions. During data acquisition, the source and detectors rotate around the patient and acquire a series of X-ray images from different angles. The X-rays are transmitted through the body and are recorded by the detectors. Then, a high-speed computer processes the data to provide cross-sectional images (slices). A CT scan can provide a more detailed image of the spine morphologies and structures [50] and is considered as a standard for the 3D reconstruction assessment method compared to the EOS imaging [51]. CT scans are recommended for complicated surgical cases, pre-operative imaging to evaluate the size of pedicles for surgical instrumentation, and post-operative imaging to check the correction [52].

2.5. Use of X-ray Systems to Image Curve Characteristics of AIS

Since AIS has a 3D nature it is important to evaluate this condition in the coronal, axial, and sagittal planes.

2.5.1. Cobb Angle

The methods that are used to measure the Cobb angle include manual methods such as using a protractor and pencil on the film radiography, and computer-aided methods on both CR and DR that identify the spine landmarks. All the methods have reproducibility within 5 degrees [53]. Good to excellent intra- and inter-rater reliabilities have been reported for both manual [54] and digital [55] Cobb angle measurements. However, an improvement on Cobb angle measurement was reported on digital measurement versus manual [53], [56]. This shows the importance of digital measurement since it is becoming the more prevalent method in scoliosis clinics [56]. A phantom study compared the Cobb angle measurement between the three X-ray methods EOS (2D), CR, and DR. The results showed that Cobb angle measurements using CR and DR methods had a high correlation with the EOS measurement with the R² value (coefficient of determination) close to 1 for both [57]. Also, the reliability results showed excellent intra-rater reliability, with an intraclass

correlation coefficient (ICC) > 0.9 and good inter-rater reliability (ICC \ge 0.74) for all three imaging methods [57]. In addition to the curve severity measurement, the Cobb angle information is also used to monitor the progression of the curve and to inform a decision on the treatment options.

2.5.2. Spinal Flexibility Assessment

Additional radiographs of children with AIS may be required for assessment of spinal flexibility. Spinal flexibility informs clinicians about the biomechanics of the curvature which is particularly helpful for surgical cases. This information advises surgeons about the structural extent of the curve, the surgical approach, and the level of fusion [58]. Yet, some studies suggest using flexibility information to estimate in-brace correction for bracing treatment, and as a predictor of curve progression [43, 44]. There are different radiographic techniques that are used to estimate spinal flexibility including:

(a) Supine side-bending: In this method, the patient voluntarily bends to the sides as far as she can. This method is the gold standard for spinal flexibility assessment [61]. The spinal flexibility is estimated by comparing the curvature in bending position versus the standing X-ray Cobb angle.

b) Push-prone: The push-prone method is performed by technicians applying forces at the apex or apices of curvature while counter forces are applied in the opposite direction around the pelvis and axilla [62]. However, it is difficult to determine what forces should be applied to obtain optimal correction with this method.

(c) Fulcrum-bending: In this method, a radiolucent fulcrum is placed under the curve apex while the patient is in a lateral position on the fulcrum. The results of a systematic review showed that the fulcrum bending method has the most accurate estimation of post-operative outcomes compared to other methods with the support of very low to low quality evidence [63]. The accuracy of the results for this method may vary based on size, rigidity, and placement of the fulcrum [64].

(d) Traction: The traction method has two options: with anesthesia or without. Greater curve correction is observed during traction under anesthesia versus without. Traction under general anesthesia is implemented by divergent forces applied on the ankles and axilla. Then translational pressure is applied at the apex of curvature [61], [62]. This method may provide better estimated outcomes for severe curves (Cobb angle >60°) compared with the side-bending method [61], [62]. However, since these images are acquired right before surgery, surgeons may not have sufficient time for thoughtful surgical planning.

In addition, Hirsch et al. used a new technique using EOS modality called the suspension test to investigate the spinal flexibility in 50 AIS surgical children [65]. In the Hirsch et al. study, EOS radiographs were captured while progressive traction forces were applied to patients through a rigid collar attached to cables. The EOS radiographs were taken when the patients were on tiptoes. This method was compared with the supine traction test. Higher forces were applied during this test and forces were more standardized than the traction method. Also, using the suspension method, it was possible to analyze the curvature changes in the three coronal, axial, and sagittal planes. The disadvantages were that the tolerance of the patient was lower in the suspension test and special equipment was required which prevented this method from being widely used [66].

Although the benefits of spinal flexibility assessments are undeniable, the extra radiographs and radiation children are exposed to is undesirable. This makes clinicians reluctant to request extra radiographs to obtain flexibility assessments for non-surgical cases.

2.5.3. Axial Vertebral Rotation (AVR)

The axial vertebral rotation (AVR) happens when the vertebra rotates around its longitudinal axis when projected in the axial plane [67]. The landmarks that are used to measure the AVR include the pedicles, spinous process, and vertebral body that are imaged on posteroanterior (PA) X-rays. The AVR measurement on plain radiographs was first introduced by Dr. John Robert Cobb [68]. In his method, the vertebral body and spinous

process are used to measure the AVR. Depending on the position of the spinous process, the degree of AVR is measured in five grades. However, the spinous process may be poorly visible on radiographs making the measurements difficult. Also, it is not possible to quantify the AVR using a grade system.

Another method was developed called the Nash-Moe method [69]. In this method, the halfwidth of the vertebral body reflected on the convex side of the curve is divided into three equal segments and the pedicle shadow on the convex side is considered the landmark. Therefore, depending on the location of the pedicles in these segments the AVR is graded from 0 to 4+. The visibility of the pedicle is an advantage in this method. However, the pedicle displacement percentage is used to estimate the AVR. Also, the size of the vertebral body is not accounted for.

In a different method, the pedicle at the convex side is used and a torsion meter is placed on an anteroposterior (AP) X-ray in an attempt to measure the AVR [70]. The AVR is quantified with ± 5 degrees in this method [71]. Still, the differences of the vertebrae sizes particularly for irregular vertebral geometry based on different locations is not accounted for.

Stokes et al. developed a method that accounted for the pedicle offsets as well as vertebral asymmetry and the differences in vertebral body sizes from T4 to L4 [72]. As Figure 2.6 shows, the projected distances from the center of both pedicles to the center of the vertebra (a and b) are measured from the X-ray; w is the width of the lamina, and d is the maximum width of the vertebra from the center to the edge [73]. In this method, the errors in measurements were \pm 1mm for identifying the landmarks and 2.7° for AVR measurement. However, outlining the vertebral edges still account for a large source of random errors in this method [73].

Another study also used the pedicle shadows along with the geometry of the vertebra and dimensional proportions to measure AVR [74]. A limitation of this method is when the AVR is over 30°, making it impossible to identify the pedicle.

Overall, the measurements of AVR on plain X-rays are not directly performed on an axial plane which limits the accuracy of the 2D AVR measurements on a coronal plane. Due to

the 3D ability of CT, AVR can be measured using this modality as well. The standard method with the most reliable results is the Ho method which measures the angle between a line going through the posterior surface of the vertebral body and the junction of the inner surface of the laminae, and the reference sagittal plane as demonstrated in Figure 2.7 [75]. Due to high levels of ionizing radiation, CT is normally not considered a routine image modality for children with AIS. Also, CT data acquisition is performed in the supine position which may affect the magnitude of the curvature in both the coronal and axial planes [76], [77]. Using the EOS 3D reconstruction, output parameters called vertebra vectors are found. This information helps to visualize the spine in 3D, providing a top view of the spine which allows for a better understanding of the vertebra size, position, and rotation [66]. The concept of the vertebra vectors was validated by comparison with the 2D radiograph measurements. The reliability was even higher than 2D measurements when the scoliosis was severe [78]. However, as mentioned earlier, 3D reconstruction by EOS is very time-consuming.



Figure 2.6 The axial vertebral rotation using Stokes' method. The projected distances of both pedicles from the center of vertebra (a and b) are measured. The "w" which is the width of the laminar and "d" is the maximum distance of the vertebra from the center to the edge. These measurements are implemented into Stokes' method equation and θ which is the vertebral rotation angle is calculated [73]



Figure 2.7 Ho's method of measuring AVR angle on a transverse view CT image of a vertebra [79]. The dashed lines are from a point at the laminar inner junction with the pedicle-lamina junction bilaterally, and the angle between these lines is bisected by a dotted line. The rotation angle (curved line) is measured between the bisector line and the vertical line.

2.5.4. Sagittal Plane Parameters

2.5.5. Kyphotic Angle (KA)

The most-reported sagittal parameters include kyphotic angle (KA) and lordotic angle (LA) which are important for AIS treatment management. Several radiographic techniques have been suggested to measure the KA such as the Cobb method, vertebral centroid method, the posterior tangent method, as well as various computer-aided methods [80]–[82]. Among all of these techniques, the Cobb method is the most popular and considered the gold standard to measure the KA [83]. As Figure 2.8 shows, the KA is measured on a standing lateral radiography by identifying the superior endplate of T1 and inferior endplate of T12. There are a few KA measurement approaches which use different end vertebral levels including T1-T12 [84], [85], T3-T12 [86], T5-T12 [87]. However, the reliability of the kyphotic measurement reported from most studies was not adequate for most tested levels. Of these levels, T1-T12 seems to be the most reported method. Carmen

et al reported partial reliability results of a kyphotic angle at T1 to T12 on 20 radiographs and the measurement error was 11 degrees with a 95% confidence interval [84]. Ohrt-Nissen et al. reported that the kyphotic angle measurements difference on intra- and interrater results from 8° to 13° [82]. They also reported that the best reproducibility in terms of agreement and reliability was achieved for KA measurements at the T4-T12 and T5-T12 levels.

According to the scoliosis research society (SRS), KA values ranging between 20 to 45 degrees are considered normal when using the Cobb method with the T1-T12 level approach. The limitation of the Cobb T1-T12 method relates to difficulties in identifying endplates, particularly at T1 due to overlap with the shoulder girdle on a conventional X-ray. Therefore, SRS recommends measurements of KA between the superior endplate of the highest measurable thoracic vertebra, usually T2 or T3, and the inferior endplate of T12.

2.5.6. Lordotic Angle (LA)

The techniques used to measure the lordotic angle are not universally agreed upon. The most used method is the Cobb method. The levels that are used to measure the LA are variable. Based on the Cobb method, the angle is measured between the perpendicular lines drawn from the superior endplate of L1 [88], or sometimes inferior endplate of T12 [89], and the inferior end plate of L5 [88], [90], with some studies using the superior end plate of S1 [85], [91] as an alternative. Most authors choose L1 as the superior end plate and S1 as the inferior level. These are the levels recommended by the SRS (Figure 2.8). For the cases where S1 is not visible, the inferior endplate of L5 is chosen. The normal range for the lordotic angle has been reported to be between 50 to 60° [92]. A study reported that the intra-rater and inter-rater reliabilities coefficients for LA measurements are within the range of 0.83 to 0.92 and 0.81 to 0.92, respectively, and 10 degrees of difference was considered an acceptable variation [93]. In that study, the best agreement of 92%, was obtained for measurements using the L1-L5 level. Sources of variation are the selection of vertebra as well as the vertebra end plate architecture.



Figure 2.8 The measurement of kyphotic (blue lines) and lordotic angles (orange lines) using T1-T12 and L1-S1 endplates, respectively

2.5.7. The Negative Effects of Ionizing Radiation

On average, a 10 year old child who is diagnosed with scoliosis requires 10 to 22 radiographs over the course of the entire treatment period [94]. However, Soucacos et al. reported that only 14.7% of scoliotic cases would progress [4]. This means most patients receive unnecessary ionizing radiation because the purpose of radiographs is to determine if the curve progresses.

Ionizing radiation from X-rays has high enough energy to break molecular bonds in the body. When this bond damage is repaired incorrectly, it can affect chromosomes and may induce cancer [95], [96]. Additionally, the accumulated ionizing radiation increases the probability of adverse health issues including cancer and abnormal pregnancies [42].

Retrospective studies indicated children with AIS who received repetitive X-rays have approximately 2 and 3% increases in lifetime risk of breast cancer and heritable defects, respectively [2], [3]. Levy et al. estimated an increase in the risk of cancer to 0.01-0.07% per 100,000 men and 0.04-0.23% per 100,000 women for scoliotic patients from conventional spinal radiographs [97]. Ronckers et al. followed 5,513 females who were exposed to an average of 22.9 radiographs per person during treatment and follow-up for scoliosis [98]. Overall, the risk of mortality was 46% higher than the general population. Cancer was identified as the primary reason for 23% of these deaths. In terms of prevalence, breast cancer was the most common, followed by lung and then ovarian cancer. Enormous efforts have been made to reduce the radiation dose for X-ray imaging. Changing the exposure orientation from AP to PA exposures can reduce the dose to organs such as breasts and thyroid from 3 to 8 times [99], [100]. The latest technology implemented in detectors of EOS 2D/3D imaging lowers radiation significantly. A study showed that the detector structures used in EOS imaging enable the reduction of entrance skin dose up to 13 times for PA and 15 times for a lateral acquisition versus conventional film radiography without degrading the image quality [101]. A new feature of the EOS using micro-doses reduces the dose up to 5.5 and 45 times [42] when compared with standard low dose and conventional radiography, respectively.

As children have a longer lifetime to reflect the radiation damage than adults, the adverse effects of radiation can appear years after exposure. Nevertheless, the cumulative radiation matters even though the exposure dose is low. Therefore, it is important to consider radiation protection and minimize radiation for these children in order to meet the ALARA principle (as low as reasonably achievable) [42].

2.6. Non-ionizing Radiation Imaging Modalities for AIS

The non-ionizing radiation methods that are described in this thesis include:

- Magnetic resonance imaging (MRI)
- Ultrasound (US) imaging method

2.6.1. Magnetic Resonance Imaging (MRI)

Magnetic resonance imaging (MRI) is an ionizing radiation-free method that uses powerful magnets to force the nuclei (in particular, hydrogen nuclei which consist of a proton) of the body to align with that magnetic field. When a radiofrequency current is pulsed through the patient, the protons are stimulated, and spin out of equilibrium. By turning off the radiofrequency field, the protons realign with the magnetic field and energy is released as radio signals which are detected by MRI sensors. The time it takes protons to realign with the magnetic field, as well as the amount of energy released, helps clinicians distinguish between different types of body tissue. MRI is recognized as a suitable method to image soft tissues such as the brain, spinal nerve, and muscles, rather than bony structures, as they are seen much more clearly on the images. The MRI systems are either open or closed. The closed ones fully surround the body, while the open MRI is open on the sides. An open MRI has a lower image quality than a closed one because of a weaker magnetic field [102]. Both closed and open MRIs require patients to be in a supine position which can affect the application of this modality for AIS imaging. Since the effect of gravity is eliminated in supine position, the magnitude of the curvature is reduced. Recently, MRI machines that scan in standing or sitting positions have also been developed but their image quality is not as good as closed systems. MRI also provides images in three-dimensions including the coronal, axial and sagittal planes. The AVR and torsion of the vertebral body and discs may be measured using MRI [103], [104]. A study showed that the AVR could be measured accurately and reproducibly with a mean difference of 3 degrees between MRI and plain radiography [104]. However, MRI is more frequently used for scoliosis cases related to neurologic abnormalities such as tumors. It is uncommon for monitoring or diagnosis of AIS [105], [106]. Compared to CT scans, MRI scans typically take a longer time and are louder, and they need patients to enter a narrow tunnel. Therefore, subjects who are uncomfortable with confined spaces, called "claustrophobic", do not feel comfortable using the closed MRI systems that fully surround the body [107]. In addition, people with medical implants or other metal inside the body may be unable to undergo an MRI examination safely.

2.6.2. Ultrasound (US) Imaging Method

Ultrasound (US) is a non-ionizing imaging modality that uses a sound wave with a frequency >20 KHz. The range of ultrasound frequencies for most medical applications is usually between 2 MHz to 10 MHz [108]. For ultrasonography, a pulse technique is used to transmit an US signal of mechanical waves to a region of interest. The ultrasound theory is that when the US signals travel through mediums, a fraction of energy reflects from the interface between two mediums to the transducer. The reflected energy can be processed to display images and used to indicate stiffness and elasticity of materials which are related to the acoustic impedance of the mediums [108]. The acoustic impedance of bone is significantly higher than the surrounding soft tissue, leading to a substantially stronger reflection at a tissue-bone interface when the ultrasound signal penetrates from soft tissue to bone. This implies that stiffer materials are expected to have stronger reflections.

2.6.3. Ultrasound for Imaging of AIS

Since the vertebral column, a region of interest throughout this thesis, has a bony structure covered by soft tissue, it is possible to image the posterior vertebra features using US imaging. Suzuki et al. was the first group to use ultrasound techniques for imaging patients with scoliosis [109]. They were able to identify the transverse processes and the laminae on US images. However, their approach only scanned patients in the prone position to measure the AVR. Dr. Lou's team is pioneering the proposal to measure the Cobb angle from US spinal images. A phantom feasibility study was first conducted to investigate the application of US for imaging of the spine [110]. That study showed that the posterior arch structures of the vertebra, which include the spinous process, transverse process, laminae, and superior articulate process of the vertebra, were detected on the image. The dimensions of these landmarks were within 4% error when compared with the phantom measurements. After the feasibility study, clinical trials were conducted to validate that US images could be used to assess and monitor AIS disorder [109]–[112]. The landmarks used in clinical practice to measure the curve severity on US images were the center of laminae (COL) [7], [111]–[113], spinous process (SP) [110], [114], transverse process (TP) [115], [116] and

superior articulate process (SAP) of the vertebra [115]. In terms of the reliability of these landmarks, high intra-rater and inter-rater reliabilities were reported when SP was used to measure the severity; however, the curve magnitude was not measured directly by using this method [110]. Also, limited evidence was available on using TP and SAP to measure curve severity. Using the TP method, the variation of the reliability coefficient was relatively large, with the range for intra-rater and inter-rater reliability coefficients varying from 0.57–0.98 and 0.75–0.96, respectively [117]. A systematic review analysis on the reliability of all available ultrasound imaging methods for severity measurements on AIS revealed that among all of the proposed methods for measurement of Cobb angle, the COL was the most reliable [117]. Figure 2.9 illustrates the COL method on an AIS case on both the transverse and coronal views. In the COL method, the proxy Cobb angle is measured by the angle between the two most tilted lines, which connect the COL at each vertebral level, within a curve. The intra-rater and inter-rater reliabilities of US Cobb measurements using the COL showed high ICC [2, 1] > 0.80 and the measurement difference between the radiography and the US method was between 3 to 5 degrees, which is within the clinically accepted error (<5°) [111], [113]. Moreover, a significant improvement in the Cobb angle measurement occurred with the aid of previous radiograph methods, and the intra-rater and inter-rater ICC [2, 1] were increased to >0.90 [7], [111].

The COL method was also used to measure the AVR in both *in-vitr*o and *in-vivo* studies. The results showed high intra-rater and inter-rater reliabilities for both (ICCs > 0.91, MAD < 1.4°) [112]. The US AVR measurements were compared with the Stokes' method on radiography and the results showed a good correlation for the *in-vitro* study (ICC = 0.84-0.85) with the mean absolute difference (MAD) within $4.5^{\circ}-5.0^{\circ}$; while the agreement for the *in-vivo* study was poor to moderate (ICC = 0.49-0.54, MAD = $2.7^{\circ}-3.5^{\circ}$). However, these studies were pilot studies with limited datasets, and this emphasizes the need for conducting a large clinical study to confirm the use of US for AVR measurement.

In addition, a study was conducted to investigate reliabilities of using US to measure the Cobb angle on the plane of maximum curvature (PMC). In this study, three raters were involved in measuring the PMC on US images. The US PMC Cobb values were compared

to the EOS PMC Cobb. The ICC values for intra-rater, inter-rater and inter-method reliabilities were all over 0.90. The MAD and the standard error of measurements (SEM) for both PMC Cobb were $< 4^{\circ}$ [118].

In this thesis, kyphotic angles on US images were investigated. Details of the studies are reported in Chapter 5.



Figure 2.9 Ultrasound imaging of the spine for a case with AIS: (a) An US transverse plane of a vertebra showing the selection of the centers of laminae, and (b) measurement of the curve severity by using the center of laminae (COL) method

2.7. Bone Property Assessment

Approximately 27% to 38% of children with AIS have been identified as osteopenic or have low bone mass [8], [119]. Bone is composed of two main parts, cancellous (also called trabecular or spongy) and cortical (compact). The basic material comprising cancellous and compact bone appears identical, while the degree of porosity and the organization is different. The architecture of cancellous bone includes its porosity and connectivity, the extent to which structural elements are connected together [120]. Osteopenia is a condition

that affects bone and is characterized by low bone mass [121]. Bone mineral density (BMD) is the ratio of bone mineral content to bone size. The average BMD values of girls with AIS were 4.5% lower than that of the age- and sex-matched controls [122]. A study compared the BMD scores from the lumbar and the three regions of the femur of AIS children with the matched control group and the results showed that BMD was low in all these regions versus the control group [123].

Two methods of bone property assessments, dual-energy X-ray absorptiometry (DXA) and quantitative ultrasound (QUS) imaging methods have been described in following sections.

2.7.1. Dual-energy X-ray absorptiometry (DXA)

Currently, the most common method to assess BMD is DXA which exposes patients to ionizing radiation. This method uses two beams of X-rays with different levels of energy. One beam with lower energy which, is absorbed by soft tissue and the other with a higher level of energy which is absorbed by bone. When the soft tissue X-ray absorption is subtracted, it is possible to estimate the absorption of each beam by bone; this is related to bone density. BMD is usually measured on the hip, spine, and forearm by using DXA. In this method, the BMD status of the patients is evaluated by Z scores. The Z scores compare children's BMD with age-matched norms and give clinicians a sense of the ageappropriateness of bone loss. When Z scores are less than or equal to -1 standard deviation, bone status is classified as osteopenia. When age-adjusted bone mineral density is above – 1 standard deviation, it is counted as normal bone status. However, some drawbacks are prominent with DXA. It uses ionizing radiation which is harmful to children. The radiation dose from a DXA exam on the hip and spine is estimated to be between 10 to 15 microsieverts [Sv is a unit of dose equivalent (the biological effect of ionizing radiation)]. Although it is considered low radiation, for children with AIS who already must undergo many X-rays during the treatment, it is still worrisome. Also, DXA is a projectional technique, which means imaging a 3D object in 2D. For children with AIS who have large vertebral rotation, it may lead to an inaccurate analysis of BMD [124], [125]. Furthermore, DXA is only able to quantify the bone mass but not qualify it, and it is relatively costly and

may not be readily available in clinical settings [120]. Also, BMD may not truly represent bone strength. The strength of an object depends on the amount of material in that object, the internal structure, geometry, and the mechanical properties of the constituting material [126]. A term defined as bone quality might better explain bone strength than bone density [121]. Bone quality is described as a composite of properties including the internal structure, geometry, and the mechanical properties of the material that determine how well the bone can resist the fracture [126].

2.7.2. Quantitative Ultrasound (QUS)

Quantitative Ultrasound (QUS) has been introduced to assess bone quality. QUS systems are different in terms of the emitted frequency, pathways of US transmission inside the bone, and the skeletal site as the region of interest. The range of frequencies used in QUS devices is within 500 kHz and 1.25 MHz [127]. The majority of QUS devices can be used only for one skeletal site such as the proximal phalanges and calcaneus bone. QUS devices work based on the transmission of the US wave through the region of interest while the transmitter and receiver of the US are placed on the opposite side of the target.

The advantages of US over DXA are the lack of ionizing radiation, low cost, and portability. More importantly, the elastic nature of the ultrasound waves makes the US imaging method an ideal option to evaluate the bone quality [128]. The main QUS parameters associated with bone quality or bone strength are broadband ultrasound attenuation (BUA), the ultrasound velocity of sound (VOS), and stiffness index (SI) [129], [130]. BUA is a measure of frequency dependent ultrasound attenuation. It provides information about the cortical thickness, elasticity, and micro-architecture of bone. The VOS indicates the transmission velocity of ultrasound passing through bone. The VOS evaluates bone elasticity and density. The SI is a composite of BUA and VOS parameters, and it provides information about bone elasticity. Recently, researchers have investigated the use of QUS parameters for AIS. A study used QUS parameters to evaluate the bone quality of children with AIS and its correlation with BMD measured with DXA at the femoral neck. The QUS parameters BUA and SI measured in the calcaneus were lower

than the normal control group for mild curves (10 to 25 degrees). The results of QUS were then confirmed by the BMD measured at the femoral neck which was also significantly lower than normal controls [131]. Lam et al. investigated the correlation between QUS measurements at the heel area and curve progression in children with AIS [130]. They used the QUS parameters BUA, VOS, and SI and compared them with the BMD from the hip. They found low bone quality indicated by SI was a significant prognostic factor of curve progression with the odds ratio of 2 (1.0–3.7, 95% CI) while BUA and VOS were not [129].

Zheng et al. used a US pulse echo technique to measure the US reflection signals directly at the spine area [132]. In their study, the frequency amplitude index (FAI), a parameter dictated by the reflection coefficient of the soft tissue/bone interface for bone quality assessment, was captured. Only 18 subjects were recruited, and the results showed that the reflection index decreased as the curve severity increased. However, the correlation coefficient (\mathbb{R}^2) was only 0.14. In that study, the effect of soft tissue was ignored which may have affected the results significantly.

Although studies have shown that US may provide information about bone quality, DXA is still the method of choice to estimate BMD for clinical practice. Further studies are required to evaluate the ability of US to estimate bone quality. In Chapter 4, a new US parameter that helps to estimate bone stiffness has been described.

2.8. Prediction of Curve Progression

Prognostic factors are the variables that are used to predict whether a health condition will get worse or not. A list of different prognostic factors including radiographic, demographic, physiologic, and genetic-related parameters have been reported and related with curve progression [5], [123], [130], [133]–[137]. A comprehensive systematic review on the prognostic factors of curve progression has been described in Chapter 3.

2.8.1. Methods for Prediction of Curve Progression

The methods for prediction of progression include linear regression analysis and classification analysis. Under classification methods, the logistic regression analysis was

selected to perform the prediction model development analysis for this PhD thesis. Logistic regression analysis is a popular method for developing prediction models in medical research because it is an easy, fast and simple classification method compared to other classification methods which require complicated programming [138]–[141]. The dependent variable is whether a patient show curve progression or not. This simply implies that the dependent variable is binary (assumes a value of either 0 or 1) or dichotomous [142]. The predicted dependent variable is a function of the probability that a particular patient will be in one of the categories (0, 1). Probability is the likelihood something will happen, and it takes a value between 0 to 1. The odds are the ratio determined by dividing the probability that an event will occur by the probability that an event will not occur. The key component of logistic regression analysis is logit which is simply the natural logarithm (ln) of the odds (Equation 2.1). Therefore, the logistic regression will be:

Equation 2.1 Shows the logistic regression equation.

Log of odds: ln [prob(an event occurs)/prob(event not occur)]= $B_0 + B_1(X_1) + B_2(X_2)$ +...+ $B_n(X_n)$ (2.1)

The B_0 is the intercept and B_1 , B_2 ..., B_1 indicate the effect of the predictor variable on the odds of the predicted variable.

2.9. Summary

This chapter describes spine anatomy, scoliosis, different imaging modalities to image scoliosis, bone property assessment methods, and the method of choice to develop the prediction model. A detailed analysis on predictive factors of curve progression as an essential part of this PhD thesis have been described in Chapter 3.

Chapter 3: A Systematic Review on Prognostic Factors of Curve Progression in AIS

3.1. Overview

This chapter reports a systematic review study on identifying prognostic factors of curve progression in AIS. Systematic review studies stand at the highest position compared to other study designs based on the hierarchy of evidence. The reason could be related to the design of systematic reviews that focus on answering a specific question and following a step-by-step comprehensive approach that assess the quality of studies to ensure a non-biased conclusive statement. To answer one of the main objectives of this PhD thesis as "identifying the prognostic factors of curve progression", it was necessary to conduct this systematic review to learn about current literature, deficits and strength and tailor this PhD research project accordingly. Section 3.2 to section 3.6 include the manuscript with the title "Identifying Prognostic Factors of Curve Progression in Adolescents with Idiopathic Scoliosis: A Systematic Review" which has been submitted to PLOS One journal. Section 3.7 summaries the key findings which lead to the final predictors of progression supported by different level of evidence and for different progression criteria and follow-up durations.

3.2. Introduction

Adolescent idiopathic scoliosis (AIS) defined as a lateral curvature of the spine is the most common type of scoliosis affecting 2-4% of children [1]. The standard measurement of the severity of AIS is the Cobb angle on a standing posteroanterior radiograph of the spine [2]. A curve change $>5^{\circ}$ between two consecutive radiographs (within 6 to 12 months) is considered curve progression [1]. However, only 14.7% of children show curve progression which implies that most radiographs do not detect any difference [2]. These unnecessary radiographs expose children to ionizing radiation and can increase risk of cancer [3].

Literature suggests that scoliosis progression may be predicted by clinical parameters [4-9]. Potential predictors include: large Cobb angle, osteopenia, age<13 years at diagnosis, skeletal immaturity, pre-menarche status, and multiple indices combining radiographic, demographic, and physiologic characteristics [4-9]. Lonstein et al. reported that the Risser sign, magnitude of the curve and patient's age at the initial examination are highly correlated with curve progression [6]. However, it was not clarified whether the prediction was for the next visit or during later follow-ups. In another study, a Risser sign of 0 or 1, an apical level cephalad to the T12, absence of coronal imbalance (\leq 10 millimetre), and lower chronological age predicted progression over a short-term interval (within one year follow-up) [10]. Tan et al. found that a larger initial Cobb angle was the most important predictor for prediction of long-term progression [9].

There is a wide variety of parameters which have been reported as predictors for curve progression. The quality of studies to identify predictors has been questioned. Also, studies used different definitions of curve progression measured at different follow-up intervals. Therefore, a comprehensive systematic review study to investigate what are the best predictors for different progression thresholds as well as for short- and long-term follow-up intervals is needed. A systematic review and meta-analysis on identifying predictors on curve progression for scoliosis was reported in 2015 [8]. The keywords and database used in this study did not include all the common clinical parameters and instead included only a few parameters acquired through complex medical, physiologic, biochemical, and genetic tests. Considering that physicians are interested to identify predictive factors of progression for different thresholds and follow-up intervals, in their review, studies were analyzed based on different categories of predictive factors rather than progression criteria or follow-up intervals. Moreover, their review was not registered which makes it unclear whether a standard systematic review protocol determined a priori was followed.

Thus, the goals of this study were to identify predictive factors for three curve progression categories including curve magnitude a) increases by $\geq 5^{\circ}$, b) reaching $\geq 30^{\circ}$ or surgery threshold, and c) changes in Cobb angle as $^{\circ}$ /year on any two radiographs over both short-term (within ≤ 1 -year) and long-term (>1-year) follow-up intervals.

3.3. Methods

3.3.1. Protocol and Registration

The systematic review protocol was registered in PROSPERO (CRD42019116211) and followed the guidelines of the Preferred Reporting Items for Systematic Review and Meta-Analysis-PRISMA checklist [11].

3.3.2. Inclusion Criteria

The eligibility criteria for each phase starting from abstracts to full-texts review and extraction phase contain: 1) human participants, 2) diagnosis of AIS, 3) Mean±SD age between 10 and 18 years, 4) includes a follow-up, 5) Mean±SD of Cobb $\geq 10^{\circ}$, 6) radiographic or ultrasonic imaging of the spine at follow-up. For the full-text screening, one more criterion was assessed: 7) reporting on the prediction of curve progression.

3.3.3. Exclusion Criteria

Exclusion criteria were: 1) congenital, neuromuscular, syndromic, and secondary diagnoses of scoliosis, 2) using only specialized parameters as predictors (such as laboratory test results including level of enzymes or hormones, brain-function, genetic, electromyography tests), 3) using only non-radiographic imaging to assess curve progression such as in vitro studies, engineering methods (simulations, finite element method, and machine learning), magnetic resonance imaging (MRI), computed tomography (CT), and questionnaires, 4) having scoliosis surgery at baseline, 5) samples with less than 10 participants, 6) case reports, conference proceedings, commentary and opinion letters, and reviews, and 7) non-English studies. These criteria were used for all phases.

3.3.4. Information Sources and Search Strategy

The search terms were determined after consultation with scoliosis experts (EP and EL) and a medical librarian. Five databases including MEDLINE (OVID), EMBASE (OVID),

Cochrane Central Register of Controlled Trials (CENTRAL), Web of Science (Web of knowledge), and CINAHL (EBSCO) were selected. The search was performed on the same day, from inception to October 18th, 2018, for all databases. The search strategy developed for Medline by the librarian was adjusted for each database (Appendix 3.1). Zotero was used to manage the references [12].

3.3.5. Study Selection

Five independent reviewers were involved in the abstract and full-text screening stages. The primary reviewer (MK) screened all the articles for the first round paired with one of the four other reviewers (AC, JW, DH, EP). For the second round of screening, again the main reviewer (MK) assessed all the articles, and each full-text article was screened by one of the four other reviewers (AC, JW, DH, EP). Three students and two experts contributed for the study selection part. All students were graduate trainees working on scoliosis field with more than 2 years of experience. The two experts (DH, EP) had more than 10 years of experience working on scoliosis. The reviewers were trained prior to screening. For the training, the criteria were explained clearly, and the reviewers were asked to screen the first 10 articles. Then, the votes were compared and ensured that all the reviewers were confident to continue the screening. The Covidence software (Veritas Health Innovation, Melbourne, Australia) which is specifically devised to conduct systematic reviews was used for the abstract screening. Three options were presented to reviewers for each decision about each article: yes, no, unsure. For the full-text screening, Microsoft Excel forms were designed to capture screening decisions. Reviewers decided which articles to include according to the selection criteria and chose included, excluded or unclear in separate forms for each reviewer. The articles that were excluded along with the reasons have been provided in the Appendix 3.2.

3.3.6. Data Collection Process and Data Items

The eligible articles were included for the full-text extraction phase. An extraction form devised based on the key characteristics including study origin, study design, baseline

description of the population, descriptions of the potential prognostics predictors, descriptions of the outcomes as curve progression, and the reports on the statistical analysis and results. Six reviewers contributed for the extraction. The primary reviewer (MK) extracted the information for all the articles and the rest of reviewers (DH, JW, EG, KN, CC) helped for the second round of extraction. The results were compared for reliability assessment. For the extraction phase, two more reviewers were added to the same group with two of them (EG, KN) as graduate (over two years of experience in scoliosis) and one undergraduate (CC, as a novice researcher) students. The same training procedure was followed for the extraction phase.

3.3.7. Risk of Bias of Studies

The Quality in Prognostic Studies (QUIPS) tool [13] was used for quality appraisal. As recommended, six different domains were appraised and tailored for risk of bias assessment (ROB) for this study (Appendix 3.3). The QUIPS domains include study participation, study attrition, prognostic factor measurement, outcome measurement, study confounding, and statistical analysis and results. For each domain a list of key characteristics determined by consensus of the authors were defined and the questions of each domain were scored according to the available information (Appendix 3.3). An overall score as high, moderate, and low risk of bias was assigned to each article. To determine the overall score for each article, the questions under each domain were prioritized based on their importance for our study and reviewers relied most on the information provided for those specific questions (the questions highlighted in Appendix 3.3, as red, yellow, and green illustrate on the importance, respectively) to assign the overall score for each article. Articles that were missing the information about those questions achieved lower scores.

3.3.8. Summary Measures

Results were summarized for each prognostic factor of three curve progression definitions over two types of follow-up intervals.

3.3.9. Synthesis of Results

Studies were classified based on the curve progression definition used and the follow-up interval duration. The curve progression categories were: i) increase in curve magnitude $\geq 5^{\circ}$, ii) curve progression to $\geq 30^{\circ}$ or requiring surgery, iii) change in Cobb angle reported as $^{\circ}$ /year. A short-term follow-up interval was defined as measuring two radiographs captured within \leq 1-year interval. A long-term follow-up interval was defined as any two radiographs captured at > 1-year interval. Levels of evidence were also qualified as strong, moderate, limited, unclear, conflicting, and no evidence based on the risk of bias and the consistency of the research findings for each article (adapted from Cornelius, et al.) (Table 3.1) [14].

 Table 3.1 Rules for determining of the level of evidence when formulating summary statements about whether a variable was predictive of progression or not.

Level of Evidence	Description
Strong	Consistent results (≥80%) from at least 2 high-quality studies
Moderate	One high-quality study and consistent findings (≥80%) in 1 or more low-quality studies
Limited	Findings in 1 high-quality cohort or consistent results (≥80%) among low-quality studies
Unclear	Findings in only one study among low-quality studies
Conflicting	Inconsistent results irrespective of study quality
No	No study identified

Adapted from Cornelius et al. [14]

3.3.10. Additional Analyses

The inter-rater reliability was analyzed by using Kappa coefficients for the decisions during the full-text screening and for the Risk of bias (RoB) assessment [15].

3.4. Results

The PRISMA flowchart (Figure 3.1) shows a total of 1448 articles were acquired after duplicated articles were removed.



Figure 3.1 The PRSIMA flow chart

After screening, 58 full-text articles were included. Table 3.2 presents the included articles with their year of publication, country of origin, sample sizes and risk of bias score.

Author	Country	Sample size	QUIPS		QUIPS 3	QUIPS 4	QUIPS 5	QUIPS 6	Overall RoB
Cheung et al, 2004 [16]	The Netherlands	30	M	N/A	L	L	Н	L	L
Davis et al, 2018 [17]	USA	56	L	N/A	L	L	L	М	L
Hung, 2005 [4]	China	324	М	N/A	L	L	Н	L	L
Lara et al, 2017 [18]	USA	295	L	Н	L	L	М	М	L
Mao et al, 2016 [19]	Germany	95	L	N/A	L	М	М	L	L
O'Neill et al, 2005 [20]	USA	276	L	N/A	L	L	М	L	L
Ohrt-Nissen et al, 2016 [21]	Denmark	63	L	N/A	L	L	L	L	L
Sitoula et al, 2015 [22]	USA	161	L	N/A	L	L	L	М	L
Sun et al, 2013 [23]	China	68	L	N/A	L	М	L	L	L
Zaina et al, 2017 [24]	Italy	351	L	N/A	М	L	L	М	L
Duval- Beaupere, 1992 [25]	France	262	Н	N/A	М	Н	Н	Н	Н
Konieczny et al, 2017 [26]	Germany	72	L	Н	Н	М	Н	Н	Н
Kotwicki et al, 2008 [27]	Poland	158	Н	N/A	L	М	Н	Н	Н
Aulisa et al, 2009 [28]	Italy	50	Н	N/A	L	М	Н	М	М
Aulisa et al, 2014 [29]	Italy	522	М	N/A	М	М	М	М	М
Bohl et al, 2014 [30]	USA	34	Н	N/A	М	М	М	М	М
Bunnell, 1986 [31]	USA	123	М	N/A	М	Н	М	Н	М
Charles et al, 2017 [32]	France	372	М	N/A	М	М	М	М	М
Cheung et al, 2018 [33]	China	318	L	N/A	L	М	L	Н	М
Cheung, et al, 2018 [34]	China	513	L	N/A	М	М	М	М	М
D'Amato et al, 2001 [35]	USA	102	М	N/A	М	М	М	М	М
Danielsson et al, 2007 [36]	Sweden	92	L	L	L	М	М	М	М
Fang et al, 2015 [37]	China	32	Н	N/A	М	М	Н	М	М
Gepstein et al, 2002 [38]	Israel	122	М	N/A	М	М	L	М	М
Goldberg et al, 2001 [39]	Ireland	153	L	N/A	М	М	М	М	М
Goodbody et al, 2016 [40]	USA	182	L	N/A	М	L	L	Н	М
Guo et al, 2012 [41]	China	60	М	N/A	L	М	М	М	М
Guo et al, 2014 [42]	China	38	М	Н	L	М	М	М	М
Karol et al, 1993 [43]	USA	210	L	Н	М	L	М	М	М
Karol et al, 2016 [44]	USA	168	М	L	М	М	М	М	М

 Table 3.2 Included articles and RoB assessments

Karol, 2001	USA	112	L	N/A	L	L	М	Н	М
[45] Katz et al, 2001	USA	51	M	Н	M	M	M	M	M
[46] Katz et al, 2010	USA	100	L	H	M	L	M	M	M
[47] Kuroki et al,	Japan	31	L	N/A	M	M	M	M	M
2015 [48] Lam et al, 2013	-								
[49] LeBlanc, 1998	Hong Kong	294	M	N/A	M	М	М	L	M
[50] Lee et al, 2012	Canada	146	М	N/A	М	М	Н	М	М
[51] Nault et al,	China	2308	М	Н	М	М	М	М	М
2013 [7]	Canada	37	L	N/A	М	М	М	М	М
Nault et al, 2014 [52]	Canada	133	М	N/A	М	М	L	М	М
Pasquini et al, 2016 [53]	Italy	67	М	М	М	Н	М	М	М
Peterson et al, 1995 [10]	Sweden	159	L	М	М	М	М	L	М
Ploumis et al, 2018 [54]	USA	73	L	N/A	L	L	L	М	М
Ryan et al, 2007 [55]	USA	62	L	N/A	М	М	L	L	М
Shi et al, 2016 [56]	China	200	L	N/A	М	М	М	М	М
Sun et al, 2010 [57]	China	142	L	N/A	М	М	М	М	М
Tan et al, 2009 [9]	Singapore	186	L	М	М	М	Н	L	М
Thompson et al, 2017 [58]	USA	168	L	Н	М	L	М	М	М
Trivedi and Thomson, 2001 [59]	USA	42	М	М	М	М	Н	L	М
Vijvermans et al, 2004 [60]	Belgium	151	L	N/A	М	Н	М	М	М
Weiss, 1992 [61]	Germany	118	М	N/A	М	М	М	Н	М
Wu et al, 2011 [62]	Canada	35	М	N/A	М	Н	М	М	М
Yamauchi et al, 1988 [63]	Japan	122	Н	N/A	М	М	М	М	М
Yip et al, 2016 [64]	Hong Kong	513	L	N/A	М	М	М	L	М
Ylikoski, 1993 [65]	Finland	110	L	N/A	L	М	L	Н	М
Yrjonen and Ylikoski, 2006 [66]	Finland	80	L	N/A	L	М	М	М	М
Yrjonen et al, 2006 [67]	Finland	66	L	N/A	М	L	L	М	М
Zhang et al, 2015 [68]	China	89	М	N/A	М	М	L	L	М
Zhu et al, 2017 [69]	China	54	L	N/A	L	М	М	L	М

Abbreviations: L: Low, M: Moderate, H: High, N/A: not applicable, QUIPS1: Study Participation, QUIPS2: Study Attrition, QUIPS3: Prognostic Factor Measurement, QUIPS4: Outcome, QUIPS5: Study Confounding, QUIPS6: Statistical Analysis and Reporting

3.4.1. RoB Within Studies

Of the 58 included articles, only 10 (17%) showed low RoB, 45 (77%) presented moderate RoB, and 3 (5%) presented high RoB [16-18] (Table 3.2). The study participants RoB domain had the greatest number of articles with low RoB 32 (55%). The confounding and statistical analysis-reporting domains had the most articles presenting high RoB with 10 (17%) and 9 (16%), respectively.

3.4.2. Study Characteristics

Most studies were retrospective [35 (60%)] and used an unknown [35 (60%)] or consecutive [18 (31%)] recruitment strategy (Table 3.3). Of the 58 included articles, 34 (59%) were cohort, 17 (29%) case series, 5 (9%) case-control, and 2 (3%) randomized controlled trial (RCT) studies. Most articles [31 (53%)] reported follow-up \geq 2 years. For 8 (14%) articles, completion of bracing or having an age of at least 15 years old were used to determine the time for the follow-up.

		St	tudy Characteristic	28				
Recruitment timeline n (%) of articles		Recruitment Strategy n (%) of articles		Type of Study n (%) of articles		Range of the Mean Clinical FU duration n of articles (%)		
Prospective	15	Consecutive	18 (31%)	Cohort	34	<2 years	5 (9%)	
1	(26%)		10 (5170)		(59%)	\geq 2 years	31 (5%)	
	35	Randomized	3 (5%)	Case-series	17 (29%)	Skeletal maturity	11 (19%)	
Retrospective	55 (60%)	Non- randomized	1 (2%)	Case-control	5 (9%)	Beyond skeletal maturity	1 (2%)	
Unknown	8 (14%)	nknown	Convenience	1 (2%)	Randomized controlled trial	2 (3%)	Others (completion of bracing or up to age at least 15 years old)	8 (14%)
		Unknown	35 (60%)			Not specified	2 (3%)	
	Sample Characteristics							
	Total number of participants 11,134			is, number of pants	8			

Table 3.3 Study characteristic and baseline information summary

		N of participants %)	N of participants (n of articles, %)		N of participants (n of articles, %)		
Range of the mean ages for participants (n of articles, %)				0	1334 (38, 66%)		
11.8-14.9 (55, 95%)				1	339 (41, 71%)		
Gender	N of participants (n of articles, %)	Pre-menarche 1988 (17, 29%)		2	463 (38, 66%)		
Male	1980 (57, 98%)	Post-menarche	2130 (26, 45%)	3	302 (15, 26%)		
Female	8986 (57, 98%)	Both Pre- and post- menarche reported	2821 (16, 28%)	4	273 (11, 19%)		
	Curve classification,		31, 53%	5	98 (5, 9%)		
N of participants	s (n of articles, %)	Not specified	51, 5570	Others	136 (3, 5%)		
Thoracic	1269 (37, 62%)	Treatment, number of participants N of participants (n of articles, %)		Not specified	13, 22%		
Thoracolumbar	349 (34, 59%)	Observation /non- treated	1582 (12, 21%)	Range of the mean Cobb angle (°) N of participants (n of articles			
Lumbar	361 (20, 34%)	Brace	4642 (42, 72%)				
Thoracolumbar/Lumbar	617 (12, 21%)	Surgery	475 (20, 34%)				
Double	873 (27, 43%)	Brace+ surgery	18 (1, 2%)	Initial,			
Triple	19 (4, 7%)	Exercise	118 (2, 3%)	No of articles (%)	22-47 (54, 93%)		
Others	1213 (6, 10%)	Electrical simulation 30 (1, 2%)		Follow- up, No	15-49 (16, 28%)		
Not specified	16, 28%	Not specified	8, 14%	articles (%)	13-77 (10, 2070)		

3.4.3. Sample Characteristics

The studies included a total of 11,134 participants (Table 3.3). Most participants were female 8986 (81%) (Table 3.3). The number of pre-and post-menarche female participants were 1988 vs 2130 participants reported in 17 (29%) vs 26 (45%) articles. Most participants had a thoracic curve type 1269 (37, 62%). Risser 0 was the most common skeletal maturity grade reported across the studies at baseline (38, 66%). The mean initial Cobb angles ranged 22 to 47 degrees.

3.4.4. Reviewer Agreement Analyses

The percentages of agreement were 84% and 59% for full-text screening and RoB ratings, respectively. The Kappa coefficients were 0.55 and 0.77 for full-text screening and RoB assessments, respectively.

3.4.5. Synthesis of the Results

Most studies defined curve progression as an increment of curve magnitude $\geq 5^{\circ}$ [$\geq 5^{\circ}$ (26 studies, 45%), $\geq 6^{\circ}$ (17, 29%), or $\geq 10^{\circ}$ (3, 5%)] (Appendix 3.4). A second group of studies defined progression as curve magnitude increases to $\geq 30^{\circ}$ or requiring surgery: $\geq 30^{\circ}$ (2, 3%), $\geq 40^{\circ}$ (12, 21%), $\geq 50^{\circ}$ (5, 9%), and surgery (13, 22%) (Appendix 3.5). Four articles (7%) reported on change in Cobb angle as °/year (Appendix 3.6). Thirteen articles (22%) mentioned more than one progression criterion. Of the 58 articles, 20 (34%) articles studied progression over short-term follow-up interval and 34 (59%) over long-term intervals (Appendix 3.4-3.6). For 4 (7%) articles the follow-up interval studied was unclear (Appendix 3.4 and 3.5).

I) Predictive factors for curve progression $\geq 5^{\circ}$ over short- and long-term follow-up intervals

Forty-six articles used $\geq 5^{\circ}$ as the criteria of progression (Appendix 3.4). To predict progression $\geq 5^{\circ}$, a total of 32 predictive factors were studied over short-term and 87 over long-term follow-up intervals (Table 3.4).

Level of Evidence	Number of studies with ROB predictor+/ nonpredictor-)		Predictor/Non- predictor	Follow-up Term	
Strong	2+/0 Low, 4+/2- Moderate, 1+/0 Moderate*	High Cobb angle*	Р	Short-term, unclear*	
Strong	2+/0 Low, 2+/1- Moderate	Pre-menarche status	Р	Short-term	
Limited	3+/0 Moderate	Apex location (at higher vertebral levels)	Р	Short-term	
Limited	1+/0 Low, 0/1- Moderate	High spinal growth	Р	Short-term	
Limited	1+/0 Low	Low bone mineral density (BMD) of the femoral neck in concave-side hip	Р	Short-term	
Limited	1+/0 Low	Low BMD of the femoral neck in convex-side hip	Р	Short-term	

Table 3.4 Levels of evidence summary statements about whether each predictive factor investigated does or does not predict curve progression $\geq 5^{\circ}$ over short-term/long-term follow-up intervals

Limited	1+/0 Low	Low BMD of the Trochanter in concave-	Р	Short-term
Liiniteu	-	side hip	r	
Limited	1+/1- low, 6+/0 Moderate, 0/2- Moderate*	Low stages of Risser sign* (0-3)	Р	Short-term, unclear*
Unclear	1+/0 Moderate	Body morphology (Anthropometric measurements and morphologic classification) #	Р	Short-term
Unclear	1+/0 Moderate	High height	Р	Short-term
Unclear	1+/0 Moderate	Initial apex lateral deviations	Р	Short-term
Unclear	1+/0 Moderate	Low ASIS circumference (mm)	Р	Short-term
Unclear	1+/0 Moderate	Low hip width (mm)	Р	Short-term
Unclear	1+/0 Moderate	Low proportion of bone volume of body	Р	Short-term
Unclear	1+/0 Moderate	Maximal apex lateral deviations	Р	Short-term
Unclear	1+/0 Moderate	Maximal Cobb angle	Р	Short-term
Unclear	1+/0 Moderate	Open Triradiate cartilage (TRC)	Р	Short-term
Strong	0/2- Low, 2+/3- Moderate	Curve pattern	NP	Short-term
Moderate	0/1- Low, 0/1- Moderate, 0/1- High	Age at menarche	NP	Short-term
Moderate	0/1- Low, 0/1- Moderate, 0/1- High	Body mass index (BMI)	NP	Short-term
Limited	0/1- Low	BMD of the femoral neck on the non- dominant side	NP	Short-term
Limited	0/1- Low	BMD of the trochanter in convex-side hip	NP	Short-term
Limited	0/1- Low, 0/1- Moderate	Family history of scoliosis	NP	Short-term
Limited	0/2- Moderate	Kyphosis	NP	Short-term
Limited	0/1- Low, 0/1- Moderate	Weight	NP	Short-term
Unclear	0/1- Moderate	Apical vertebral rotation	NP	Short-term
Unclear	0/1- Moderate	Initial apex lateral deviations	NP	Short-term
Unclear	0/1- Moderate	Lordosis	NP	Short-term
Unclear	0/1- Moderate	Lumbosacral transitional abnormalities	NP	Short-term
Unclear	0/1- Moderate	Spinal balance (alignment of C7)	NP	Short-term
Unclear	0/1- Moderate	Rib hump	NP	Short-term
Conflicting	1+/1- Low	BMD in the spine	P/NP	Short-term
Conflicting	2+/2- Moderate	Gender	P/NP	Short-term
Conflicting	1+/-1 Low, 4+/3- Moderate	Initial age	P/NP	Short-term
Limited	1+/0 Low, 3+/2- Moderate	High apical vertebral rotation	Р	Long-term
Limited	3+/0 Moderate	High torsion	Р	Long-term
Limited	1+/0 Low, 0/1- Low*	High weight*	Р	Long-term, Unclear*
Limited	1+/0 Low, 0/1- Moderate	Low curve flexibility	Р	Long-term
Limited	2+/0 Moderate	Low slenderness (height/width ratio of regional T1-L5)	Р	Long-term
Limited	2+/0 Moderate	Open Triradiate cartilage (TRC)	Р	Long-term
Limited	1+/0 Low, 3+/4- Moderate	Pre-menarche status	Р	Long-term
Limited	2+/0 Moderate	Spinopelvic inclination (T1 \leq 3.5°, T9 \leq 6.5°)	Р	Long-term
Unclear	1+/0 Moderate	Apical intervertebral rotation (>2.4 °)	Р	Long-term
	1+/0 Low *, 1+/0	BMI (high [BMI (85th percentile) and	Р	Long-term,
Unclear	Moderate	low (<20 th percentile)]	r	unclear*
Unclear	1+/0 Moderate	High axial angle of the plane in which the Cobb angle is maximal	Р	Long-term
Unclear	1+/0 Moderate	High axial rotation of lower junctional vertebra	Р	Long-term
Unclear	1+/0 Moderate	High Cobb angle at brace weaning	Р	Long-term
Unclear	1+/0 Moderate	High sacral takeoff angle (>5°)	Р	Long-term
Unclear	1+/0 Moderate	Higher coronal rotation of lower junctional vertebra	Р	Long-term
Unclear	1+/0 Moderate	Iliac crest height difference (>1 cm)	Р	Long-term
Unclear	1+/0 Moderate	Low bone mineral density (BMD) of	Р	Long-term
Unclear	1+/0 Moderate	Low bone quality (broadband ultrasound attenuation (BUA))	Р	Long-term
--------------------	--------------------------------	--	----------	------------------------
Unclear	1+/0 Moderate	Low bone quality (stiffness index (SI))	Р	Long-term
Unclear	1+/0 Moderate	Low bone quality (velocity of sound (VoS))	Р	Long-term
Unclear	1+/0 Moderate	Low kyphosis angle	Р	Long-term
Unclear	1+/0 Moderate	Lower sagittal intervertebral rotation at the apical level	Р	Long-term
Unclear	1+/0 Moderate	Lower slenderness (height/depth ratio of local L4)	Р	Long-term
Unclear	1+/0 Moderate	Lower slenderness (height/depth ratio regional T1–L5)	Р	Long-term
Unclear	1+/0 Moderate	Lower slenderness (height-depth ratio of local T12)	Р	Long-term
Unclear	1+/0 Moderate	Lumbar pelvic relationship (LPR) (>12°)	Р	Long-tern
Unclear	1+/0 Moderate	Vertebral translations (at apical level)	Р	Long-tern
Moderate	1+/3- Low, 3+/16- Moderate	Age	NP	Long-tern
Moderate	0/1- Low, 0/3- Moderate	Age at menarche	NP	Long-tern
Moderate	0/1- Low, 0/2- Moderate	Apex location	NP	Long-tern
Moderate	1+/2- Low, 2+/11- Moderate	Curve pattern	NP	Long-tern
Moderate	0/1- Low, 0/2- Moderate	Height (standing)	NP	Long-tern
Limited	0/2- Moderate	3D Cobb angle	NP	Long-tern
Limited	1+/3- Low, 7+/14- Moderate	Cobb angle	NP	Long-tern
Limited	1+/3- Low, 3+/3- Moderate,	Gender	NP	Long-tern
Unclear	0/1- Moderate	Age at bracing	NP	Long-tern
Unclear	0/1- Moderate	Age at maturity	NP	Long-tern
Unclear	0/1- Moderate	Apical disk wedging	NP	Long-tern
Unclear	0/1- Moderate	Apical vertebral translation (AVT) ratio (AVT-R)	NP	Long-tern
Unclear	0/1- Moderate	Apical vertebral translation -lumbar (AVT-L)	NP	Long-tern
Unclear	0/1- Moderate	Apical vertebral translation -thoracic (AVT-T)	NP	Long-tern
Unclear	0/1- Moderate	Axial intervertebral rotation at the apical level	NP	Long-tern
Unclear	0/1- Moderate	Axial rotation of upper junctional vertebra	NP	Long-tern
Unclear	0/1- Moderate	Cobb ratio (thoracic Cobb/ lumbar Cobb)	NP	Long-tern
Unclear	0/1- Moderate	Coronal decompensation	NP	Long-tern
Unclear	0/1- Moderate	Coronal intervertebral Rotation at the apical level	NP	Long-tern
Unclear	0/1- Moderate	Coronal rotation of upper junctional vertebra	NP	Long-tern
Unclear	0/1- Moderate	Difference of inclination between the proximal and the distal vertebra	NP	Long-tern
Unclear	0/1- Moderate	Disc index (wedging of the disk spaces in the curve)	NP	Long-tern
Unclear	0/1- High	Educational level of the parents	NP	Long-tern
Unclear	0/1- High	Educational level of the patients	NP	Long-tern
Unclear	0/1- High	Family history of scoliosis	NP	Long-tern
Unclear	0/1- Moderate	Femoral head height difference (mm)	NP	Long-tern
Unclear	1-/0 Moderate	Height (sitting)	NP	Long-tern
Unclear	0/1- High	Hours of self-training per week Hours of sports per week outside of the	NP	Long-tern
Unclear	0/1- High	school	NP	Long-tern
Unclear Unclear	0/1- Moderate 0/1- Moderate	Lordosis Low curve disk wedging	NP NP	Long-term Long-term

Unclear	0/1- High	Nutritional behavior	NP	Long-term
Unclear	0/1- Moderate	Pelvic incidence (PI)	NP	Long-term
Unclear	0/1- Moderate	Relative apical distance (RAD)	NP	Long-term
Unclear	0/1- Moderate	Rib-vertebra angle - concave side (RVA-cv)	NP	Long-term
Unclear	0/1- Moderate	Rib-vertebra angle- convex side (RVA- cx)	NP	Long-term
Unclear	0/1- Moderate	Rib-vertebra angle difference (RVAD)	NP	Long-term
Unclear	0/1- Moderate	Sacral slope (SS)	NP	Long-term
Unclear	0/1- Moderate	Sagittal rotation of lower junctional vertebra	NP	Long-term
Unclear	0/1- Moderate	Sagittal rotation of upper junctional vertebra	NP	Long-term
Unclear	0/1- Moderate	Slenderness (height/depth ratio of local T6)	NP	Long-term
Unclear	0/1- Moderate	Slenderness (height/width ratio of local T12)	NP	Long-term
Unclear	0/1- Moderate	T12-L1 intervertebral rotation	NP	Long-term
Unclear	0/1- Moderate	Trunk shift	NP	Long-term
Unclear	0/1- High	Type of school bag	NP	Long-term
Unclear	0/1- Moderate	Upper curve disk wedging	NP	Long-term
Unclear	0/1- Moderate	Upper curve intervertebral rotation	NP	Long-term
Unclear	0/1- Moderate	Vertebral tilt angles (VTA) for lumbar- inferior [VTA-L-I])	NP	Long-term
Unclear	0/1- Moderate	Vertebral tilt angles (VTA) for lumbar– superior [VTA-L-S	NP	Long-term
Unclear	0/1- Moderate	Vertebral tilt angles (VTA) for thoracic- inferior [VTA-T-I]	NP	Long-term
Unclear	0/1- Moderate	Vertebral tilt angles (VTA) for thoracic- superior [VTA-T-S]	NP	Long-term
Unclear	0/1- Moderate	Vertebral translations (at first thoracic level)	NP	Long-term
Conflicting	1+/1- Moderate	Apical vertebral body wedging	P/NP	Long-term
Conflicting	1+/1- Moderate	Low slenderness (height/width ratio of local L4)	P/NP	Long-term
Conflicting	1+/1- Moderate	Low slenderness (height/width ratio of local T6)	P/NP	Long-term
Conflicting	+1/1-Moderate	Pelvic tilt (PT)	P/NP	Long-term
Conflicting	2+/2- Low, 7+/8- Moderate	Risser sign	P/NP	Long-term
Conflicting	1+/1- Moderate	Spinal growth	P/NP	Long-term
Conflicting	1+/1- Moderate	Vertebral translation	P/NP	Long-term

* indicates the unclear follow-up interval reported for that specific predictive factor. P: predictor, NP: non-predictor. "The parameters considered under body morphology were: Wrist thickness (mm), Deltoid circumference (mm), Under breast circumference (mm), Thigh circumference (mm), Arm circumference (mm), Fat (%), Muscle volume (%), Head, face, and neck ectomorphism score, Head, face, and neck mesomorphism score, Thorax ectomorphism score, Thorax mesomorphism score, Superior limbs ectomorphism score, Superior limbs mesomorphism score, Forearm length (mm)-All were significantly related with progression.

There were unclear follow-up intervals for 4 predictive factors.

The summary statements for prediction of progression $\geq 5^{\circ}$ over a short-term interval are:

- Strong evidence that large Cobb angle and pre-menarche status were predictive.
- Strong evidence that curve pattern was not predictive.
- Moderate evidence that age at menarche for girls and BMI were not predictive.

• Limited evidence that apex location (at higher vertebral levels), high spinal growth, low BMD of the femoral neck in the hips and the trochanter in the concave-side hip, and low Risser sign were predictive.

The summary statements for prediction of progression $\geq 5^{\circ}$ over a long-term interval are:

- Moderate evidence that age, age at menarche, curve pattern, apex location, and height were not predictive.
- Limited evidence that high apical/axial vertebral rotation, high torsion, high weight, low curve flexibility, low slenderness (height/width ratio of local T6, of regional T1-L5), open Triradiate cartilage (TRC), pre-menarche status, and spinopelvic inclination (T1≤ 3.5°, T9≤ 6.5°) were predictive.

II) Predictive factors for curve progression to a magnitude $\geq 30^{\circ}$ or requiring surgery over short- and long-term follow-up intervals

Thirty-two articles used progression $\geq 30^{\circ}$ or surgical thresholds as a progression criterion (Appendix 3.5). There were 22 predictive factors studied over short-term, 13 for long-term and 4 for unclear follow-up intervals (Table 3.5).

Level of Evidence	Number of studies with ROB (predictor+/ nonpredictor-)	Parameters	Predictor/Non- predictor	Follow-up Term
Moderate	1+/0 Low, 2+/0 Moderate	Pre-menarche status	Р	Short-term
Limited	4+/0 Moderate, 1+/0 Moderate*	High Cobb angle*	Р	Short-term, Unclear*
Limited	2+/0 Moderate	Low age	Р	Short-term
Limited	1+/0 Low	Sanders stages of maturity (SS1, SS2, and SS3)	Р	Short-term
Unclear	1+/0 Moderate	Low areal bone mineral density (aBMD) at proximal of both femurs (zBMD < -1)	Р	Short-term
Unclear	1+/0 Moderate	Gender (female)	Р	Short-term
Unclear	1+/0 Moderate	Low height (< 153.9 cm)	Р	Short-term
Unclear	1+/0 Moderate	Low bone morphometry (cortical bone area, CrAr/ total bone area, TotAr (%)) at non- dominant distal radius	Р	Short-term
Unclear	1+/0 Moderate	Low bone morphometry (cortical thickness, CtTh) at non-dominant distal radius	Р	Short-term

Table 3.5 Levels of evidence summary statements about whether each predictive factor investigated does or does not predict curve progression when attempting to predict progression $>30^{\circ}$ or surgery

Unclear	1+/0 Moderate	Low Olecranon stages of maturity	Р	Short-term
Unclear	1+/0 Moderate	Low Risser sign (0-3)	Р	Short-term
		Low trabecular bone micro-architecture	-	Short term
Unclear	1+/0 Moderate	(trabecular thickness, Tb.Th) at non-dominant distal radius	Р	Short-term
Unclear	1+/0 Moderate	Low volumetric BMD (volumetric density of cortical bone measured at distal radius, Dcort) at non-dominant distal radius	Р	Short-term
Unclear	1+/0 Moderate	Open Triradiate cartilage (TRC)	Р	Short-term
Unclear	0/1- Moderate	Bone morphometry (cortical periosteal perimeter, (CtPm)) at non-dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Bone morphometry (total bone area, Tot Area (cm ²)) at non-dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Bone morphometry (trabecular bone area, Tb.Ar (cm ²)) at non-dominant distal radius	NP	Short-term
Unclear	0/1- High*	Hip joint asymmetries	NP	Unclear*
Unclear	0/1- Moderate	Trabecular bone micro-architecture (bone volume over total volume, BV/TV) at non- dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Trabecular bone micro-architecture (trabecular number, TrN) at non-dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Trabecular bone micro-architecture (trabecular spacing, Tb.Sp) at non-dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Volumetric BMD (overall volumetric density measured at distal radius, Dtot) at non- dominant distal radius	NP	Short-term
Unclear	0/1- Moderate	Volumetric BMD (volumetric density of trabecular bone measured at distal radius, Dtrab) at non-dominant distal radius	NP	Short-term
Moderate	2+/1- Low, 6+/0 Moderate	High Cobb angle	Р	Long-term
Limited	1+/0 Low	Apex location (curve apex cephalad to T10)	Р	Long-term
Limited	1+/0 Low	High weight	Р	Long-term
Limited	1+/0 Low, 2+/3- Moderate	Low age	Р	Long-term
Unclear	1+/0 Moderate	Body mass index [BMI (>85 th percentile) and <20 th percentile)]	Р	Long-term
Unclear	1+/0 Moderate	High Cobb angles at brace weaning	Р	Long-term
Unclear	1+/0 Moderate	Low grades of distal Radius and Ulna (DRU) classification (R7/U5)	Р	Long-term
Limited	1+/2- Low, 0/1- Moderate	Gender	NP	Long-term
Unclear	0/1- Moderate	Age at menarche	NP	Longterm
Unclear	0/1- Moderate	Initial puberty status	NP	Long-term
Conflicting	1+/1- Low, 2+/1- Moderate, 1+/1- Moderate*	Curve Pattern	P/NP	Long-term, Unclear*
Conflicting	2+/2- Moderate	Menarche status	P/NP	Long-term
Conflicting	1+/2- Low, 3+/0 Moderate, 0/1- Moderate*	Risser sign	P/NP	Long-term, Unclear*

* indicates the unclear follow-up interval reported for that specific predictive factor. Abbreviations: #P: predictor, NP: non-predictor, #bone parameters: zBMD: z score of bone mineral density, CrAr: cortical bone area, TotAr (%): total bone area; CtTh: cortical thickness, Tb.Th: trabecular thickness, Dcort: volumetric density of cortical bone measured at distal radius, TRC: triradiate cartilage, CtPm: cortical periosteal perimeter, Tot Area: total bone area, Tb.Ar: trabecular bone area, BV/TV: bone volume over total volume, TrN: trabecular number, Tb.Sp: trabecular spacing, Dtot: overall volumetric density measured at distal radius, Dtrab: volumetric density of trabecular bone measured at distal radius. The summary statements for prediction of progression $\geq 30^{\circ}$ or requiring surgery over a short-term interval are:

- Moderate evidence that pre-menarche predicted progression.
- Limited evidence that large Cobb angle, low age, and Sanders stages of maturity (SS1-SS3) predicted progression.

The summary statements for prediction of progression $\geq 30^{\circ}$ or requiring surgery over a long-term interval are:

- Moderate evidence that large Cobb angle predicted progression.
- Limited evidence that low age, high weight, and apex location cephalad to T10 predicted progression.

III) Predictive factors for *curve progression reported as change in Cobb angle (°/year) over short- and long-term follow-up intervals*

The degree of curve progression per year was reported in only 4 articles (Appendix 3.6). Nine predictors were studied for prediction of progression as change in Cobb angle (°/year) over short-term and 3 over long-term intervals (Table 3.6).

Table 3.6 Levels of evidence summary statements about whether each predictive factor
investigated does or does not predict curve progression defined as change in Cobb angle
(°/vear) over short-term and long-term follow-up intervals

Level of Evidence	Number of studies with ROB (predictor+/ nonpredictor-)	Parameters	Predictive+/Non- predictive	Follow-up Term
Limited	2+/0 Moderate, 1+/0 High	Curve pattern (Thoracic, double, triple curves)	Р	Short-term
Limited	1+/0 High, 1+/0 Moderate	High Cobb angle	Р	Short-term
Limited	2+/0 Moderate	Low age	Р	Short-term
Unclear	1+/0 Moderate	Growth velocity ($\geq 4 \text{ cm/year}$)	Р	Short-term
Unclear	1+/0 Moderate	High growth of the T4-L4 segment	Р	Short-term
Unclear	1+/0 High	High rib hump	Р	Short-term
Unclear	1+/0 Moderate	Pre-menarche	Р	Short-tern
Unclear	1+/0 High	State II of maturity (between start of puberty and menarche)	Р	Short-tern
Conflicting	1+/1- Moderate	Risser sign	P/NP	Short-tern
Unclear	1+/0 Moderate	Distal Radius and Ulna classification (DRU) (R7/U5 stages)	Р	Long-term
Unclear	1+/0 Moderate	High arm span	Р	Long-tern
Unclear	1+/0 Moderate	High height	Р	Long-tern

[#]Abbreviation: P: predictor, NP: non-predictor

For progression defined as change in Cobb Angle (°/year) the summary statements are:

• Limited evidence that curve pattern (thoracic, double, triple curves), large Cobb, and low age predicted progression over short-term follow-up intervals.

3.5. Discussion

Much of the evidence about prediction of curve progression is still limited or unclear and conflicting evidence is still present. The most cited parameters among studies were age, Cobb, Risser sign, menarche status, and gender. This emphasizes that these classical factors are still the most popular to use for prediction of progression for clinicians, which is probably due to their importance and feasibility to measure. However, our results showed that, of these factors, only large Cobb and premenarche were identified as predictive with moderate to strong evidence. The list of predictors with supporting evidence varied for each progression criteria and follow-up interval suggesting that research should use clear definitions and specific follow-up intervals.

The number of studies was insufficient to reach higher levels of evidence for most of the variables. The reason may be that some of these factors such as bone quality parameters have only recently been considered as potential prognostic predictors of progression with promising results from some studies [4], [5], [23]. The same may apply to 3D parameters of spine [7], [37], [52]. Those may also be important as a curve progression outcome since they could reflect the true 3D nature of the AIS.

In the review by Noshchenko et al., it was not explored whether predictors differed depending on progression definitions and follow-up durations which may limit our ability to rely on their results [8]. In their review, most parameters that were associated with progression were related to complex genetic, brain function and laboratory tests. However, clinicians require more convenient and simple prognostic tools if such clinical parameters can achieve good prediction accuracy. Nevertheless, our review shows that the current

literature suffers from lack of high quality studies for predicting progression which is consistent with the results of Noshchenko et al [8].

To standardize comparison on the predictive value of variables and avoid heterogeneity among studies, we recommend three key standard thresholds progression criteria which are consistent with the Scoliosis Research Society's (SRS) [1]:

i) Curve change $>5^{\circ}$ between two consecutive radiographs can be used to detect progression for short-term intervals (<1-year). Predicting curve progression for the next visit may guide treatment decision and could significantly reduce the number of radiographs required at upcoming visits.

 ii) Progression to a magnitude of 25 to 40° can be considered for initiating brace treatment.
 Short-term prediction may help for timely initiating treatment or adjust the treatment plan for a more efficient outcome.

iii) Progression to a magnitude of >45° during growth or greater than 45° after growth stops can be considered as a threshold for prediction of progression to surgical treatment over long-term follow-up. Learning about predictors over long-term follow-up intervals may also help clinicians plan possible treatment options or discuss expectations at maturity.

Based on the RoB assessment, the study confounding, and statistical analysis domains had the most articles presenting high RoB bias. The potential key confounding factors including age, curve severity, curve type, skeletal maturity, gender, treatment, and menarche were often not appropriately accounted for increasing the risk of bias of the relationship between prognostic factor and outcome. Also, the lack of an appropriate conceptual framework or insufficient statistics to decide about predictors were evident. The ideal study design for conducting prognostic studies is a prospective cohort study. The most common method used for prediction of progression was linear regression analysis which is recommended due to simplicity and readability. It is suggested that studies report a robust step-by step approach from univariate to multivariate analysis and model development by reporting on level of significance, cut off values, coefficients, accuracy tests, and preferably a validity study afterwards. In this review, only English language articles were included. Results showed moderate to good agreement between reviewers for full-text screening and RoB assessment, respectively. The agreement for full-text screening was (k=0.55) at first assessment which improved to 0.63 after adjusting the criteria and reassessing the articles. Data homogeneity is being examined to determine if a meta-analysis will be possible.

3.6. Conclusions

Our review showed that high Cobb angle and pre-menarche status were predictive of progression $\geq 5^{\circ}$ over short-term intervals and for long-term intervals, only limited evidence was available for the predictive factors. For progression $\geq 30^{\circ}$ or requiring surgery, pre-menarche was predictive over a short-term interval and large Cobb angle was predictive over the long-term. For progression in Cobb angle (°/year), only limited evidence showed that low age, high Cobb, and curve pattern were predictive over the short-term. More high-quality studies are required to conclude about majority of parameters presenting only limited, unclear or conflicting level of evidence.

3.7. Summary

The results of the systematic review study showed that only high Cobb angle and premenarche status were predictive of progression $\geq 5^{\circ}$ over short-term intervals with a strong support from evidence. Other parameters with lower levels of evidence should be further investigated.

References

[1] S. Negrini, T. M. Hresko, J. P. O'Brien, N. Price, SOSORT Boards, and SRS Non-Operative Committee, "Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee," *Scoliosis*, vol. 10, p. 8, 2015, doi: 10.1186/s13013-014-0025-4.

- P. N. Soucacos *et al.*, "Assessment of curve progression in idiopathic scoliosis," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 7, no. 4, pp. 270–277, 1998, doi: 10.1007/s005860050074.
- [3] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine*, vol. 21, no. 13, pp. 1540–1547, 1996.
- [4] V. W. Y. Hung *et al.*, "Osteopenia: a new prognostic factor of curve progression in adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 87, no. 12, pp. 2709–2716, Dec. 2005, doi: 10.2106/JBJS.D.02782.
- [5] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for detection of dranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS)-a prospective cohort study till skeletal maturity," *Osteoprosis International*, 2010, vol. 21, pp. S686-S687.
- [6] J. E. Lonstein and J. M. Carlson, "The prediction of curve progression in untreated idiopathic scoliosis during growth," J. Bone Joint Surg. Am., vol. 66, no. 7, pp. 1061–1071, Sep. 1984.
- M.-L. Nault, J.-M. Mac-Thiong, M. Roy-Beaudry, J. deGuise, H. Labelle, and S. Parent, "Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis," *J. Pediatr. Orthop.*, vol. 33, no. 6, pp. 618–623, Sep. 2013, doi: 10.1097/BPO.0b013e318292462a.
- [8] A. Noshchenko *et al.*, "Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis," *World J. Orthop.*, vol. 6, no. 7, p. 537, 2015.
- [9] K.-J. Tan, M. M. Moe, R. Vaithinathan, and H.-K. Wong, "Curve progression in idiopathic scoliosis: follow-up study to skeletal maturity," *Spine*, vol. 34, no. 7, pp. 697–700, 2009.

- [10] L. E. Peterson and A. L. Nachemson, "Prediction of progression of the curve in girls who have adolescent idiopathic scoliosis of moderate severity. Logistic regression analysis based on data from The Brace Study of the Scoliosis Research Society," *J. Bone Joint Surg. Am.*, vol. 77, no. 6, pp. 823–827, Jun. 1995, doi: 10.2106/00004623-199506000-00002.
- [11] D. Moher, A. Liberati, J. Tetzlaff, and D. G. Altman, "Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement," Ann. Intern. Med., vol. 151, no. 4, pp. 264–269, Aug. 2009, doi: 10.7326/0003-4819-151-4-200908180-00135.
- [12] T. S. Zotero, "George Mason University," *Fairfax VA USA Cent. Hist. New Media*, 2016.
- [13] J. A. Hayden, D. A. van der Windt, J. L. Cartwright, P. Côté, and C. Bombardier, "Assessing bias in studies of prognostic factors," *Ann. Intern. Med.*, vol. 158, no. 4, pp. 280–286, Feb. 2013, doi: 10.7326/0003-4819-158-4-201302190-00009.
- [14] L. R. Cornelius, J. J. L. van der Klink, J. W. Groothoff, and S. Brouwer, "Prognostic factors of long term disability due to mental disorders: a systematic review," J. Occup. Rehabil., vol. 21, no. 2, pp. 259–274, Jun. 2011, doi: 10.1007/s10926-010-9261-5.
- [15] M. L. McHugh, "Interrater reliability: the kappa statistic," *Biochem. Medica*, vol. 22, no. 3, pp. 276–282, 2012.
- [16] J. Cheung *et al.*, "The relation between electromyography and growth velocity of the spine in the evaluation of curve progression in idiopathic scoliosis," *Spine*, vol. 29, no. 9, pp. 1011–1016, May 2004, doi: 10.1097/00007632-200405010-00012.
- [17] L. Davis, J. S. Murphy, K. A. Shaw, K. Cash, D. P. Devito, and M. L. Schmitz, "Nighttime bracing with the Providence thoracolumbosacral orthosis for treatment of adolescent idiopathic scoliosis: A retrospective consecutive clinical series,"

Prosthet. Orthot. Int., vol. 43, no. 2, pp. 158–162, Apr. 2019, doi: 10.1177/0309364618792727.

- [18] T. Lara *et al.*, "The Risk of Curve Progression and Surgery in African Americans With Adolescent Idiopathic Scoliosis," *Spine Deform.*, vol. 5, no. 4, pp. 250–254, Jul. 2017, doi: 10.1016/j.jspd.2017.01.013.
- [19] S. Mao *et al.*, "Initial Cobb angle reduction velocity following bracing as a new predictor for curve progression in adolescent idiopathic scoliosis," *Eur. Spine J.*, vol. 25, no. 2, pp. 500–505, Feb. 2016, doi: 10.1007/s00586-015-3937-7.
- [20] P. J. O'Neill *et al.*, "Decreased orthotic effectiveness in overweight patients with adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 87, no. 5, pp. 1069– 1074, May 2005, doi: 10.2106/JBJS.C.01707.
- [21] S. Ohrt-Nissen, D. W. Hallager, M. Gehrchen, and B. Dahl, "Flexibility Predicts Curve Progression in Providence Nighttime Bracing of Patients With Adolescent Idiopathic Scoliosis," *Spine*, vol. 41, no. 22, pp. 1724–1730, Nov. 2016, doi: 10.1097/BRS.00000000001634.
- [22] P. Sitoula *et al.*, "Prediction of Curve Progression in Idiopathic Scoliosis: Validation of the Sanders Skeletal Maturity Staging System," *Spine*, vol. 40, no. 13, pp. 1006– 1013, Jul. 2015, doi: 10.1097/BRS.00000000000952.
- [23] X. Sun *et al.*, "Osteopenia predicts curve progression of adolescent idiopathic scoliosis in girls treated with brace treatment," *J. Pediatr. Orthop.*, vol. 33, no. 4, pp. 366–371, Jun. 2013, doi: 10.1097/BPO.0b013e31827b7b5f.
- [24] F. Zaina, S. Donzelli, and S. Negrini, "Overweight is not predictive of bracing failure in adolescent idiopathic scoliosis: results from a retrospective cohort study," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 26, no. 6, pp. 1670–1675, Jun. 2017, doi: 10.1007/s00586-017-4985y.

- [25] G. Duval-Beaupère, "Rib hump and supine angle as prognostic factors for mild scoliosis," *Spine*, vol. 17, no. 1, pp. 103–107, Jan. 1992, doi: 10.1097/00007632-199201000-00015.
- [26] M. R. Konieczny, P. Hieronymus, and R. Krauspe, "Time in brace: where are the limits and how can we improve compliance and reduce negative psychosocial impact in patients with scoliosis? A retrospective analysis," *Spine J. Off. J. North Am. Spine Soc.*, vol. 17, no. 11, pp. 1658–1664, Nov. 2017, doi: 10.1016/j.spinee.2017.05.010.
- [27] T. Kotwicki, A. Walczak, and A. Szulc, "Trunk rotation and hip joint range of rotation in adolescent girls with idiopathic scoliosis: does the" dinner plate" turn asymmetrically?," *Scoliosis*, vol. 3, no. 1, pp. 1–11, 2008.
- [28] A. G. Aulisa, V. Guzzanti, M. Galli, C. Perisano, F. Falciglia, and L. Aulisa, "Treatment of thoraco-lumbar curves in adolescent females affected by idiopathic scoliosis with a progressive action short brace (PASB): assessment of results according to the SRS committee on bracing and nonoperative management standardization criteria," *Scoliosis*, vol. 4, no. 1, pp. 1–8, 2009.
- [29] A. G. Aulisa, M. Giordano, F. Falciglia, E. Marzetti, A. Poscia, and V. Guzzanti, "Correlation between compliance and brace treatment in juvenile and adolescent idiopathic scoliosis: SOSORT 2014 award winner," *scoliosis*, vol. 9, no. 1, pp. 1–9, 2014.
- [30] D. D. Bohl, C. J. Telles, N. S. Golinvaux, B. A. Basques, P. A. DeLuca, and J. N. Grauer, "Effectiveness of Providence nighttime bracing in patients with adolescent idiopathic scoliosis," *Orthopedics*, vol. 37, no. 12, pp. e1085–e1090, 2014.
- [31] W. P. Bunnell, "The natural history of idiopathic scoliosis before skeletal maturity.," *Spine*, vol. 11, no. 8, pp. 773–776, 1986.
- [32] Y. P. Charles, F. Canavese, and A. Diméglio, "Curve progression risk in a mixed series of braced and nonbraced patients with idiopathic scoliosis related to skeletal

maturity assessment on the olecranon," *J. Pediatr. Orthop. B*, vol. 26, no. 3, pp. 240–244, 2017.

- [33] W. H. P. Cheung, K. D. K. Luk, D. Samartzis, and J. P. Y. Cheung, "Curve progression matching with skeletal growth in adolescent idiopathic scoliosis using the distal radius and ulna (DRU) classification," 2018.
- [34] J. P. Y. Cheung, P. W. H. Cheung, D. Samartzis, and K. D.-K. Luk, "APSS-ASJ Best Clinical Research Award: Predictability of Curve Progression in Adolescent Idiopathic Scoliosis Using the Distal Radius and Ulna Classification," *Asian Spine J.*, vol. 12, no. 2, pp. 202–213, Apr. 2018, doi: 10.4184/asj.2018.12.2.202.
- [35] C. R. D'Amato, S. Griggs, and B. McCoy, "Nighttime bracing with the Providence brace in adolescent girls with idiopathic scoliosis," *Spine*, vol. 26, no. 18, pp. 2006– 2012, Sep. 2001, doi: 10.1097/00007632-200109150-00014.
- [36] A. J. Danielsson, R. Hasserius, A. Ohlin, and A. L. Nachemson, "A prospective study of brace treatment versus observation alone in adolescent idiopathic scoliosis: a follow-up mean of 16 years after maturity," *Spine*, vol. 32, no. 20, pp. 2198–2207, Sep. 2007, doi: 10.1097/BRS.0b013e31814b851f.
- [37] M.-Q. Fang, C. Wang, G.-H. Xiang, C. Lou, N.-F. Tian, and H.-Z. Xu, "Long-term effects of the Chêneau brace on coronal and sagittal alignment in adolescent idiopathic scoliosis," *J. Neurosurg. Spine*, vol. 23, no. 4, pp. 505–509, Oct. 2015, doi: 10.3171/2015.2.SPINE14970.
- [38] R. Gepstein *et al.*, "Effectiveness of the Charleston bending brace in the treatment of single-curve idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 22, no. 1, pp. 84–87, Feb. 2002.
- [39] C. J. Goldberg, D. P. Moore, E. E. Fogarty, and F. E. Dowling, "Adolescent idiopathic scoliosis: the effect of brace treatment on the incidence of surgery," *Spine*, vol. 26, no. 1, pp. 42–47, Jan. 2001, doi: 10.1097/00007632-200101010-00009.

- [40] C. M. Goodbody, I. B. Asztalos, W. N. Sankar, and J. M. Flynn, "It's not just the big kids: both high and low BMI impact bracing success for adolescent idiopathic scoliosis," *J. Child. Orthop.*, vol. 10, no. 5, pp. 395–404, Oct. 2016, doi: 10.1007/s11832-016-0763-3.
- [41] J. Guo *et al.*, "Pelvic tilt and trunk inclination: new predictive factors in curve progression during the Milwaukee bracing for adolescent idiopathic scoliosis," *Eur. Spine J.*, vol. 21, no. 10, pp. 2050–2058, Oct. 2012, doi: 10.1007/s00586-012-2409-6.
- [42] J. Guo *et al.*, "A prospective randomized controlled study on the treatment outcome of SpineCor brace versus rigid brace for adolescent idiopathic scoliosis with followup according to the SRS standardized criteria," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 23, no. 12, pp. 2650–2657, Dec. 2014, doi: 10.1007/s00586-013-3146-1.
- [43] L. A. Karol, C. E. Johnston, R. H. Browne, and M. Madison, "Progression of the curve in boys who have idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 75, no. 12, pp. 1804–1810, Dec. 1993, doi: 10.2106/00004623-199312000-00010.
- [44] L. A. Karol, D. Virostek, K. Felton, C. Jo, and L. Butler, "The Effect of the Risser Stage on Bracing Outcome in Adolescent Idiopathic Scoliosis," *J. Bone Joint Surg. Am.*, vol. 98, no. 15, pp. 1253–1259, Aug. 2016, doi: 10.2106/JBJS.15.01313.
- [45] L. A. Karol, "Effectiveness of bracing in male patients with idiopathic scoliosis," *Spine*, vol. 26, no. 18, pp. 2001–2005, Sep. 2001, doi: 10.1097/00007632-200109150-00013.
- [46] D. E. Katz and A. A. Durrani, "Factors that influence outcome in bracing large curves in patients with adolescent idiopathic scoliosis," *Spine*, vol. 26, no. 21, pp. 2354–2361, Nov. 2001, doi: 10.1097/00007632-200111010-00012.

- [47] D. E. Katz, J. A. Herring, R. H. Browne, D. M. Kelly, and J. G. Birch, "Brace wear control of curve progression in adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 92, no. 6, pp. 1343–1352, Jun. 2010, doi: 10.2106/JBJS.I.01142.
- [48] H. Kuroki, N. Inomata, H. Hamanaka, K. Higa, E. Chosa, and N. Tajima, "Efficacy of the Osaka Medical College (OMC) brace in the treatment of adolescent idiopathic scoliosis following Scoliosis Research Society brace studies criteria," *Scoliosis*, vol. 10, p. 12, Apr. 2015, doi: 10.1186/s13013-015-0036-9.
- [49] T. P. Lam *et al.*, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound Med. Biol.*, vol. 39, no. 3, pp. 381–387, Mar. 2013, doi: 10.1016/j.ultrasmedbio.2012.09.012.
- [50] R. LeBlanc, H. Labelle, F. Forest, and B. Poitras, "Morphologic discrimination among healthy subjects and patients with progressive and nonprogressive adolescent idiopathic scoliosis," *Spine*, vol. 23, no. 10, pp. 1109–1115, 1998.
- [51] C. S. Lee *et al.*, "Effectiveness of the Charleston night-time bending brace in the treatment of adolescent idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 32, no. 4, pp. 368–372, 2012.
- [52] M.-L. Nault *et al.*, "Three-Dimensional Spinal Morphology Can Differentiate Between Progressive and Nonprogressive Patients With Adolescent Idiopathic Scoliosis at the Initial Presentation," *Spine*, vol. 39, no. 10, pp. E601–E606, May 2014, doi: 10.1097/BRS.00000000000284.
- [53] G. Pasquini *et al.*, "The outcome of a modified version of the Cheneau brace in adolescent idiopathic scoliosis (AIS) based on SRS and SOSORT criteria: a retrospective study," *Eur J Phys Rehabil Med*, vol. 52, no. 5, pp. 618–629, 2016.
- [54] A. Ploumis, V. Trivedi, J.-H. Shin, K. B. Wood, and B. E. Grottkau, "Progression of idiopathic thoracic or thoracolumbar scoliosis and pelvic obliquity in adolescent

patients with and without limb length discrepancy," *Scoliosis Spinal Disord.*, vol. 13, no. 1, pp. 1–7, 2018.

- [55] P. M. Ryan, E. G. Puttler, W. M. Stotler, and R. L. Ferguson, "Role of the triradiate cartilage in predicting curve progression in adolescent idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 27, no. 6, pp. 671–676, 2007.
- [56] B. Shi *et al.*, "Curve progression in adolescent idiopathic scoliosis with a minimum of 2 years' follow-up after completed brace weaning with reference to the SRS standardized criteria," *Spine Deform.*, vol. 4, no. 3, pp. 200–205, 2016.
- [57] X. Sun *et al.*, "Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis," *Orthop. Surg.*, vol. 2, no. 4, pp. 285–290, 2010.
- [58] R. M. Thompson, E. W. Hubbard, C.-H. Jo, D. Virostek, and L. A. Karol, "Brace Success Is Related to Curve Type in Patients with Adolescent Idiopathic Scoliosis," *JBJS*, vol. 99, no. 11, pp. 923–928, Jun. 2017, doi: 10.2106/JBJS.16.01050.
- [59] J. M. Trivedi and J. D. Thomson, "Results of Charleston bracing in skeletally immature patients with idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 21, no. 3, pp. 277–280, Jun. 2001.
- [60] V. Vijvermans, G. Fabry, and J. Nijs, "Factors determining the final outcome of treatment of idiopathic scoliosis with the Boston brace: a longitudinal study," J. Pediatr. Orthop. Part B, vol. 13, no. 3, pp. 143–149, May 2004, doi: 10.1097/00009957-200405000-00001.
- [61] H. R. Weiss, "The Progression of Idiopathic Scoliosis under the Influence of a Physiotherapy Rehabilitation Programme," *Physiotherapy*, vol. 78, no. 11, pp. 815– 821, Nov. 1992, doi: 10.1016/S0031-9406(10)60452-X.
- [62] H. Wu, J. L. Ronsky, F. Cheriet, J. Harder, J. C. Küpper, and R. F. Zernicke, "Time series spinal radiographs as prognostic factors for scoliosis and progression of spinal deformities," *Eur. Spine J.*, vol. 20, no. 1, pp. 112–117, Jan. 2011, doi: 10.1007/s00586-010-1512-9.

- [63] Y. Yamauchi, T. Yamaguchi, and Y. Asaka, "Prediction of curve progression in idiopathic scoliosis based on initial roentgenograms. A proposal of an equation," *Spine*, vol. 13, no. 11, pp. 1258–1261, Nov. 1988, doi: 10.1097/00007632-198811000-00009.
- [64] B. H. K. Yip *et al.*, "Prognostic Value of Bone Mineral Density on Curve Progression: A Longitudinal Cohort Study of 513 Girls with Adolescent Idiopathic Scoliosis," *Sci. Rep.*, vol. 6, p. 39220, Dec. 2016, doi: 10.1038/srep39220.
- [65] M. Ylikoski, "Spinal growth and progression of adolescent idiopathic scoliosis," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 1, no. 4, pp. 236–239, Mar. 1993, doi: 10.1007/BF00298366.
- [66] T. Yrjönen and M. Ylikoski, "Effect of growth velocity on the progression of adolescent idiopathic scoliosis in boys," *J. Pediatr. Orthop. Part B*, vol. 15, no. 5, pp. 311–315, Sep. 2006, doi: 10.1097/01202412-200609000-00001.
- [67] T. Yrjönen, M. Ylikoski, D. Schlenzka, R. Kinnunen, and M. Poussa, "Effectiveness of the Providence nighttime bracing in adolescent idiopathic scoliosis: a comparative study of 36 female patients," *Eur. Spine J.*, vol. 15, no. 7, pp. 1139–1143, Jul. 2006, doi: 10.1007/s00586-005-0049-9.
- [68] Y. Zhang *et al.*, "Factors relating to curve progression in female patients with adolescent idiopathic scoliosis treated with a brace," *Eur. Spine J.*, vol. 24, no. 2, pp. 244–248, Feb. 2015, doi: 10.1007/s00586-014-3674-3.
- [69] Z. Zhu *et al.*, "Is Brace Treatment Appropriate for Adolescent Idiopathic Scoliosis Patients Refusing Surgery With Cobb Angle Between 40 and 50 Degrees," *Clin. Spine Surg.*, vol. 30, no. 2, pp. 85–89, Mar. 2017, doi: 10.1097/BSD.0b013e3182a1de29.

Appendices

Appendix 3.1 The Search Strategy for the Selected Databases

Medline (n=890)

1. spinal curvatures/ or scoliosis/

2. (scolio* or spin* curvature).mp.

3. idiopathic.mp.

4. Adolescent/

5. (teen* or adolescen* or youth or youths or young people or young adult*).mp.

6. (1 or 2) and 3 and (4 or 5)

7. (radiograph* adj3 (indicat* or predict* or parameter* or factor*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

8. (clinical adj3 (indicat* or predict* or parameter* or factor*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

9. (demograph* adj3 (indicat* or predict* or parameter* or factor*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

10. (curve* adj3 (angle* or magnitude* or severity or thoracic or location or thoracolumbar or thoraco lumbar or lumbar or direction* or type* or mild or moderate or severe or pattern* or Lenke)).mp.

11. Cobb angle*.mp.

12. (lateral adj3 (curve* or curvature* or deviation*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

13. (Apical adj3 (level* or vertebra* rotation* or location* or translation*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

14. (Apex adj3 (level* or vertebra* rotation* or location* or translation*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

15. AVR.mp.

16. axial rotation*.mp.

17. axial vertebra* rotation*.mp.

18. coronal balance.mp.

19. decompensation*.mp.

20. Rib* vertebra* angle*.mp.

21. coronal imbalance.mp.

22. frontal imbalance.mp.

23. radiograph* imbalance.mp.

24. (rib hump or rib prominence).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating subheading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

25. shoulder asymmetry.mp.

26. height velocity.mp.

27. leg length discrepanc*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

28. spin* growth velocity.mp.

29. growing index.mp.

30. growth index.mp.

31. osteopenia.mp. or exp Bone Diseases, Metabolic/

32. bone mineral densit*.mp.

33. BMD.mp.

34. bone quality.mp.

35. bone densit*.mp.

36. bone stiffness.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 37. forward bend*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

38. surface topograph*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 39. Plumb line*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 40. Scoliomet*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 41. inclinomet*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 42. Moire topography.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 43. hypermobility.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 44. spin* flexibilit*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 45. spin* movement*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 46. spin* morpholog*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 47. 3D spin* shape*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 48. kyphosis angle*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 49. kyphotic.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

50. lordosis angle*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 51. lordotic.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

52. spin* movement*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 53. spinopelvic balance.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 54. sagittal alignment*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 55. pelvi* incidence.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 55. pelvi* incidence.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 56. sacr* slope mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 57. age.mp.

58. gender.mp.

59. sex.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

60. sexual maturity.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 61. skeletal maturity.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 62. Risser.mp. [mp=title, abstract, original title, name of substance word, subject heading word, unique identifier, synonyms] 62. Risser.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

63. Tri-radiate cartilage.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 64. wrist x-ray.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

65. Tanner stage*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 66. Sanders.ab.

67. menarche.mp.

68. premenarche.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]
69. (wedg* adj3 vertebra*).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

70. plane of maximal deformit*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating subheading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

71. or/7-70

72. disease progression/

73. progression.mp.

74. curve progression*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

76. (Cobb adj3 (increase* or change* or difference*)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

77. curve angle change*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

78. curve deterioration*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

79. spin* deformit* progression*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating subheading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

80. progressive adolescent idiopathic scoliosis.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

81. progressive AIS.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 82. brac* prescription*.mp.

83. surgery* prescription*.mp.

84. or/72-83

85. (predict* or prognos* or risk factor*).ti.

86. model*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 87. (significant* adj6 (association* or relation* or predict* or prognos* or risk factor*)).mp.

88. predict* value of test*.mp.

89. progress*.mp.

90. (prognostic index or prognostic model*).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

91. (predict* adj4 (outcome* or curve* or angle* or Cobb)).mp.

92. regression.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]
 93. (Fisher* adj exact).mp.

94. (regression analys* or logistic regression).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

95. Multiple logistic regression model*.mp.

96. multivari*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 97. Cox regression*.mp.

98. ((multivaria* or performance or cox or proportional hazards) adj3 (model or models)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

99. regression tree*.mp.

100. Kaplan-Meier Estimate.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

101. Risk assessment*.mp.

102. risk ratio*.mp.

103. Odds ratio*.mp.

104. hazard ratio*.mp.

105. likelihood ratio*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms] 106. positive likelihood ratio*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

107. negative likelihood ratio*.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating subheading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

108. sensitivit*.mp.

109. specificit*.mp.

110. (receiver operating characteristic* or ROC).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

111. (recalibration or receiver operating curve or roc).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

112. (scatter diagram* or scatter plot* or covariance decomposition or youden index).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

113. (area under curve or AUC or bootstrapping or Brier Score or "c index" or c-index).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

114. (calibration or CCR or concordance index or correct classification rate or cross validation).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

115. ("D statistic" or D-statistic).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating subheading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

116. (Murphy's decomposition or Nagelkerke* or negative predictive value or npv or net reclassification improvement or nomogram*).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

117. (positive predict* value* or PPV).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

118. (negative predict* value* or NPV).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

119. machine learning model*.mp.

120. cluster analys*.mp.

121. Discriminant analysis.mp. or exp Discriminant Analysis/

122. (decision curve analysis or discrimination or external validation).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

123. ("goodness of fit" or Hosmer-Lemeshow test or internal validation).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

124. (misclassification adj3 (probability or rate)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]

125. or/85-124 126. 6 and 71 and 84 and 125

120. 0 and /1 and 64 and 1

Embase (n=995)

1. exp scoliosis/

2. (scolio* or spin* curvature).mp.

3. idiopathic.mp.

4. adolescent/

5. (teen* or adolescen* or youth or youths or young people or young adult*).mp.

6. (1 or 2) and 3 and (4 or 5)

7. (radiograph* adj3 (indicat* or predict* or parameter* or factor*)).mp.

8. (clinical adj3 (indicat* or predict* or parameter* or factor*)).mp.

9. (demograph* adj3 (indicat* or predict* or parameter* or factor*)).mp.

10. (curve* adj3 (angle* or magnitude* or severity or thoracic or location or thoracolumbar or thoraco lumbar or lumbar or direction* or type* or mild or moderate or severe or pattern* or Lenke)).mp.

11. Cobb angle*.mp.

12. (lateral adj3 (curve* or curvature* or deviation*)).mp.

13. (Apical adj3 (level* or vertebra* rotation* or location* or translation*)).mp.

14. (Apex adj3 (level* or vertebra* rotation* or location* or translation*)).mp.

15. AVR.mp.

16. axial rotation*.mp.

17. axial vertebra* rotation*.mp.

18. coronal balance.mp.

19. decompensation*.mp.

20. Rib* vertebra* angle*.mp.

21. coronal imbalance.mp.

22. frontal imbalance.mp.

23. radiograph* imbalance.mp.

24. (rib hump or rib prominence).mp.

25. shoulder asymmetry.mp.

26. height velocity.mp.

27. leg length discrepanc*.mp.

28. spin* growth velocity.mp.

29. growing index.mp.

30. growth index.mp.

31. osteopenia.mp. or exp Bone Diseases, Metabolic/

32. bone mineral densit*.mp.

33. BMD.mp.

34. bone quality.mp.

35. bone densit*.mp.

36. bone stiffness.mp.

37. forward bend*.mp.

38. surface topograph*.mp.

39. Plumb line*.mp.

40. Scoliomet*.mp.

41. inclinomet*.mp.

42. Moire topography.mp.

43. hypermobility.mp.

44. spin* flexibilit*.mp.

45. spin* movement*.mp.

46. spin* morpholog*.mp.

47. 3D spin* shape*.mp.48. kyphosis angle*.mp.

49. kyphotic.mp.

50. lordosis angle*.mp.

51. lordotic.mp.

52. spin* movement*.mp.

53. spinopelvic balance.mp.

54. sagittal alignment*.mp.

55. pelvi* incidence.mp.

56. sacr* slope.mp.

57. age.mp.

- 58. gender.mp.
- 59. sex.mp.
- 60. sexual maturity.mp.
- 61. skeletal maturity.mp.
- 62. Risser.mp.
- 63. Tri-radiate cartilage.mp.
- 64. wrist x-ray.mp.
- 65. Tanner stage*.mp.
- 66. Sanders.ab.
- 67. menarche.mp.
- 68. premenarche.mp.
- 69. (wedg* adj3 vertebra*).mp.
- 70. plane of maximal deformit*.mp.
- 71. or/7-70
- 72. disease progression/
- 73. progression.mp.
- 74. curve progression*.mp.
- 75. scoliosis progression*.mp.
- 76. (Cobb adj3 (increase* or change* or difference*)).mp.
- 77. curve angle change*.mp.
- 78. curve deterioration*.mp.
- 79. spin* deformit* progression*.mp.
- 80. progressive adolescent idiopathic scoliosis.mp.
- 81. progressive AIS.mp.
- 82. brac* prescription*.mp.
- 83. surgery* prescription*.mp.
- 84. or/72-83
- 85. (predict* or prognos* or risk factor*).ti.
- 86. model*.mp.
- 87. (significant* adj6 (association* or relation* or predict* or prognos* or risk factor*)).mp.
- 88. predict* value of test*.mp.
- 89. progress*.mp.
- 90. (prognostic index or prognostic model*).mp.
- 91. (predict* adj4 (outcome* or curve* or angle* or Cobb)).mp.
- 92. regression.mp.
- 93. (Fisher* adj exact).mp.
- 94. (regression analys* or logistic regression).mp.
- 95. Multiple logistic regression model*.mp.

96. multivari*.mp.

- 97. Cox regression*.mp.
- 98. ((multivaria* or performance or cox or proportional hazards) adj3 (model or models)).mp.
- 99. regression tree*.mp.
- 100. Kaplan-Meier Estimate.mp.

101. Risk assessment*.mp.

- 102. risk ratio*.mp.
- 103. Odds ratio*.mp.
- 104. hazard ratio*.mp.
- 105. likelihood ratio*.mp.
- 106. positive likelihood ratio*.mp.
- 107. negative likelihood ratio*.mp.
- 108. sensitivit*.mp.
- 109. specificit*.mp.
- 110. (receiver operating characteristic* or ROC).mp.
- 111. (recalibration or receiver operating curve or roc).mp.
- 112. (scatter diagram* or scatter plot* or covariance decomposition or youden index).mp.
- 113. (area under curve or AUC or bootstrapping or Brier Score or "c index" or c-index).mp.
- 114. (calibration or CCR or concordance index or correct classification rate or cross validation).mp.
- 115. ("D statistic" or D-statistic).mp.
- 116. (Murphy's decomposition or Nagelkerke* or negative predictive value or npv or net reclassification improvement or nomogram*).mp.
- 117. (positive predict* value* or PPV).mp.
- 118. (negative predict* value* or NPV).mp.
- 119. machine learning model*.mp.
- 120. cluster analys*.mp.
- 121. Discriminant analysis.mp. or exp Discriminant Analysis/
- 122. (decision curve analysis or discrimination or external validation).mp.
- 123. ("goodness of fit" or Hosmer-Lemeshow test or internal validation).mp.
- 124. (misclassification adj3 (probability or rate)).mp.
- 125. or/85-124

126. 6 and 71 and 84 and 125

CENTRAL (n=118)

ID	Search
#1	scolio* or spin* curvature*
#2	idiopathic
#3	Adolescent
#4	(teen* or adolescen* or youth or young people or young adult*)
#5	#1 and #2 and (#3 or #4)
#6	(radiograph* AND (indicat* or predict* or parameter* or factor*))
#7	(clinical AND (indicat* or predict* or parameter* or factor*))
#8	(demograph* AND (indicat* or predict* or parameter* or factor*))
#9 or mild or 1	(curve* AND (angle* or magnitude* or severity or thoracic or location or thoracolumbar or thoraco lumbar or lumbar or direction* or type* moderate or severe or pattern* or Lenke))
#10	Cobb angle*
#11	(lateral AND (curve* or curvature* or deviation*))
#12	(Apical AND (level* or vertebra* rotation* or location* or translation*))
#13	(Apex AND (level* or vertebra* rotation* or location* or translation*))
#14	AVR

#15 axial rotation*

ease*

#59	sexual maturity
#60	skeletal maturity
#61	Risser
#62	Tri-radiate cartilage
#63	wrist x-ray
#64	Tanner stage*
#65	Sanders:ab
#66	menarche
#67	premenarche
#68	(wedg* AND vertebra*)
#69	plane of maximal deformit*
#70	{OR #6-#69}
#71	disease progression
#72	progression
#73	curve progression*
#74	scoliosis progression*
#75	(Cobb AND (increase* or change* or difference*))
#76	curve angle change*
#77	curve deterioration*
#78	spin* deformit* progression*
#79	progressive adolescent idiopathic scoliosis
#80	progressive AIS
#81	brac* prescription*
#82	surgery* prescription*
#83	{OR #71-#82}
#84	(predict* or prognos* or risk factor*):ti
#85	model*
#86	(significant* AND (association* or relation* or predict* or prognos* or risk factor*))
#87	predict* value of test*
#88	progress*
#89	(prognostic index or prognostic model*)
#90	(predict* AND (outcome* or curve* or angle* or Cobb))
#91	regression
#92	(Fisher* AND exact)
#93	(regression analys* or logistic regression)
#94	Multiple logistic regression model*
#95	multivari*
#96	Cox regression*
#97	((multivaria* or performance or cox or proportional hazards) AND (model or models))
#98	regression tree*
#99	Kaplan-Meier Estimate
#100	Risk assessment*
#101	risk ratio*

#102	Odds ratio*
#103	hazard ratio*
#104	likelihood ratio*
#105	positive likelihood ratio*
#106	negative likelihood ratio*
#107	sensitivit*
#108	specificit*
#109	(receiver operating characteristic* or ROC)
#110	(recalibration or receiver operating curve or roc)
#111	(scatter diagram* or scatter plot* or covariance decomposition or youden index)
#112	(area under curve or AUC or bootstrapping or Brier Score or "c index" or c-index)
#113	(calibration or CCR or concordance index or correct classification rate or cross validation)
#114	("D statistic" or D-statistic)
#115	(Murphy's decomposition or Nagelkerke* or negative predictive value or npv or net reclassification improvement or nomogram*)
#116	(positive predict* value* or PPV)
#117	(negative predict* value* or NPV)
#118	machine learning model*
#119	cluster analys*
#120	Discriminant analysis
#121	(decision curve analysis or discrimination or external validation)
#122	("goodness of fit" or Hosmer-Lemeshow test or internal validation)
#123	(misclassification AND (probability or rate))
#124	{OR #84-#123}
#125	#5 and #70 and #83 and #124
Web of Sci	ence (n=565)

Web of Science (n=565)

TOPIC 1: TS=((scolio* OR "spin* curvature") AND (idiopathic) AND (teen* OR adolescen* OR youth OR youths OR "young adult*"))

Indexes=SCI-EXPANDED, SSCI, A&HCI Timespan=All years

TOPIC 2: TS=(((radiograph* OR clinical OR demograph*) NEAR/3 (indicat* OR predict* OR parameter* OR factor*)) OR (curve* NEAR/3 (angle* OR magnitude* OR severety OR thoracic OR location OR thoracolumbar OR thoraco-lumbar OR lumbar OR direction* OR type* OR mild OR moderate OR severe OR pattern* OR Lenke)) OR "Cobb angle*" OR (lateral NEAR/3 (curve* OR curvature* OR deviation*)) OR ((Apical OR Apex) NEAR/3 (level* OR "vertebra* rotation*" OR location* OR translation*)) OR AVR OR "axial rotation*" OR "axial rotation*" OR "coronal balance" OR decompensation* OR "radiograph* imbalance" OR "rib hump" OR "rib prominence" OR "shoulder asymmetry" OR "height velocity" OR "leg length discrepanc*" OR "spin* growth velocity" OR "growing index" OR "growth index" OR osteopenia OR "surface topograph*" OR "Plumb line*" OR Scoliomet* OR "Moire topography" OR hypermobility OR "spin* flexibilit*" OR "spin* movement*" OR "spin* morpholog*" OR "3D spin* shape*" OR "kyphosis angle*" OR kyphotic OR "lordosis angle*" OR lordotic OR "sagital al algumet"" OR "spin* growth core soft oR severe or soft or "sagital al maturity" OR "spin* diate and the core of the cardiate cardiage" OR "spin* and the cardiate or the cardiate cardiage" OR "spin* shape*" OR "Moire topography" OR angle*" OR lordotic OR "spin* movement*" OR "spin* morpholog*" OR "sagital alignmet*" OR "spin* slope" OR "sage*" OR severe or soft or s

Indexes=SCI-EXPANDED, SSCI, A&HCI Timespan=All years

TOPIC 3: TS=(progression OR (Cobb NEAR/3 (increase* OR change* OR difference*)) OR "curve angle change*" OR "curve deterioration*" OR "progressive adolescent idiopathic scoliosis" OR "progressive AIS" OR "brac* prescription*" OR "surgery* prescription*")

Indexes=SCI-EXPANDED, SSCI, A&HCI Timespan=All years

TOPIC 4: TI=(predict* OR prognos* OR "risk factor*") OR TS=(model* OR (significant* NEAR/6 (association* OR relation* OR predict* OR prognos* OR "risk factor*")) OR "predict* value of test*" OR progress* OR "prognostic index" OR "prognostic model*" OR (predict* NEAR/4 (outcome* OR curve* OR angle* OR Cobb)) OR regression OR (Fisher* NEAR exact) OR multivari* OR "Kaplan-Meier Estimate" OR "Risk assessment*" OR "risk ratio*" OR "Odds ratio*" OR "Odds ratio*" OR "likelihood ratio*" OR "positive likelihood ratio*" OR "negative likelihood ratio*" OR "secificit* OR "receiver operating characteristic*" OR OC OR recalibration OR "receiver operating curve" OR "scatter plot*" OR "covariance decomposition" OR "youden index" OR "area under curve" OR AUC OR bootstrapping OR "Brier Score" OR "c index" OR "correct classification rate" OR "cross validation" OR "D statistic" OR "Murphy's decomposition" OR

Nagelkerke* OR "negative predictive value" OR npv OR "net reclassification improvement" OR nomogram* OR "positive predict* value*" OR PPV OR "negative predict* value*" OR NPV OR "machine learning model*" OR "cluster analys*" OR "Discriminant analysis")

Indexes=SCI-EXPANDED, SSCI, A&HCI Timespan=All years

TOPIC 5: #1 AND #2 AND #3 AND #4

Indexes=SCI-EXPANDED, SSCI, A&HCI Timespan=All years

CINAHL (n=319)

Query	Limiters/Expanders
S15	S5 AND S8 AND S11 AND S14
S14	\$12 OR \$13
S13	(MH "Discriminant Analysis")
S12	TI ((predict* OR prognos* OR "risk factor*")) OR ((model* OR (significant* n6 (association* OR relation* OR predict* OR prognos* OR "risk factor*")) OR "predict* value of test*" OR progress* OR "prognostic index" OR "prognostic model*" OR (predict* n4 (outcome* OR curve* OR angle* OR Cobb)) OR regression OR (Fisher* n exact) OR multivari* OR "Kaplan-Meier Estimate" OR "Risk assessment*" OR "risk ratio*" OR "Odds ratio*" OR "hazard ratio*" OR "likelihood ratio*" OR "positive likelihood ratio*" OR "negative likelihood ratio*" OR "positive likelihood ratio*" OR "negative likelihood ratio*" OR sensitivit* OR specificit* OR "receiver operating characteristic*" OR ROC OR recalibration OR "receiver operating curve" OR "scatter diagram*" OR "Brief Score" OR "c index" OR "c-index" OR calibration OR "CCR OR "concordance index" OR "negative predictive value" OR "Murphy's decomposition" OR Nagelkerke* OR "negative predictive value" OR not or "D statistic" OR "Murphy's decomposition" OR "positive predict* value*" OR PPV OR "negative predict* value*" OR NPV OR "machine learning model*" OR "cluster analys*" OR "D scriminant analysis"))
S11	S9 OR S10
S10	(MH "Disease Progression")
S9	(progression OR (Cobb n3 (increase* OR change* OR difference*)) OR "curve angle change*" OR "curve deterioration*" OR "progressive adolescent idiopathic scoliosis" OR "progressive AIS" OR "brac* prescription*" OR "surgery* prescription*")
S8	S6 OR S7
S7	(MH "Bone Diseases, Metabolic+")
S6	 (((radiograph* OR clinical OR demograph*) n3 (indicat* OR predict* OR parameter* OR factor*)) OR (curve* n3 (angle* OR magnitude* OR severity OR thoracic OR location OR thoraco-lumbar OR lumbar OR direction* OR type* OR mild OR moderate OR severe OR pattern* OR Lenke)) OR "Cobb angle*" OR (lateral n3 (curve* OR curvature* OR deviation*)) OR ((Apical OR Apex) n3 (level* OR "vertebra* rotation*" OR location* OR translation*)) OR AVR OR "axial rotation*" OR "axial rotation*" OR "coronal balance" OR decompensation* OR "Rib* vertebra* angle*" OR "coronal imbalance" OR "frontal imbalance" OR "radiograph* imbalance" OR "rib hump" OR "rib prominence" OR "shoulder asymmetry" OR "height velocity" OR "leg length discrepanc*" OR "spin* growth velocity" OR BMD OR "bone quality" OR "bone densit*" OR "bone stiffness" OR "forward bend** OR "surface topograph* OR "Plumb line*" OR Scoliomet* OR "spin* morpholog*" OR "3D spin* shape*" OR "kyphosis angle*" OR kyphotic OR "lordosis angle*" OR "sort al alignment*" OR "spin* age OR gender OR sex OR "sexual maturity" OR "skeletal maturity" OR Risser OR "Tranalize cartilage" OR "wrist x-ray" OR "Tanner stage*" OR Sanders OR menarche OR premenance OR (wedg* n3 vertebra*) OR "plane of maximal deformit*")
\$5	((S1 OR S2) AND S3 AND S4) OR (MH "Scoliosis, Idiopathic, Adolescent")
S4	(teen* or adolescen* or youth or youths or young people or young adult*)
S3	idiopathic
S2	(scolio* OR "spin* curvature")
S1	(MH "Spinal Curvatures+") OR (MH "Scoliosis+")

Appendix 3.2 List of Excluded Articles with the Reason of Exclusion

Reasons for exclusion

1-Congenital, neuromuscular, syndrome related, and secondary reasons of scoliosis

2-Using only specialized parameter to predict the curve progression such as laboratory test results including level of enzymes or hormones, brain-function tests, genetic-related tests or analysis, electromyography, exercise

3-Using only non-radiographic imaging methods to assess curve progression such as in vitro studies, engineering methods (simulations, finite element method, and machine learning methods), magnetic resonance imaging (MRI), Computed tomography (CT), and questionnaires,

4-Having had scoliosis surgery at baseline

5-Samples less than 10 participants, case reports, conference proceedings, commentary and opinion letters, and reviews,

6-Non-English papers

7-Non-human subjects

8-Non AIS

9- The Mean±SD for age was more or less than 10 to 18 years old

10- The Mean±SD for the initial Cobb angle was < 10 degree

11-Study without follow-up

12-No radiographic/ultrasound follow-up of spine deformity

13-No report on prediction of curve progression

	Excluded Reference	Reason for Exclusi on
1.	Lam, T.P., et al., <i>Vitamin D level and its correlation with bone mineral density in girls with adolescent idiopathic scoliosis (AIS)</i> . Osteoporosis International, 2013. 24 (SUPPL 4): p. S640.	5
2.	Carr, W.A., et al., Treatment of idiopathic scoliosis in the Milwaukee brace. Long-term results. Journal of Bone and Joint Surgery - Series A, 1980. 62(4): p. 599-612.	13
3.	Kelly, J.J., et al., Treatment of adolescent idiopathic scoliosis and evaluation of the adolescent patient. Current Orthopaedic Practice, 2018. 29(5): p. 424-429.	5
4.	Hamill, C.L., The treatment of idiopathic scoliosis. Current Opinion in Orthopaedics, 2003. 14(3): p. 119-126.	5
5.	Kotwicki, T., et al., Conservative management of idiopathic scoliosisguidelines based on SOSORT 2006 Consensus. Ortopedia, traumatologia, rehabilitacja, 2009. 11(5): p. 379-95.	5
6.	Maruyama, T., et al., Conservative treatment for adolescent idiopathic scoliosis: can it reduce the incidence of surgical treatment? Pediatric rehabilitation, 2003. 6(3-4): p. 215-9.	13
7.	Jiang, J., et al., [Comparison of the sagittal profiles between thoracic idiopathic scoliosis patients with different curve progression]. Zhonghua wai ke za zhi [Chinese journal of surgery], 2011. 49(9): p. 812-5.	6
8.	Kennaway, D., Cause of idiopathic scoliosis. Spine, 2000. 25(19): p. 2552-3.	5
9.	Sanders, J.O., et al., Bracing in adolescent idiopathic scoliosis, surrogate outcomes, and the number needed to treat. Journal of pediatric orthopedics, 2012. 32 Suppl 2(hsw, 8109053): p. S153-7.	5
10.	Lou, E., et al., Brace wear characteristics during the first 6 months for the treatment of scoliosis. Studies in health technology and informatics, 2012. 176(ck1, 9214582): p. 346-9.	13
11.	Negrini, S., et al., Characteristics of patients with more than 20 degree of improvement or worsening during conservative treatment of adolescent idiopathic scoliosis. Studies in health technology and informatics, 2012. 176 (ck1, 9214582): p. 354-7.	13
12.	Ng, S.Y., et al., Bracing Scoliosis - State of the Art. Current pediatric reviews, 2015(101240290).	5
13.	Schultz, A.B., <i>Biomechanical factors in the progression of idiopathic scoliosis</i> . Annals of biomedical engineering, 1984. 12 (6): p. 621-30.	5
14.	Grivas, T.B., G.I. Rodopoulos, and N.V. Bardakos, <i>Biomechanical and clinical perspectives on nighttime bracing for adolescent idiopathic scoliosis</i> . Studies in health technology and informatics, 2008. 135 (ck1, 9214582): p. 274-90.	5
15.	Adler, N., et al., Balance reactions and eye-hand coordination in idiopathic scoliosis. Journal of orthopaedic research : official publication of the Orthopaedic Research Society, 1986. 4(1): p. 102-7.	13
16.	Pries, P., et al., [Ambulatory management of idiopathic scoliosis using the Lyon orthopedic treatment. 70 cases reviewed]. Prise en charge ambulatoire des scolioses idiopathiques par le traitement orthopedique lyonnais. 70 dossiers revus., 1990. 31 (2): p. 119- 24.	5

 Burwell, R.G., Aetiology of idiopathic scoliosis: current concepts. Pediatric rehabilitation, 2003. 6(3-4): p. 137-7/ Biarcy, G.L., Advances in idiopathic scoliosis in children and adolescents. Advances in pediatrics, 2007. 54(20): 55-66. Rinsky, L.A. and J.G. Gamble, Adolescent idiopathic scoliosis. The Western journal of medicine, 1988. 148(2): p. 20. Gunnoe, B.A., Adolescent idiopathic scoliosis. Orthopaedic review, 1990. 19(1): p. 35-43. Shindle, M.K., et al., Adolescent idiopathic scoliosis: modern management guidelines. Journal of surgical orthop 2006. 15(1): p. 43-52. Parent, S., P.O. Newton, and D.R. Wenger, Adolescent idiopathic scoliosis: etiology, anatomy, natural histor Instructional course lectures, 2005. 54(ifc, 7507149): p. 529-36. Rinsky, L.A., Advances in management of idiopathic scoliosis. Neurosurgery clinics of North America, 2007. 18(2): p. 2 Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: Lenke type I-VI case studies. Neurosurgery clinics of our of the scolio of the scoliosis. Current prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. 	oo, 0370436): p. p. 182-91. paedic advances, ry, and bracing. 55. 2255-9.	5 5 5 5 5 5 5
 Rinsky, L.A. and J.G. Gamble, Adolescent idiopathic scoliosis. The Western journal of medicine, 1988. 148(2): p Gunnoe, B.A., Adolescent idiopathic scoliosis. Orthopaedic review, 1990. 19(1): p. 35-43. Shindle, M.K., et al., Adolescent idiopathic scoliosis: modern management guidelines. Journal of surgical orthop 2006. 15(1): p. 43-52. Parent, S., P.O. Newton, and D.R. Wenger, Adolescent idiopathic scoliosis: etiology, anatomy, natural histor Instructional course lectures, 2005. 54(ifc, 7507149): p. 529-36. Rinsky, L.A., Advances in management of idiopathic scoliosis. Hospital practice (Office ed.), 1992. 27(4): p. 49-24. Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. 	paedic advances, ry, and bracing. 55. 255-9.	5 5 5 5
 Gunnoe, B.A., Adolescent idiopathic scoliosis. Orthopaedic review, 1990. 19(1): p. 35-43. Shindle, M.K., et al., Adolescent idiopathic scoliosis: modern management guidelines. Journal of surgical orthop 2006. 15(1): p. 43-52. Parent, S., P.O. Newton, and D.R. Wenger, Adolescent idiopathic scoliosis: etiology, anatomy, natural histor Instructional course lectures, 2005. 54(ifc, 7507149): p. 529-36. Rinsky, L.A., Advances in management of idiopathic scoliosis. Hospital practice (Office ed.), 1992. 27(4): p. 49-24. Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. 	paedic advances, ry, and bracing. 55. 255-9.	5 5 5
 21. 2006. 15(1): p. 43-52. Parent, S., P.O. Newton, and D.R. Wenger, Adolescent idiopathic scoliosis: etiology, anatomy, natural histor Instructional course lectures, 2005. 54(ifc, 7507149): p. 529-36. 23. Rinsky, L.A., Advances in management of idiopathic scoliosis. Hospital practice (Office ed.), 1992. 27(4): p. 49-51 24. Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis. Neurosurgery clinics of North America, 2007. 18(2): p. 2 25. Weinstein, S.L., Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. 26. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: Lenke type LVL case studies. Neurosurgery clinics of 	ry, and bracing. 55. 255-9.	5 5
 Instructional course lectures, 2005. 54(ifc, 7507149): p. 529-36. Rinsky, L.A., Advances in management of idiopathic scoliosis. Hospital practice (Office ed.), 1992. 27(4): p. 49-5 Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis. Neurosurgery clinics of North America, 2007. 18(2): p. 2 Weinstein, S.L., Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. 	55. 255-9.	5
 Rinsky, L.A., Advances in management of idiopathic scoliosis. Hospital practice (Office ed.), 1992. 27(4): p. 49-3 Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis. Neurosurgery clinics of North America, 2007. 18(2): p. 2 Weinstein, S.L., Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. Arlet, V. and V. Beddi, Adolescent idiopathic scoliosis: Lenke type LVL case studies. Neurosurgery clinics of 	255-9.	
 Weinstein, S.L., Adolescent idiopathic scoliosis: prevalence and natural history. Instructional course lecture 7507149): p. 115-28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ck1, 9214582): p. 59-63. Arlet V, and V, Beddi. Adolescent idiopathic scoliosis: Lenke type LVL case studies. Neurosurgery clinics of the studies of the studies of the studies of the studies. Studies in the studies of the studies of the studies of the studies of the studies. 		
 25. 7507149): p. 115-28. 26. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: natural history and prognosis. Studies in health technology 2002. 91(ek1, 9214582): p. 59-63. Arlet V, and V, Beddi. Adolescent idiopathic scoliosis: Lenke type LVL case studies. Neurosurgery clinics of a studies. 	es, 1989. 38(ifc,	5
 200. 91(ck1, 9214582): p. 59-63. Arlet V and V Beddi. Adolescent idionathic scoliosis: Lenke type LVL case studies. Neurosurgery clinics of 		5
Arlet, V. and V. Reddi, Adolescent idiopathic scoliosis: Lenke type I-VI case studies. Neurosurgery clinics of	and informatics,	5
2/. 2007. 18 (2): p. e1-24.	North America,	5
28. Goldberg, C.J., et al., Adolescent idiopathic scoliosis: is the search for aetiology constrained by the orthosis? Stechnology and informatics, 2002. 88(ck1, 9214582): p. 222-5.	Studies in health	5
29. Castro, F.P., Jr., <i>Adolescent idiopathic scoliosis, bracing, and the Hueter-Volkmann principle.</i> The spine journal of the North American Spine Society, 2003. 3 (3): p. 180-5.	: official journal	13
 Hansen, T.B., [Adolescent idiopatisc scoliosis among girls in the Herning region. A follow-up of girls with adole scoliosis found in an earlier screening at school]. Adolescent idiopatisk skoliose hos piger i Herningomradet. En et af piger med adolescent idiopatisk skoliose fundet i en tidligere skolescreening, 1994. 156(35): p. 4979-82. 		6
31. Gurr, J.F., A school screening program that works. The Canadian nurse, 1977. 73(12): p. 24-9.		13
32. Rowe, D.E., et al., A meta-analysis of the efficacy of non-operative treatments for idiopathic scoliosis. The Jour joint surgery. American volume, 1997. 79 (5): p. 664-74.		5
33. Negrini, S., et al., A controlled prospective study on the efficacy of SEAS.02 exercises in preventing progression mild idiopathic scoliosis. Studies in health technology and informatics, 2006. 123 (ck1, 9214582): p. 523-6.	n and bracing in	13
34. Scott, M.M. and H. Piggott, <i>A short-term follow-up of patients with mild scoliosis.</i> The Journal of bone and joint volume, 1981. 63B (4): p. 523-5.		9
35. Zhang, Y., et al., [A clinical follow-up study on treatment of adolescent idiopathic scoliosis with brace]. Zhongh [Chinese journal of surgery], 2007. 45 (8): p. 529-32.	ua wai ke za zhi	6
36. Chen, Z. and L. Rong, <i>Comparison of combined anterior-posterior approach versus posterior-only approach in tre-</i> <i>idiopathic scoliosis: a meta-analysis.</i> European Spine Journal, 2016. 25 (2): p. 363-371.	eating adolescent	5
37. Will, R.E., et al., <i>Cobb Angle Progression in Adolescent Scoliosis Begins at the Intervertebral Disc.</i> Spine, 2009. 2786.	34 (25): p. 2782-	9
38. Kadoury, S. and H. Labelle, <i>Classification of three-dimensional thoracic deformities in adolescent idiopathic .</i> multivariate analysis. European Spine Journal, 2012. 21 (1): p. 40-49.	scoliosis from a	11
39. De Giorgi, S., et al., <i>Chneau brace for adolescent idiopathic scoliosis: long-term results. Can it prevent surgery?</i> Journal, 2013. 22 : p. S815-S822.	European Spine	9
40. Corradin, M., et al., <i>Cervical sagittal alignment variations in adolescent idiopathic scoliosis patients treated with sacral orthosis.</i> European Spine Journal, 2017. 26 (4): p. 1217-1224.	thoraco-lumbo-	13
41. Maruyama, T., Bracing adolescent idiopathic scoliosis: A systematic review of the literature of effective conser- looking for end results 5 years after weaning. Disability and Rehabilitation, 2008. 30 (10): p. 786-791.	vative treatment	5
42. Lusini, M., et al., Brace treatment is effective in idiopathic scoliosis over 45 degrees: an observational procontrolled study. Spine Journal, 2014. 14(9): p. 1951-1956.	ospective cohort	13
43. Schiller, J.R., N.A. Thakur, and C.P. Eberson, <i>Brace Management in Adolescent Idiopathic Scoliosis</i> . Clinical O Related Research, 2010. 468 (3): p. 670-678.	Orthopaedics and	5
44. Gilbert, S.R., et al., BMI and Magnitude of Scoliosis at Presentation to a Specialty Clinic. Pediatrics, 2015. 13 E1424.	35(6): p. E1417-	5
45. Wang, X., et al., Biomechanical modeling and analysis of a direct incremental segmental translation system for the of scoliotic deformities. Clinical Biomechanics, 2011. 26(6): p. 548-555.	e instrumentation	13
46. Shaughnessy, W.J., Advances in scoliosis brace treatment for adolescent idiopathic scoliosis. The Orthopedic America, 2007. 38 (4): p. 469-v.	clinics of North	5
 Allington, N.J. and J.R. Bowen, Adolescent idiopathic scoliosis: Treatment with the Wilmington brace - A compar and part-time use. Journal of Bone and Joint Surgery-American Volume, 1996. 78A(7): p. 1056-1062. 	rison of full-time	9
 48. Osteoprosis international : a journal established as result of cooperation between the European Foundation for C the National Osteoprosis Secondation of the USA, 2005. 16(12): p. 1926-32. 		11
 49. Reamy, B.V. and J.B. Slakey, Adolescent idiopathic scoliosis: Review and current concepts. American Family 1 64(1): p. 111-116. 	Physician, 2001.	5
50. Bowen, J.R., K.A. Keeler, and S. Pelegie, <i>Adolescent idiopathic scoliosis managed by a nighttime bending brac</i> 2001. 24 (10): p. 967-70.	ce. Orthopedics,	13
 51. Skaggs, D.L. and G.S. Bassett, Adolescent idiopathic scoliosis: an update. American family physician, 1996. 53(7): p 2327-35	5
 Skagg, D.L. and O.S. Basset, Adorescent haloplantic scoress, an apparent Anteneral name physician, 1990. 30 (Seifert, J., F. Thielemann, and P. Bernstein, [Adolescent idiopathic scoliosis : Guideline for practical application idiopathische Skoliose : Leitfaden fur die praktische Anwendung., 2016. 45(6): p. 509-17. 		6
 Ajemba, P.O., et al., A support vectors classifier approach to predicting the risk of progression of adolescent idio, IEEE transactions on information technology in biomedicine : a publication of the IEEE Engineering in Medici Society, 2005. 9(2): p. 276-82. 		13
2,	pathic scoliosis.	13

55.	Telang, S.S., et al., A large adolescent idiopathic scoliosis curve in a skeletally immature patient: is early surgery the correct approach? Overview of available evidence. Journal of spinal disorders & techniques, 2006. 19 (7): p. 534-40.	5
56.	Luk, K.D.K., et al., Assessment of skeletal maturity in scoliosis patients to determine clinical management: a new classification scheme using distal radius and ulna radiographs. Spine Journal, 2014. 14(2): p. 315-325.	13
57.	Samdani, A.F., et al., Anterior Vertebral Body Tethering for Idiopathic Scoliosis Two-Year Results. Spine, 2014. 39(20): p. 1688-1693.	4
58.	Schulze, A., et al., Adolescent scoliosis. From deformity to treatment. Orthopade, 2015. 44(11): p. 836-844.	6
59.	Ede, M.M.P.N. and S.W. Jones, Adolescent idiopathic scoliosis: evidence for intrinsic factors driving aetiology and progression. International Orthopaedics, 2016. 40(10): p. 2075-2080.	2
60.	Connolly, P., et al., Adolescent Idiopathic Scoliosis - Long-Term Effect of Instrumentation Extending to the Lumbar Spine. Journal of Bone and Joint Surgery-American Volume, 1995. 77A(8): p. 1210-1216.	4
61.	Haasbeek, J.F., Adolescent idiopathic scoliosis. Postgraduate Medicine, 1997. 101(6): p. 207-&.	5
62.	Cheng, J.C., et al., Adolescent idiopathic scoliosis. Nature Reviews Disease Primers, 2015. 1: p. 15030.	5
63.	Fayssoux, R.S., R.H. Cho, and M.J. Herman, A History of Bracing for Idiopathic Scoliosis in North America. Clinical Orthopaedics and Related Research, 2010. 468 (3): p. 654-664.	5
64.	Zhang, J., et al., Vitamin D enhanced the osteogenic response of adolescent idiopathic scoliosis primary osteoblasts to vibration. Journal of orthopaedic translation. Conference: 2016 international combined meeting of orthopaedic research societies, ICORS 2016. China, 2016. 7: p. 113.	5
65.	Vira, S., et al., <i>The Interobserver and Intraobserver Reliability of the Sanders Classification Versus the Risser Stage</i> . Journal of pediatric orthopaedics, 2016. (no pagination).	13
66.	Wong, M.S., et al., The effect of rigid versus flexible spinal orthosis on the clinical efficacy and acceptance of the patients with adolescent idiopathic scoliosis. Spine, 2008. 33(12): p. 1360-1365.	13
67.	Den Boer, W.A., et al., Treatment of idiopathic scoliosis with side-shift therapy: an initial comparison with a brace treatment historical cohort. European spine journal, 1999. 8(5): p. 406-410.	13
68.	Yip, B.H.K., et al., <i>Treating low bone mass with calcium and Vitamin D supplementation in girls with adolescent idiopathic scoliosis (AIS) - A randomized double-blinded placebo-controlled trial.</i> Journal of bone and mineral research. Conference: 2015 annual meeting of the american society for bone and mineral research, ASBMR 2015. United states, 2015. 30 (Supplement 1) (no pagination).	5
69.	Bettany-Saltikov, J., et al., Surgical versus non-surgical interventions in people with adolescent idiopathic scoliosis. Cochrane Database of Systematic Reviews, 2015(4).	5
70.	Schreiber, S., et al., Schroth physiotherapeutic scoliosis-specific exercises for adolescent idiopathic scoliosis: how many patients require treatment to prevent one deterioration? - results from a randomized controlled trial - "SOSORT 2017 Award Winner". Scoliosis and spinal disorders, 2017. 12 (1) (no pagination).	13
71.	Wiemann, J.M., S.A. Shah, and C.T. Price, Nighttime bracing versus observation for early adolescent idiopathic scoliosis. Journal of pediatric orthopaedics, 2014. 34(6): p. 603-606.	13
72.	Bettany-Saltikov, J., et al., <i>Surgical versus non-surgical interventions for adolescent idiopathic scoliosis: a Cochrane review protocol.</i> Scoliosis. Conference: 10th international conference on conservative management of spinal deformities - SOSORT 2013 annual meeting united states. Conference start: 20130508 conference end: 20130511, 2013. 8(no pagination).	5
73.	Negrini, S., et al., Specific exercises reduce brace prescription in adolescent idiopathic scoliosis: a prospective controlled cohort study with worst-case analysis. Journal of rehabilitation medicine, 2008. 40 (6): p. 451-455.	13
74.	Schreiber, S., et al., Schroth Physiotherapeutic Scoliosis-Specific Exercises Added to the Standard of Care Lead to Better Cobb Angle Outcomes in Adolescents with Idiopathic Scoliosis - an Assessor and Statistician Blinded Randomized Controlled Trial. Plos one, 2016. 11(12): p. e0168746.	13
75.	Danielsson, A.J. and A.L. Nachemson, Radiologic findings and curve progression 22 years after treatment for adolescent idiopathic scoliosis: comparison of brace and surgical treatment with matching control group of straight individuals. Spine, 2001. 26(5): p. 516-525.	13
76.	Yankey, J., et al., <i>Practical implications of inter-rater agreement in a randomized controlled clinical trial.</i> Clinical trials., 2010. 7(4): p. 436.	5
77.	Deng, M., et al., MRI-based morphological evidence of spinal cord tethering predicts curve progression in adolescent idiopathic scoliosis. Spine journal, 2015. 15 (6): p. 1391-1401.	13
78.	Negrini, S. and G. Marchini, <i>Efficacy of the symmetric, patient-oriented, rigid, three-dimensional, active (SPoRT) concept of bracing for scoliosis: a prospective study of the Sforzesco versus Lyon brace.</i> Europa medicophysica, 2007. 43 (2): p. 171-81; discussion 183-4.	13
79.	Schreiber, S., et al., Feasibility and three months preliminary results of an RCT on the effect of Schroth exercises in adolescent idiopathic scoliosis (AIS). Scoliosis. Conference: 9th international conference on conservative management of spinal deformities - SOSORT 2012 annual meeting italy. Conference start: 20120510 conference end: 20120512, 2013. 8(no pagination).	5
80.	Cobetto, N., et al., Effectiveness of braces designed using computer-aided design and manufacturing (CAD/CAM) and finite element simulation compared to CAD/CAM only for the conservative treatment of adolescent idiopathic scoliosis: a prospective randomized controlled trial. European spine journal, 2016. 25(10): p. 3056-3064.	13
81.	Bunge, E.M. and H.J. de Koning, Bracing patients with inopathic scoliosis: design of the Dutch randomized controlled treatment trial. BMC musculoskeletal disorders, 2008. 9: p. 57.	8
82.	Sanders, J.O., et al., Bracing for idiopathic scoliosis: how many patients require treatment to prevent one surgery? Journal of bone and joint surgery - american volume, 2014. 96(8): p. 649-653.	13
83.	Sponseller, P.D., Bracing for adolescent idiopathic scoliosis in practice today. Journal of pediatric orthopaedics, 2011. 31(1 SUPPL.): p. S53-S60.	5
84.	Weinstein, S.L., et al., <i>Effects of bracing in adolescents with idiopathic scoliosis</i> . New England journal of medicine, 2013. 369 (16): p. 1512-1521.	13
85.	Weinstein, S.L., et al., Design of the Bracing in Adolescent Idiopathic Scoliosis Trial (BrAIST). Spine, 2013. 38(21): p. 1832-1841.	13
86.	Weiss, H.R. and G. Weiss, Curvature progression in patients treated with scoliosis in-patient rehabilitationa sex and age matched controlled study. Studies in health technology and informatics, 2002. 91: p. 352-356.	9
87.	Nct, Calcium + Vitamin D Supplementation for Low Bone Mass in Adolescent Idiopathic Scoliosis (AIS). Https://clinicaltrials.gov/show/nct01103115, 2010.	5
88.	Negrini, S., et al., Braces for idiopathic scoliosis in adolescents. A cochrane review. Scoliosis, 2010. 5(no pagination).	5

89.	Negrini, S., et al., Braces for Idiopathic Scoliosis in Adolescents. Spine, 2016. 41(23): p. 1813-1825.	5
90.	Schwieger, T., et al., Body Image and Quality of Life and Brace Wear Adherence in Females With Adolescent Idiopathic Scoliosis. Journal of pediatric orthopedics, 2017. 37(8): p. e519-e523.	13
91.	Schwieger, et al., Body image and quality-of-life in untreated versus brace-treated females with adolescent idiopathic scoliosis. Spine, 2016. 41 : p. 311-319.	13
92.	Liang, J., et al., <i>Efficacy of three-dimensionally integrated exercise for scoliosis in patients with adolescent idiopathic scoliosis:</i> study protocol for a randomized controlled trial. Trials, 2018. 19 : p. 485.	13
93.	Zi-qiang, C., et al., Factors as predictors for thoracic and thoracolumbar/lumbar structural curves in adolescent idiopathic scoliosis. Chinese Medical Journal, 2012. 125 (8): p. 1439-1442.	9
94.	Negrini, S., et al., Exercises reduce the progression rate of adolescent idiopathic scoliosis: Results of a comprehensive systematic review of the literature. Disability and Rehabilitation, 2008. 30 (10): p. 772-785.	5
95.	Pesenti, S., et al., Evolution of adolescent idiopathic scoliosis: Results of a multicenter study at 20 years' follow-up. Orthopaedics & Traumatology-Surgery & Research, 2015. 101(5): p. 619-622.	13
96.	Jaremko, J.L., et al., <i>Estimation of spinal deformity in scoliosis from torso surface cross sections</i> . Spine, 2001. 26(14): p. 1583-1591.	8
97.	Zheng, Y., et al., <i>EPIDEMIOLOGICAL STUDY OF ADOLESCENT IDIOPATHIC SCOLIOSIS IN EASTERN CHINA</i> . Journal of Rehabilitation Medicine, 2017. 49 (6): p. 512-519.	5
98.	Lateur, G., et al., <i>Efficacy of nighttime brace in preventing progression of idiopathic scoliosis of less than 25 degrees</i> . Orthopaedics & Traumatology-Surgery & Research, 2017. 103 (2): p. 275-278.	13
99.	Mordecai, S.C. and H.V. Dabke, <i>Efficacy of exercise therapy for the treatment of adolescent idiopathic scoliosis: a review of the literature</i> . European Spine Journal, 2012. 21 (3): p. 382-389.	5
100.	Korovessis, P., et al., Effects of thoracolumbosacral orthosis on spinal deformities, trunk asymmetry, and frontal lower rib cage in adolescent idiopathic scoliosis. Spine, 2000. 25(16): p. 2064-2071.	13
101.	Yrjonen, T., et al., Effectiveness of the Providence nighttime bracing in adolescent idiopathic scoliosis: a comparative study of 36 female patients. European Spine Journal, 2006. 15(7): p. 1139-1143.	13
102.	Allard, P., et al., Effect of body morphology on standing balance in adolescent idiopathic scoliosis. American Journal of Physical Medicine & Rehabilitation, 2004. 83(9): p. 689-697.	5
103.	Anwer, S., et al., Effects of Exercise on Spinal Deformities and Quality of Life in Patients with Adolescent Idiopathic Scoliosis. Biomed Research International, 2015: p. 123848.	5
104.	Nachemson, A., et al., Effectiveness of Treatment with a Brace in Girls Who Have Adolescent Idiopathic Scoliosis - a Prospective, Controlled-Study Based on Data from the Brace Study of the Scoliosis-Research-Society. Journal of Bone and Joint Surgery- American Volume, 1995. 77A(6): p. 815-822.	13
105.	Coillard, C., et al., Effectiveness of the SpineCor brace based on the new standardized criteria proposed by the Scoliosis Research Society for adolescent idiopathic scoliosis. Journal of Pediatric Orthopaedics, 2007. 27(4): p. 375-379.	13
106.	Lee, C.S., et al., <i>Effectiveness of the Charleston Night-time Bending Brace in the Treatment of Adolescent Idiopathic Scoliosis.</i> Journal of Pediatric Orthopaedics, 2012. 32 (4): p. 368-372.	13
107.	Wiley, J.W., et al., <i>Effectiveness of the Boston brace in treatment of large curves in adolescent idiopathic scoliosis.</i> Spine, 2000. 25 (18): p. 2326-2332.	13
108.	Karol, L.A., et al., <i>Effect of Compliance Counseling on Brace Use and Success in Patients with Adolescent Idiopathic Scoliosis.</i> Journal of Bone and Joint Surgery-American Volume, 2016. 98 (1): p. 9-14.	13
109.	Courtois, I., et al., Bone mineral density at the femur and lumbar spine in a population of young women treated for scoliosis in adolescence. Revue du rhumatisme (English ed.), 1999. 66(12): p. 705-10.	13
110.	Theologis, T.N., et al., Early detection of progression in adolescent idiopathic scoliosis by measurement of changes in back shape with the integrated shape imaging system scanner. Spine, 1997. 22(11): p. 1223-1227.	13
111.	Modi, H.N., et al., Drooping of Apical Convex Rib-vertebral Angle in Adolescent Idiopathic Scoliosis of More Than 40 Degrees A Prognostic Factor for Progression. Journal of Spinal Disorders & Techniques, 2009. 22(5): p. 367-371.	13
112.	Lyu, Q., et al., Does spinal deformity correction of non-dystrophic scoliosis in neurofibromatosis type I with one-stage posterior pedicle screw technique produce outcomes similar to adolescent idiopathic scoliosis? Spine Journal, 2017. 17(12): p. 1850-1858.	4
113.	Roye, B.D., et al., Does ScoliScore Provide More Information Than Traditional Clinical Estimates of Curve Progression? Spine, 2012. 37(25): p. 2099-2103.	2
114.	Wang, W., et al., Different Curve Pattern and Other Radiographical Characteristics in Male and Female Patients With Adolescent Idiopathic Scoliosis. Spine, 2012. 37(18): p. 1586-1592.	13
115.	Yang, J.H., et al., Does delaying surgery in immature adolescent idiopathic scoliosis patients with progressive curve, lead to addition of fusion levels? European Spine Journal, 2014. 23(12): p. 2672-2679.	13
116.	Ersen, O., et al., Difference between Spinecor brace and Thoracolumbosacral orthosis for deformity correction and quality of life in adolescent idiopathic scoliosis. Acta Orthopaedica Belgica, 2016. 82(4): p. 710-714.	13
117.	Burton, M.S., Diagnosis and Treatment of Adolescent Idiopathic Scoliosis. Pediatric Annals, 2013. 42(11): p. 224-228.	5
118.	Wever, D.J., et al., <i>Curve progression and spinal growth in brace treated idiopathic scoliosis</i> . Clinical Orthopaedics and Related Research, 2000(377): p. 169-179.	9
119.	Cassella, M. and J. Hall, Current Treatment Approaches in the Nonoperative and Operative Management of Adolescent Idiopathic Scoliosis. Physical Therapy, 1991. 71(12): p. 897-909.	5
120.	Djurasovic, M. and S.D. Glassman, Correlation of radiographic and clinical findings in spinal deformities. Neurosurgery Clinics of North America, 2007. 18 (2): p. 223-+.	5
121.	Aulisa, A.G., et al., Correlation Between Hump Dimensions and Curve Severity in Idiopathic Scoliosis Before and After Conservative Treatment. Spine, 2018. 43 (2): p. 114-119.	13
122.	Lechner, R., et al., Comparison of two- and three-dimensional measurement of the Cobb angle in scoliosis. International Orthopaedics, 2017. 41(5): p. 957-962.	8
123.	Sud, A. and A.I. Tsirikos, Current concepts and controversies on adolescent idiopathic scoliosis: Part I. Indian Journal of Orthopaedics, 2013. 47(2): p. 117-128.	5
124.	Wong, M.S. and W.C. Liu, Critical review on non-operative management of adolescent idiopathic scoliosis. Prosthetics and Orthotics International, 2003. 27(3): p. 242-253.	5
125.	Korovessis, P., et al., Vertical transmission of the hip rolls due to wearing of TLSO for scoliosis. Journal of Spinal Disorders, 1996. 9(4): p. 326-333.	9

126.	Asher, M. and L. Cook, <i>The Transverse Plane Evolution of the Most Common Adolescent Idiopathic Scoliosis Deformities - a Cross-Sectional Study of 181 Patients.</i> Spine, 1995. 20 (12): p. 1386-1391.	5
127.	Lonstein, J. and R. Winter, <i>The Milwaukee Brace for the Treatment of Adolescent Idiopathic Scoliosis - a Review of 1,020 Patients</i> . Journal of Bone and Joint Surgery-American Volume, 1994. 76A (8): p. 1207-1221.	13
128.	Scherrer, SA., et al., Three-Dimensional Vertebral Wedging in Mild and Moderate Adolescent Idiopathic Scoliosis. Plos One, 2013. 8(8): p. e71504.	13
129.	Cho, R.H., et al., Which Lenke 1A Curves Are at the Greatest Risk for Adding-On and Why? Spine, 2012. 37(16): p. 1384-1390.	4
130.	Aulisa, A.G., et al., Treatment of Lumbar Curves in Scoliotic Adolescent Females With Progressive Action Short Brace A Case Series Based on the Scoliosis Research Society Committee Criteria. Spine, 2012. 37 (13): p. E786-E791.	13
131.	Kim, H., et al., Scoliosis Imaging: What Radiologists Should Know. Radiographics, 2010. 30(7): p. 1823-1842.	5
132.	Jiang, J., et al., The influence of elastic orthotic belt on sagittal profile in adolescent idiopathic thoracic scoliosis: a comparative radiographic study with Milwaukee brace. Bmc Musculoskeletal Disorders, 2010. 11: p. 219.	13
133.	Wong, HK. and KJ. Tan, <i>The natural history of adolescent idiopathic scoliosis</i> . Indian Journal of Orthopaedics, 2010. 44(1): p. 9-13.	5
134.	Cilli, K., et al., School screening for scoliosis in Sivas, Turkey. Acta Orthopaedica Et Traumatologica Turcica, 2009. 43(5): p. 426-430.	13
135.	Kim, H.J., J.S. Blanco, and R.F. Widmann, <i>Update on the management of idiopathic scoliosis</i> . Current Opinion in Pediatrics, 2009. 21 (1): p. 55-64.	5
136.	Goldberg, C.J., et al., <i>Scoliosis: a review</i> . Pediatric Surgery International, 2008. 24(2): p. 129-144.	5
137.	Kouwenhoven, JW.M. and R.M. Castelein, <i>The Pathogenesis of Adolescent Idiopathic Scoliosis Review of the Literature</i> . Spine, 2008. 33 (26): p. 2898-2908.	5
138.	Janicki, J.A. and B. Alman, Scoliosis: Review of diagnosis and treatment. Paediatrics & Child Health, 2007. 12(9): p. 771-776.	5
139.	Zeh, A., et al., <i>The flexible Triac (TM)-Brace for conservative treatment of idiopathic scoliosis. An alternative treatment option</i> ? Acta Orthopaedica Belgica, 2008. 74(4): p. 512-521.	13
140.	Dohnert, M.B. and E. Tomasi, Validity of computed photogrammetry for detecting idiopathic scoliosis in adolescents. Revista Brasileira De Fisioterapia, 2008. 12 (4): p. 290-297.	5
141.	McIntire, K.L., et al., <i>Treatment of adolescent idiopathic scoliosis with quantified trunk rotational strength training - A pilot study</i> . Journal of Spinal Disorders & Techniques, 2008. 21 (5): p. 349-358.	13
142.	Pazos, V., et al., <i>Reliability of trunk shape measurements based on 3-D surface reconstructions</i> . European Spine Journal, 2007. 16 (11): p. 1882-1891.	13
143.	Dolan, L.A. and S.L. Weinstein, <i>Surgical rates after observation and bracing for adolescent idiopathic scoliosis</i> . Spine, 2007. 32 (19): p. S91-S100.	5
144.	Van Goethem, J., et al., Scoliosis. Neuroimaging Clinics of North America, 2007. 17(1): p. 105-+.	5
145.	Bunnell, W.P., <i>Selective screening for scoliosis</i> . Clinical Orthopaedics and Related Research, 2005(434): p. 40-45. Di Felice, F., et al., <i>The Natural History of Idiopathic Scoliosis During Growth A Meta-Analysis</i> . American Journal of Physical	5
146.	Medicine & Rehabilitation, 2018. 97(5): p. 346-356.	5
147.	Hung, A.LH., et al., Validation Study of the Thumb Ossification Composite Index (TOCI) in Idiopathic Scoliosis. Journal of Bone and Joint Surgery-American Volume, 2018. 100 (13): p. e88.	13
148.	Grossman, D.C., et al., Screening for Adolescent Idiopathic Scoliosis US Preventive Services Task Force Recommendation Statement. Jama-Journal of the American Medical Association, 2018. 319 (2): p. 165-172.	13
149.	Richards, B.S., et al., Standardization of criteria for adolescent idiopathic scoliosis brace studies - SRS Committee on Bracing and Nonoperative Management. Spine, 2005. 30 (18): p. 2068-2075.	5
150.	Rudnick, S.B., et al., <i>Scoliosis severity does not impact the risk of scoliosis in family members</i> . Journal of Pediatric Orthopaedics-Part B, 2018. 27 (2): p. 147-151.	13
151.	Dunn, J., et al., Screening for Adolescent Idiopathic Scoliosis Evidence Report and Systematic Review for the US Preventive Services Task Force. Jama-Journal of the American Medical Association, 2018. 319 (2): p. 173-187.	5
152.	Li, Z., et al., <i>The radiographic parameter risk factors of rapid curve progression in Lenke 5 and 6 adolescent idiopathic scoliosis:</i> A retrospective study. Medicine, 2017. 96 (52): p. e9425.	9
153.	Illes, T.S., et al., <i>The horizontal plane appearances of scoliosis: what information can be obtained from top-view images?</i> International Orthopaedics, 2017. 41 (11): p. 2303-2311.	13
154.	Hong, A., et al., Surface Topography Classification Trees for Assessing Severity and Monitoring Progression in Adolescent Idiopathic Scoliosis. Spine, 2017. 42 (13): p. E781-E787.	13
155.	Sun, X., et al., Rib-vertebral angle measurements predict brace treatment outcome in Risser grade 0 and premenarchal girls with adolescent idiopathic scoliosis. European Spine Journal, 2016. 25(10): p. 3088-3094.	9
156.	Karimi, M.T., et al., The influence of trunk bracing on time, spatial parameters, symmetry of ground reaction, force and loading rate in adolescent idiopathic scoliosis. Journal of Mechanics in Medicine and Biology, 2016. 16(2): p. 1650009.	13
157.	Cheng, J.C.Y. and X. Guo, Osteopenia in adolescent idiopathic scoliosis - A primary problem or secondary to the spinal deformity? Spine, 1997. 22(15): p. 1716-1721.	13
158.	Garcia, M.R., et al., Ossification of the phalanges of the foot and its relationship to peak height velocity and the calcaneal system. Journal of Childrens Orthopaedics, 2018. 12 (1): p. 84-90.	13
159.	Cottalorda, J., et al., Orthoses for mild scoliosis: A prospective study comparing traditional plaster mold manufacturing with fast, noncontact, 3-dimensional acquisition. Spine, 2005. 30 (4): p. 399-405.	13
160.	Price, C.T., et al., Nighttime bracing for adolescent idiopathic scoliosis with the Charleston Bending Brace: Long-term follow-up. Journal of Pediatric Orthopaedics, 1997. 17(6): p. 703-707.	13
161.	Upadhyay, S., et al., New Prognostic Factors to Predict the Final Outcome of Brace Treatment. Spine, 1995. 20(5): p. 537-545.	9
162.	Schlosser, T.P.C., et al., Natural sagittal spino-pelvic alignment in boys and girls before, at and after the adolescent growth spurt. European Spine Journal, 2015. 24(6): p. 1158-1167.	9
163.	Agabegi, S.S., et al., <i>Natural History of Adolescent Idiopathic Scoliosis in Skeletally Mature Patients: A Critical Review</i> . Journal of the American Academy of Orthopaedic Surgeons, 2015. 23(12): p. 714-723.	5
164.	Komeili, A., et al., Monitoring for idiopathic scoliosis curve progression using surface topography asymmetry analysis of the torso in adolescents. Spine Journal, 2015. 15 (4): p. 743-751.	13
165.	Chen, P.Q., Management of scoliosis. Journal of the Formosan Medical Association, 2003. 102(11): p. 751-761.	5

166.	Haber, L.L., et al., Long-term Retrospective of the Kaneda Anterior Scoliosis System in Thoracic Adolescent Idiopathic Scoliosis. Journal of Pediatric Orthopaedics, 2012. 32 (4): p. 362-367.	4
167.	Sanders, J.O., et al., <i>Maturity assessment and curve progression in girls with idiopathic scoliosis</i> . Journal of Bone and Joint Surgery-American Volume, 2007. 89A (1): p. 64-73.	9
168.	Aulisa, A.G., et al., Lyon bracing in adolescent females with thoracic idiopathic scoliosis: a prospective study based on SRS and SOSORT criteria. Bmc Musculoskeletal Disorders, 2015. 16: p. 316.	13
169.	Frez, R., J.C. Cheng, and E.M. Wong, Longitudinal changes in trunkal balance after selective fusion of King II curves in adolescent idiopathic scoliosis. Spine, 2000. 25(11): p. 1352-9.	4
170.	Lee, C.S., et al., <i>Is it enough to stop distal fusion at L3 in adolescent idiopathic scoliosis with major thoracolumbar/lumbar curves?</i> European Spine Journal, 2016. 25 (10): p. 3256-3264.	9
171.	Hardesty, C.K., et al., Interobserver Variability Using a Commercially Available System of Archived Digital Radiography With Integrated Computer-assisted Measurements for Scoliosis Cobb Angles. Journal of Pediatric Orthopaedics, 2013. 33(2): p. 163- 169.	13
172.	Dang, N.R., et al., Intra-observer reproducibility and interobserver reliability of the radiographic parameters in the Spinal Deformity Study Group's AIS Radiographic Measurement Manual. Spine, 2005. 30(9): p. 1064-1069.	5
173.	Xu, L., et al., Initial Correction Rate Can be Predictive of the Outcome of Brace Treatment in Patients With Adolescent Idiopathic Scoliosis. Clinical Spine Surgery, 2017. 30(4): p. E475-E479.	13
174.	Jaremko, J.L., et al., Indices of torso asymmetry related to spinal deformity in scoliosis. Clinical Biomechanics, 2002. 17(8): p. 559-568.	13
175.	Calman, R., T. Smithers, and R. Rowan, Impact of surgical waiting time on paediatric spinal deformity patients. Anz Journal of Surgery, 2013. 83(12): p. 929-932.	2
176.	Negrini, S., et al., Idiopathic scoliosis patients with curves more than 45 Cobb degrees refusing surgery can be effectively treated through bracing with curve improvements. Spine Journal, 2011. 11(5): p. 369-380.	13
177.	Escalada, F., et al., Growth and curve stabilization in girls with adolescent idiopathic scoliosis. Spine, 2005. 30(4): p. 411-417.	9
178.	Hacquebord, J.H. and S.S. Leopold, In Brief: The Risser Classification: A Classic Tool for the Clinician Treating Adolescent Idiopathic Scoliosis. Clinical Orthopaedics and Related Research, 2012. 470 (8): p. 2335-2338.	5
179.	Cassar-Pullicino, V.N. and S.M. Eisenstein, <i>Imaging in scoliosis: What, why and how?</i> Clinical Radiology, 2002. 57(7): p. 543-562.	5
180.	Angevine, P.D. and H. Deutsch, <i>Idiopathic scoliosis</i> . Neurosurgery, 2008. 63(3): p. A86-A93.	5
181.	Courvoisier, A., et al., <i>Growth tethering devices for idiopathic scoliosis</i> . Expert Review of Medical Devices, 2015. 12 (4): p. 449-456.	5
182.	Dimeglio, A., F. Canavese, and P. Charles, <i>Growth and Adolescent Idiopathic Scoliosis: When and How Much?</i> Journal of Pediatric Orthopaedics, 2011. 31 : p. S28-S36.	5
183.	Cheung, J., et al., Geometric and electromyographic assessments in the evaluation of curve progression in idiopathic scoliosis. Spine, 2006. 31 (3): p. 322-329.	9
184.	Lee, W.T.K., et al., Generalized low bone mass of girls with adolescent idiopathic scoliosis is related to inadequate calcium intake and weight bearing physical activity in peripubertal period. Osteoporosis International, 2005. 16 (9): p. 1024-1035.	5
185.	Souder, C., et al., Factors in Surgical Decision Making for Thoracolumbar/Lumbar AIS: It's About More Than Just the Curve Magnitude. Journal of Pediatric Orthopaedics, 2017. 37 (8): p. E530-E535.	4
186.	Jada, A., et al., Evaluation and management of adolescent idiopathic scoliosis: a review. Neurosurgical Focus, 2017. 43(4): p. E2.	5
187.	Kong, QJ., et al., Evaluating the Rotation Correction of the Main Thoracic Curve in Severe Adolescent Idiopathic Scoliosis: Bending and Traction vs. Fulcrum - A Preliminary Report. Medical Science Monitor, 2017. 23: p. 4981-4988.	4
188.	Hung, A.L.H., et al., Thumb Ossification Composite Index (TOCI) for Predicting Peripubertal Skeletal Maturity and Peak Height Velocity in Idiopathic Scoliosis A Validation Study of Premenarchal Girls with Adolescent Idiopathic Scoliosis Followed Longitudinally Until Skeletal Maturity. Journal of Bone and Joint Surgery-American Volume, 2017. 99(17): p. 1438-1446.	11
189.	Bumpass, D.B., et al., Vertebral Body Stapling for Moderate Juvenile and Early Adolescent Idiopathic Scoliosis. Spine, 2015. 40(24): p. E1305-E1314.	9
190.	Deurloo, J.A. and P.H. Verkerk, To screen or not to screen for adolescent idiopathic scoliosis? A review of the literature. Public Health, 2015. 129 (9): p. 1267-1272.	5
191.	Matussek, J., et al., Surgical treatment of scoliosis in childhood and adolescence. Age group and etiology-related indications and choice of instrumentation. Orthopade, 2015. 44(7): p. 577-590.	6
192.	Begon, M., et al., Three-dimensional vertebral wedging and pelvic asymmetries in the early stages of adolescent idiopathic scoliosis. Spine Journal, 2015. 15 (3): p. 477-486.	13
193.	Young, M., et al., Reliability and accuracy of ultrasound measurements with and without the aid of previous radiographs in adolescent idiopathic scoliosis (AIS). European Spine Journal, 2015. 24(7): p. 1427-1433.	13
194.	Keenan, B.E., et al., Segmental torso masses in adolescent idiopathic scoliosis. Clinical Biomechanics, 2014. 29(7): p. 773-779.	13
195.	Pino-Almero, L., et al., <i>Quantification of topographic changes in the surface of back of young patients monitored for idiopathic scoliosis: correlation with radiographic variables.</i> Journal of Biomedical Optics, 2016. 21(11): p. 116001.	2
196.	Villemure, I., et al., Progression of vertebral and spinal three-dimensional deformities in adolescent idiopathic scoliosis - A longitudinal study. Spine, 2001. 26(20): p. 2244-2250.	13
197.	Chalmers, E., et al., Predicting success or failure of brace treatment for adolescents with idiopathic scoliosis. Medical & Biological Engineering & Computing, 2015. 53(10): p. 1001-1009.	8
198.	Ferrero, E., et al., Proximal junctional kyphosis in thoracic adolescent idiopathic scoliosis: risk factors and compensatory mechanisms in a multicenter national cohort. European Spine Journal, 2018. 27(9): p. 2241-2250.	4
199.	Hwang, C.J., et al., Progression of trunk imbalance in adolescent idiopathic scoliosis with a thoracolumbar/lumbar curve: is it predictable at the initial visit? Journal of Neurosurgery-Pediatrics, 2017. 20(5): p. 450-455.	13
200.	Busscher, I., F.H. Wapstra, and A.G. Veldhuizen, Predicting growth and curve progression in the individual patient with adolescent idiopathic scoliosis: design of a prospective longitudinal cohort study. Bmc Musculoskeletal Disorders, 2010. 11: p. 93.	13
201.	Cheng, J.C.Y., X. Guo, and A.H.L. Sher, Persistent osteopenia in adolescent idiopathic scoliosis - A longitudinal follow-up study. Spine, 1999. 24(12): p. 1218-1222.	13
	Spine, 1799. 24(12). p. 1210-1222. Song, K.M. and D.G. Little, <i>Peak height velocity as a maturity indicator for males with idiopathic scoliosis.</i> Journal of Pediatric	

203.	Dalleau, G., et al., Pelvis Morphology, Trunk Posture and Standing Imbalance and Their Relations to the Cobb Angle in Moderate and Severe Untreated AIS. Plos One, 2012. 7(7): p. e36755.	11
204.	Beausejour, M., et al., Patient characteristics at the initial visit to a scoliosis clinic - A cross-sectional study in a community without school screening. Spine, 2007. 32 (12): p. 1349-1354.	5
205.	Gennari, E., et al., Orthotic treatment for adolescent idiopathic scoliosis: evaluation of efficacy based on clinical outcomes. Minerva Ortopedica E Traumatologica, 2012. 63(4): p. 257-261.	13
206.	Harrop, J.S., J. Birknes, and C.I. Shaffrey, Noninvasive measurement and screening techniques for spinal deformities. Neurosurgery, 2008. 63 (3): p. A46-A53.	5
207.	Seoud, L., et al., Multilevel Analysis of Trunk Surface Measurements for Noninvasive Assessment of Scoliosis Deformities. Spine, 2012. 37(17): p. E1045-E1053.	13
208.	Burwell, R.G. and P.H. Dangerfield, <i>Re: Lowe T, Lawellin D, Smith D, et al. Platelet calmodulin levels in adolescent idiopathic scoliosis. Do the levels correlate with curve progression and severity?</i> Spine, 2003. 28 (14): p. 1621-1622.	2
209.	Spoonamore, M.J., L.A. Dolan, and S.L. Weinstein, Use of the Rosenberger brace in the treatment of progressive adolescent idiopathic scoliosis. Spine, 2004. 29(13): p. 1458-1464.	13
210.	Delorme, S., H. Labelle, and CE. Aubin, <i>The Crankshaft Phenomenon Is Cobb Angle Progression a Good Indicator in Adolescent Idiopathic Scoliosis</i> ? Spine, 2002. 27 (6): p. E145-E151.	4
211.	Lenke, L.G., C.C. Edwards, and K.H. Bridwell, <i>The Lenke classification of adolescent idiopathic scoliosis: How it organizes curve patterns as a template to perform selective fusions of the spine</i> . Spine, 2003. 28 (20): p. S199-S207.	4
212.	LeBlanc, R., et al., Relation between adolescent idiopathic scoliosis and morphologic somatotypes. Spine, 1997. 22(21): p. 2532-2536.	13
213.	Hilibrand, A., et al., <i>The Sagittal Alignment of the Cervical-Spine in Adolescent Idiopathic Scoliosis.</i> Journal of Pediatric Orthopaedics, 1995. 15 (5): p. 627-632.	4
214.	Adankon, M.M., et al., <i>Scoliosis Follow-Up Using Noninvasive Trunk Surface Acquisition</i> . Ieee Transactions on Biomedical Engineering, 2013. 60 (8): p. 2262-2270.	2
215.	Stokes, O.M. and K.D.K. Luk, <i>The current status of bracing for patients with adolescent idiopathic scoliosis</i> . Bone & Joint Journal, 2013. 95B (10): p. 1308-1316.	5
216.	Lonstein, J.E., <i>Natural history and school screening for scoliosis</i> . The Orthopedic clinics of North America, 1988. 19 (2): p. 227-37.	5
217.	Guillaumat, M., [Natural history of scoliosis from childhood to old age]. Histoire naturelle des scolioses de l'enfance a la vieillesse., 1999. 183 (4): p. 705-9.	6
218.	Wong, C., Mechanism of right thoracic adolescent idiopathic scoliosis at risk for progression; a unifying pathway of development by normal growth and imbalance. Scoliosis, 2015. 10 (101271527): p. 2.	5
219.	Montgomery, F., S. Willner, and G. Appelgren, <i>Long-term follow-up of patients with adolescent idiopathic scoliosis treated conservatively: an analysis of the clinical value of progression.</i> Journal of pediatric orthopedics, 1990. 10 (1): p. 48-52.	9
220.	Burwell, R.G., et al., <i>Leg-arm length ratios correlate with severity of apical vertebral rotation in girls after school screening for adolescent idiopathic scoliosis (AIS): a dynamic pathomechanism in the initiation of the deformity?</i> Studies in health technology and informatics, 2008. 140 (ck1, 9214582): p. 189-93.	13
221.	Weisz, I., et al., <i>ISIS scanning: a useful assessment technique in the management of scoliosis.</i> Spine, 1988. 13 (4): p. 405-8. Delorme, S., H. Labelle, and CE. Aubin, <i>Is Cobb angle progression a good indicator in adolescent idiopathic scoliosis?</i> Spine,	11
222.	2002. 27(6): p. E145-51. Grivas, T.B., et al., Intervertebral disc biomechanics in the pathogenesis of idiopathic scoliosis. Studies in health technology and	4
223.	informatics, 2006. 123 (ck1, 9214582): p. 80-3. Chan, A.C.Y., et al., Intra- and Interobserver Reliability of the Cobb Angle-Vertebral Rotation Angle-Spinous Process Angle for	9
224.	Adolescent Idiopathic Scoliosis. Spine deformity, 2014. 2(3): p. 168-175. Shi, B., et al., Integrated Multidimensional Maturity Assessments Predicting the High-risk Occurrence of Peak Angle Velocity	13
225.	During Puberty in Progressive Female Idiopathic Scoliosis. Clinical spine surgery, 2017. 30 (4): p. E491-E496. Thompson, G.H. and B.S. Richards Iii, <i>Inclusion and assessment criteria for conservative scoliosis treatment</i> . Studies in health	9
226.	technology and informatics, 2008. 135 (ck1, 9214582): p. 157-63. Weiss, HR., G. Weiss, and F. Petermann, <i>Incidence of curvature progression in idiopathic scoliosis patients treated with scoliosis</i>	5
227.	in-patient rehabilitation (SIR): an age- and sex-matched controlled study. Pediatric rehabilitation, 2003. 6(1): p. 23-30. Donzeli, S., et al., In favour of the definition "adolescents with idiopathic scoliosis": juvenile and adolescent idiopathic scoliosis	2
228.	bonzein, s., et al., in Javour of the definition adolescents with talopathic scottosis : Juvenite and adolescent talopathic scottosis braced after ten years of age, do not show different end results. SOSORT award winner 2014. Scoliosis, 2014. 9(101271527): p. 7.	13
229.	Xu, Hg., et al., Imaging study of wedge changes in the vertebral bodies and intervertebral discs in adolescent idiopathic scoliosis. Orthopaedic surgery, 2009. 1(4): p. 300-4.	13
230.	Dobosiewicz, K., et al., Influence of method of asymmetric trunk mobilization on shaping of a physiological thoracic kyphosis in children and youth suffering from progressive idiopathic scoliosis. Studies in health technology and informatics, 2002. 91 (ck1, 9214582): p. 348-51.	13
231.	Gauchard, G.C., et al., Influence of different types of progressive idiopathic scoliosis on static and dynamic postural control. Spine, 2001. 26(9): p. 1052-8.	13
232.	Cahill, P.J., et al., Infection after spinal fusion for pediatric spinal deformity: thirty years of experience at a single institution. Spine, 2010. 35 (12): p. 1211-7.	4
233.	Yamane, K., et al., Impact of Rotation Correction after Brace Treatment on Prognosis in Adolescent Idiopathic Scoliosis. Asian spine journal, 2016. 10(5): p. 893-900.	5
234.	Diard, F., et al., [Imaging of chilhood and adolescent scoliosis]. Imagerie des scolioses de l'enfant et de l'adolescent., 2002. 83(9 Pt 2): p. 1117-2.	6
235.	Andrews, G. and G.D. MacEwen, <i>Idiopathic scoliosis. An 11-year follow-up study of the role of the Milwaukee brace in curve control and trunco-pelvic alignment.</i> Orthopedics, 1989. 12 (6): p. 809-16.	13
236.	Fisher, D.A., G.F. Rapp, and M. Emkes, <i>Idiopathic scoliosis: transcutaneous muscle stimulation versus the Milwaukee brace</i> . Spine, 1987. 12 (10): p. 987-91.	13
237.	Renshaw, T.S., Idiopathic scoliosis in children. Current opinion in pediatrics, 1993. 5(4): p. 407-12.	13
238.	Weinstein, S.L., <i>[Idiopathic scoliosis in adolescence. Incidence and progression of untreated scoliosis].</i> Die idiopathische Adoleszentenskoliose. Haufigkeit und Progression unbehandelter Skoliosen, 1989. 18 (2): p. 74-86.	6

239.	Andersen, M.O., et al., [Idiopathic scolioses treated with the Boston brace]. Idiopatiske skolioser behandlet med Boston-korset., 1993. 155 (26): p. 2044-6.	6							
240.	Parent, E.C., et al., <i>Identifying the best surface topography parameters for detecting idiopathic scoliosis curve progression</i> . Studies in health technology and informatics, 2010. 158 (ck1, 9214582): p. 78-82.	13							
241.	Kadoury, S., J. Shen, and S. Parent, Global geometric torsion estimation in adolescent idiopathic scoliosis. Medical & biological engineering & computing, 2014. 52(4): p. 309-19.	13							
242.	Howell, F.R., J.K. Mahood, and R.A. Dickson, Growth beyond skeletal maturity. Spine, 1992. 17(4): p. 437-40.	13							
243.	Brox, J.I., et al., Good brace compliance reduced curve progression and surgical rates in patients with idiopathic scoliosis. European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 2012. 21 (10): p. 1957-63.	13							
244.	Focarile, F.A., et al., <i>Effectiveness of nonsurgical treatment for idiopathic scoliosis. Overview of available evidence.</i> Spine, 1991. 16 (4): p. 395-401.	5							
245.	Miller, J.A., A.L. Nachemson, and A.B. Schultz, Effectiveness of braces in mild idiopathic scoliosis. Spine, 1984. 9(6): p. 632-5.	13							
246.	Zaborowska-Sapeta, K., et al., <i>Effectiveness of Cheneau brace treatment for idiopathic scoliosis: prospective study in 79 patients followed to skeletal maturity.</i> Scoliosis, 2011. 6(1): p. 2.	13							
247.	Korovessis, P., et al., Effect of the Cheneau Brace in the Natural History of Moderate Adolescent Idiopathic Scoliosis in Girls: Cohort Analysis of a Selected Homogenous Population of 100 Consecutive Skeletally Immature Patients. Spine deformity, 2018. 6(5): p. 514-522.	13							
248.	Steen, H., J.E. Lange, and J.I. Brox, <i>Early weaning in idiopathic scoliosis</i> . Scoliosis, 2015. 10 (101271527): p. 32.	9							
249.	Pellios, S., et al., Curve progression 25 years after bracing for adolescent idiopathic scoliosis: long term comparative results between two matched groups of 18 versus 23 hours daily bracing. Scoliosis and spinal disorders, 2016. 11(101675716): p. 3.	9							
250.	Cruickshank, J.L., M. Koike, and R.A. Dickson, <i>Curve patterns in idiopathic scoliosis. A clinical and radiographic study.</i> The Journal of bone and joint surgery. British volume, 1989. 71 (2): p. 259-63.	5							
251.	Shannon, T.M.L., Development of an apparatus to evaluate Adolescent Idiopathic Scoliosis by dynamic surface topography. Studies in health technology and informatics, 2008. 140 (ck1, 9214582): p. 121-7.	13							
252.	De Korvin, G., et al., Detection of progressive idiopathic scoliosis during growth using back surface topography: a prospective study of 100 patients. Annals of physical and rehabilitation medicine, 2014. 57 (9-10): p. 629-39.	2							
253.	Weinstein, S.L. and I.V. Ponseti, <i>Curve progression in idiopathic scoliosis</i> . The Journal of bone and joint surgery. American volume, 1983. 65 (4): p. 447-55.	13							
254.	Donzelli, S., et al., Consistent and regular daily wearing improve bracing results: a case-control study. Scoliosis and spinal disorders, 2018. 13 (101675716): p. 16.	13							
255.	Matussek, J., et al., [Conservative treatment of idiopathic scoliosis with effective braces: early response to trunk asymmetry may avoid curvature progress]. Konservative Therapie der idiopathischen Skoliose mit dem Korsett : Hohe Korrekturraten bei fruhzeitiger Beobachtung und Behandlung der Rumpfasymmetrie., 2014. 43 (7): p. 689-2.	6							
256.	Chiu, Y.L., T.J. Huang, and R.W. Hsu, Curve patterns and etiologies of scoliosis: analysis in a university hospital clinic in Taiwan. Changgeng yi xue za zhi, 1998. 21(4): p. 421-8.	9							
257.	Izatt, M.T., et al., CT and radiographic analysis of sagittal profile changes following thoracoscopic anterior scoliosis surgery. Scoliosis, 2012. 7(1): p. 15.	4							
258.	Hopf, C., [Criteria for treatment of idiopathic scoliosis between 40 degrees and 50 degrees. Surgical vs. conservative therapy]. Kriterien zur Behandlung idiopathischer Skoliosen zwischen 40 degrees und 50 degrees. Operative vs. konservative Therapie., 2000. 29 (6): p. 500-6.	6							
259.	de Mendonca, R.G.M., J.R. Sawyer, and D.M. Kelly, <i>Complications After Surgical Treatment of Adolescent Idiopathic Scoliosis</i> . The Orthopedic clinics of North America, 2016. 47 (2): p. 395-403.	4							
260.	Fishchenko, I.V. and I.V. Roi, [Complex conservative treatment of idiopathic scoliosis]. Klinichna khirurhiia, 2012(1): p. 64-7.	6							
261.	Studer, D., <i>Clinical investigation and imaging</i> . Journal of children's orthopaedics, 2013. 7(1): p. 29-35.	5							
262.	Negrini, S., et al., <i>Clinical and postural behaviour of scoliosis during daily brace weaning hours</i> . Studies in health technology and informatics, 2008. 140 (ck1, 9214582): p. 303-6.	5							
263.	Lykissas, M.G., A.H. Crawford, and V.V. Jain, <i>Complications of surgical treatment of pediatric spinal deformities</i> . The Orthopedic clinics of North America, 2013. 44 (3): p. 357-ix.	4							
264.	Veintemillas Araiz, M.T., et al., <i>Changes in spinal alignment</i> . Alteraciones de la alineacion vertebral., 2016. 58 Suppl 1 (0120775): p. 115-27.	6							
265.	Dworkin, B., et al., <i>Behavioral method for the treatment of idiopathic scoliosis</i> . Proceedings of the National Academy of Sciences, 1985. 82 (8): p. 2493-2497.	2							
266.	Weinstein, S.L., Idiopathic scoliosis. Natural history. Spine, 1986. 11(8): p. 780-3.	13							
267.	Bollini, G., et al., <i>[Idiopathic scoliosis: evaluation of the results]</i> . Scoliose idiopathique: evaluation des resultats., 1999. 183 (4): p. 757-8.	6							
268.	Ferraro, C., et al., <i>Hump height in idiopathic scoliosis measured using a humpmeter in growing subjects: relationship between the hump height and the Cobb angle and the effect of age on the hump height.</i> European journal of physical and rehabilitation medicine, 2017. 53 (3): p. 377-389.	13							
269.	Keenan, B.E., et al., Gravity-induced coronal plane joint moments in adolescent idiopathic scoliosis. Scoliosis, 2015. 10(101271527): p. 35.	5							
270.	Yamauchi, Y., et al., [Follow-up results of brace treatment of adolescent idiopathic scoliosis]. Nihon Seikeigeka Gakkai zasshi, 1986. 60(11): p. 1079-85.	6							
271.	Villemure, I., et al., Evolution of 3D deformities in adolescents with progressive idiopathic scoliosis. Studies in health technology and informatics, 2002. 91 (ck1, 9214582): p. 54-8.	2							
	Bernard, J.C., et al., [Evaluation of the efficacy of a carbon brace ("Corset monocoque carbone respectant la respiration"								
	[CMCR]) preserving lung capacity to treat idiopathic scoliosis in children and adolescents: a retrospective study of 115 patients]. Evaluation du resultat d'un corset monocoque carbone respectant la respiration (CMCR) dans la scoliose idiopathique chez l'enfant	6							
272. 273.	[CMCR]) preserving lung capacity to treat idiopathic scoliosis in children and adolescents: a retrospective study of 115 patients].	6							
 Soulanis Research Society Issue sufficience relation Society 2015. 10(10):127327); p. 12. Kessler, H., Elffordy of new computer-aduld subjective archives in the treatment of adolescent inpathic scolensis. Journal of guidance entropedses. Part H, 2008. 1144; p. 2017.11. Minas, M.K., et al., Effectiveness of the Right Cheman versus Nathoness for adolescent idiopathic scolensis. a 113 Borck, K., et al., Effectiveness of The Right Society is by Spaced ro Dynamic Brecing with Special Physiotherapy Programme in Space of System Oropedia, transmologin, rehability, 2016. 18(5): p. 234-34. Manayama, T., T.B. Grivas, and A. Kapitis, Effectiveness and outcomes of brace treatment: a systematic review. Physiotherapy theory and practice, 2011. 21(1): p. 24-62. Burnell, W.P., The natural history of isolatis: curve progression of sutrevent of radio symmetries in adolescent 113 Burnell, M.P., The natural history of scolatis: curve progression of sutrevent of radio symmetries in adolescent 113 Chuah, S.L., et al., The natural history of scolatis: curve progression of sutrevent of scalatis curve progression of sutrevent of scalatis of strength reads and symmetries. Physiotherapy theory and program of null disputities teclensis: a carops functional symmetries. The scalatis and the curve of radio symmetries and bolascent 113. Chuah, S.L., et al., The natural history of scalatis: curve progression of sutrevents with disputities. Scalatis, Scalatis, and the scalatis is advected curves of different aetiology, with early function application of the bolascing and the scalatis is advected curves of different aetiology, with early function application of the scalatis and curves and scalatis and									
--	------	---	----	--	--	--	--	--	--
276. Kessler, 11. Efficacy of a new compare-raided design/compare-raided manifecture orthost in the treatment of adolescent 13 277. Minsk, M.K., et al., Effectiveness of the Rigo Chemica versus Baston-spie arthoses for adolescent idiopathic scolusis: a 13 278. Programme in Spinol and yound discover spinol. 131 278. Programme in Spinol and yound discover spinol. 131 279. Minsk, M.K., et al., Effectiveness and and automes of brace treatment a systemic review. Physiotherapy theory and practice, 2011. 271(1): p. 26–42. 132 270. Maroyana, T. T.B. Givas, and A. Kapits, Effectiveness and automes of brace treatment a systemic review. Physiotherapy theory and practice, 2011. 27(1): p. 26–42. 132 280. Burnell, R.G., et al., The posterior soleladi flores: rel-vertebral angle and cative verbal rotation asymmetries in adolescent along solution. The media for scolusits score programme on different attology, with evely mask approximate in a specific evertebral angle and cative effect and scolus, scolusits approximate in a specific evertebral angle and cative effect and cative scolusits. 2015. Spinol (141, 2011). 132 281. Chanks, M. M. Dakovs, and Dispatic. The media prof of columbar control and specific evertebral angle and cative effect and angle scolus scolus and specific evertebral angle and cative effect and angle scolus scolus angle and cative effect and a specific evertebral angle scolus scolus and angle scolus angle scolus and scolus angle scolus angle scolus angle scolus angle scolus angle scolus a	275.	Kuroki, H., et al., Efficacy of the Osaka Medical College (OMC) brace in the treatment of adolescent idiopathic scoliosis following Scoliosis Research Society brace studies criteria. Scoliosis, 2015. 10(101271527): p. 12.	13						
 Minsk, M.K., et al., Effectiveness of the Rigo Chemon versus Booton-sple orthous for adolecent idiopathe scalinsis: a revergence and scalina and spinal disorder. 2017. 121015716 p. 7. Rozck, K., et al., Effectiveness of Treatment of Minpathic Scalinsis by SplorCor Dynamic Bracing with Special Physiotherapy Programme in Space of System Cregodin, transmologia, elability, 2016. 18(5): p. 125-343. Maroyana, T., T.B. Givas, and A. Kaspits, Effectiveness and outcomes of brace treatment: a systematic review. Bysiotherapy Scaling Physiotherapy Between theory and a scaling and theory of the State Control Scaling Scale Science Scale Scale Scale Scale Scale Scale Science Scale Scale Science Scale Scale Science Scale Science Scale Science Scale Science Scienc	276.	Kessler, J.I., Efficacy of a new computer-aided design/computer-aided manufacture orthosis in the treatment of adolescent	13						
Programme in Spine Costsom. Creptodia: Lumanologia, Publicity, 2016. 18(3): p. 225-314. Projecture in Spine Costsom. Creptodia: Lumanologia, Publicity, 2016. 18(3): p. 242-314. Projecture in Spine Costsom. Creptodia: Lumanologia, Publicity, 2016. 18(3): p. 242-314. Projecture in adolecture in the spine in the spin	277.	Minsk, M.K., et al., Effectiveness of the Rigo Cheneau versus Boston-style orthoses for adolescent idiopathic scoliosis: a	13						
279 Manyama, T., T.B. Grivas, and A. Kaspins, Effectiveness and autoeme of brace treatment: a systematic review. Physiotherapy theory and practice, 2011 27(1): p. 26-3. 5 280 Baunell, W.P., The autoral history of disopathic scolasts before selectal materiny. Spinc. 1986. 11(8): p. 773-6. 5 281 Bauvell, W.P., The autoral history of disopathic scolasts before selectal matering. The Medical Journal of Malaysia, 2001. 56 8 282 Opt, K.S., L. Chuak, and N. Starwart, The medical formatics, 2008. 140(c), 2145821 p. 26-3. 4 283 Spared follows on in segregative treated carres. The Medical Journal of Malaysia, 2001. 56 8 283 Spared follows on in segregative treated carres. The Medical Journal of Malaysia, 2001. 55 4 284 Addomis, M.M. Dakovic, and I. Points: The numeral history of mild idiopathic scolasis. Acta medica Conatica : casops Brave Swinks, 2014. 5(10271327); p. 5. 13 286 programme on idiopathic scolosis curvature: camparative study. SOMOR 2014 neuroal wature Associatios: Lourantal technology. SWIN SWIN and anary Wature Scolosis. 2014. 5(10271327); p. 5. 13 287 Program on idiopathic scolosis: cavature: camparative study. SOMOR 2014 neuroal wature Manosescolosis: con of the Carried Spine Research Sociatios: Marking and anary Mannes Casolissis. 2014. 5(10271327); p. 5. 13 288 Robana, T., et al., The astoplicas canding style descinti and anary medic scolasis:	278.	Rozek, K., et al., Effectiveness of Treatment of Idiopathic Scoliosis by SpineCor Dynamic Bracing with Special Physiotherapy							
 Indexly and pictures. 2011. <i>1</i>(1): P. 24-22. Francell, W. P., The attainty of alignable scalasis before skeletal maturity. Spine, 1986. 11(8): p. 773-6. Burwell, R. G., et al., <i>The posterior skeletal thorax: rhv-rebrint angle and scalal vertebral rotation asymmetrics in adolescent individual scalasis and scalar vertebral rotation asymmetrics in adolescent individual scalasis and scalar vertebral rotation asymmetrics in adolescent individual scalar scalar</i>	279.	Maruyama, T., T.B. Grivas, and A. Kaspiris, Effectiveness and outcomes of brace treatment: a systematic review. Physiotherapy	5						
281 Burwell, R.G., et al., The posterior skeletal thorac: rob-vertebral angle and actal vertebral rotation asymmetries in adulescent the discretion sciences of the second science of the second sciences of the second sciences of the second science science of the second science of the second science of									
 Idiopathic scolutors: Studies in health technology and informatics, 2008. 140(ck1, 9214582); p. 263-8. Oh, K.S., S.L. Chuha, And S. Harvant, The need for scolusis screening in Malaysia, 2001. 56 Chuha, S.L., et al., The neural history of scolusis: curve progression of unretated curves of different actiology, with early (mean 2 year) follow up in surgically requested curves. The Medical journal of Malaysia, 2001. 56 Perink, M.D. Bakovie, and L. Bogini, The metatrol history of scolusis: curve progression of unretated curves of different actiology, with early (mean 2 year) follow up in surgically requested curves. The Medical journal of Malaysia, 2001. 56 Suppl C(m2m, 0651547); p. 37-40. Perink, M.D. Bakovie, and J. Dottricin, The effect of compliance to a Rigo System Cheneau brace and a specific exercise programme on hiopathic scolusis curvature: a comparative study. SOSOR7 2014 award winner. Scolusis, 2014 (19)(11211217); p. 113 Perink, L.A., and V. Barting, and J. Dottricin, The effect of compliance and outcome for patients with idiopathic scolusis: an evidence-final inclusion of the European Spine Society, 1996. 5(6): p. 374-9. Paliman, T., et al., The association between Prace compliance and outcome for patients with idiopathic scolusis: an evidence-face were, Spine, 2007. 3(21) Suppl): p. 591-5100. Shi, B., L.A. and S.L. Weinstein, Surgical rates glare observation and Pracing for adolescent idiopathic scolusis: an evidence-face action. Spine Sciety, 199. 5(10). Shi, B., et al., Spined converts equives and the production scandinavian, 1989. 6(2); p. 177-80. Shi, E., et al., Spine downed weight volcasity in practicing anne progression in peripheral gifts with idiopathic scolusis: A renetical clinical triat. Amala Sci Presenta Sci									
 Suppl C(m2m, 0801547; p. 26-30. Chush, S.L., et al., <i>Panetural history of scolinis: curve progression of untreated curves of different actiology, with early (mean 2 year) follow up in surgically reacted curves. The Medical journal of Malaysia, 2001. 56 Suppl C(m2m, 0561547; p. 37-40.</i> Peerina, M., D Bakovie, and L Sojnis, T. Fan <i>effect of compliance to a Rige System Chemau brace and a specific curcuise programme on idiopathic scolinis curvature: a comparative study: SOSORT 2014 award winner.</i> Scoliosis, 2014. 9(101271527): 19. Rivett, L. A. Stewart, and J. Dotterion. <i>The effect of compliance to a Rige System Chemau brace and a specific curcuise programme on idiopathic scolinis curvature: a compliance and outcome for patients with idiopathic scolinis: a meridence study: SOSORT 2014 award winner. Scoliosis, 2014. 9(101271527): 19.</i> Prilis, J.E., et al., <i>The benefits of school screening for scolinis in the central part of The Netherlands. European Specia Define and outcome for patients with idiopathic scolinis: a meridence study and the European Specia Define and outcome for patients with idiopathic scolinis: a meridence study and the European Specia Define and outcome for patients with idiopathic scolinis: an evidence-based review. Spine, 2007. 32(19 Suppl): p. 591-8100.</i> Carri, A.J. et al., <i>Stoface treephonizgrammer of theracic hyphosis.</i> Acta orthopasdica Scandinavica, 1989. 60(2): p. 177-80. Shi, B., et al., Spinel, <i>Corris 2</i>(16): F1(1): p. 530. Braditod, D.S., A. Tangyo, and J. Vanselow, <i>Surglue electrical stimulation in the treatment of idiopathic scolinis: Therm results for the application of the scolinis and the analytic scolinis is 2015 (11): p. 10.</i> Zama, F., et al., Spine Corr, exercise and SPoiff T. 101410: p. 350. Zama, F., et al., Spine Corr, exercise and SPoiff 2014 (11): p. 320. Zama, F., et al., Spine Corr, exercise and SPoiff 2014 (11): p. 320. 	281.	idiopathic scoliosis. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 263-8.	13						
 2) <i>earl follow up in surgically treated carves.</i> The Makical journal of Malaysia, 2001. 56 Suppl C(m.m. 0361547): p. 57-40. Pecink, M., M. Dakovis, and I. Bojnic, The <i>entard history of mild idiopathic scoliosis</i>. Acta medica: casepis Harvatske akademije medicinskih ranaosti, 1992. 46(2): p. 75-8. Rivett, L., A. Stewart, and J. Potterton, <i>The effect of compliance to a Rigo System Cheneut brace and a specific exercise programme on idiopathic scoliosis curvature: a comparative study: SOSORT 2014 avard winner. Scoliosis, 2014. 9(101271527): p. 5.</i> Pruijs, J.E., et al., <i>The benefits of school screening for scoliasis in the central part of The Netherlands:</i> European spine Journal : official publication of the European Spine Journal of Decomposition of the Cervical Spine Research Society, 1996. 5(6): p. 374-9. Bolina, T., et al., <i>The association between brace compliance and outcome for patients with idiopathic scoliosis: an evidence bared areview. Spine, 2007.</i> 32(19 Suppl): p. 591-5100. Carr, A.J. et al., <i>Suffer estrengohogrammery of thracic kephosis.</i> Acta orthopaedica Scandinavica, 1989. 60(2): p. 177-80. Shi, B., et al., <i>Spinal growth velocity wasis beight velocity in predicting carve progression in pert-puberal girls with idiopathic scoliosis: an evidence bared areview. Spina, 1980.</i> 106.171 (p. 187-4). Bradford, D.S., A. Tanguy, and J. Vaneslow. <i>Sufface electrical simulation in the treatment of idiopathic Scoliosis? Short term results for basis in Molancheschela disonders.</i> 2016. 171(p. p. 187-4). Zaim, F., et al., <i>Spinal Corv. curctue and SNORT regula bandermatics</i>, 2012. 170(ck1, 214532): p. 501-4. Zaim, F., et al., <i>Spinal Corv. curctue and SNORT regula bandermatics</i>, 2016. 171(p. p. 124532). Meiss, H., et al., <i>Spinal Corv. curctue and SNORT regula bandermatics</i>, 2012. 170(ck1, 214532): p. 501-4. Zaim, F., et al., <i>Spinal Corv. curctue and SNORT r</i>	282.	Suppl C(m2m, 0361547): p. 26-30.	8						
 kademije medicinskih zanosti, 1992. 46(2): p. 75-8. 13 Rivett, L., A. Stewart, and J. Potetton, <i>The effect of compliance to a Rigo System Cheneui brace and a specific exercise programme on idopathic scoliosis curvature: a comparative study: SOSORT 2014 award winner.</i> Scoliosis, 2014. 9(101271527): p. 5. Pruijs, J.E., et al., <i>The benefits of school screeming for scoliosis in the central part of The Netherlands.</i> European spine journal: Cervical Spine Research Society, 1996. 5(6): p. 374-9. Rahman, T., et al., <i>The association between brace compliance and outcome for patients with idiopathic scoliosis: an evidence-based review.</i> Spine, 2007. 32(1): Spip1): p. 591-5100. Carr, A.J. et al., <i>Suffict artes of the observation and bracing for adolescent idiopathic scoliosis: an evidence-based review.</i> Spine, 2007. 32(1): Spip1): p. 591-5100. Carr, A.J. et al., <i>Suffict artes of the observation and bracing for adolescent idiopathic scoliosis: an evidence-based review.</i> Spine, 2007. 32(1): Suppli): p. 591-5100. Carr, A.J. et al., <i>Spine Growth selectity versus height velocity in predicting curve progression in pert-puberial girls with idiopathic scoliosis: BMC misculosices end SPoRT rigid brace: what is the best for Adolescent Idiopathic scoliosis? Short term result in a Dipatients. Spine, 1983. 8(7), p. 73-64.</i> Zaina, F., et al., <i>SpineCore, exercise and SPoRT rigid brace: what is the best for Adolescent Idiopathic Scoliosis? Short term result in orbifosis Ostoris and the themlogy and informatics, 2012.1176(4), 21212176(4), 2121327(1), 224.</i> Amado J. Dyuscial and enabilitation medicine, 2018(10127157), p. 23. Medisori, S., et al., <i>SpoinCare exercise reduce the need for bracing in adolescent with idiopathic Scoliosis? Abort term result for hybricia add reliabilitation medicine, 2018(10127157), p. 23.</i> Medisori, R., et al., <i>SociORT Award Winner 2015: a multicentre stu</i>	283.	2 year) follow up in surgically treated curves. The Medical journal of Malaysia, 2001. 56 Suppl C(m2m, 0361547): p. 37-40.	4						
 programme on idiopathic scoliosis curvature: a comparative study: SOSOR7 2014 award winner. Scoliosis, 2014. 9(101271527): praije, J.E., et al., <i>The benefits of school screening for scoliosis in the central part of The Vetherlands.</i> European Section of the Carvinal publication of the Furopean Spins Oxiety, the European Spinal Deformity Society, and the European Section of the Carvinal Spine Research Society, 1996. 5(0): p. 734-9. Rahman, T., et al., <i>The association between brace compliance and outcome for patients with idiopathic scoliosis: an evidence-based review.</i> Spine, 2007. 32(4): p. 420-2. Dolan, L.A. and S.L. Weinstin, <i>Surgical rates after observation and bracing for adolescent idiopathic scoliosis: an evidence-based review.</i> Spine, 2007. 32(1): Suppl. p. S91-8100. Sti, B., et al., Spinal growth velocity versus height velocity in predicting curve progression in peri-pubertal girls with idiopathic scoliosis: BMC mesculoskletal disorders, 2016. 17(1): p. 368. Bradford, D.S., A. Traguy, and J. Vanselow, <i>Surface electrical stimulation in the treatment of idiopathic scoliosis? Nort term results in 3D patients.</i> Spine, 1988. 3(P): p. 757-64. Zaina, F., et al., <i>Spinelfee vereixes and SPoRT rigid brace: what is the best for Adolescent Idiopathic scoliosis? A practical clinical trial.</i> Annols of physical and rehabilitation medicine. 2018(1012)15277; p. 23. Weiss, HR. and M. Werkmann, <i>Soft braces in unit:centres study comparing the SPORT and ART braces effectiveness according to the suppression Scoliosis.</i> 2012. 17(c): p. 12-937. Brodser, R.J., G. Arnstrong, and M.S. Moreland, Spinal deformity and bock surface asymmetry in idiopathic scoliosis: A practical clinical trial. Annobe Scoliosis. 2012. 17(c): p. 12-937. Brodses, H.A., et al., <i>Scoliosis AP prospective epidemiological study</i>. The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 12-37. Br	284.		13						
 Pruijs, J.E., et al., <i>The benefits of school screening for scoliots: in the central part of The Netherlands</i>. European Spine journal: Official Spine Research Society, 1996. 5(0): p. 374-9. Rahman, T., et al., <i>The association between brace compliance and outcome for patients with idiopathic scoliosis: an evidence-brace deriver.</i> Spine, 2007. 32(19 Suppl.): p. S91-S100. Dolan, L.A., and S.L. Woinstein. <i>Streigent artes: after observation and bracing for adolescent idiopathic scoliosis: an evidence-brace deriver.</i> Spine, 2007. 32(19 Suppl.): p. S91-S100. Carr, A.J., <i>et al.</i>, <i>Spinel and score observa brace hypotectry in predicting curve progression in peri-pubertal gifts with idiopathic scoliosis: an evidence-brace deriver.</i> Spine, 2007. 32(19 Suppl.): p. S91-S100. Bradford, D.S., A. Tangy, and J. Yanesoh. <i>Suffer electrical stimulation in the treatment of idiopathic scoliosis: preliminory results in 30 patients.</i> Spine, 1993. 8(7): p. 157-64. Bradford, D.S., A. Tangy, and J. Yanesoh. <i>Suffer electrical stimulation in the treatment of idiopathic scoliosis: Apractical clinical trial.</i> Annals of physical and rehabilitation medicine, 2018. (1150:2713). Zunin, S., et al., <i>Spenel fevercises realists</i> 2015. 10(10121): 2717. p. 23. Kuest, H, et al., Spine (PAS). And Thing at the scolie in addolescent idiopathic scoliosis: A practical clinical trial. Annals of physical and rehabilitation medicine, 2018. (1150:2713). Zunin, F., et al., Spine (PAS). And Winner 2015. an unificent study. Comparing the SPORT and ART braces effectiveness according to the SOSORT-SRS recommendations. Scoliosis, 2015. 10(10121): 271, p. 23. Weise, H, et al. <i>Clinical science the need for biarcing in addolescent idiopathic scoliosis (AIS) - Review of the literature and description of a new approach.</i> Scoliosis, 2015. 10(10121): 271, p. 23. Weise, H, et al., Scoliosis Annog, and M.S. Mo	285.	programme on idiopathic scoliosis curvature: a comparative study: SOSORT 2014 award winner. Scoliosis, 2014. 9(101271527):	13						
 Rahman, T., et al., <i>The association between brace compliance and outcome for patients with idiopathic scoliosis.</i> Journal of patients in control of scolos. Sci V: p. 420-2. Dolan, L.A. and S.L. Weinstein, <i>Surgical rates after observation and bracing for adolescent idiopathic scoliosis: an evidence-base deview.</i> Spine, 2007. 32(1) Supply: p. S91-S100. Carr, A.J., et al., <i>Spinal growth velocity versus height velocity in predicting curve progression in peri-pubertal girls with idiopathic scoliosis: BMC musceloskeletal disorders, 2016.</i> 17(1): p. 368. Bradford, D.S., A. Tanguy, and J. Vanselow, <i>Surface electrical stimulation in the treatment of idiopathic scoliosis: Preliminary results in 0 patients, Spine,</i> 1983. 4(7): p. 575-64. Zann, F., et al., <i>SpineCor, exercise and SPORT rigid brace: what is the best for Adolescent Idiopathic scoliosis: Proteiminary results in 0 patients.</i> Scoliosis, 2010. 17(1): p. 136. Anads of physical and erabilitation medicine. 2018(101029277). Anads of physical and erabilitation medicine. 2018(101029277). Zaina, F., et al., <i>SOCORT Award Winner 2015: a multicentre study comparing the SPORT and ART braces effectiveness according to ethosyonery and eraphilic scoliosis is accionsis. 2015.</i> 10(1021757); p. 23. Weiss, HR. and M. Werkmann, Soft braces in the treatment of Adolescent Midpathic Scoliosis. Journal of orthopaetic research: official publication of the Orthopaetic Research Scoliosis. 4(1): p. 129-37. Brods, H.L., et al., Scoliosis accolissi, colisis, colisis, and the stream scale and natural history. A prospective epidemiological study. The 2 Journal of braces in the advective and natural history. A prospective epidemiological study. The 2 Journal of braces in the research collisis, colisis scolissis, 2015. 2015. 6(2): p. 102-37. Brods, H.L., et al., Scoliosis detection, pattent characteristis, referral patterns and treatment in the ab	286.	official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the	13						
 Dolan, L.A. and S.L. Weinstein, Surgical rates after observation and bracing for adolescent idiopathic scoliosis: an evidence- scoliosis. BMC mescalophyle, ps91-S100. Carr, A.J., et al., Spinal growth velocity versus height velocity in predicting curve progression in peri-inducting fuels with idiopathic scoliosis: preliminary realistics. BMC mescaloskeleal disorders, 2016. 17(1): p. 368. Bradford, D.S., A. Tanguy, and J. Vanselow, Surface electrical stimulation in the treatment of idiopathic scoliosis: preliminary realistics in 0 printers, Spine. 1938. NCI: p. 575-64. Zania, F., et al., Spinel Cyst. RVI; p. 575-64. Zania, F., et al., Spiceffic exercises and SPORT rigid brace: what is the best for Adolescent Idiopathic scoliosis? Stort term results from 2 tertospecific secories rathee. Studes in health technology and informatics, 2012. 176(64, 9214582); p. 361-4. Angatio a Ophysical and rehabilitation medicine, 2018 (101029273). Zania, F., et al., SOSORT Award Winner 2015: a multicentre study comparing the SPORT and ART braces effectiveness according to the SOORT study and medicine, 2018 (101029277); p. 23. Weiss, HR. and M. Werkmann, Soft braces in the treatment of Adolescent Idiopathic Scoliosis, 101 (2015) (2017). Brods, H.L., et al., Softotiss: a prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. Brods, H.L., et al., Scoliosis 2012. 7019; p. 12-6. Gore, D.R., et al., Scoliosis science, prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. Books, H.L., et al., Scoliosis science, prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 40(2): p. 198-60(2): p. 192-37. Books, H.L., et al., Scoliosis science and matural history. A prospective epidemiological study. The Journal of bon	287.	Rahman, T., et al., The association between brace compliance and outcome for patients with idiopathic scoliosis. Journal of	13						
 Carr, A.J., et al., <i>Suface streeophotogrammetry of thoracic typhosis.</i> Acta orthopaedica Scandinavica, 1989. 60(2): p. 177-80. Shi, B., et al., <i>Spinal growth velocity versus height velocity in predicting curve progression in peri-pubertal girls with idiopathic scolosis.</i> BMC musculoskeletal disorders. 2016. 17(1): p. 368. Bradford, D.S., A. Tanguy, and J. Vanselow, <i>Surface electrical stimulation in the treatment of idiopathic scoliosis: preliminary results in 30 patients.</i> Spine, 1983. 8(7): p. 757-64. Zaina, F., et al., <i>SpineCor, exercise and SPoRT regid brace: what is the best for Adolescent lidopathic Scoliosis? Short term results in a statistic in a drehabilitation medicine, 2018 (19(1)): p. 2368.</i> Annals of physical and rehabilitation medicine, 2018 (19(1)): p. 237. Zaina, F., et al., <i>Sportife exercises reluce the need for bracing in adolescents with idiopathic scoliosis: A practical clinical trial.</i> Anals of physical and rehabilitation medicine, 2018 (19(1)): p. 233. Zaina, F., et al., <i>SOSORT-sRS recommendations</i>, Steoliosis, 2015. 10(10121752): p. 23. Weiss, HR. and M. Werkmann, <i>Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature and description of a new approach</i>. Scoliosis, 2012. 1(1): p. 11. Stokes, I.A., J.G. Amstrong, and M.S. Moreland, <i>Spinal deformity and back surface asymmetry in idiopathic scoliosis.</i> Journal of ontopaedic research : official publication of the Orthopaedic Research Society, 1988. (61): p. 129-37. Brostosk, F.L., et al., <i>Scolosis A prospective epidemiological study</i>. The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. Adobber, R.D., et al., <i>Scolosis and epidemicological study</i>. The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. Robosk, R.D., et al., <i>Scolosis and ereal matereteristis: referral patterns and treatment i</i>	288.	Dolan, L.A. and S.L. Weinstein, Surgical rates after observation and bracing for adolescent idiopathic scoliosis: an evidence-	5						
 Shi, B., et al., <i>Spinal growth velocity versus height velocity in predicting curve progression in peri-pubertal girls with idiopathic scoliosis</i>. BMC musculoskeletal disorders, 2016. 17(1): p. 368. Bradford, D.S., A. Tanguy, and J. Vanselow. <i>Surface electrical simulation in the treatment of idiopathic scoliosis: preliminary results in 30 patients</i>. Spine, 1983. 8(7): p. 757-64. Ziana, F., et al., <i>Spine Cor. cercics end SPoRT rigid brace: what is the best for Adolescent Idiopathic Scoliosis? Short term results from 2 retrospective studies</i>. Studies in health technology and informatics, 2012. 176(ek1, 9214582): p. 361-4. Zaina, F., et al., <i>Spine effice services: reduce the need for bracing in adolescents with idiopathic scoliosis: A practical clinical trial</i>. Annals of physical and rehabilitation medicine, 2018(101502773). Zaina, F., et al., <i>Spine additions</i>. Scoliosis, 2015. 10(101217527): p. 23. Weiss, HR. and M. Werkmann, <i>Soft braces in the reatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature adversity of a new approach</i>. Scoliosis, 2017. 101: p. 11. Stokes, I.A., I.G. Armstrong, and M.S. Moreland, <i>Spinal deformity and back surface asymmetry in idiopathic scoliosis</i>. Journal of onthopacitic research. Society, 1988. 6(1): p. 129-7. Brooks, H.L., et al., <i>Scoliosis is Clive spinal deformity ropect</i>. Pediatrics, 1981. 67(2): p. 196-200. Rogala, E.J., D.S. Drummond, and J. Gurr, <i>Scoliosis: incidence and natural history. A prospective epidemiological study</i>. The Journal of bone and joint surgery. American volume, 1975. 54(7): p. 264-72. Madord, D.C., M.A. Asher, and S.M. Lai, <i>Scoliosis: correction maintenance in skeletally immature patients with idiopathic scoliosis. J and scolid science and natural history. A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 64(2): p. 174.</i> Burton, D.C., M.A. A	289		13						
 Bradford, D.S., A. Tanguy, and J. Vanselow, <i>Surface electrical stimulation in the treatment of idiopathic scoliosis: preliminary results in 30 pattents.</i> Spine, 1983. 8(7): p. 757-64. Zaina, F., et al., <i>SpineCor, exercise and SPoRT rigid brace: what is the best for Adolescent Idiopathic Scoliosis? Short term results from 2 retrospective studies.</i> Studies in health technology and informatics, 2012. 176(ck1, 9214582): p. 361-4. Negrifi, S., et al., <i>Specific exercises reduce the need for bracing in adolescents with idiopathic scoliosis? A practical clinical trial.</i> Annals of physical and rehabilitation medicine, 2018(101502773). Zaina, F., et al., <i>NOSDRT Award Wimmer 2013: a multicentre study comparing the SPoRT and ART braces effectiveness according to the SOSORT-SRS recommendations.</i> Scoliosis, 2015. 10(101271527): p. 23. Weiss, HR. and M. Werkmann, <i>Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature description of a new approach.</i> Scoliosis, 2012. 7(1): p. 11. Stokes, I.A., J.G. Armstrong, and M.S. Moreland, <i>Spinal deformity and back surface asymmetry in idiopathic scoliosis.</i> Journal of orthopaedic research : official publication of the Orthopaedic Research Society, 1988. 6(1): p. 109-37. Brooks, H.L., et al., <i>Scoliosis creening: results of a community project.</i> Pediatries, 1981. 67(2): p. 196-200. Rogala, E.J., D.S. Drummond, and J. Gurr, <i>Scoliosis: incidence and natural history. A prospective epidemiological study.</i> The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 12-6. Adobor, R.D., et al., <i>Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway.</i> Scoliosis, 2012. 7(1): p. 11-6. Burton, D.C., M.A. Asher, and S.M. Lai, <i>Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. <i>A spatial </i></i>		Shi, B., et al., Spinal growth velocity versus height velocity in predicting curve progression in peri-pubertal girls with idiopathic							
 Zaina, F., et al., Spine Cor. exercise and SPORT rigid brace: what is the best for Adolescent Idiopathic Scoliosis? Short term results from 2 retrospective studies. Studies in health technology and informatics, 2012. 176(ekt.l 49214582); p. 361-4. Negrini, S., et al., Specific exercises reluce the need for bracing in adolescents with Idiopathic Scoliosis? A practical clinical trial. Annals of physical and relabilitation medicine, 2018(101502773). Zaina, F., et al., SOSORT Award Winner 2015: a multicentre study comparing the SPORT and ART braces effectiveness according to the SOSORT-SRS recommendations. Scoliosis, 2015. 10(101271527); p. 23. Weiss, HR. and M. Werkmann, Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature and description of a new approach. Scoliosis, 2012. 7(1); p. 11. Stokes, I.A., J.G. Armstrong, and M.S. Moreland, Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of orthopaedic research is official publication of the Orthopaedic Research Society, 1988. 6(1); p. 129-37. Brooks, H.L., et al., Scoliosis screening: results of a community project. Pediatrics, 1981. 67(2); p. 196-200. Rogala, E.J., D.S. Drummod, and J. Gur, Scoliosis: incidence and natural history. A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1978. 60(2); p. 173-6. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening results of a community project. Pediatrics, 1981. 67(2); p. 196-200. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. Informatics, 102, 102, p. 113-6. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. I a scoliosis and spinal flexibilit	291.	Bradford, D.S., A. Tanguy, and J. Vanselow, Surface electrical stimulation in the treatment of idiopathic scoliosis: preliminary	13						
 Negrini, S., et al., Specific exercises reduce the need for bracing in adolescents with idiopathic scoliosis: A practical clinical trial. Annals of physical and rehabilitation medicine, 2018(101502773). Zaina, F., et al., SOSORT Award Winner 2015: a multicentre study comparing the SPORT and ART braces effectiveness according to the SOSORT-SRS recommendations. Socioiss, 2015. 10(101271527): p. 23. Weiss, HR. and M. Werkmann. Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature and description of a new approach. Scoliosis, 2012. 7(1): p. 11. Stokes, I.A., J.G. Armstrong, and M.S. Moreland. Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of orthopaedic research Societs, 1988. 6(1): p. 129-37. Brooks, H.L., et al., Scoliosis: A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. Gore, D.R., et al., Scoliosis screening: results of a community project. Pediatrics, 1981. 67(2): p. 196-200. Rogala, E.J., D.S. Drummond, and J. Gurr, Scoliosis: incidence and natural history. A prospective epidemiological study. The 2 Hournal of bone and joint surgery. American volume, 1978. 60(2): p. 174-6. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. J Scoliosis. Progressing results of the spine ingiris with idiopathic scoliosis. Progressing results of patient scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 103-43. Rigo, M., G. Quera-Salva, and M. Villagrasa. Sagittal configuration of the spine ingiris with idiopathic scoliosis: progressing results with idiopathic scoliosis: Sprogressing results accol	292.	Zaina, F., et al., SpineCor, exercise and SPoRT rigid brace: what is the best for Adolescent Idiopathic Scoliosis? Short term results	13						
 Annais of physical and renabilitation medicine, 2018(101502/15). 294. Zaina F., et al., SOSORT Award Wimer 2015. 10(101271527): p. 23. 295. Weiss, HR. and M. Werkmann, Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature and description of a new approach. Secoliosis, 2012. 7(1): p. 11. 296. Stokes, I.A., J.G. Armstrong, and M.S. Moreland, Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of orthopaetic research: official publication of the Orthopaetic Research Society, 1988. 6(1): p. 129-37. 297. Brooks, H.L., et al., Scoliosis: A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 113 298. Gore, D.R., et al., Scoliosis: A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. 299. Rogala, E.J., D.S. Drummond, and J. Gurr, Scoliosis: incidence and natural history. A prospective epidemiological study. The 2 unanal of bone and joint surgery. American volume, 1978. 60(2): p. 174-6. 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. Janal flexibility in patients with idiopathic scoliosis. In program in Norway. Scoliosis, 2012. 7(1): p. 18. 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing results of a case control study. Segittal spinal flexibility in patients with idiopathic scoliosis: Progressing results of a case control study. Scoliosis and spinal disorders, 2018. 132(101675716): p. 10. 303. Ridpe, M., G. Quera-Salva, and M. Villagrasa, Sagittal configura	293.	Negrini, S., et al., Specific exercises reduce the need for bracing in adolescents with idiopathic scoliosis: A practical clinical trial.	13						
10 the SOSORT-NG recommendations. Scoliosis, 2012. 10 (1012/1): p. 25. 295. Weiss, HR. and M. Werkmann, Sofb braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature and description of a new approach. Scoliosis, 2012. 7(1): p. 11. 296. Stokes, I.A., J.G. Armstrong, and M.S. Moreland, Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of 13 297. Brooks, H.L., et al., Scoliosis: A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. 13 298. Gore, D.R., et al., Scoliosis creening: results of a community project. Pediatrics, 1981. 67(2): p. 196-200. 2 299. Rogala, E.J., D.S. Drummond, and J. Gur, Scoliosis: incidence and natural history. A prospective epidemiological study. The 2 11 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis: progressing rather in sition really necessary? Spine, 2000. 25(1): p. 61-8. 5 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagitta		Zaina, F., et al., SOSORT Award Winner 2015: a multicentre study comparing the SPORT and ART braces effectiveness according							
and description of a new approach. Scoliosis, 2012. 7(1): p. 11. 11 296. Stokes, I.A., J.G. Armstrong, and M.S. Moreland, Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of orthopaedic research : official publication of the Orthopaedic Research Society, 1988. 6(1): p. 129-37. 113 297. Brooks, H.L., et al., Scoliosis: A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 175, 57(7): p. 968-72. 12 298. Gore, D.R., et al., Scoliosis screening: results of a community project. Pediatrics, 1981. 67(2): p. 196-200. 2 299. Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. 12 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asker, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. Is anterior fusion really necessary? Spine, 2000. 25(1): p. 61-8. 4 302. IoasA-43. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and Informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies		Weiss, HR. and M. Werkmann, Soft braces in the treatment of Adolescent Idiopathic Scoliosis (AIS) - Review of the literature							
orthopaedic research : orthical publication of the Orthopaedic Research Society, 1988. 6(1): p. 129-37. 13 297. Brooks, H.L., et al., <i>Scoliosis: A prospective epidemiological study.</i> The Journal of bone and joint surgery. American volume, 1975. 57(7): p. 968-72. 13 298. Gore, D.R., et al., <i>Scoliosis screening: results of a community project.</i> Pediatrics, 1981. 67(2): p. 196-200. 2 299. Rogala, E.J., D.S. Drummond, and J. Gurr, <i>Scoliosis: incidence and natural history. A prospective epidemiological study.</i> The journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. 2 300. <i>Program in Norway.</i> Scoliosis, 2012. 7(1): p. 18. 301. Burton, D.C., M.A. Asher, and S.M. Lai, <i>Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis.</i> 4 301. Burton, D.C., M.A. Asher, and S.M. Lai, <i>Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis.</i> 4 302. Nakakohji, T., <i>Sagittal spinal flexibility in patients with idiopathic scoliosis.</i> 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, <i>Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing</i> 5 303. Rigo, M., et al., <i>Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke IC Adolescent Idiopathic Scoliosis: applica assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis during</i>		Stokes, I.A., J.G. Armstrong, and M.S. Moreland, Spinal deformity and back surface asymmetry in idiopathic scoliosis. Journal of							
1975. 37(7): p. 968-72. 2 298. Gore, D.R., et al., Scoliosis screening: results of a community project. Pediatrics, 1981. 67(2): p. 196-200. 2 299. Rogala, E.J., D.S. Drummond, and J. Gurr, Scoliosis: incidence and natural history. A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. 2 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. Is anterior fusion really necessary? Spine, 2000. 25(1): p. 61-8. 4 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa. Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis: a case control study. Scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 305. Burkus, M., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 1									
299. Rogala, E.J., D.S. Drummond, and J. Gurr, Scoliosis: incidence and natural history. A prospective epidemiological study. The Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. Is anterior fusion really necessary? Spine, 2000. 25(1): p. 61-8. 4 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 5 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke IC Adolescent Idiopathic scoliosis: a case control study. Scoliosis and spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2013. 7(1): p. 43-9. 13 <td< td=""><td></td><td></td><td></td></td<>									
299. Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6. 2 300. Adobor, R.D., et al., Scoliosis detection, patient characteristics, referral patterns and treatment in the absence of a screening program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. 4 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 5 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke 1C Adolescent Idiopathic scoliosis. Spine deformity, 2015. 3(6): p. 560-565. 4 305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis a case control study. Scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Prelintary validation of curve progression in adol	298.		2						
300. program in Norway. Scoliosis, 2012. 7(1): p. 18. 11 301. Burton, D.C., M.A. Asher, and S.M. Lai, Scoliosis correction maintenance in skeletally immature patients with idiopathic scoliosis. 4 301. Is anterior fusion really necessary? Spine, 2000. 25(1): p. 61-8. 4 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 1033-43. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke 1C Adolescent Idiopathic Scoliosis: Asigittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 306. Burkus, M., et al., Sagittal plane assessment of not progression? How to deal with adolescent idiopathic scoliosis during puberty. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 13 308. Noshchenko, A., et al., Preliminary results and worst-case analysis of in patient scoliosis: A systematic review with second progression model for brace treatment. Studies in health technology and informatics,	299.	Journal of bone and joint surgery. American volume, 1978. 60(2): p. 173-6.	2						
301. Is anterior fusion really necessary? Spine, 2000. 25(1): p. 61-8. 4 302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 1033-43. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke 1C Adolescent Idiopathic Scoliosis. Spine deformity, 2015. 3(6): p. 560-565. 4 305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis: A systematic review with second study. Scoliosis, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - Sta	300.		11						
302. Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p. 1033-43. 5 303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke 1C Adolescent Idiopathic Scoliosis. Spine deformity, 2015. 3(6): p. 560-565. 4 305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis during puberty. 9 306. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 13 308. Noshchenko, A., et al., Preliminary results and worst-case analysis of in patient scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 </td <td>301.</td> <td></td> <td>4</td>	301.		4						
303. Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing rather than initiating factor. Studies in health technology and informatics, 2006. 123(ck1, 9214582): p. 90-4. 5 304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke IC Adolescent Idiopathic Scoliosis. Spine deformity, 2015. 3(6): p. 560-565. 4 305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis: a case control study. Scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 1(1): p. 35-40. 13 308. Noshchenko, A., et al., Preliminary results and worst-case analysis of in patient scoliosis: A systematic review with 5 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 310. Borysov,	302.	Nakakohji, T., Sagittal spinal flexibility in patients with idiopathic scoliosis. Nihon Seikeigeka Gakkai zasshi, 1994. 68(12): p.	5						
304. Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke IC Adolescent Idiopathic Scoliosis. Spine deformity, 2015. 3(6): p. 560-565. 4 305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis: a case control study. Scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 1(1): p. 35-40. 13 308. Noshchenko, A., et al., Preliminary results and worst-case analysis of in patient scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 Nearpini, S., et al., Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric 5	303.	Rigo, M., G. Quera-Salva, and M. Villagrasa, Sagittal configuration of the spine in girls with idiopathic scoliosis: progressing	5						
305. Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis: a case control study. Scoliosis and spinal disorders, 2018. 13(101675716): p. 10. 9 306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 1(1): p. 35-40. 13 308. Noshchenko, A., et al., Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 Nearpini, S., et al., Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric 5	304.	Celestre, P.C., et al., Sagittal Alignment Two Years After Selective and Nonselective Thoracic Fusion for Lenke 1C Adolescent	4						
306. Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty. Journal of children's orthopaedics, 2013. 7(1): p. 43-9. 13 307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 1(1): p. 35-40. 13 308. Noshchenko, A., et al., Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 Nearini S. et al. Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric 5	305.	Burkus, M., et al., Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic	9						
307. Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997. 13 308. Noshchenko, A., et al., Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 Negrini S. et al. Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric 5	306.	Dimeglio, A. and F. Canavese, Progression or not progression? How to deal with adolescent idiopathic scoliosis during puberty.	13						
1(1): p. 35-40. 1(1): p. 35-40. 308. Noshchenko, A., et al., Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 5 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5 Negrini S. et al. Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric 5		Weiss, H.R., et al., Preliminary results and worst-case analysis of in patient scoliosis rehabilitation. Pediatric rehabilitation, 1997.							
meta-analysis. World journal of orthopedics, 2015. 6(7): p. 537-58. 309. Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics, 2008. 140(ck1, 9214582): p. 294-8. 5 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. 5		Noshchenko, A., et al., Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with							
2008. 140(ck1, 9214582): p. 294-8. 310. Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p. 12-6. Negrini S. et al. Physical exercises as a treatment for adolescent idiopathic scoliosis. A systematic review. Pediatric		Lou, E., et al., Preliminary validation of curve progression model for brace treatment. Studies in health technology and informatics,							
12-0. Negrini S, et al. Physical avarcises as a treatment for adolescent idionathic scoliosis. A systematic review Pediatric		Borysov, M., et al., Postural Re-Education of Scoliosis - State of the Art (Mini-review). Current pediatric reviews, 2016. 12(1): p.							
	311.		5						

312.	De Maio, F., et al., Posterior spine fusion with Moss-Miami instrumentation for adolescent idiopathic scoliosis: radiographic,	4						
312.	clinical and patient-centered outcomes. The Iowa orthopaedic journal, 2007. 27 (8908272, b90): p. 28-39.	4						
313.	Geoffrey Burwell, R. and P.H. Dangerfield, <i>Platelet calmodulin levels in adolescent idiopathic scoliosis</i> . Do the levels correlate with curve progression and severity? Spine, 2003. 28 (17): p. 2036-8.	5						
314.	Fusco, C., et al., <i>Physical exercises in the treatment of adolescent idiopathic scoliosis: an updated systematic review.</i> Physiotherapy theory and practice, 2011. 27 (1): p. 80-114.	5						
315.	Andersen, M.O., K. Thomsen, and K.O. Kyvik, <i>Perceived health status in self-reported adolescent idiopathic scoliosis: a survey based on a population of twins.</i> Spine, 2010. 35 (16): p. 1571-4.							
316.	Dobosiewicz, K., et al., Pathomechanic basics of conservative treatment of progressive idiopathic scoliosis according to Dobosiewicz method based upon radiologic evaluation. Studies in health technology and informatics, 2002. 91 (ck1, 9214582): p. 336-41.	13						
317.	de Seze, M. and E. Cugy, <i>Pathogenesis of idiopathic scoliosis: a review</i> . Annals of physical and rehabilitation medicine, 2012. 55 (2): p. 128-38.	13						
318.	Vialle, R., [Orthopedic and surgical treatment of idiopathic scoliosis in childhood and adolescence]. Traitement orthopedique et chirurgical des scolioses idiopathiques chez l'enfant et l'adolescent., 2006. 56(2): p. 147-56.	6						
319.	Goldberg, M.S., et al., Observer variation in assessing spinal curvature and skeletal development in adolescent idiopathic scoliosis. Spine, 1988. 13(12): p. 1371-7.	13						
320.	Green, N.E., <i>Part-time bracing of adolescent idiopathic scoliosis</i> . The Journal of bone and joint surgery. American volume, 1986. 68 (5): p. 738-42.	13						
321.	Hensinger, R.N. and G.H. Thompson, Orthotic management in adolescent idiopathic scoliosis: leveling the playing field. Journal of pediatric orthopedics, 2007. 27(4): p. 367-8.	5						
322.	Hasler, C., et al., No effect of osteopathic treatment on trunk morphology and spine flexibility in young women with adolescent idiopathic scoliosis. Journal of children's orthopaedics, 2010. 4(3): p. 219-26.	11						
323.	Grivas, T.B., G.I. Rodopoulos, and N.V. Bardakos, Night-time braces for treatment of adolescent idiopathic scoliosis. Disability and rehabilitation. Assistive technology, 2008. 3(3): p. 120-9.	5						
324.	Brigham, E.M. and D.G. Armstrong, Motivations for Compliance With Bracing in Adolescent Idiopathic Scoliosis. Spine deformity, 2017. 5(1): p. 46-51.	5						
325.	Bettany-Saltikov, J., et al., Management of Spinal Deformities and Evidence of Treatment Effectiveness. The open orthopaedics journal, 2017. 11(101480491): p. 1521-1547.	5						
326.	Frez, R., J.C.Y. Cheng, and E.M.C. Wong, Longitudinal changes in trunkal balance after selective fusion of King. Spine, 2000. 25(11): p. 1352-1359.	4						
327.	Willers, U., et al., Long-term results of Boston brace treatment on vertebral rotation in idiopathic scoliosis. Spine, 1993. 18(4): p. 432-5.	13						
328.	Seifert, J. and A. Selle, [Is night-time bracing still appropriate in the treatment of idiopathic scoliosis?]. Hat die Nachtorthese in der Skoliosetherapie noch eine Berechtigung?, 2009. 38(2): p. 146-50.	6						
329.	Wu, H., et al., Scoliotic progression patterns in prognostic factors and future prediction of spinal deformity progression. Studies in health technology and informatics, 2006. 123 (ck1, 9214582): p. 40-6.	9						
330.	Lonstein, J.E., Scoliosis: surgical versus nonsurgical treatment. Clinical orthopaedics and related research, 2006. 443(0075674, dfy): p. 248-59.	5						
331.	Emans, J.B., Scoliosis: diagnosis and current treatment. Women & health, 1984. 9(2-3): p. 81-102.	5						
332.	Gross, C., et al., Scoliosis and growth. An analysis of the literature. Clinical orthopaedics and related research, 1983(175): p. 243- 50.	5						
333.	Carman, D., et al., <i>Role of exercises in the Milwaukee brace treatment of scoliosis</i> . Journal of pediatric orthopedics, 1985. 5 (1): p. 65-8.	13						
334.	Dickson, R.A., et al., <i>School screening for scoliosis: cohort study of clinical course.</i> British medical journal, 1980. 281 (6235): p. 265-7.	8						
335.	Adobor, R.D., et al., School screening and point prevalence of adolescent idiopathic scoliosis in 4000 Norwegian children aged 12 years. Scoliosis, 2011. 6(101271527): p. 23.	5						
336.	Hackenberg, L., et al., [Scanning stereographic surface measurement in idiopathic scoliosis after VDS (ventral derotation spondylodesis)]. Rasterstereografische Oberflachenvermessung idiopathischer Skoliosen nach VDS., 2000. 138(4): p. 353-9.	6						
337.	Rigo, M., et al., <i>Retrospective results in immature idiopathic scoliotic patients treated with a Cheneau brace.</i> Studies in health technology and informatics, 2002. 88 (ck1, 9214582): p. 241-5.	13						
338.	Pizones, J., et al., Relationship between the different torsion-related thoracic deformity parameters of adolescent idiopathic scoliosis. European journal of orthopaedic surgery & traumatology : orthopedie traumatologie, 2016. 26(7): p. 763-9.	5						
339.	Minami, S., [Results of brace treatment in idiopathic scoliosisevaluation of the patients treated for over 2 years or those who completed the treatment]. Nihon Seikeigeka Gakkai zasshi, 1982. 56(6): p. 471-85.	6						
340.	Klos, S.S., et al., Reliability of a functional classification system in the monitoring of patients with idiopathic scoliosis. Spine, 2007. 32 (15): p. 1662-6.	13						
341.	Cimic, M., et al., Relationship between clinical contourometric measurements and vertebral rotation in adolescent idiopathic scoliosis. Collegium antropologicum, 2009. 33 Suppl 2 (8003354): p. 127-33.	13						
342.	Schulte, T.L., et al., Raster stereography versus radiography in the long-term follow-up of idiopathic scoliosis. Journal of spinal disorders & techniques, 2008. 21(1): p. 23-8.	13						
343.	Honeyman, C., Raising awareness of scoliosis among children's nurses. Nursing children and young people, 2014. 26(5): p. 30-38.	5						
344.	Simony, A., et al., Radiological Outcomes in Adolescent Idiopathic Scoliosis Patients More Than 22 Years After Treatment. Spine deformity, 2015. 3(5): p. 436-439.	13						
345.	Hui, S.C.N., et al., Radiation does of digital radiography (DR) versus micro-dose x-ray (EOS) on patients with adolescent idiopathic scoliosis: 2016 SOSORT- IRSSD "John Sevastic Award" Winner in Imaging Research. Scoliosis and spinal disorders, 2016. 11(101675716): p. 46.							
346.	Bullmann, V., et al., [Prospective evaluation of braces as treatment in idiopathic scoliosis]. Prospektive Untersuchung zur Korsettbehandlung bei idiopathischen Skoliosen., 2004. 142(4): p. 403-9.	6						
347.	Heine, J. and H. Reher, [Progression of untreated idiopathic scoliosis up to the end of growth]. Die Progredienz der unbehandelten idiopathischen Skoliose bis Wachstumsabschluss, 1975. 113(1): p. 87-96.	6						

348.	Meade, K.P., et al., Progression of unsupported curves in adolescent idiopathic scoliosis. Spine, 1987. 12(6): p. 520-6.	5						
349.	Bjerkreim, I. and I. Hassan, Progression in untreated idiopathic scoliosis after end of growth. Acta orthopaedica Scandinavica, 1982. 53(6): p. 897-900.	9						
350.	Hassan, I. and I. Bjerkreim, Progression in idiopathic scoliosis after conservative treatment. Acta orthopaedica Scandinavica, 1983. 54(1): p. 88-90.	11						
351.	Blaha, J. and V. Hybnerova, [Prognostics in idiopathic scoliosis]. Prognostika idiopaticke skoliozy., 2005. 48(1): p. 27-30.	6						
352.	Ohashi, M., et al., Predicting Factors at Skeletal Maturity for Curve Progression and Low Back Pain in Adult Patients Treated Non-Operatively for Adolescent Idiopathic Scoliosis with Thoracolumbar/Lumbar Curves: A Mean 25-Year Follow-up. Spine, 2018(7610646, uxk, 7610649).							
353.	Beaulieu, M., et al., <i>Postural imbalance in non-treated adolescent idiopathic scoliosis at different periods of progression</i> . European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 2009. 18 (1): p. 38-44.							
354.	Rigo, M., Patient evaluation in idiopathic scoliosis: Radiographic assessment, trunk deformity and back asymmetry. Physiotherapy theory and practice, 2011. 27(1): p. 7-25.	13						
355.	Muminagic, S., et al., Way of life as emphasizing factors in the progression of idiophatic scoliosis in adolescence era. Materia socio-medica, 2012. 24(3): p. 182-5.	13						
356.	Hodson, G.C., Vertebral body dimensions: an aid to diagnosis of severely progressive adolescent idiopathic scoliosis? The Australian journal of physiotherapy, 1984. 30 (2): p. 39-41.	5						
357.	Fortin, C., et al., Validity of a quantitative clinical measurement tool of trunk posture in idiopathic scoliosis. Spine, 2010. 35 (19): p. E988-94.	9						
358.	Noonan, K.J., et al., <i>Use of the Milwaukee brace for progressive idiopathic scoliosis.</i> The Journal of bone and joint surgery. American volume, 1996. 78 (4): p. 557-67.	13						
359.	Ono, T., [Trunk deformity in scoliosis studied by surface measurement]. Nihon Seikeigeka Gakkai zasshi, 1995. 69(10): p. 915-26.	6						
360.	Davidson, D., M. Letts, and J. Jarvis, <i>Triple major curves in children</i> . Canadian journal of surgery. Journal canadien de chirurgie, 2003. 46 (3): p. 193-8.	13						
361.	Bassett, G.S., W.P. Bunnell, and G.D. MacEwen, <i>Treatment of idiopathic scoliosis with the Wilmington brace. Results in patients with a twenty to thirty-nine-degree curve.</i> The Journal of bone and joint surgery. American volume, 1986. 68 (4): p. 602-5.	13						
362.	Bergoin, M., [Treatment of idiopathic scoliosis in children]. Le traitement de la scoliose idiopathique de l'enfant., 1993. 40 (4): p. 259-69.	6						
363.	Cook, S.D., et al., <i>Trabecular bone mineral density in idiopathic scoliosis</i> . Journal of pediatric orthopedics, 1987. 7(2): p. 168-74.	13						
364.	Peltonen, J., M. Poussa, and M. Ylikoski, <i>Three-year results of bracing in scoliosis</i> . Acta orthopaedica Scandinavica, 1988. 59 (5): p. 487-90.	13						
365.	Christine, C., C. Alin, and C.H. Rivard, <i>Treatment of early adolescent idiopathic scoliosis using the SpineCor System</i> . Studies in health technology and informatics, 2008. 135 (ck1, 9214582): p. 341-55.	13						
366.	Brink, R.C., et al., <i>Three-dimensional pelvic incidence is much higher in (thoraco)lumbar scoliosis than in controls.</i> European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 2018(9301980, b9y).	13						
367.	Bunnell, W.P., G.D. MacEwen, and S. Jayakumar, <i>The use of plastic jackets in the non-operative treatment of idiopathic scoliosis</i> . <i>Preliminary report.</i> The Journal of bone and joint surgery. American volume, 1980. 62 (1): p. 31-8.	13						
368.	De Smet, A.A., et al., <i>The top view for analysis of scoliosis progression</i> . Radiology, 1983. 147(2): p. 369-72.	5						
369.	Stokes, I.A., L.C. Bigalow, and M.S. Moreland, <i>Three-dimensional spinal curvature in idiopathic scoliosis</i> . Journal of orthopaedic research : official publication of the Orthopaedic Research Society, 1987. 5 (1): p. 102-13.	13						
370.	Perdriolle, R. and J. Vidal, Thoracic idiopathic scoliosis curve evolution and prognosis. Spine, 1985. 10(9): p. 785-91.	13						
371.	Tredwell, S.J. and M. Bannon, <i>The use of the ISIS optical scanner in the management of the braced adolescent idiopathic scoliosis patient</i> . Spine, 1988. 13 (10): p. 1104-5.	13						
372.	Denton, T.E., F.M. Randall, and D.A. Deinlein, <i>The use of instant moire photographs to reduce exposure from scoliosis radiographs</i> . Spine, 1992. 17 (5): p. 509-12.	13						
373.	Tomory, I. and M. Orosz, [<i>The significance of biological age in the progression and treatment of idiopathic scoliosis</i>]. A biologiai kor jelentosge az idiopathias scoliosis progressiojaban es kezeleseben., 1979. 120 (31): p. 1861-3.	6						
374.	Weiss, H.R., [The Schroth scoliosis-specific back schoolinitial results of a prospective follow-up study]. Die skoliosespezifische Ruckenschule nach Schrotherste Ergebnisse einer prospektiven Verlaufskontrolle., 1995. 133 (2): p. 114-9.	6						
375.	Sun, X., et al., [The role of initial bone mineral status in predicting the early outcome of brace treatment in girls with adolescent idiopathic scoliosis]. Zhonghua wai ke za zhi [Chinese journal of surgery], 2008. 46 (14): p. 1066-9.	6						
376.	Weatherley, C.R., et al., <i>The rib deformity in adolescent idiopathic scoliosis. A prospective study to evaluate changes after</i> <i>Harrington distraction and posterior fusion.</i> The Journal of bone and joint surgery. British volume, 1987. 69 (2): p. 179-82.	4						
377.	Hopf, C., E. Sandt, and J. Heine, [<i>The progression of untreated idiopathic scoliosis in the x-ray image</i>]. Die Progredienz unbehandelter idiopathischer Skoliosen im Rontgenbild., 1989. 151 (3): p. 311-6.	6						
378.	Dickson, R.A., et al., <i>The pathogenesis of idiopathic scoliosis. Biplanar spinal asymmetry</i> . The Journal of bone and joint surgery. British volume, 1984. 66 (1): p. 8-15.	13						
379.	Palmisani, M., et al., <i>The epidemiology of idiopathic scoliosis in the city of Bologna. A three-year review of positive cases.</i> La Chirurgia degli organi di movimento, 1990. 75 (4): p. 353-60.	8						
380.	Eardley-Harris, N., et al., <i>The effectiveness of selective thoracic fusion for treating adolescent idiopathic scoliosis: a systematic review protocol.</i> JBI database of systematic reviews and implementation reports, 2015. 13 (11): p. 4-16.	5						
381.	Murray, D.W. and C.J. Bulstrode, <i>The development of adolescent idiopathic scoliosis</i> . European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 1996. 5 (4): p. 251-7.	13						
382.	Lee, C.S. and A.L. Nachemson, <i>The crankshaft phenomenon after posterior Harrington fusion in skeletally immature patients with thoracic or thoracolumbar idiopathic scoliosis followed to maturity.</i> Spine, 1997. 22 (1): p. 58-67.	4						
383.	Weiss, H. and G.M. Weiss, <i>Brace treatment during pubertal growth spurt in girls with idiopathic scoliosis (IS): a prospective trial comparing two different concepts.</i> Pediatric Rehabilitation, 2005. 8 (3): p. 199-206.							
	Beauséjour, M., et al., Pathways of healthcare utilisation in patients with suspected adolescent idiopathic scoliosis: a cross-	5						

385.	Borysov, M. and T. Mogiliantseva, Rehabilitation of Adolescents with Scoliosis During Growth - Preliminary Results Using a Novel Standardized Approach in Russia. Current Pediatric Reviews, 2016. 12(1): p. 31-35.	13						
386.	Janusz, P., et al., Xbal and PvulI polymorphisms of estrogen receptor 1 gene in females with idiopathic scoliosis: no association with occurrence or clinical form. PloS one, 2013. 8(10): p. e76806.	9						
387.	Betz, R.R., et al., Vertebral body stapling: a fusionless treatment option for a growing child with moderate idiopathic scoliosis. Spine, 2010. 35 (2): p. 169-76.	4						
388.	Grivas, T.B., et al., <i>What a school screening program could contribute in clinical research of idiopathic scoliosis aetiology.</i> Disability and rehabilitation, 2008. 30 (10): p. 752-62.							
389.	Cuddihy, L., et al., Vertebral Body Stapling versus Bracing for Patients with High-Risk Moderate Idiopathic Scoliosis. BioMed	4						
390.	research international, 2015. 2015 (101600173): p. 438452. Ding, Lx., et al., [Vertebra morphology measurement in adolescent idiopathic scoliosis population]. Zhonghua wai ke za zhi [Chianga Guranga 2000, 47(2): n. 1227.2]							
391.	[Chinese journal of surgery], 2009. 47 (17): p. 1327-31. Stewart, D.G., Jr., et al., <i>Consultation with the specialist: adolescent idiopathic scoliosis</i> . Pediatrics in Review, 2006. 27 (8): p.	5						
392.	299-306. Salvador-Esteban, E. and P. Luengo-González, <i>Etiopathogeny and natural history of idiopathic scoliosis</i> . Rehabilitacion, 2009.	6						
	43 (6): p. 258-264. Chun, E.M., et al., <i>The change in ratio of convex and concave lung volume in adolescent idiopathic scoliosis: a 3D CT scan based</i>							
393.	cross sectional study of effect of severity of curve on convex and concave lung volumes in 99 cases. European Spine Journal, 2008. 17(2): p. 224-229.	5						
394.	Escalada, F., et al., Growth, maturity and prognosis of scoliosis. Maturity evaluation methods. Rehabilitation, 2009. 43(6): p. 276-280.	6						
395.	Wong, M.S., Early intervention in non-operative management of adolescent idiopathic scoliosis. 2000, Hong Kong Polytechnic (People's Republic of China). p. 186 p.	13						
396.	Wright, A., The conservative management of adolescent idiopathic scoliosis. Physical Therapy Reviews, 1997. 2(3): p. 153-163.	13						
397.	Rooney, R.C. and B.A. Akbarnia, <i>Idiopathic scoliosis in children: an update on screening and bracing</i> . Journal of Musculoskeletal Medicine, 2004. 21 (5): p. 268-275.	5						
398.	Nault, M.L., et al., US and European Risser grading systems: which one best predict [sic] the curve acceleration phase in adolescent idiopathic scoliosis? Studies in Health Technology & Informatics, 2008. 140: p. 343-343.	5						
399.	Nault, M., et al., 3D spine morphologic differences at first visit between non evolutive and evolutive adolescent idiopathic scoliosis. Studies in Health Technology & Informatics, 2010. 158 : p. 189-189.	5						
400.	Haasbeek, J.F., Adolescent idiopathic scoliosis: recognizing patients who need treatment. Postgraduate Medicine, 1997. 101(6): p. 207-216.	5						
401.	Naranjo, J.J.A., et al., <i>Prognostic value of the end vertebra angle in the treatment of idiopathic scoliosis.</i> Rehabilitation, 2003. 37 (4): p. 201-206.	6						
402.	Lonstein, J.E., Scoliosis update: managing school screening referrals: verify asymmetry on forward bending before ordering films. Journal of Musculoskeletal Medicine, 1999. 16(10): p. 593-605.	13						
403.	Cheuk, K.Y., et al., Abnormal bone micro-architecture and rod-plate configuration in osteopenic adolescent idiopathic scoliosis (AIS). Scoliosis, 2015. 10(SUPPL. 1).	5						
404.	Cheuk, K.Y., et al., Abnormal tabecular plate and rod structure in adolescent idiopathic scoliosis - a study with individual trabecula segmentation (ITS). Journal of Orthopaedic Research, 2016. 34 (Supplement 1).	5						
405.	Kaced, H., H. Belabbassi, and A. Haddouche, <i>Abnormal skeletal growth patterns in adolescent idiopathic scoliosis</i> . Scoliosis, 2013. 8(Supplement 2).	5						
406.	Zaina, F., et al., A retrospective controlled study comparing Spinecor vs exercises for Adolescent Idiopathic Scoliosis. Scoliosis, 2013. 8(Supplement 2).	5						
407.	Coillard, C., A. Circos, and C.H. Rivard, A prospective interventional cohort study of 175 patients treated by the SpineCor orthosis, following the Scoliosis Research Society Criteria. Scoliosis, 2013. 8(Supplement 1).	5						
408.	Veldhuizen, A.G., et al., <i>A new orthotic device in the non-operative treatment of idiopathic scoliosis.</i> Medical Engineering and Physics, 2002. 24 (3): p. 209-218.	13						
409.	Aulisa, A.G., et al., A prospective study in adolescent idiopathic scoliosis affected by thoraco-lumbar and lumbar curves in treatment with a progressive action short brace (PASB): Assessment of results according to the SRS criteria. Journal of Orthopaedics and Traumatology, 2013. 14(1 SUPPL. 1): p. S41.	5						
410.	Lambert, A., et al., A preoperative radiological measurement that can predict poor results after a selective thoracic fusion (STF) in patients with adolescent idiopathic scoliosis (AIS). Spine Journal, 2016. 16(4 SUPPL. 1): p. S99.	5						
411.	Brzeczek, M., K. Slonka, and L. Hyla-Klekot, A new look at the etiopathogenesis of adolescent idiopathic scoliosis. Pediatria i Medycyna Rodzinna, 2016. 12 (2): p. 134-140.	13						
412.	Zheng, Y.P., 3d ultrasound imaging for assessment of scoliosis. Spine Journal, 2012. 12(9 SUPPL. 1): p. 164S.	5						
413.	Heary, R.F., S. Kumar, and C.M. Bono, Bracing for scoliosis. Neurosurgery, 2008. 63(3 SUPPL.): p. A125-A130.	13						
414.	Machida, M., <i>Cause of idiopathic scoliosis</i> . Spine, 1999. 24 (24): p. 2576-2583.	5						
415.	Cobetto, N., et al., Braces optimized with computer-assisted design and simulations are lighter, more comfortable, and more efficient than plaster-cast braces for the treatment of adolescent idiopathic scoliosis. Spine Deformity, 2014. 2(4): p. 276-284.	13						
416.	Wang, Z.W., et al., Bone mineral density as a prognostic factor for curve progression in adolescent idiopathic scoliosis: A longitudinal validation study. Scoliosis, 2015. 10(SUPPL. 1).	5						
417.	Schuller, H.M. and B.E.E.M.J. Veraart, Brace treatment of progressive idiopathic scoliosis; a follow-up study. Nederlands Tijdschrift voor Geneeskunde, 1985. 129 (24): p. 1145-1150.	6						
418.	Bilgic, S., O. Ersen, and A. Schirloglu, <i>Brace treatment in adolescent idiopathic scoliosis</i> . Journal of Clinical and Analytical Medicine, 2012. 3 (1): p. 112-116.	5						
419.	Hill, D.L., et al., Brace treatment for adolescent idiopathic scoliosis-protocols of the canadian spinal deformity study group surgeons. Scoliosis, 2010. 5(SUPPL. 1).	5						
	Surgeons: econosis, 2010. 5(30112, 1). Kuroki, H., Brace treatment for adolescent idiopathic scoliosis. Journal of Clinical Medicine, 2018. 7(6): p. 136.	13						
420		1.5						
420. 421.	Batista, R., et al., Association between vitamin D serum levels and adolescent idiopathic scoliosis. Scoliosis, 2014. 9(Supplement 1).	5						

423.	Zheng, R., et al., Assessment of curve progression on children with idiopathic scoliosis using ultrasound imaging method. European Spine Journal, 2018. 27(9): p. 2114-2119.	13						
424.	Van Rhijn, L.W., B.E.E.M.J. Veraart, and C.M.T. Plasmans, <i>Application of a lumbar brace for thoracic and double thoracic lumbar scoliosis: A comparative study.</i> Journal of Pediatric Orthopaedics Part B, 2003. 12 (3): p. 178-182.	13						
425.	Konieczny, M., P. Hieronymus, and R. Krauspe, Adolescent idiopathic scoliosis-analysis of 16 versus 23 hours in brace-what is the minimal time of brace wear per day? European Spine Journal, 2017. 26(11): p. 3015.	5						
426.	Lam, T.P., et al., Adolescent Idiopathic Scoliosis (AIS) is associated with deranged bone quality - A case-control study using quantitative ultrasound on 636 AIS subjects and 269 normal controls. Bone, 2009. 45(Suppl. 2): p. S62.							
427.	Parent, E., et al., Differences in measures of full-torso surface topography among healthy teenagers are independent of growth indicators. Scoliosis, 2010. 5(SUPPL. 1).	5						
428.	Ersen, O., et al., Difference between spinecor brace and rigid brace during treatment. Scoliosis, 2013. 8(Supplement 1).	5						
429.	Haumont, T., et al., <i>Could posturographic parameters predict the evolution of idiopathic scoliosis?</i> Annals of Physical and Rehabilitation Medicine, 2013. 56 (SUPPL. 1): p. e281.	5						
430.	Hershkovich, O., <i>Correlation between the body mass index, body height and the prevalence of spinal deformities in young adults.</i> Spine Journal, 2011. 11 (10 SUPPL. 1): p. 128S.	5						
431.	Coillard, C., A.B. Circo, and C.H. Rivard, <i>Correlation between maturation, growth spurt, and the progression of adolescent idiopathic scoliosis.</i> Scoliosis, 2013. 8(Supplement 1).	5						
432.	Aulisa, A.G., et al., Correlation between compliance and results of brace treatment in juvenile and adolescent idiopathic scoliosis. Scoliosis, 2014. 9(Supplement 1).	5						
433.	Yeh, N., J.H. Wynne, and T. Hresko, Coronal plane alignment in orthotic treatment of adolescent idiopathic scoliosis: Evaluation of balance parameters in orthotic outcomes. Scoliosis, 2010. 5(SUPPL. 1).	5						
434.	Aulisa, A.G., et al., Conservative treatment in adolescent idiopathic scoliosis with curves over 45degree: Is the measurement in Cobb degrees the only parameter to be considered? Journal of Orthopaedics and Traumatology, 2014. 15 (SUPPL. 1): p. S80.	5						
435.	Lenke, L.G., Commentary: Continuing the quest for identifying specific criteria for the progression of adolescent idiopathic scoliosis. Spine Journal, 2012. 12 (11): p. 996-997.	5						
436.	Durmala, J., T. Kotwicki, and J. Piotrowski, Cheneau bracing with dobomed physiotherapy for thoracic scoliosis: Prospective evaluation of 25 patients followed to skeletal maturity. Scoliosis, 2010. 5(SUPPL. 1).	5						
437.	Ji, X.R., et al., Influence of pre-operative disease course on operation and post-operative quality of life in adolescents with idiopathic scoliosis. Academic Journal of Second Military Medical University, 2010. 31 (6): p. 638-641.	5						
438.	Siebel, T. and J. Heisel, <i>Idiopathic scoliosis - Treatment with the Cheneau corset. Results in 69 patients with advanced scoliosis.</i> Fortschritte der Medizin, 1996. 114 (24): p. 29-32.	6						
439.	Simony, A., et al., <i>How effective is providence night-time bracing in treatment of adolescent idiopathic scoliosis?</i> Spine Journal, 2015. 15 (10 SUPPL. 1): p. S175-S176.	5						
440.	Samdani, A.F. and R.R. Betz, <i>Growth modulation techniques for adolescent idiopathic scoliosis</i> . Seminars in Spine Surgery, 2015. 27 (1): p. 52-57.	5						
441.	Geoffrey, B.R., et al., Girls with right thoracic adolescent idiopathic scoliosis (RT-AIS): Patterns of upper arm length asymmetry and early pubertal skeletal overgrowth both of which decrease with age suggest two initiating abnormal growth processes and aggravating factors. Clinical Anatomy, 2013. 26(3): p. 405-406.	5						
442.	Karimi, M.T., M. Kavyani, and M.R. Etemadifar, <i>Gait analysis in adolescent idiopathic scoliosis walking with Boston brace</i> . Scoliosis, 2014. 9(Supplement 1).	5						
443.	Tan, K.J., Factors associated with spinal deformity progression in adolescent idiopathic scoliosis. Pediatric Health, 2009. 3 (5): p. 445-449.	5						
444.	Ng, S.Y., J. Bettany-Saltikov, and M. Moramarco, <i>Evidence for conservative treatment of adolescent idiopathic scoliosis - Update 2015 (Mini-review)</i> . Current Pediatric Reviews, 2016. 12 (1): p. 6-11.	5						
445.	Dolan, L.A., K.A. Haggerty, and S.L. Weinstein, <i>Evaluation of TLSOS in the bracing in adolescent idiopathic scoliosis trial</i> (<i>BrAIST</i>). Scoliosis, 2013. 8(Supplement 1).	5						
446.	Negrini, S., Efficacy of rehabilitation treatment of adolescent idiopathic scoliosis (braces and exercises). Annals of Physical and Rehabilitation Medicine, 2010. 53(SUPPL. 1): p. e15-e16.	5						
447.	Filipovic, M., Etiology, natural history and conservative treatment of adolescent idiopathic scoliosis. Acta Orthopaedica Iugoslavica, 1999. 30 (1): p. 5-12.	5						
448.	Courvoisier, A., R. Vialle, and W. Skalli, EOS 3D Imaging: Assessing the impact of brace treatment in adolescent idiopathic scoliosis. Expert Review of Medical Devices, 2014. 11(1): p. 1-3.	5						
449.	Donzelli, S., F. Zaina, and S. Negrini, End growth results analysis related to Risser score, Cobb degrees, and curve types at the beginning of the treatment. Scoliosis, 2013. 8(Supplement 1).	5						
450.	Dhawale, A.A., et al., <i>Effectiveness of cross-linking posterior segmental instrumentation in adolescent idiopathic scoliosis: A 2-year follow-up comparative study.</i> Spine Journal, 2013. 13 (11): p. 1485-1492.	4						
451.	Coillard, C., A. Circo, and C. Rivard, Effectiveness of the SpineCor treatment for large scoliotic curves compared to moderate and small curves. Scoliosis, 2012. 7(Supplement 1).	5						
452.	Lam, T.P., et al., <i>Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective study of 294 cases followed-up beyond skeletal maturity.</i> Ultrasound in Medicine and Biology, 2013. 39 (3): p. 381-387.	5						
453.	Hopf, C., E. Sandt, and J. Heine, <i>Progression of untreated idiopathic scoliosis as seen on x-ray film</i> . RoFo Fortschritte auf dem Gebiete der Rontgenstrahlen und der Neuen Bildgebenden Verfahren, 1989. 151 (3): p. 311-316.	6						
454.	Ng, W.Y.W., et al., Preliminary results of the spinecor technique in the treatment of adolescent idiopathic scoliosis in a south- east Asian population. Proceedings of Singapore Healthcare, 2012. 21(SUPPL. 1): p. S168.	5						
455.	Maruyama, T., et al., Outcomes of brace treatment for adolescent idiopathic scoliosis with curve magnitude of 41 to 50 degrees. Scoliosis, 2014. 9(Supplement 1).	5						
456.	Zaina, F., et al., Overweight is not predictive of reduced effectiveness of orthosis treatment in Adolescent Idiopathic Scoliosis: Results from a retrospective cohort. Scoliosis, 2013. 8(Supplement 2).	5						
457.	Meurer, A., C. Hopf, and J. Heine, Natural course of idiopathic scoliosis. Orthopade, 1994. 23(3): p. 228-235.	6						
458.	Bobin, V., et al., Monitoring of progressive idiopathic scoliosis by BIOMOD TM-L, back surface topography device, during plaster treatment and progressive felting. Annals of Physical and Rehabilitation Medicine, 2016. 59 (Supplement): p. e91-e92.	5						
459.	Ledonio, C.G.T., et al., Minimum 20-year radiographic outcomes for treatment of adolescent idiopathic scoliosis: Preliminary results from a novel cohort of us patients. Spine Journal, 2014. 14(11 SUPPL. 1): p. S36.	5						

r	Charles in D. D. Biston d. D. Frenchet Learning and and interaction of Learning the (227 articula). Analysis in the detection	
460.	Charlopain, P., B. Biot, and R. Fauchet, <i>Lyonnais orthopedic treatment: Long-term results (237 patients)</i> . Annales de Readaptation et de Medecine Physique, 1998. 41 (3): p. 147-153.	6
461.	Aulisa, A.G., et al., <i>Lyon bracing in adolescent females with idiopathic scoliosis: Assessment of results according to the SRS committee on bracing and non-operative management standardization criteria.</i> Journal of Orthopaedics and Traumatology, 2012. 13 (SUPPL.1): p. S32-S33.	5
462.	Allard, P., et al., Is standing balance control in adolescent idiopathic scoliotic girls associated to the horizontal body and trunk center of mass offset? Scoliosis, 2010. 5(SUPPL. 1).	5
463.	Haleem, S. and C. Nnadi, Scoliosis: a review. Paediatrics and Child Health (United Kingdom), 2018. 28(5): p. 209-217.	5
464.	Guneri, B., et al., Selective posterior lumbar fusion (COBB to COBB) in Lenke type 5 adolescent idiopathic scoliosis: Radiographic and clinical results. Spine Journal, 2016. 16(10 Supplement 1): p. S242-S243.	5
465.	Zottarelli, L., G.A. La Maida, and B. Misaggi, <i>Risser's plaster corset for the treatment of adolescent idiopathic scoliosis: Is it still relevant today?</i> Journal of Orthopaedics and Traumatology, 2013. 14 (1 SUPPL. 1): p. S80.	5
466.	Goetze, H.G. and J. Heine, Results of conservative treatment of scoliosis with the Milwaukee frame. Zeitschrift für Orthopadie und Ihre Grenzgebiete, 1974. 112 (4): p. 1010-1013.	6
467.	Little, D.G. and M.D. Sussman, The Risser sign: A critical analysis. Journal of Pediatric Orthopaedics, 1994. 14(5): p. 569-575.	5
468.	Shannon, T., N. Chockalingam, and N. Jevtic, <i>The radiological and clinical assessment of a cohort of AIS patients in Serbia and Bulgaria</i> . Spine Journal, 2017. 17 (11 Supplement 1): p. S329.	5
469.	de Mauroy, J.C., et al., The new Lyon ARTbrace versus the historical Lyon brace: A prospective case series of 148 consecutive scoliosis with short time results after 1 year compared with a historical retrospective case series of 100 consecutive scoliosis; SOSORT award 2015 winner. Scoliosis, 2015. 10(1): p. 26.	13
470.	Toh, S.H., et al., <i>The efficacy of spinecor brace for adolescent idiopathic scoliosis in Singapore according to the Srs criteria</i> . Physiotherapy (United Kingdom), 2015. 101 (SUPPL. 1): p. eS1075-eS1076.	5
471.	Volz, R., et al., The effect of radiographic vertebral body and intervertebral disc wedging on curve progression in idiopathic scoliosis, Scoliosis, 2013. 8(Supplement 1).	5
472.	Parent, E.C., et al., The ability of surface topography postural measurements to detect cobb angle progression in adolescents with idiopathic scoliosis (AIS) and a main thoracic curve: Full torso scans compared to back only parameters. Scoliosis, 2014. 9(Supplement 1).	5
473.	Zaina, F., et al., SpineCor vs rigid brace for Adolescent Idiopathic Scoliosis: End of growth results from a retrospective controlled study. Scoliosis, 2013. 8(Supplement 2).	5
474.	Zaina, F., et al., Spinecor vs exercises for adolescent idiopathic scoliosis: Short term results. Scoliosis, 2010. 5(SUPPL. 1).	5
475.	Herrero, C. and E. Herrero, SpineCor treatment - The Spanish experience. First results. Scoliosis, 2012. 7(Supplement 1).	5
476.	Shi, B., et al., Spine velocity provides more accurate assessment of curve progression than height velocity in progressive female idiopathic scoliosis undergoing bracing treatment. Scoliosis, 2015. 10 (SUPPL. 1).	5
477.	Willems, P.C., <i>Spinal disorders in children</i> . European Journal of Epidemiology, 2015. 30 (8): p. 754-755.	5
478.	Dickson, R.A., Spinal deformity - Adolescent idiopathic scoliosis: Nonoperative treatment. Spine, 1999. 24(24): p. 2601-2606. Chu, W.C.W., et al., Spinal cord morphology predicts curve progression in adolescent idiopathic scoliosis treated with bracing?	5
479.	A prospective cohort study with magnetic resonance imaging. Scoliosis, 2015. 10(SUPPL. 1). Negrini, S., et al., Specific exercises reduce failure and brace prescription rates in adolescent idiopathic scoliosis patients. A	5
480.	prospective controlled end-growth cohort study of 254 patients. European Spine Journal, 2011. 20(4): p. S543.	5
481.	Escalada, F., et al., Assessment of angle velocity in girls with adolescent idiopathic scoliosis. Scoliosis, 2009. 4(101271527): p. 20.	5
482.	Huitema, G., et al., Anterior versus posterior spinal correction and fusion for adolescent idiopathic scoliosis. Cochrane Database of Systematic Reviews, 2014(9).	5
483.	Fong, D.Y.T., et al., An alternative to a randomised control design for assessing the efficacy and effectiveness of bracing in adolescent idiopathic scoliosis. The bone & joint journal, 2015. 97-B (7): p. 973-81.	13
484.	Comite Nacional de Adolescencia, S.A.P., et al., [Adolescent idiopathic scoliosis]. Consenso de escoliosis idiopatica del adolescente., 2016. 114(6): p. 585-594.	6
485.	Weinstein, S.L., et al., Adolescent idiopathic scoliosis. Lancet, 2008. 371 North American Edition(9623): p. 1527-1537.	5
486.	Price, C.T., Adolescent Idiopathic Scoliosis: Immaturity Predicts Progression, but Is More Brace Wear Time Beneficial? Commentary on an article by Lori A. Karol, MD, et al.: "The Effect of the Risser Stage on Bracing Outcome in Adolescent Idiopathic Scoliosis". The Journal of bone and joint surgery. American volume, 2016. 98 (15): p. e65.	5
487.	Horne, J.P., R. Flannery, and S. Usman, Adolescent idiopathic scoliosis: diagnosis and management. American Family Physician, 2014. 89(3): p. 193-198.	5
488.	Hamad, A., E.B. Ahmed, and A.I. Tsirikos, Adolescent idiopathic scoliosis: a comprehensive approach to aetiology, diagnostic assessment and treatment. Orthopaedics & Trauma, 2017. 31 (6): p. 343-349.	5
489.	Nct, 6 Years Study on Changes in Bone Quality, Bone Turnover and Curve Severity in AIS With and Without Calcium and Vit-D Supplementation. <u>Https://clinicaltrials.gov/show/nct02904421</u> , 2016.	5
490.	Kadoury, S., et al., 3-D Morphology Prediction of Progressive Spinal Deformities From Probabilistic Modeling of Discriminant Manifolds. IEEE transactions on medical imaging, 2017. 36(5): p. 1194-1204.	2
491.	Cho, W., et al., Additional Risk Factors for Adding-On After Selective Thoracic Fusion in Adolescent Idiopathic Scoliosis: Implication of Lowest Instrumented Vertebra Angle and Lumbosacral Takeoff. Spine deformity, 2018. 6(2): p. 164-169.	4
492.	Yim, A.P.Y., et al., Abnormal skeletal growth patterns in adolescent idiopathic scoliosisa longitudinal study until skeletal maturity. Spine, 2012. 37(18): p. E1148-54.	5
493.	Brink, R.C., et al., A reliability and validity study for different coronal angles using ultrasound imaging in adolescent idiopathic scoliosis. The spine journal : official journal of the North American Spine Society, 2018. 18(6): p. 979-985.	5
494.	Coillard, C., A.B. Circo, and C.H. Rivard, A Prospective Randomized Controlled Trial of the Natural History of Idiopathic Scoliosis versus treatment with the Spinecor brace. Sosort Award 2011 Winner. European Journal of Physical & Rehabilitation Medicine, 2014. 50(5): p. 479-487.	13
495.	Lu, L., et al., A novel therapeutic strategy for adolescent idiopathic scoliosis based on osteoporotic concept. Medical hypotheses, 2013. 80 (6): p. 773-5.	5
496.	Zheng, X., et al., A novel evaluation index for adolescent idiopathic scoliosis progression measurement and diagnosis. Technology and health care : official journal of the European Society for Engineering and Medicine, 2014. 22(2): p. 225-41.	5

497.	Gammon, S.R., et al., A comparison of thoracolumbosacral orthoses and SpineCor treatment of adolescent idiopathic scoliosis patients using the Scoliosis Research Society standardized criteria. Journal of pediatric orthopedics, 2010. 30 (6): p. 531-8.	13
498.	Janicki, J.A., et al., A comparison of the thoracolumbosacral orthoses and providence orthosis in the treatment of adolescent idiopathic scoliosis: Results using the new SRS inclusion and assessment criteria for bracing studies. Journal of Pediatric Orthopaedics, 2007. 27(4): p. 369-374.	13
499.	Bettany-Saltikov, J., et al., A Comparison of Patient-Reported Outcome Measures Following Different Treatment Approaches for Adolescents with Severe Idiopathic Scoliosis: A Systematic Review. Asian spine journal, 2016. 10(6): p. 1170-1194.	5
500.	Hasler, C.C., A brief overview of 100 years of history of surgical treatment for adolescent idiopathic scoliosis. Journal of children's orthopaedics, 2013. 7(1): p. 57-62.	5
501.	Parent, E.C., et al., <i>Effects of Schroth exercises added to standard care in adolescents with idiopathic scoliosis (AIS) on surface topography parameters-a randomized controlled trial (RCT).</i> Scoliosis and spinal disorders. Conference: 13th international conference on conservative management of spinal deformities and 1st joint meeting of the international research society on spinal deformities and the society on scoliosis orthopaedic and rehabilitation treatment, SOSORT-IRSSD 2016 meeting. Canada, 2017. 12 (Supplement 1) (no pagination).	13
502.	Fortin, C., et al., Usefulness of the clinical measure of trunk imbalance in adolescent idiopathic scoliosis. Scoliosis and Spinal Disorders, 2016. 11(Supplement 1).	5
503.	Cheuk, K.Y., et al., Unique correlation pattern between bone qualities and handgrip strength in adolescent idiopathic scoliosis (AIS) girls. Scoliosis and Spinal Disorders, 2018. 13(Supplement 1).	5
504.	Cinnella, P., et al., Treatment with cheneau brace in adolescent idiopathic scoliosis (AIS): Retrospective study with follow up of at least 5 years. European Spine Journal, 2012. 21(4): p. 765.	5
505.	Aulisa, A.G., et al., Treatment of lumbar curves in adolescent females affected by idiopathic scoliosis with a progressive action short brace (PASB): Assessment of results according to the SRS committee on bracing and nonoperative management standardization criteria. Scoliosis, 2012. 7(Supplement 1).	5
506.	Lee, T.Y., et al., A novel approach to sagittal profiling of adolescent idiopathic scoliosis using 3D ultrasound. Scoliosis and Spinal Disorders, 2018. 13(Supplement 1).	5
507.	Aubin, CE., J. Clin, and J. Rawlinson, Biomechanical simulations of costo-vertebral and anterior vertebral body tethers for the fusionless treatment of pediatric scoliosis. Journal of Orthopaedic Research, 2018. 36(1): p. 254-264.	5
508.	Brink, R., et al., "Anterior spinal overgrowth" is not exclusive for idiopathic scoliosis, and is located in the disc. Global Spine Journal, 2017. 7(2 Supplement 1): p. 96S.	5
509.	Czaprowski, D., et al., <i>Diurnal variation of body height in children with idiopathic scoliosis</i> . Scoliosis and Spinal Disorders, 2018. 13(Supplement 1).	5
510.	Cheung, P.W.H., et al., Curve progression matching with skeletal growth in adolescent idiopathic scoliosis using the distal radius and ulna (dru) classification. Global Spine Journal, 2018. 8(1 Supplement 1): p. 141S.	5
511.	Quisth, L., I. Beuschau, and A. Simony, Initial experience with the providence nighttime bracing in adolescent idiopathic scoliosis. Prosthetics and Orthotics International, 2015. 39(SUPPL. 1): p. 526.	5
512.	Beuschau, I., L. Quisth, and A. Simony, Initial experience with the providence nighttime bracing in adolescent idiopathic scoliosis. Scoliosis, 2014. 9(Supplement 1).	5
513.	Landauer, F., Improvement in 20 years of AIS brace treatment. Prosthetics and Orthotics International, 2015. 39 (SUPPL. 1): p. 438.	5
514.	Schrander, D., et al., First end-term results of the maastricht brace in the treatment of adolescent idiopathic scoliosis. European Spine Journal, 2016. 25(Supplement 3): p. S432-S433.	5
515.	Rioseco, F., et al., Does proximal hook fixation prevent mid-term proximal junctional kyphosis in patients operated for adolescent idiopathic scoliosis? European Spine Journal, 2017. 26(2 Supplement 1): p. S294.	5
516.	Qian, B., et al., Does disc wedging contribute to the clinical results of brace treatment in adolescent idiopathic scoliosis? European Spine Journal, 2017. 26(2 Supplement 1): p. S402.	5
517.	Simony, A., M.O. Andersen, and S.B. Christensen, Radiological characteristics associated with successfully outcome, after boston brace treatment of adolescent idiopathic scoliosis - A retrospective review. Global Spine Journal, 2018. 8(1 Supplement 1): p. 55S.	5
518.	Simony, A., et al., <i>Providence nighttime bracing, in treatment of adolescent idiopathic scoliosis.</i> European Spine Journal, 2015. 24(6 SUPPL. 1): p. S728.	5
519.	Schrander, D., et al., <i>Predictors of failure in bracing adolescent idiopathic scoliosis</i> . Scoliosis and Spinal Disorders, 2018. 13(Supplement 1).	5
520.	Cheung, J.P.Y., et al., Predicting the risk of curve progression by the radius and distal ulna (DRU) classification for patients with adolescent idiopathic scoliosis. Global Spine Journal, 2018. 8(1 Supplement 1): p. 53S.	5
521.	Korbel, K., et al., <i>Outcome of Cheneau brace treatment for idiopathic scoliosis in adolescents</i> . Scoliosis and Spinal Disorders, 2018. 13 (Supplement 1).	5
522.	Cebeci, B.O., et al., Non-fusion growth modulation with anterior vertebral body tethering for adolescent idiopathic scoliosis: A promising minimal invasive alternative to traditional treatment. European Spine Journal, 2017. 26(2 Supplement 1): p. S291.	5
523.	Stejpien, A. and K. Graff, Rotation of the trunk and pelvis and coupled movements in the sagittal plane in double support stance in adolescent girls with idiopathic scoliosis. Scoliosis and Spinal Disorders, 2016. 11(Supplement 1).	5
524.	Landauer, F. and K. Vanas, Scoliosis as the "first" sign of various diseases. Scoliosis and Spinal Disorders, 2016. 11(Supplement 1).	5
525.	Schrander, D., et al., Results of the Maastricht brace in the treatment of adolescent idiopathic scoliosis. Scoliosis and Spinal Disorders, 2018. 13(Supplement 1).	5
526.	Assi, A., et al., Relationship between spinal deformity in the axial plane and the coronal Cobb angle in adolescent idiopathic scoliosis. European Spine Journal, 2017. 26 (2 Supplement 1): p. S257.	5
527.	Aulisa, A.G., et al., <i>The importance of the lateral profile in the treatment of idiopathic scoliosis.</i> Scoliosis and Spinal Disorders, 2016. 11 (Supplement 1).	5
528.	Parent, E. and A. Richter, The functional properties of paraspinal muscles in adolescents with idiopathic scoliosis (AIS): A systematic review of the literature. Scoliosis and Spinal Disorders, 2016. 11(Supplement 1).	5
529.	Blyme, A.K., et al., The association between severity of scoliosis and lung clearance index (LCI) in patients with adolescent idiopathic scoliosis (AIS). European Spine Journal, 2015. 24(6 SUPPL. 1): p. S697.	5
530.	Borghi, R., et al., Surgical treatment of adolescent idiopathic scoliosis treatment with asymmetric bars and direct vertebral derotation. European Spine Journal, 2017. 26(4): p. 1343.	5

532. <i>controlled trial protocol.</i> BMC musculosk	ONTRAIS: cONservative TReatment for Adolescent Idiopathic Scoliosis: a randomised						
	Abbott, A., H. Moller, and P. Gerdhem, CONTRAIS: cONservative TReatment for Adolescent Idiopathic Scoliosis: a randomised controlled trial protocol. BMC musculoskeletal disorders, 2013. 14.						
533. Abel, M.F., et al., <i>Selective versus nonsel</i> 2011. 36 (14): p. 1103-1112.							
534. Commentary on an article by Rachel M.	Abel, M.F., Brace Success as Related to Curve Type, Compliance, and Maturity in Adolescents with Idiopathic Scoliosis: Commentary on an article by Rachel M. Thompson, MD, et al.: "Brace Success Is Related to Curve Type in Patients with Adolescent Idiopathic Scoliosis". Journal of Bone & Joint Surgery, American Volume, 2017. 99 (11): p. 1-2.						
535. progression? European spine journal : offic	ogarty, <i>Adolescent idiopathic scoliosis: is rising growth rate the triggering factor in</i> cial publication of the European Spine Society, the European Spinal Deformity Society, Spine Research Society, 1993. 2 (1): p. 29-36.	10					
536. Skalli, W., et al., Early Detection of Progr	essive Adolescent Idiopathic Scoliosis. Spine, 2017. 42(11): p. 823-830.	2					
	ogression from skeletal maturity: A simplified classification during adolescence. Journal	8					
	ight velocity to other maturity indicators in idiopathic scoliosis in girls. Journal of Bone	8					
	scoliosis in males. A natural history study. Spine, 1988. 13(10): p. 1091-5.	9					
540. Lonstein, J.E. and J.M. Carlson, <i>The predi</i> of bone and joint surgery. American volum	ction of curve progression in untreated idiopathic scoliosis during growth. The Journal ne, 1984. 66 (7): p. 1061-71.	9					
	y, Scoliosis: a review. The Australian journal of physiotherapy, 1982. 28(3): p. 20-5.	8					
	British medical journal (Clinical research ed.), 1983. 286(6365): p. 615-8.	8					
543. Soucacos, P.N., et al., <i>Risk factors for idio</i> 8.	pathic scoliosis: review of a 6-year prospective study. Orthopedics, 2000. 23(8): p. 833-	8					
	 curve progression in Adolescent Idiopathic Scoliosis using Random Forest regression. 8. 103((Garcia-Cano, Duong) Ecole de technologie superieure, Universite du Quebec, Quebec H3C 1K3, Canada): p. 34-43. 	2					
	adolescent idiopathic scoliosis in girls. Journal of Pediatric Orthopaedics-Part B, 2005.	9					
	progression in idiopathic scoliosis. European spine journal : official publication of the nal Deformity Society, and the European Section of the Cervical Spine Research Society,	9					
	D analysis of mild scoliosis. European Spine Journal, 2013. 22(11): p. 2427-2432.	9					
	gression after treatment with the Wilmington brace for idiopathic scoliosis. Journal of	9					
549. Fernandez-Feliberti, R., et al., <i>Effectivene</i> pediatric orthopedics, 1995. 15 (2): p. 176-	ess of TLSO bracing in the conservative treatment of idiopathic scoliosis. Journal of	9					
	SO treatment of idiopathic scoliosis. An analysis of the Wilmington jacket. Spine, 1988.	9					
	ied Boston brace with anti-rotatory blades on the progression of curves in idiopathic c rehabilitation, 2003. 6 (3-4): p. 237-42.	9					
	ew method using top views of the spine to predict the progression of curves in idiopathic	9					
	he SpineCor brace for the conservative treatment of adolescent idiopathic scoliosis.	13					
	ak, Maturity Indicators and Adolescent Idiopathic Scoliosis Evaluation of the Sanders	13					
	comparative study of TLSO, Charleston, and Milwaukee braces for idiopathic scoliosis.	13					
	ent outcomes by monitoring brace usage. Studies in health technology and informatics,	5					

Appendix 3.3 The Quality Appraisal Tailored for each Domain of the QUIPS Tool

	Description of each domain
Domains of the Quality Appraisal	Key Characteristics: Age, gender, curve Severity, curve type, skeletal maturity, and menarche,
	Must describe key characteristics in population. (Complete score if 4+ of 6, partial if 2 of 6, 1 is no i.e. not enough)
	Must specify the recruitment type (consecutive, randomized, non-randomized) and the source of population (e.g.: database search), partial if only one
	Must describe time from beginning of recruitment for the study to end of recruitment in body of the article. No partial.
	Must either mention hospital/center in body of text, or for single center, have single address in contacts. No partial.
1. Study Participation	Must include both inclusion and exclusion. Partial if only one. NOTE: must mention diagnosis, age, sex, in their criteria plus mention Surgery excluded.
i oliu j i i i opinoli	if <10 patients participated in the study per predictor considered, No.
	Must describe 4/6 key characteristics. Partial if 2/6
	Summary Study participation: The study sample represents the population of interest on key characteristics, sufficient to limit potential bias of the observed relationship between PF and outcome:
	Key Characteristics: Age, gender, radiographic baseline: curve Severity, curve type, skeletal maturity, and menarche,
	Must describe if any participants lost to follow up. If no attrition exists, then boxes are N/A
	If the participants lost to follow up, any attempts for follow-up must be described. No partial
2 Study Attrition	Must give one reason for the participants lost to follow up. No partial
2. Study Attrition	Must describe 2/6 characteristics of the participants lost to follow up
	If lost to follow up must account for 2/6 characteristics
	Study Attrition Summary: Loss to follow-up (baseline sample to study population analyzed) is not associated with key characteristics sufficient to limit potential bias to the observed relationship between PF and outcome:
3. Prognostic Factor Measurement: demographic	Methods and Setting: Surgeon/researcher/rater/reader/reviewer/ clinician, Hospital/center

factors (age, gender, etc.), radiographic factors (Cobb angle, Risser, sign etc.), bone quality, treatment (observation,	The PF must be described clearly (e.g., the Cobb method, the bone quality, skeletal maturity (Risser), curve type (Lenke, King) or cite a paper which described the methods clearly. No if only mentioned cobb method, Risser sign etc. NOTE: give partial if only good definition of some of the predictors studied.
bracing)	Must mention validity and reliability measures if multiple readers (surgeons/researchers), or if blinded. If a researcher did the measurements, his/her experience required). Partial if only 1/2 reliability measures. NOTE: Validity is supported by Citation to validity study or correlations with measurement gold standard or another measurement previously validated for the same goal.
	Need a justification for the choice of cut-off selected a priori. Could cite a paper to justify use of a cutoff.
	Must provide the description of the place (hospital, center), and the researcher, If one described only, partial. NOTE: for self-reported variable we do not need to have rater information.
	At least 75% of the participants must have the data for PF (mark down if less than 75% had) NOTE: unsure if not reported.
	Must describe the methods used for the missing data (replaced by mean of the data, etc.) If no missing data, N/A
	PF Measurement Summary: PF is adequately measured in study participants to sufficiently limit potential bias.
	Methods and Setting: Surgeon/researcher/rater/reader/reviewer/ clinician, Hospital/center
	Must define curve progression (e.g. curve change >6, etc.) / cite a paper that explained it clearly, must mention the method used to measure the curve progression (Cobb method), must mention the duration of the follow-ups to measure the curve progression if 1/3, partial
4. Outcome Measurement: curve progression	Need also validity data. Reliability measures if multiple readers (surgeons/researchers), or if blinded. If a researcher did the measurements, his/her experience required). Partial if only one of validity or reliability of measures provided.
	Must provide the description of the place (hospital, center), and the researcher, etc. If one described only, partial.
	Outcome Measurement Summary: Outcome of interest is adequately measured in study participants to sufficiently limit potential bias.
	Key Confounders: Age, radiographic baseline: Curve Severity, curve type, skeletal maturity, gender, treatment (brace, observation) and menarche if not as the "PF"
	Must mention 4/7 of the important confounders measured. If 3/7 measured, partial NOTE: Give yes reported adequately if recruited only one level or the confounders.
5. Study Confounding	Must clearly define the important confounders listed. If 3/7 measured, partial
crossing consuming	Must mention the reliable method of measurements for the curve type (Lenke, King) and skeletal maturity (Risser, sanders, tanner). If other methods without demonstrated reliability was used for either curve type/skeletal maturity, partial. if either blinded/multiple readers' reliability of X-ray information mentioned, partial
	Must provide the description of the place (hospital, center) and the surgeon. If one described only, partial.
	Must provide methods of imputation if certain patients missing data. If no missing data, N/A.

	For listed confounders, must either only include patients of those confounders, or separate patients in method.
	For listed confounders, if not accounted in study design, must analyze patients of each confounder separately.
	Study Confounding Summary: Important potential confounders are appropriately accounted for, limiting potential bias with respect to the relationship between PF and outcome.
	Key statistics: name of analysis, the significant threshold, for multivariable analyses state the variable selection method
	Must have sufficient statistics, partial if report only 1/3 for the listed statistics
	The model (regression) must provide appropriate conceptual framework. Otherwise, N/A.
6. Statistical Analysis and Reporting	The developed model must have the appropriate statistics. Otherwise, N/A.
	If in methods describe reporting of data related to curve progression must report the results
	Statistical Analysis/Presentation Summary: The statistical analysis is appropriate for the design of the study, limiting potential for presentation of invalid or spurious results.

Note: QUIPS is Quality in Prognostic Studies, red cells are the most important, yellow is moderate, and green is less critical. To determine the overall score for each article, the questions under each domain were prioritized based on their importance (the questions highlighted as red, yellow, and green based on the importance, respectively). Articles that were missing the information about those questions achieved lower scores.

Appendix 3.4 Summary of the Parameters for the Articles with Progression Criteria $\geq 5^{\circ}$

Predictive factors	Authors	Parameters	Non- pre	Pre	Univariate analysis	Multivariate analysis	ROB	Follow- up interval
	Aulisa, et al. [1]	Age (mean \pm SD=11.8 \pm 0.5)	×		Spearman correlation=0.01, P=NS		М	L
	Aulisa, et al. [2]	Age at initial (mean=12.6)		~		Reg, B=0.87, SE=0.25, R ² =0.48, P=0.001	М	S
	Bohl, et al. [3]	Age (mean=12.9, range=10-15)	×		Chi square, P=0.256		М	L
	Bunnell [4]	Age at initial (mean=12), Age at menarche (mean=12.5)	×	~	Age at diagnosis reported to be a prognostic factor		М	S
	Danielsson, et al. [5]	Age at initial Age at maturity	× ×		Mann-Whitney, U test, P=0.82, U test, P=0.1		М	L
	Guo, et al. [6]	Initial Age in three bracing groups (SpineCor (pro during treatment) SpineCor (pro beyond maturity) (10-14) Rigid (pro beyond maturity),	×		Test: Unknown: SpineCor (pro during treatment), P=0.664, SpineCor (pro beyond maturity), P=0.616 Rigid (pro beyond maturity), P=0.322		М	L
Age (years)	Guo, et al. [7]	Age (mean=pro:12-non- pro:13)	×		T-tests, P>0.05		М	L
Ÿ	Hung [8]	initial age (range=11-13 yr)		~	Model=SN=76% (95% CI, 69%- 83%), SP=70% (95% CI, 62%- 77%), (younger age at the time of diagnosis, higher risk)	Log reg, B=0.72, OR- =2.1 (95% CI,1.1-4), P=0.03, AUC=0.8 (95% CI, 0.75- 0.85);	L	S
	Karol [9]	Age (range=10-16)	×		Fisher's exact test (two-tailed), P> 0.05		М	L
	Karol, et al. [10]	Age: categories=8- 10, 11-12, 13-14, \geq 15, age total (younger age, 8-10 to be as a high risk group)		~	Kolmogorov- Smirnov test, P< 0.005		М	S
	Katz, et al.[11]	Age (mean=Pro: 12.6, Non-pro: 12.7)	×		T-test, P=0.68		М	L
	Konieczny, et al. [12]	Age at menarche (mean \pm SD =12.3 \pm 0.2)	×		No correlation, P>0.05		Н	S
	Kuroki, et al. [13]	Age (mean =12, range=10-14)	×		Difference test=NS		М	L

LAM, et al. [14]	Age (≤13, >13)		~	Chi, P<0.001, SN=0.874, SP=0.665	Log reg= - 1.756, AOR=5.79 (95% CI,3.02- 11.1), AUC=0.831 (95% CI, 0.785-0.877)	М	L
Lara[15]	Age at initial (mean ±SD=14.1 ±1.8 (range, 10.0- 18.7)	×			Log reg, P=0.342,	L	L
LeBlanc [16]	Age (mean=14, range=9-20)	×		Discriminant analysis, P=0.1662		М	S
Mao, et al. [17]	Age at initial visit, mean \pm SD =Pro: 12.0 \pm 1.1 and non- pro: 12.4 \pm 1.0 (10- 14) and Age at menarche, mean \pm SD =Pro: 12.3 \pm 1.1 and non- pro: 12.3 \pm 1.0 (10- 14)	×		T-test, P=0.132 T-test, P=0.974	Reg, B=-0.555, R ² = 0.043, OR=1.742 (age at initial visit)	L	L
Nault, et al. [18]	Age (Pro: mean=13.3 (range=11-15), Non-pro: mean=12.5 (range=10-14))	×		T-test, P=0.49		М	L
Nault, et al. [19]	Age (mean ±SD =Pro and Non-pro: 12.6 ± 1.2)	×		T-test, P=0.8		М	L
O'Neill, et al. [20]	Age at initial (mean=12.8, range=11-17)	×		Cor Coef (r)=-0.118		L	L
Ohrt- Nissen, et al. [21]	Age (mean ±SD =13.3±1.5)		*	Unpaired T- test=P <0.001, reg, B=-2.86(95% CI, -4.441.28), P< <0.001	Reg, B= flexibility: - 0.21 (-0.32 to - 0.10), P<0.001, R2= 0.23; Premenarchal: 5.95 (0.44 to 11.54), P=0.035, R2= 0.39; Cobb (degrees): - 0.24 (-0.77 to 0.29), P=0.366, R ² = 0.40; (all significant variables); age: -1.16 (-3.0 to 0.67), P=0.209, R ² =0.42	L	L
Peterson, et al. [22]	Age (≤ 13 , >13) (younger ages, ≤ 13)		√*	SN, SP= 81%; +LR=82%, -LR= 80%;	B= 0.83, SE=0.67, P= 0.22	М	S
Shi, et al. [23]	Age at initial visit (mean ±SD =12.1±1.2, range= 10-14), Age at menarche (mean ±SD	* *		Chi square, P= 0.496, Chi square, P= 0.517		М	L

		=12.7±1.1,	1	1			1	
		-12.7 ± 1.1 , range=10-14)						
	Sun, et al.	Age (mean \pm SD: Pro=12.8 \pm 1.5,	×		t- test, P=0.383			
	[24]	Non-pro=13.1±1.3)	×		t- test, P= 0.521			
		and Age at menarche					L	S
		(mean ±SD:						
		Pro=12.8±1.2,						
		Non-pro=12.6±1.2) Age at initial	×		Chi square,			
	Sun, et al.	Age at initial			P=0.118,			
	[25]				SN=0.59,			
		A co of monorcho	×		SP=0.57,			
		Age at menarche (years)	~		ACC=0.57, OR=1.95,		М	L
		0)			+LR=1.39, -			
					LR=0.70,			
					+PV=0.24, - PV=0.85			
					Chi, P=NS			
	Vijvermans,	Age at initial			Age at menarche	Age at initial:		
	et al. [26]	Age at bracing Age at menarche			and bracing did not significantly	Reg, B=- 0.160,		
		Age at menarene	×	~	differ between pro	SE=0.08,		
			×		and non pro	P=0.048, Age	Μ	L
						at bracing: Reg, B=0.003,		
						SE= 0.137,		
						P=0.98		
	Weiss [27]	Age 10-15			13 to 15: 23.8% progression rate			
				✓	between cobb 20-		Μ	L
					29 deg			
	Wu, et al.	Initial age (12.3, 8-	×		T-test, P=NS		М	S
	[28]	12)					141	5
	Yamauchi,	Age (Mean=12-13)			Cor Coef=-0.42, (younger age (12	Partial Cor= - 0.11		
	et al. [29]			✓	years at higher	0.11	Μ	L
					risk)			
	Yrjonen, et	Age (mean: 13.1-	×		T-test, P>0.05		М	L
	al. [30]	14.8)		<u> </u>		T	191	
	Zhang, et al.	Age (≤13, >13)	×			Log reg= NS	М	L
	[31]	T: 4: -1			D =0.1 OD 1.02	D D 0.02		
	Zhu, et al.	Initial age (13.7±1.8, 10-15)	×		P=0.1, OR= 1.02 (95%CI, 0.76-	Reg, B=0.03	М	S
	[32]	(10.1.=1.0, 10 10)			1.38)		111	
	Aulisa, et	Gender				Reg, B=0.85,		
	al. [2]		×			SE=1.11, R ² =0.48,	М	S
						R =0.48, P=0.44		
	Bohl, et al.	Gender (male)			Chi=0.01,			
5	[3]				SN=29%, SP=100%,			
Gender					ACC=65%, -			
ಲೆ				✓	LR=71%, +PV=		Μ	L
					100%, -PV=59% (Male was			
					(Male was considered as			
					positive variable)			
	Bunnell [4]	Gender		~	Pro: M=53%		М	S
					F=35%,			-

	1		1	-	(1	r		
					(when Cobb > 30			
		Candan			deg, $F=M$)			
	Davis, et al. [33]	Gender	×		Chi, P= 0.17		L	L
	Lara, et al. [15]	Gender	×		Bivariate analysis, P= 0.53	OR=1.54, (95% CI, 0.75- 3.19), P=0.24	L	L
	Nault, et al. [19]	Gender	×		SN=10%, SP=89%, ACC=49%, OR=0.94, +LR=0.95, -LR=1, +PV=0.5, - PV=0.48, (Male was considered as positive variable)		М	L
	Nault, et al. [18]	Gender	×		SN=17%, SP= 91%, ACC= 61%, OR=2.13, +LR= 1.94, -LR= 0.90, +PV=0.56, - PV=0.62 (Male was considered as positive variable)		М	L
	O'Neill, et al. [20]	Gender (male)		~	Pearson's product moment=0.15	Reg, B=0.14, R ² =18%	L	L
	Ohrt- Nissen, et al. [21]	Gender	×		SN=3%, SP=93%, ACC=41%, OR=0.36, +LR=0.37, -LR= 1.05, +PV=33.33%, - PV=41.67% (Male was considered as positive variable)		L	L
	Vijvermans, et al. [26]	Gender	×			Reg, B= 0.46, SE, 0.43, P=0.28	М	L
	Wu, et al. [28]	Gender (male)		~	$5, \ge 10: M=60\%,$ $26\%, 5, \ge 10:$ F=30%, 10%		М	S
	Yrjonen, et al. [30]	Gender (male)		V	SN=40%, SP=47%, ACC=45%, OR=0.59, +LR= 0.75, -LR=1.27, +PV=18.18%, - PV= 72.73%, (Male was considered as positive variable)		М	L
	Zhu, et al. [32]	Gender	×		OR= 2.45 (95% CI, 0.47-7.48)	Log reg, B=0.89	М	S
Menarche Status	Bunnell [4]	Menarche status (premenarche)		~	Pre menarche, Pro=53%, Non- Pro=47%, Post menarche, Pro=11%, Non- Pro=89%		М	S
Mei	Danielsson, et al. [5]	Premenarche at inclusion		~	Chi square P=0.049		М	L

		1	1				
Guo, [6]	et al. Menarche in three bracing groups (SpineCor, Rigid (Pro beyond maturity), SpineCor (pro beyond maturity)	×		Chi square, P= 0.675, 0.872, 0.178,		М	L
Hung	[8] Menarche status (premenarche as high risk)		v	SN=76% (95% CI, 69%-83%), SP=70% (95% CI, 62%-77%)	Log reg, B=0.91, OR=2.5 (95% CI, 1.00-6.00), P=0.04, AUC=0.8 (95% CI, 0.75- 0.85);	L	S
Katz, [11]	et al. Menarche status	×		Chi square, P=0.24, SN=28%, SP=55%, ACC=45%, OR=0.47, +LR=+0.62, - LR=1.31, +PV=27.77%, - PV=55.17% (post menarche was considered as positive test)		М	L
LAM, [21]	et al. Menarche status (NM)		~	Chi square, P< 0.001, SN=84%, SP=66%, AOR= 8.58 (95% CI, 3.86-19.06)	Log reg, B=2.15, AUC=0.83 (95% CI, 0.78- 0.87),	М	L
Ohrt- Nisser al. [21	111211115K/		×	Chi square, P<0.001, SN=47%, SP=29%, ACC=38%, OR=14.08 (95% CI, 2.87-95.26), P=0.002, +LR=0.65, -LR= 1.85, +PV= 42.86%, - LR=32%, (post menarche was considered as positive test) Reg, B=8.88 (95% CI, 3.92-13.83), P< 0.001	Reg, B= 5.95 (95% CI, 0.44- 11.54), R ² = 0.39, P=0.035	L	L
Peters al. [22	on, et [] Menarche status (premenarche as high risk)		√*	Significant association with progression	Log reg, B=0.65, SE=0.69, P=0.34 (menarche not significant so didn't put not into the model)	М	S
Shi, [23]	et al. Menarche status	×		Chi square, P=0.51		М	L
	et al. Menarche status (premenarche as high risk)		~	SN=74.5%, SP=64.7%,	OR=6.67 (95% CI, 1.31-33.8), P=0.022	L	S

r	1					r	-	
	Sun, et al. [25]	Menarche status (premenarche as high risk)		~	Chi square, P<0.001, Fisher's exact, P< 0.001, SN=51%, SP=23%, ACC=27%, OR=0.21, +LR=0.53, - LR=2.52, +PV=0.11, - PV=0.63, (post menarche was considered as positive test)		М	L
	Zhang, et al. [31]	Menarche status	×			NS (not into the log reg model)	М	L
	Aulisa, et al. [1]	Risser sign (0 to 2)	×		Spearman= 0.07, P=NS		М	L
	Bohl, et al. [3]	Risser sign (0-2)	×		Chi square=P=0.10		М	L
	Bunnell [4]	Risser (0-4) (lower showed higher risk)		~	Risser 0: Pro=68%, Non- pro=32%, Risser 1,2: Pro=52%, Non- pro: 48%, Risser 3,4: Pro=18%, Non- pro: 82%		М	S
	D'Amato, et al. [34]	Risser sign (≤1) (higher risk)		~	Chi square=P=0.05 (Risser=2 associated with non-progression)		М	S
5	Danielsson, et al. [5]	Risser sign (0-4)	×		Mann-Whitney nonparametric U test, P=0.25 (NS)		М	L
Risser Sign	Davis, et al. [33]	Risser sign (0, 1)		~	Chi square=0.012 (Risser 0 associated with higher progression)		L	L
	Gepstein, et al. [35]	Charleston: (TH, TL, L+, Risser 0-2, (TH, TL, L)+ Risser 3-4, Boston: (TH, TL, L)+ Risser 0-2, (TH, TL, L)+ Risser 3-4,	×		Chi square, P= (0.79, NM, 0.75), (0.57, 0.51, 0.78); (0.79, NM, 0.75), (0.57, 0.51, 0.78), Risser (0-2) and (3-4) were not associated with success rate of both methods		М	L
	Guo, et al. [6]	Risser sign (0-2) in two bracing groups SpinCor (progressed during treatment, progressed beyond skeletal maturity), rigid (progressed beyond skeletal maturity)	×		Chi, square P=0.63, 0.09, and 0.36		М	L

			r				
Guo, et al. [7]	Risser sign (0, 1, 2)		~	T- tests, $P < 0.05$ (<1 was at higher risk versus ≥ 1)		М	L
Hung [8]	Lower Risser (0-1)		v	Model=SN=76% (95% CI, 69%- 83%), SP=70% (95% CI, 62%- 77%),	Log Reg, B= 1.54, OR= 4.7 (95% CI, 2.2- 9.9), P<0.001, AUC= 0.8 (95% CI, 0.75- 0.85);	L	S
Karol [9]	Risser sign (≤2)		~	Fisher's exact test, P< 0.05		М	L
Karol, et al. [10]	Risser (earlier stage 0, 1)		~	Kolmogorov- Smirnov test= P< 0.002		М	S
Karol, et al. [36]	Risser 0, 1, 2		~	Chi square P<0.0001, 0.51, 0.51 (Risser 0 at higher risk)		М	S
Katz, et al. [11]	Risser sign (0-2)	×		Chi square P= 0.30		М	L
Katz, et al. [37]	Risser 0, 1, 2		~	Risser stage 0 are at greatest risk for progression		М	L
Kuroki, et al. [13]	Risser sign (0-2)	×		Chi square P=NS		М	L
Lara, et al. [15]	Risser (0-5)+TRC (open/close)	×		Chi square P=0.141		L	L
Mao, et al. [17]	Risser sign (0-2)	×		T-test, P= 0.426		L	L
Nault, et al. [18]	Risser 0+open TRC, Risser 0+ closed TRC, Risser 1	×		Risser 0+open TRC, Pro=20%, Non-pro=16%, Risser 0+close TRC Pro=13%, Non-pro=23%, Risser 1, Pro=7%, Non-pro21%		М	L
O'Neill, et al. [20]	Lower Risser sign		~	Pearson's product moment= -0.262 , (p < 0.05) (n=276 patients)	Reg, B= -0.241, R ² =18%, P< 0.01	L	L
Pasquini, et al. [38]	Risser (0-2)	×		Chi square P=0.643		М	U
Peterson, et al. [22]	Risser sign (0, 1)		~		Log reg, B=2.28, SE=0.69, P<0.001	М	S
Sun, et al. [24]	Risser sign (<2)	×		T-test, P=0.054		L	S
Sun, et al. [25]	Risser sign (0-1)		V	Chi square, P<0.05, SN=51%, SP=64%, ACC=62%, OR=1.94, +LR= 1.45, -LR=0.75, +PV= 0.25, -PV= 0.85, (Risser 0-1 considered as positive test)		М	L

l	1			1		1		
	Thompson, et al. [39]	Risser 0+TH, Risser 0+L, Risser 1,2+Th, R1,2+L	×		Chi square, P=0.21, 0.34		М	U
	Vijvermans, et al. [26]	Risser sign (≤2)		~	Chi square, P <0.001	R0=Log reg, B= -0.39, SE=0.16, R ² =4.21%, P=0.017	М	L
	Yamauchi, et al. [29]	Risser sign (NM)		~	Correlation: -0.47 (Low grades at higher risk) Partial correlation: -0.28		М	L
	Zhang, et al.[31]	Risser sign (≤2)		~	OR=6.137	Log reg, B=1.814, SE=0.76, P=0.01	М	L
	Zhu, et al. [32]	Risser sign (0 or 1)		~	Chi square, P=0.02, SN=51.20, SP- 88.50%, OR=7.51 (95% CI, 1.27- 24.43)	Log reg, B=0.82, AUC=1.2	М	S
()	Karol, et al. [36]	TRC, open (higher risk), closed+ Risser 0		~	Open: Pro=13%, Non-pro=24%, close: Pro=23%, Non-pro=45%		М	S
lge (TRe	Katz, et al.[37]	Triradiate cartilage (TRC) closed open (higher risk)		~	T- test, P= 0.004 (open)		М	L
Triradiate Cartilage (TRC)	Ryan, et al. [40]	TRC, closed, open (higher risk)		¥	Chi square, P= 0.027 SN=68%, SP= 68%, OR=4.45, +LR= 2.09, -LR= 0.47, +PV= 53.57%, - PV= 79.41%, (TRC open considered as positive)		М	L
	Bunnell [4]	Height-weight ratio (NM)	×		Not significant		М	S
x (BMI) (kg/m²)	Goodbody, et al. [41]	High-age group (BMI [85th percentile), mid- BMI group (BMI 20th–85th percentile), low- BMI group (BMI<20 th percentile)		~	High: OR=2.6, P=0.02, Low BMI: OR=3.2, P< 0.01 (as compared to the mid-BMI group)	High: OR=1.2, P=0.71, Low: OR=2.8, P=0.03,	М	L
Body Mass Index (BMI) (k	Konieczny, et al. [12]	Body mass index (BMI) (at skeletal maturity) mean±SD= 20.9± 0.38	*		Pearson correlation, P=0.14		Н	S
Bo	Sun, et al. [24]	BMI, mean±SD= Pro: 16.9±1.5, Non-pro: 16.8±1.4	×		T-test, P= 0.90		L	S
	Zaina, et al. [42]	BMI, mean \pm SD= 27.2 \pm 2.5, \geq 85th percentile		~	Correlation Coef=0.16, P=0.02		L	U

		D 1			D'			-
Body Morpholog y	LeBlanc [16]	Body shape/compositions (List provided in a footnote)		~	Discriminant analysis, P< 0.0001-0.028, ACC=84%		М	S
	O'Neill, et al. [20]	Overweight		~	T-test, P< 0.01, Pearson's product moment= 0.203	Reg, B=0.153, R ² = 18%	L	L
(kg)	Sun, et al. [24]	Weight, mean±SD= Pro: 40.9±6, Non-pro: 41.4±5.5	×		T-Test, P=0.74		L	S
Weight (kg)	Zaina, et al. [42]	Normal weight, mean \pm SD= 49 \pm 8 and overweight BMI \ge 85th percentile, 66 \pm 11	×		Chi square, P=NS		L	U
	Zhu, et al. [32]	Weiht, mean±SD= Pro: 42.2±9.7 , Non-pro: 40.9±11.2	×		T-test, P=0.15		М	S
	Katz, et al.[11]	Peak height velocity ≥9, <9 cm/year)	×		Chi square, P=0.29, SN=58%, SP=59%, ACC=59%, OR=2.04, +LR= 1.43, -LR= 0.70, +PV=35%, -PV= 70.17% (Pre-PHV considered as positive test)		М	L
Height	LeBlanc [16]	Sitting height (mm), mean= Pro: 808.46, Non-pro: 823.06		~	Discriminant analysis, P< 0.0001, 84%		М	S
	Mao, et al. [17]	Height at initial visit, Non-pro: 154.7 ± 7.2 , Pro: 152.4 ± 7.8 (cm)	×		T- test, P=0.249		L	L
	Zhang, et al. [31]	Increasing velocity of standing height $(\geq 30, <30)$, Increasing velocity of sitting height (mm/ year) $\geq 20, < 20$	×			Log reg: P=NS Log reg: P=NS	М	L
	Cheung, et al. [43]	Spinal Growth Velocity at start > 15 mm/year		~	SN=79%, SP= 79%, +PV= 3.78, - PV= 0.26, correlation=0.40	Change in Cobb=1.177*ln (spinal growth velocity)	L	S
Spinal Growth	Ryan, et al. [40]	Growth velocity cm/mo (range) 0.28 (0-1.2) with non-progression	×		T-test, P=0.14 (with non- progression or success rate)		М	L
Spii	Wu, et al. [28]	Spinal growth in coronal and 3D, Maximal spinal length in coronal plane and 3D	* *		High spinal growth velocity was not a clear cause for high incidence of progression.		М	S

	71 4 1	The spinal length			Spinal growth in coronal and 3D: Non-pro: (coronal: r = 0.0-0.38 and 3D: 0.09-0.43), Pro: (coronal: $r = -$ 0.64 to 0.59 and 3D: -0.52 to 0.01); Maximal spinal length in coronal plane and 3D: Non-pro: (coronal: r = 0.08-0.43, 3D: 0.03-0.32), Pro: (coronal: 0.0- 0.29, 3D: -0.07 to 0.2)	Log reg,		
	Zhang, et al. [31]	increasing velocity (≥30, <30 mm/year)		~		B=1.808, SE= 0.77, P=0.02 OR=6.098 (higher length velocity higher risk)	М	L
	Kuroki, et al. [13]	Curve flexibility (upright Cobb vs hanging Cobb)	×		Chi square, P=NS		М	L
Curve Flexibility	Ohrt- Nissen, et al. [21]	Supine bending flexibility (lower flexibility higher progression)		v	T- test, P<0.001, Reg, B= -0.22 (95% CI, -0.32 0.11), P< 0.001	$\begin{array}{c} OR=0.95 \\ (95\% \ CI, \ 0.9- \\ 0.98), \ P=0.013; \\ Reg, \ AB=- \\ 0.21 \ (95\% \ CI, - \\ 0.32- \ 0.1), \\ R^2=23\% \ (for \\ flexibility \ only, \\ the \ overall \\ R^2=42\%) \ , \\ P<0.001 \end{array}$	L	L
Hip Parameters	LeBlanc [16]	Hip width, (mean, Pro=302.6, Non- pro=308.0 (mm)		~	Discriminant analysis=0.0008, ACC=84%		М	S
H Parar	Ploumis, et al. [44]	Femoral head height difference $(\leq 10, > 10 \text{ mm})$	×		Chi, P>0.05		М	L
	Aulisa, et al. [1]	Initial Cobb, mean±SD=:29.30 ± 5.16, FU: 14.67 ± 7.56		~	Kruskal- Wallis=111.902, P< 0.0001		М	L
	Aulisa, et al. [2]	Initial Cobb, mean±SD= 29.8 ± 7.5 and FU 17.1 ± 10.9		~		Reg, B= -0.14, SE=0.04, R ² = 0.48, P=0.001	М	S
Cobb Angle (°)	Bohl, et al.[3]	Initial Cobb, range=25-40	×		Chi square, P=0.31		М	L
Cobb 2	Bunnell [4]	Initial Cobb, mean (range)= 33 (10- 49)		~	Higher progression rate reported for cases with Cobb 40-50 (78%)		М	S
	D'Amato, et al. [34]	Cobb: 20-24; 25-34; 35-42 (> 35 high risk)		~	Proportions: Pro :19%, Non- pro: 81%; Pro: 29%, Non-pro:		М	S

				71%; Pro: 37%,			
Danielsson, et al. [5]	Initial Cobb (range) 25-35	×		Non-pro: 63% Mann-Whitney nonparametric U test, P=0.1 (NS)		М	L
Davis, et al. [33]	Cobb >35 or < 35,	×		Chi square, P=0.95,		L	L
Gepstein, et al. [35]	Charleston vs Boston success rates= Cobb < 25, 25-29, 30-39	×		Chi, P=0.73, 0.59, 0.62 (association with non- progression or brace success rate)		М	L
Guo, et al.[6]	Primary Cobb 20– 30 (in bracing groups as SpineCor (during treatment and beyond skeletal maturity), Rigid	×		Chi square, P=0.957,0.669; 1		М	L
Guo, et al.[7]	Initial Cobb, mean±SD= Pro: 28.5 ± 6.4, Non- pro: 30.1 ± 5.1	×		T- test=NS		М	L
Hung [8]	Initial Cobb (20- 29, 30-39, ≥40) (A larger initial Cobb angle, higher risk)		~	SN=76% (95% CI, 69%-83%), SP= 70% (95% CI, 62%-77%)	Log ref, B= 0.19, 1.56, 1.52; OR= 1.2 (95%CI, 0.6- 2.6); 4.8 (95% CI, 1.9-11.9), P<0.01; 4.6 (95%CI, 1.3- 15.9), P <0.017; AUC=0.8 (95%CI, 0.75- 0.85)	L	S
Karol, et al. [9]	Cobb (10-45)		~	Fisher's exact test, P<0.05, Reported curves ≥ 30 at higher risk vs < 30		М	L
Karol, et al.[10]	Cobb (10-50) (larger curves higher risk) compared <25 Vs ≥25		~	Chi, P=0.0147		М	S
Katz, et al.[11]	Cobb angle (mean, Pro=39.9, Non- pro=39.2), Cobb ratio (thoracic Cobb/ lumbar Cobb, Mean, Pro, Non- pro=1.1)	*		T-test, P=0.35, T-test, P=0.77		М	L
Kuroki, et al. [13]	Initial Cobb (20- 29 vs 30-39)	×		Chi square, P=NS		М	L
LAM, et al. [21]	Cobb (10 to 25, ≥26) (NM)		~	Chi square, P= 0.17, SN=85%, SP=66%, AOR= 2.19 (95% CI, 1.22-3.91)	Log reg, B=0.782, AUC=0.83 (95%CI, 0.78- 0.88), P=0.008	М	L
Lara, et al. [15]	Initial Cobb, mean±SD=		~		P=0.002, OR=1.03 (95% CI, 1.01- 1.04)	L	L

		28.01±13.46 (10-				(Larger curve		1
		86)				magnitude at		
						presentation)		
	Mao, et al. [17]	Cobb at initial visit (20-40)	×		T- test, P=0.263		L	L
	Nault, et al. [18]	Initial Cobb mean±SD= Pro: 23.4 ± 8.6, Non- pro: 21.3 ± 8.3	×		Comparison test (NM), P=0.2		М	L
-	Nault, et al.[19]	Cobb angle at initial visit (11 to 40)	×		T- tests=0.99		М	L
	Ohrt- Nissen, et al. [21]	Initial Cobb mean±SD= Pro=34±4, Non- pro=33±4	×		T- test, P= 0.396; OR= 0.99 (95% CI, 0.83- 1.18), > 0.431; Reg, B= 0.19 (95% CI, - 0.40-0.78), P=0.528,	Reg, B= -0.24 (95% CI, - 0.77- 0.29), R ² = 0.40, P=0.366	L	L
	Pasquini, et al.[38]	Cobb (20-30 vs 30- 40, (high risk))		V	Chi square, P=0.022, SN=100%, SP=26%, ACC=31%, +LR= 1.35, -LR=0, +PV=9.80%, - PV= 100% (20-30 considered as positive)		М	U
	Peterson, et al. [22]	Initial Cobb angle (25 to 35)	×			Log reg, B= 0.37, SE=0.6, R ² =0.54	М	s
	Shi, et al. [23]	Initial Cobb, Cobb at brace weaning	×	~	Chi square=, P= 0.95 Chi square, P= 0.033 (Cobb at brace weaning is significant)		М	L
	Sun, et al.[24]	Initial Cobb (31 to 40)		~	T-test, P=0.047, Model, SN= 74.5%, SP=64.7%, +LR= 2.11, -LR= 1.54,	OR=6.73 (95% CI, 1.3-34.7), P=0.023	L	S
	Sun, et al.[25]	Curve magnitude >30, ≤30	×		Chi square, P=0.26, Cobb= SN= 67%, SP=45%, ACC=49%, OR= 1.65, +LR= 1.22, - LR= 0.73, +PV=0.22, -PV= 0.85 (<) 30 considered as positive)		М	L
	Trivedi and Thomson [45]	Cobb (< 30 vs 30- 40)	×		T-test = P=0.23		М	L
	Vijvermans, et al. [26]	Initial Cobb (Mean: 29.51±6.71) (high Cobb)	×			Log reg, B= 0.03, SE= 0.023, P=0.203, R ² P= 1.17%	М	L

	1	1.1.1.0.11		1	a	,		,
	Weiss [27]	Initial Cobb		~	Greater Cobb showed more progressive curves		М	L
	Wu, et al. [28]	Initial Cobb Maximal Cobb angle		✓ ✓	Significantly different between Pro and Non-pro		М	S
	Yamauchi, et al. [29]	Cobb (20-45)		~	Correlation, coef= -0.3, Partial correlation= -0.03	Reg, B=0.132	М	L
	Yrjonen, et al. [30]	Initial Cobb (mean range 33.1-32.4 in boys and girls)	×		T-test, P>0.05		М	L
	Zhang, et al. [31]	Cobb (≥35 (high risk), <35)		~	OR=13.69	Log reg, B= 2.61, SE= 0.77, P=0.001	М	L
	Zhu, et al. [32]	Initial Cobb (40- 50)	×		P=0.9, OR= 0.78 (95% CI, 0.59- 1.02)	Reg, B= -0.25	М	S
	Bohl, et al. [3]	Curve subtype (TH, TL/L, double)	×		Chi, P= 0.459		М	L
	Bunnell [4]	Curve pattern (TH, TL, L)		~	TH: Pro=77%, Non-pro=23%; TL: Pro=67%, Non-pro=33%, L: Pro=30%, Non- pro=70%; Double: Pro=66%, Non- pro=34%		М	S
et al.[34]	D'Amato, et al.[34]	Curve type (TH, TL, L, double)		~	TH: Pro=37%, Non-pro= 63%; TL: Pro=7%, Non- pro= 93%; L: Pro=6%, Non- pro= 94%; double: Pro=35%, Non- pro= 65% (TH and Double showed the highest rate of pro)		М	S
Curve Pattern	Davis, et al. [33]	Curve classification (TH, TL/L, Double major)	×		Chi, P=0.31		L	L
	Fang, et al. [46]	Curve type (TH, TL, L, Double major)	×		F test, P=0.081		М	S
	Gepstein, et al.	TH, TL, L	×		Chi, P>0.05		М	L
	Guo, et al.[6]	Curve pattern in two bracing groups SpineCor (during treatment, beyond skeletal maturity), rigid	×		SpineCor, Chi=0.238, 0.626; Rigid= 0.828		М	L
	Hung [8]	Curve pattern (TH, TL, L, triple)	×		NS		L	S
	Karol [9]	King classification (1-5)	×		NS		М	L

	atz, et .[11]	Double curves (high risk) versus a single TH, or a L or TL		~	Chi, P= significant		М	L
	uroki, et .[13]	Curve pattern (TH, TL, L)	×		Chi, P=NS		М	L
	ara, et .[15]	Curve pattern (TH, L, TL, double TH, double L)	×		Fisher exact test, P=0.484		L	L
	ault, et .[18]	Curve type (TH, TL, L, double, triple)	×		NS		М	L
Ni	hrt- issen, et .[21]	Curve type (double, TH, TL, L)		~	Curve type: Wilcoxon rank- sum test, P= 0.032,	Double: univariate reg, B= 9.2 (95%CI, -1.5- 19.92), P=0.091; TH: 13.36 (95%CI, 4.2-22.52), P=0.005; TL: 6.29 (95% CI, - 3.96-16.5), P= 0.224; L (as reference)	L	L
Sh [23	ni, et al. 3]	Curve pattern (MT, TL/L, Double)	×		Chi, P= 0.321, 0.381, 0.822		М	L
Su [2:	ın, et al. 5]	Curve pattern (TH, TH+TL/L, TL/L)		~	Chi square, P=0.044		М	L
Su al.	ın, et .[24]	Curve pattern (TH, TL, L)	×		SN=74.5%, SP=64.7%, OR= 11.25 (95% CI, 0.88-24.24), P=0.071		L	S
	rivedi and nomson 5]	Curve type (TH, TL+L)	×		Independent sample test, P= 1.00		М	L
Vi	ijvermans, al.[26]	King classification (1-4)	×			Log reg, B= - 0.028, SE=0.155, R ² P= 0.02%, P= 0.858	М	L
W ⁴ al.	['] u, et .[28]	Curve pattern (double)	×		Double curves (Right TH-Left L) did not consistently progress in the same curve region (primary or minor)		М	S
	rjonen, et .[30]	Curve pattern (TH, TL, L, double major)	×		curve pattern did not have statistical influence on the risk of progression (t -test, P > 0.05)		М	L
Zh [3	nang, et al. 1]	Types of scoliosis (TH, double major)/ curve type (right or left)	×			Log reg=NS	М	L
Zh [32	nu, et al. 2]	Curve pattern (TH, double, TL/L)	×		Chi, P=0.392		М	S

		Vymbooig	×		Both are not			
	Bunnell [4]	Kyphosis Lordosis	×		prognostic		М	S
6		Lordobis			prognostie		101	5
Sagittal Profile (°)	Nault, et al.[18] Peterson, et	Kyphosis Lordosis Kyphosis	*	~	Kyphosis, P=0.02 (S) (low values, higher risk) Lordosis, P=0.45 (NS)	Log reg, B= 0.37, SE=0.61,	M	L
	al.[22]					P=0.54 (NS)	IVI	5
Rib Vertebral Angle and Rib Hump	Katz, et al.[11]	Rib-vertebra angle- convex side [RVA-cx], Rib-vertebra angle - concave side [RVA-cv] Rib-vertebra angle difference [RVAD]	× × ×		T-test, P=0.11 T-test, P=0.75 T-test, P=0.21		М	L
) Vertebi	Peterson, et al.[22]	Rib hump	×			Log reg, B= - 0.37, SE=0.83, P=0.66	М	S
Rit	Yamauchi, et al. [29]	Rib-vertebral angle difference	×		Cor Coef=-0.17, Partial Cor Coef=- 0.06		М	L
	LeBlanc [16]	ASIS circumference (mm)		~	Discriminant analysis, P=0.0039, ACC=84%		М	S
	Ploumis, et al.[44]	Iliac crest height difference		~	Cor, Coef= 0.25, P<0.05		М	L
	Katz, et al.[11]	Lumbar pelvic relationship (LPR)		~	T-test, p=0.1(S) (>12°)		М	L
	Guo, et al.[7]	Pelvic incidence (PI)	×		T- tests, P=NS		М	L
S	Katz, et al.[11]	Pelvic tilt	×		T-test, P=0.88 (NS)		М	L
Pelvic Parameters	Guo, et al.[7]	pelvic tilt (PT)		*	T- tests, P=<0.01(<-0.5; PT: SN=64.30%, SP=83.30,	For sig ones, Log reg, B=- 8.78, P= <0.01, AUC=0.76; P=<0.01; Cut- off=<=0.5	М	L
	Guo, et al.[7]	sacral slope (SS)	×		T- tests, P=NS		М	L
	Ploumis, et al.[44]	Sacral takeoff angle		~	Cor, Coef= 0.23, P<0.05		М	L
	Guo, et al.[7]	Spinopelvic inclination (T1) Spinopelvic inclination (T9)		*	$\begin{array}{l} T- tests, P= <0.05$\\ (S), $PT1 (\le 3.5): $SN=64.30\%, $SP=71.40\%, $SP=71.40\%, T- tests, P= <0.05$\\ (S); $PT9 (\le 6.5^\circ): $SN=92.90\%, $SP=40.50\% \end{array}	For sig ones, Log reg, B: SPT1=-2.26, P<0.05, AUC=0.69, P<0.05, cut- off: ≤3.5 SPT9=-, -2.87; P<0.05, ,	М	L

						AUC=0.65;		
						P=<0.05, Cut- off:≤6.5		
t	Bunnell [4]	Spinal balance	×		Nonprognostic		М	S
Spinal Alignment	Katz, et al. [11]	Coronal decompensation, Lateral trunk shift	* *		T-test, P=0.45 T-test, P=0.42		М	L
Spinal	Yamauchi, et al. [29]	Difference of inclination between the proximal and distal vertebra	×		Cor Coef=-0.16, Partial Cor Coef=0.08		М	L
	Katz, et al.[11]	Apical vertebral rotation thoracic apex [AVR-T] Apical vertebral rotation lumbar apex [AVR-L]	*		T-test, P=0.23 T-test, P=0.56		М	L
kotation	Ohrt- Nissen, et al. [21]	Apical vertebra rotation (Nash and Moe grade 1 and 2)		~	Reg, B= 10.64 (2.47-18.81), 12.86 (4.69- 21.03); P= 0.012, 0.003 (high values)		L	L
tebral F	Peterson, et al.[22]	Apical vertebral rotation (Nash and Moe method)	×			Log reg, B= 1.15, SE=0.89, P= 0.19	М	S
Apical Vertebral Rotation	Vijvermans, et al.[26]	Apical vertebra rotation		~		Reg, B= 0.295, P=0.135, R ² = 3.43%, P=0.031 (higher rotation)	М	L
	Yamauchi, et al. [29]	Apical vertebra rotation		~	Cor Coef=-0.22, Partial Cor Coef=0.28 (higher rotation)		М	L
	Zhang, et al.[31]	Apical vertebral rotation		~	(>grade III, high risk) OR=16.34	Log reg, B= 2.78, SE=0.93, P=0.003	М	L
	Nault, et al.[18]	Apical intervertebral rotation		~	P=0.006 (S) (>2.4 °)		М	L
	Nault, et al.[18]	Upper curve intervertebral rotation	×		P=0.2 (NS)		М	L
	Nault, et al.[18]	Lower curve intervertebral rotation	×		0.3 (NS)		М	L
	Nault, et al.[18]	T12–L1 intervertebral rotation	×		0.6 (NS)		М	L
uo	Nault, et al. [19]	Axial rotation of upper junctional vertebra	×		Reg, P= 0.09,		М	L
d Rotati	Nault, et al. [19]	Sagittal rotation of upper junctional vertebra	×		Reg, P= 0.6		М	L
Axial Vertebral Rotation	Nault, et al. [19]	Coronal rotation of upper junctional vertebra	×		Reg, P =0.2		М	L
Axial 1	Nault, et al. [19]	Axial intervertebral rotation at apical level	×		Reg, P = 0.2		М	L

	Nault, et al. [19]	Sagittal intervertebral rotation at the apical level		v	Reg, P = 0.01 (S) (a difference in the sagittal plane of 1.3 degrees) (lower values, higher risk)		М	L
	Nault, et al. [19]	Coronal intervertebral Rotation at the apical level	×		Reg, P = 0.5		М	L
	Nault, et al. [19]	Axial rotation of lower junctional vertebra		~	Reg, P = 0.008 (S) (higher values of Lower junctional vertebra axial) vertebra coronal)		М	L
	Nault, et al. [19]	Sagittal rotation of lower junctional vertebra	×		Reg, P = 0.2		М	L
	Nault, et al. [19]	Coronal rotation of lower junctional vertebra		~	Reg, P = 0.03 (higher values of Lower junctional vertebra coronal, S		М	L
	Bohl, et al. [3]	Apex (T8 and cephalad Vs Caudal to T8)	x		Chi square, P= 0.163 (NS) SN=53%, SP=71%, ACC=62%, OR=2.7, +LR= 1.8, -LR= 0.67, +PV= 0.64, - PV=0.60; (T8 and cephalad considered positive)		М	L
	D'Amato, et al.[34]	Curve apex of T8: (higher vs lower)		~	Chi square, P=0.034 (higher curve location more progressive)		М	S
cation	Danielsson, et al. [5]	Apex at T9 or above	×		NS (0.20)		М	L
Apex Location	Davis, et al. [33]	Apex location (T6 -T9, T10-L3)	×		Chi square, P=0.081 (NS)		L	L
	Peterson, et al.[22]	Level of apex (T8- T11), (T12-L1)		~	Apical level cephalad to 12 th vertebra is significant	Log, re=2.73, SE=1.13, P=0.02, +PV=82%, - PV=80%, SP=81%, SN=81%	М	S
	Wu, et al.[28]	Apex Location		~	P= Apex locations in the progressed group were significantly higher (*1.2 vertebra levels) than those in the stable group		М	S
Torsion	Aulisa, et al.[1]	Torsion of the apical vertebra measured by Pedriolle degrees		~	Kruskal-Wallis= 16.9880, P< 0.01		М	L

		Torsion			T-test, P=0.02		
	Nault, et al. [18]	TOISION		~	(high torsion)	М	L
	Nault, et al.[19]	Torsion		~	T-test, P= 0.049 (S) (high torsion, higher risk)	М	L
Lumbosacr al Factors	Bunnell [4]	lumbosacral transitional anomalities	×		Non-prognostic	М	S
Vertebral Tilt	Katz, et al.[37]	Vertebral tilt angles (thoracic–superior [VTA-T-S], thoracic-inferior [VTA-T-I], lumbar–superior [VTA-L-S], lumbar–inferior [VTA-L-I])	* * * *		T-test, P= 0.33, T-test, P= 0.08, T-test, P= 0.82, T-test, P=0.85	М	L
a	Nault, et al. [18]	3D Cobb angle	×			М	L
3D Cobb angle	Nault, et al.[19]	3D Cobb angle	×		T-test, P= 0.08 (NS)	М	L
3D C	Nault, et al. [18]	Cobb angle of plane of maximum deformity		~	T-test, P= 0.001 (higher angle of plane of maximum curvature)	М	L
60 B	Nault, et al. [18]	3D apical vertebral body wedging	×		T-test, P=0.9 (NS)	М	L
Vertebral Wedging	Nault, et al.[19]	3D wedging apical vertebrae	×		T-test, P= 0.9 (NS)	М	L
	Nault, et al. [18]	T6 width slenderness	×		T-test, P=0.09 (NS)	М	L
	Nault, et al. [18]	T6 depth slenderness		~	T-test, P=0.05 (S) (low values higher risk)	М	L
	Nault, et al.[19]	Slenderness T6 vertebra		~	T-test, P= 0.05 (low values higher risk)	М	L
ss	Nault, et al. [18]	T12 width slenderness	×		T-test, P=0.1 (NS)	М	L
Slenderness	Nault, et al. [18]	T12 depth slenderness		~	T-test, P=0.03 (S) (low values higher risk)	М	L
S.	Nault, et al. [18]	L4 width slenderness		~	T-test, P= 0.009 (S) (low values higher risk)	М	L
	Nault, et al. [18]	L4 depth slenderness		~	T-test, P=0.007 (S) (low values higher risk)	М	L
	Nault, et al.[19]	Slenderness L4 vertebra	×		T-test, P= 0.5 (NS)	М	L
	Nault, et al. [18]	T1–L5 width slenderness		~	T-test, P=0.05 (S) (low values higher risk)	М	L

		T1 I5 donth		✓	T-test, P=0.005			
	Nault, et al. [18]	T1–L5 depth slenderness			(S) (low values higher risk)		М	L
	Nault, et al.[19]	Slenderness T1- L5 vertebra		~	T-test, P= 0.046 (low values higher risk)		М	L
	Nault, et al. [18]	Apical disk wedging	×		T-test, P=0.6 (NS)		М	L
a <i>c</i> i	Nault, et al. [18]	Upper curve disk wedging	×		T-test, P=0.2 (NS)		М	L
Disk Wedging	Nault, et al. [18]	Lower curve disk wedging	×		T-test, P=0.6 (NS)		М	L
Disk	Nault, et al.[19]	Mean 3D wedging of apical disks		~	T-test, P=0.03 (S) (high values)		М	L
	Yamauchi, et al. [29]	Disc index (wedging of the disk spaces in the curve)	×		Cor Coef=0.02, Partial Cor Coef= -0.04		М	L
slations	Katz, et al.[11]	Apical vertebral translation - thoracic [AVT-T] Apical vertebral translation-lumbar [AVT-L] AVT ratio [AVT- R] Relative apical distance [RAD]	× × ×		T-test, P= 0.56 T-test, P=0.39 T-test, P=0.31 T-test, P=0.85		М	L
Vertebral Translations	Wu, et al.[28]	Initial apex lateral deviations, Maximal apex lateral deviations		* *	T-test,P= S, Maximal apex lateral deviation,T-test, P=S		М	S
λ	Yamauchi, et al. [29]	Deviation of the apical vertebra, first thoracic vertebra	×	Ý	Cor Coef= Apical vertebra: -0.45 (S), first vertebra: 0.18 (NS); Partial Cor Coef= Apical vertebra: -0.25 (S), first vertebra: 0.03 (NS)	Apical vertebra: log reg, B= -0.258	М	L
Bone Density of Spine	Hung [8]	BMD (Z-score bone mineral density Spine (L2- L4)	×		T-test, P= 0.022, Chi, P=0.062,		L	S
Bone Dens	Sun, et al.[24]	BMD (lumbar spine from L2 to L4 (LSBMD) (g/cm2)		~	T test, P=0.018, LSBMD, SN= 74.50%, SP=64.70%; +PV= 2.11, - PV=1.54; Reg, P= 0.003 (lower values, higher risk)	OR= 11.24 (95% CI, 12.36- 53.43), P=0.002	L	S

	Hung [8]	BMD of femoral neck in convex- side hip		~	T-test, P=0.013, Chi, P <0.05 (lower values considered as risk		L	S
Bone Density of Hip	Hung [8]	BMD of femoral neck in concave- side hip		~	factors) T-test, P=0.014, Chi, P <0.05 (lower values considered as risk factors)	Femoral neck in concave-side hip put into the model, Log reg, B= 0.81; OR= 2.3 (95% CI,1.1- 4.5), P=0.02; AUC=0.8 (95% CI%, 0.75-0.85); Model: SN= 76% (95% CI, 69%-83%), SP= 70% (95% CI, 62%-77%); (lower values considered as risk factors)	L	S
	Hung [8]	BMD of Trochanter in convex-side hip	×		T-test, P=0.346, Chi, P=0.177		L	S
	Hung [8]	BMD of Trochanter in concave-side hip		~	T-test, P=0.261, P<0.05 (low values, higher risk)		L	S
	LAM, et al. [21]	Bone Density (BMD) both hips		√*	Chi square P= 0.002 (lower values as high risk) (level of sig P<0.2)		М	L
	Sun, et al.[24]	BMD of femoral neck on the nondominant side (FNBMD) (g/cm2)	×		P=0.129 (association test NS, test not specified)		L	S
	LAM, et al. [21]	Bone quality (broadband ultrasound attenuation (BUA)		√*	Chi square P=0.057 (lower values had higher risk) (level of sig P<0.2)		М	L
Bone Quality	LAM, et al. [21]	velocity of sound (VoS)		√*	Chi square P= 0.073 (S) (lower values had higher risk) (level of sig P<0.2)		М	L
Bone	LAM, et al. [21]	Stiffness index (SI)		~	Chi square P= 0.004 (S) (lower values had higher risk) (level of sig P<0.2); SN=85%, SP=66%; AOR= 2 (95% CI, 1.08- 3.17)	Log reg, B= 0.694, AUC= 0.83 (95%CI, 0.78-0.87)	М	L
Bone Proporti on	LeBlanc [16]	Bone volume (%) of whole body		~	Discriminant analysis, P< 0.0001, ACC=84%		М	S

	Bunnell [4]	Family history	×	Non-prognostic		М	S
	Hung [8]	Family history of scoliosis	×	Chi, P= 0.750		L	S
	Konieczny, et al.[12]	Family history of scoliosis	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
Other Factors	Konieczny, et al.[12]	Hours of sports per week outside of the school (< 1 hour, ≥ 1)	×	Hours of sport/w: Man-Whitney U test, P> 0.05	Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
	Konieczny, et al.[12]	Hours of self- training per week	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
Other	Konieczny, et al.[12]	Educational level of parents	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
	Konieczny, et al.[12]	Nutritional behavior	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
	Konieczny, et al.[12]	Type of school bag	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S
	Konieczny, et al.[12]	Educational level of the patients	×		Binary logistic regression analysis (stepwise, backward), P=NS	Н	S

abbreviations used in table were defined as follows: Non-pre: Non-predictor, Pre: Predictor, Pro: Progression, Non-pro: Nonprogression, Cor Coef: Correlation coefficient, P: P value, NS: Not significant, NM: Not mentioned, CI: Confidence interval, Log reg: Logistic regression, B: Beta coefficient, SE: Standard error, AUC: Area under curve, OR: Odds ratio, AOR: Adjusted odds ratio, SN: Sensitivity, SP: Specificity, -PV: negative predictive value, +PV: Positive predictive value, -LR: Negative likelihood ratio, +LR: Positive likelihood ratio, ACC: Accuracy, AB: adjusted B coefficient, R²P: quantifies the weight of importance of every variable in the model; the * denote that the variable was significantly associated with progression in univariate analysis but the p value of the model including that variable was not significant, cm:centimeter, TH: thoracic, TL: thoracolumbar, L: lumbar, kg=kilogram

Appendix 3.4 References

- 1. Aulisa, A.G., et al., Treatment of thoraco-lumbar curves in adolescent females affected by idiopathic scoliosis with a progressive action short brace (PASB): assessment of results according to the SRS committee on bracing and nonoperative management standardization criteria. Scoliosis, 2009. 4(101271527): p. 21.
- 2. Aulisa, A.G., et al., Correlation between compliance and brace treatment in juvenile and adolescent idiopathic scoliosis: SOSORT 2014 award winner. Scoliosis, 2014. 9(1): p. 6.

- 3. Bohl, D.D., et al., *Effectiveness of Providence nighttime bracing in patients with adolescent idiopathic scoliosis.* Orthopedics, 2014. **37**(12): p. e1085-90.
- 4. Bunnell, W.P., The natural history of idiopathic scoliosis. Clinical orthopaedics and related research, 1988(229): p. 20-5.
- 5. Danielsson, A.J., et al., A prospective study of brace treatment versus observation alone in adolescent idiopathic scoliosis: a follow-up mean of 16 years after maturity. Spine, 2007. **32**(20): p. 2198-207.
- 6. Guo, J., et al., A prospective randomized controlled study on the treatment outcome of SpineCor brace versus rigid brace for adolescent idiopathic scoliosis with follow-up according to the SRS standardized criteria. European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 2014. 23(12): p. 2650-7.
- Guo, J., et al., Pelvic tilt and trunk inclination: new predictive factors in curve progression during the Milwaukee bracing for adolescent idiopathic scoliosis. European Spine Journal, 2012. 21(10): p. 2050-2058.
- Hung, V.W.Y., et al., Osteopenia: A new prognostic factor of curve progression inadolescent idiopathic scoliosis. Journal of Bone and Joint Surgery-American Volume, 2005. 87A(12): p. 2709-2716.
- 9. Karol, L.A., Effectiveness of bracing in male patients with idiopathic scoliosis. Spine, 2001. 26(18): p. 2001-5.
- 10. Karol, L.A., et al., *Progression of the curve in boys who have idiopathic scoliosis.* The Journal of bone and joint surgery. American volume, 1993. **75**(12): p. 1804-10.
- 11. Katz, D.E. and A.A. Durrani, Factors that influence outcome in bracing large curves in patients with adolescent idiopathic scoliosis. Spine, 2001. 26(21): p. 2354-2361.
- Konieczny, M.R., P. Hieronymus, and R. Krauspe, *Time in brace: where are the limits and how can we improve compliance and reduce negative psychosocial impact in patients with scoliosis? A retrospective analysis.* Spine Journal, 2017. 17(11): p. 1658-1664.
- 13. Kuroki, H., et al., Predictive factors of Osaka Medical College (OMC) brace treatment in patients with adolescent idiopathic scoliosis. Scoliosis, 2015. 10(1).
- Lam, T.P., et al., QUANTITATIVE ULTRASOUND FOR PREDICTING CURVE PROGRESSION IN ADOLESCENT IDIOPATHIC SCOLIOSIS: A PROSPECTIVE COHORT STUDY OF 294 CASES FOLLOWED-UP BEYOND SKELETAL MATURITY. Ultrasound in Medicine and Biology, 2013. 39(3): p. 381-387.
- Lara, T., et al., The Risk of Curve Progression and Surgery in African Americans With Adolescent Idiopathic Scoliosis. Spine deformity, 2017. 5(4): p. 250-254.
- 16. LeBlanc, R., et al., Morphologic discrimination among healthy subjects and patients with progressive and nonprogressive adolescent idiopathic scoliosis. Spine, 1998. 23(10): p. 1109-15; discussion 1115-6.
- 17. Mao, S., et al., *Initial Cobb angle reduction velocity following bracing as a new predictor for curve progression in adolescent idiopathic scoliosis.* European Spine Journal, 2016. **25**(2): p. 500-505.
- 18. Nault, M.-L., et al., *Three-Dimensional Spinal Morphology Can Differentiate Between Progressive and Nonprogressive Patients With Adolescent Idiopathic Scoliosis at the Initial Presentation*. Spine, 2014. **39**(10): p. E601-E606.
- 19. Nault, M.-L., et al., *Three-dimensional Spine Parameters Can Differentiate Between Progressive and Nonprogressive Patients With AIS at the Initial Visit: A Retrospective Analysis.* Journal of Pediatric Orthopaedics, 2013. **33**(6): p. 618-623.
- O'Neill, P.J., et al., Decreased orthotic effectiveness in overweight patients with adolescent idiopathic scoliosis. Journal of Bone and Joint Surgery-American Volume, 2005. 87A(5): p. 1069-1074.
- 21. Ohrt-Nissen, S., et al., *Flexibility predicts curve progression in providence nighttime bracing of patients with adolescent idiopathic scoliosis.* Spine, 2016. **41**(22): p. 1724-1730.
- 22. Peterson, L.E. and A.L. Nachemson, *Prediction of progression of the curve in girls who have adolescent idiopathic scoliosis of moderate severity. Logistic regression analysis based on data from The Brace Study of the Scoliosis Research Society.* Journal of bone and joint surgery. American volume, 1995. **77**(6): p. 823-827.
- 23. Shi, B., et al., *Curve Progression in Adolescent Idiopathic Scoliosis With a Minimum of 2 Years' Follow-up After Completed Brace Weaning With Reference to the SRS Standardized Criteria.* Spine deformity, 2016. **4**(3): p. 200-205.
- 24. Sun, X., et al., Osteopenia Predicts Curve Progression of Adolescent Idiopathic Scoliosis in Girls Treated With Brace Treatment. Journal of Pediatric Orthopaedics, 2013. 33(4): p. 366-371.
- Sun, X., et al., Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis. Orthopaedic surgery, 2010. 2(4): p. 285-90.
- 26. Vijvermans, V., G. Fabry, and J. Nijs, *Factors determining the final outcome of treatment of idiopathic scoliosis with the Boston brace: a longitudinal study.* Journal of pediatric orthopedics. Part B, 2004. **13**(3): p. 143-9.

- 27. Weiss, H.R., *The progression of idiopathic scoliosis under the influence of a physiotherapy rehabilitation programme*. Physiotherapy, 1992. **78**(11): p. 815-821.
- 28. Wu, H., et al., *Time series spinal radiographs as prognostic factors for scoliosis and progression of spinal deformities*. European Spine Journal, 2011. **20**(1): p. 112-117.
- Yamauchi, Y., T. Yamaguchi, and Y. Asaka, Prediction of curve progression in idiopathic scoliosis based on initial roentgenograms. A proposal of an equation. Spine, 1988. 13(11): p. 1258-1261.
- 30. Yrjonen, T., et al., Results of brace treatment of adolescent idiopathic scoliosis in boys compared with girls: a retrospective study of 102 patients treated with the Boston brace. European Spine Journal, 2007. 16(3): p. 393-397.
- Zhang, Y., et al., Factors relating to curve progression in female patients with adolescent idiopathic scoliosis treated with a brace. European Spine Journal, 2015. 24(2): p. 244-248.
- 32. Zhu, Z., et al., Is Brace Treatment Appropriate for Adolescent Idiopathic Scoliosis Patients Refusing Surgery With Cobb Angle Between 40 and 50 Degrees. Clinical Spine Surgery, 2017. **30**(2): p. 85-89.
- 33. Davis, L., et al., *Nighttime bracing with the Providence thoracolumbosacral orthosis for treatment of adolescent idiopathic scoliosis: A retrospective consecutive clinical series.* Prosthetics and orthotics international, 2018(ozz, 7707720): p. 309364618792727.
- D'Amato, C.R., S. Griggs, and B. McCoy, Nighttime bracing with the Providence brace in adolescent girls with idiopathic scoliosis. Spine, 2001. 26(18): p. 2006-12.
- 35. Gepstein, R., et al., *Effectiveness of the Charleston bending brace in the treatment of single-curve idiopathic scoliosis.* Journal of Pediatric Orthopaedics, 2002. **22**(1): p. 84-87.
- Karol, L.A., et al., *The Effect of the Risser Stage on Bracing Outcome in Adolescent Idiopathic Scoliosis*. Journal of Bone and Joint Surgery-American Volume, 2016. 98(15): p. 1253-1259.
- Katz, D.E., et al., Brace wear control of curve progression in adolescent idiopathic scoliosis. The Journal of bone and joint surgery. American volume, 2010. 92(6): p. 1343-52.
- Pasquini, G., et al., The outcome of a modified version of the Cheneau brace in adolescent idiopathic scoliosis (AIS) based on SRS and SOSORT criteria: a retrospective study. European Journal of Physical and Rehabilitation Medicine, 2016. 52(5): p. 618-626.
- Thompson, R.M., et al., Brace Success Is Related to Curve Type in Patients with Adolescent Idiopathic Scoliosis. The Journal of bone and joint surgery. American volume, 2017. 99(11): p. 923-928.
- 40. Ryan, P.M., et al., *Role of the triradiate cartilage in predicting curve progression in adolescent idiopathic scoliosis.* Journal of Pediatric Orthopaedics, 2007. **27**(6): p. 671-676.
- 41. Goodbody, C.M., et al., *It's not just the big kids: both high and low BMI impact bracing success for adolescent idiopathic scoliosis.* Journal of Childrens Orthopaedics, 2016. **10**(5): p. 395-404.
- 42. Zaina, F., S. Donzelli, and S. Negrini, *Overweight is not predictive of bracing failure in adolescent idiopathic scoliosis:* results from a retrospective cohort study. European Spine Journal, 2017. **26**(6): p. 1670-1675.
- 43. Cheung, J., et al., *The relation between electromyography and growth velocity of the spine in the evaluation of curve progression in idiopathic scoliosis.* Spine, 2004. **29**(9): p. 1011-6.
- 44. Ploumis, A., et al., *Progression of idiopathic thoracic or thoracolumbar scoliosis and pelvic obliquity in adolescent patients with and without limb length discrepancy*. Scoliosis and spinal disorders, 2018. **13**(101675716): p. 18.
- 45. Trivedi, J.M. and J.D. Thomson, *Results of Charleston bracing in skeletally immature patients with idiopathic scoliosis.* Journal of pediatric orthopedics, 2001. **21**(3): p. 277-80.
- 46. Fang, M.Q., et al., Long-term effects of the Cheneau brace on coronal and sagittal alignment in adolescent idiopathic scoliosis. Journal of neurosurgery: spine, 2015. 23(4): p. 505-509.

Appendix 3.5 Summary of Articles for the Progression Criteria to a Magnitude $\geq 30^{\circ}$ or Requiring Surgery

Predictors	Author	Parameters	Non- Pre	Pre	Univariate analysis	Multivariate analysis	RoB	FU intervals
	Goldberg, et al. [1]	Age at initial (13.9±1.7)	×		T-test=1.37, P=0.11		М	L
	Karol [2]	Age (13.9, 10.6– 16.9)		~	Fisher's exact test (two-tailed), P< 0.05, (Age, < 13 yr is high risk for surgical thresholds)		М	L
	Lara, et al. [3]	Age at initial (14.2±1.9, 10-18.8		~	Younger age at presentation showed higher risk of progression	OR=0.44 (95%CI, 0.27- 0.72), P=0.0012	L	L
Age (years)	Lee, et al. [4]	Age at the initial		~	Unadjusted HR= 0.91 (95% CI, 0.84- 0.97), HR= 0.008, (11.3 years as cutoff, risk of progression decreases as age increases)	Adjusted HR = 0.4 (95% CI, 0.26-0.6), P< 0.001	М	S
Ag	Shi, et al. [5]	Age at the initial (\leq 45: 12.1 \pm 1.2 vs > 45: 12.5 \pm 1.3)	×		Chi square, P= 0.678		М	L
	Sun, et al. [6]	Age at initial (10.0– 12.9 vs 13.0–15.9), Age at menarche	×	~	Age at initial (younger age: 10.0– 12.9 yr at higher risk of progression): Chi square, P <0.05, Age at menarche: NS		М	L
	Tan, et al. [7]	Age at initial	×		Chi, square P= 0.158		М	L
	Yip, et al. [8]	Age at initial		~	HR=0.53 (95%CI, 0.41-0.96), P=0.00 (risk of progression decreased by age)	HR=0.53 (95%CI, 0.37- 0.75), P=0.00	М	S
	Davis, et al. [9]	Gender	×		Chi, P= 0.5		L	L
	Lara, et al. [3]	Gender	×		Chi, P=0.55	OR= 2.60 (95%CI, 0.56, 11.95), P=0.22	L	L
der	Lee, et al. [4]	Gender		~	Unadjusted HR= 0.81 (95%CI, 0.64- 1.03), P= 0.805 for male (females showed higher risk of progression)	Adjusted HR=0.02 (95%CI, 0- 0.24), P=0.002	М	S
Gender	Sitoula, et al. [10]	Gender		~	P=0.029, SN=27%, SP=86%, ACC=65%, OR=2.42, +LR= 2.02, -LR= 0.83, +PV= 0.53, - PV=0.68 (females showed a higher rate for progression)		L	L
	Tan, et al. [7]	Gender (females at higher risk)	×		Chi, P= 0.07, SN=5%, SP= 82%, ACC= 67%, OR= 3.8	Log reg, B= 0.931,	М	L

		1	1		10 F0 / PT	1	1	
					(95%CI, 0.87- 16.9),P=0.26, +LR=0.29, -LR= 1.14, +PV= 7.14%, - PV= 77.22% (Male was considered as positive test)			
	Cheung, et al. [11]	Menarche		~	Chi, P<0.001, (higher risk for premenarche)		М	L
	Goldberg, et al. [1]	Menarche status		~	Z score= 3.672, P< 0.01 (pre-menarche at diagnosis)		М	L
Status	Lee, et al. [4]	Initial menarche status		v	Menarche status classified as an intermediate risk group for progression (pre-menarche at higher risk)		М	S
ertal	Shi, et al. [5]	Menarche age	×		Chi, P=0.85		М	L
Menarche/Pubertal Status	Sun, et al. [12]	Menarche status		~	Fisher's exact test, P< 0.001, (pre- menarche Status at higher risk)	Log reg, OR= 12.00 (95%CI, 3.73-47.25) (pre-menarche)	L	S
Me	Tan, et al. [7]	Initial pubertal status	×		Chi square, P= 0.48	Log, reg, P=0.129, OR=2.3 (95%CI, 0.8– 6.6) (after puberty positive)	М	L
	Yip, et al. [8]	Initial menarche status		~	Adjusted HR=0.21(95%CI, 0.11-0.38), P <0.001 (pre-menarche at higher risk)	Unadjusted HR= 0.41 (95%CI, 0.19- 0.86), P=0.02	М	S
	Cheung, et al. [11]	Risser 0-3 for both		~	Chi square, P<0.001 (Risser 0-3 have a significant risk for progression to surgical threshold))		М	L
	Davis, et al.	Risser classification $(0, \ge 1)$	×		Chi square, P= 0.13		L	L
-	Karol [13]	Risser sign (0, 1, and 2)		~	Progression to surgery was related to less Mature Risser sign		М	L
Risser Sign	Karol [2]	Risser 0 vs 1, 2; Risser 1 vs 2; Risser 0 open TRC vs Closed TRC		~	Chi square, P< 0.001, 0.51, 0.005; Risser status (0) at the high risk for surgery		М	S
	Lara, et al. [3]	Risser (0-5)+TRC (open/close)	×		Chi, P=0.1093		L	L
	Sitoula, et al. [10]	Risser (0, 1, 2, 3)		~	Chi square, P=0.005 (lower Risser grade have higher rate of progression)		L	L
	Sun, et al. [6]	Risser (0-1 vs 2-4) (surgical threshold)		~	Fisher's exact test, P< 0.05, Chi square, P=0.002, (Lower Risser grade 0–1 at		М	L
r	r			1	1 . 1 . 1 . 2		1	
---------------------------------------	--------------------------	--	---	---	--	--------------------------------	---	---
					higher risk of progression to			
	Thompson, et al. [14]	Risser 0 (MT, L); Risser 1, 2 (MT, L)	×		surgery Chi, P>0.05		М	U
Triradiate Cartilage (TRC)	Karol, et al. [2]	TRC+ Risser 0 (open, close)		~	Chi square, P=< 0.005 (open triradiate cartilage higher risk)		М	S
Distal Radius and Ulna Class (DRU)	Cheung, et al. [11]	Radius grade (5-10) and Ulna grade (4- 8) for (≥40, ≥50 thresholds)		~	 ≥40: Chi square, P< 0.001 and Chi square, P< 0.001; ≥50: Chi, P=0.002, Chi, P < 0.001; ACC= ≥40: 74.5% and 71.5%, ≥50: 85.6% and 86%; (R6/U5 showed a higher risk of progression) 		М	L
Olecranon Stages of Maturity	Charles, et al. [15]	Olecranon process maturation for primary curve magnitude of 21– 30° and >30°		v	Chi square, P= 0.003 (higher risk during the accelerating pubertal growth phase for curves 21- 30), P< 0.001 (higher risk at the onset of the accelerating growth phase for >30 curves)		М	S
Sanders Stages of Maturity	Sitoula, et al. [10]	Sanders (1-7)		~	Chi square, P=0.0001, patients with SS1, SS2, and SS3 with curves > 20 ° or 25 at greatest risk for curve progression into the surgical range		L	L
Body Mass Index (BMI) (kg/m2)	Goodbody, et al. [16]	Low-BMI group (20th percentile), Mid-BMI group (20th–85th percentile) High-BMI group (85th percentile)		~	Low BMI: For progression > 45: P < 0.01, OR=3.7, Need for surgery: P<0.01, OR=4.1, and for Orthotic failure P < 0.01, OR=3.7; Mid- BMI: NM; High BMI: For progression > 45: P=0.01, OR=3.4, Need for surgery=0.13, OR=2.2, and Orthotic failure=0.04, OR=2.4 (high and low BMI at higher risk of progression)		М	L
Weight (kg)	O'Neill, et al. [17]	Weight (Being overweight was a risk factor,≥ 85 th percentile)		~	T-test, P< 0.05; Pearson's product moment= 0.20	B=0.15, R ² =18%	L	L

				1			1	
		Hip joint			Right: Mann-			
ters		asymmetries			Whitney test, P> 0.05			
net	Kotwicki,	derived parameters						
Hip Parameters	et al. [18]	# (T 1 1	×				Н	U
Pa		(Including						
lip		parameters for total						
Ξ		range of rotation right and left hip)						
		Height (≥153.9 vs <			Unadjusted HR=	Adjusted HR=		
		153.9)			0.98 (95%CI, 0.97-	0.98 (95%CI,		
Ξ		155.7)			0.99), HR,	0.97-0.99),		
(c	Lee, et al.				(P<0.001), Risk of	P=0.002		~
ght	[4]			~	progression decreases	1 0.002	М	S
Height (cm)					by increasing height			
-								
		Primary curve		✓	Chi square, P=			
		magnitude 21-30°		,	0.0034			
		and $>30^{\circ}$		✓				
	Charles, et				Chi square, P=		М	S
	al. [15]				< 0.001			
					(progression > 30 has			
					higher risk versus \leq 30)			
	Cheung, et	Initial Cobb			Chi square, P < 0.001,			
	al. [11]	Initial Cooo			$(\geq 35 \text{ showed a high})$			_
				~	risk)		М	L
					,			
	Davis, et	Initial Cobb angle	×		Chi, P<=0.091		L	L
	al. [9]	$(< 35 \text{ vs} > 35^{\circ})$	~				L	L
		Initial Cobb angle			Surgical: T- test=			
	G 1 1				9.16, P< 0.001,			
	Goldberg,			✓	Unoperated: T test=		М	L
	et al. [1]				11.66, P< 0.001 (higher Cobb at			
					presentation)			
		Curve magnitude			Fisher's exact test			
`		Curve mugintude			(two-tailed), P<			
le ((Karol [13]			\checkmark	0.0001 (initial Cobb		М	L
ngl					\geq 30 higher risk to			
V C					progress to surgery)			
Cobb Angle (°)	Karol [19]	Curve magnitude		✓	Chi, P<0.0001		М	S
C D		Initial Cobb angle,			Larger Curve	OR=2.6	7	
	Lara, et al.	32.5 ±18.9 (10-104)		✓	magnitude at	(95%CI, 0.56-	L	L
	[3]				presentation was a	11.95),		_
		Califa and			significant risk factor	P<0.0001		
		Cobb angle			Unadjusted HR	Adjusted HR=		
	Lee, et al.			√*	=0.69 (95%CI, 1.16- 1.21), P<0.001	0.84 (95%CI, 0.69-1.03),	М	S
	[4]				(Initial Cobb angle >	P=0.095	171	5
					25 and higher risk)			
		Initial curve			Two-way Chi,			
	Pasquini,	magnitude 20-30 vs		~	P=0.022 (30-40 °		М	U
	et al. [20]	30-40			higher risk of		1/1	U
					progression)			
		Initial Cobb angles,		✓	Chi, P< 0.001 (initial			
		Cobb angles at		~	Cobb)			
	G1 · · 1	brace weaning (≤45		×	C1: D < 0.001 ()			
	Shi, et al.	vs > 45 groups)			Chi, P< 0.001 (at brace weaning)		М	т
	[5]				High values of Cobb		IVI	L
					associate with higher			
					risk of progression			
					after brace weaning)			
L	1	1	1			1	1	

r	Γ	Luidial Calif			Luidial Calib	Γ		
	Sitoula, et	Initial Cobb			Initial Cobb, P=0.0001 (Cobb \geq 25			
	al. [10]			~	had higher risk of		L	L
					progression)			
	Sun, et al.	Cobb before bracing			Chi, P= 0.002 (>30°	Cobb >30:		
	[6]	$(>30^{\circ} vs \le 30)$		✓	was an independent	6.484 (95%CI,	М	L
	[0]				risk factor for curve	1.302–32.289),	101	Ľ
					progression)	P=0.022		
		Initial Cobb angle < 25 vs ≥25			Chi square, P< 0.001, SN= 68%, SP=92%,	P< 0.001, OR=27.5		
		25 VS <u>~</u> 25			ACC=87%, OR=24.6	(95%CI, 10.2-		
					(95%CI, 9.9-60.06),	73.9)		
	Tan, et al.			~	+LR=8.43, -	, ,		r
	[7]			v	LR=0.34, +PV=		М	L
					68.42%, -PV=			
					91.89%; (Cobb ≥25			
					has higher risk of progression)			
		Initial Cobb angle			HR=1.11 (95%CI,	HR=1.15		
	Yip, et al.	lintial Coob aligic		,	1.06-1.17), P< 0.001	(95%CI, 1.1-		
	[8]			✓	(high risk group had	1.2), P=0.000	М	S
					a Cobb ≥24)	~~		
		Modified Lenke			≥40: Chi square,			
		curve type (1-6)			P=0.006 (S) (TH and			
	Cheung, et			~	doubles curve at			т
	al. [11]			v	higher risk of		М	L
					progression) ≥50: Chi square,			
					P=0.012 (NS)			
	D	Curve classification			Chi square, P=0.07			
	Davis, et	(TH, TH/L, Double	×		I , ····		L	L
	al. [9]	major)						
		Curve pattern (TH,			Curve pattern: Fisher	TH: OR= 0.29		
		L, TL, Double TH,			exact test, P=0.0014	(95%CI, 0.03-		
		Double L (reference))				2.79), P= 0.28; L: OR=		
		(Telefence))				0.13(95%CI,		
						<0.001-> 999),		
	Lana at al					P= 0.83; TL:		
	Lara, et al. [3]			√*		OR= 0.03	L	L
E	[3]					(95%CI, <		
tte						0.001-45.25),		
Pa						P= 0.34; Double TH		
Curve Pattern						OR=1.51		
Сп						(95%CI,0.18-		
						12.32), P=0.70		
	Pasquini,	Curve type (TH,			Chi square, P=0.312			
	et al. [20]	TL, L, double	×				М	U
	- []	major)						
	Shi, et al.	Curve patterns (MT,	×		Chi square, P=1.00,		м	т
	[3]	TL, Double)	Â		0.78, 0.21		М	L
		Curve pattern (TH,			Chi square, P=0.278	Log Reg,		
		Double TL/L)			(TH higher risk)	B=1.256,		
	Sun, et al.	,		~		OR=2.671	М	т
	[6]			`		(95%CI, 1.064-	1V1	L
						11.591, P=		
						0.039		
		Modified Lenke	×		mLenke (I-VI) : Chi, P= 0.0866,			
	Thompson,	(mLenke) classification		~	1 - 0.0000,		М	U
	et al. [14]	system (I-VI),		.			141	U
				1	1			
		Curve type Main						

		TH (mLenke I-III) vs main L (mLenke V or VI) Apex location (T6 - T9, T10-L3)			Curve type TH vs L : Chi, P=0.0129 (main TH (mLenke I, II, or III) at higher risk) Chi square, P=0.004, (apex cephalad to	T10-L3: OR= 0.21 (95%CI,		
Apex Location	Davis, et al. [9]	17, 110-L3)		~	(apex cephalad to T10 at higher risk)	0.21 (95 %C1, 0.06-0.75), P=0.016	L	L
	Yip, et al. [8]	Osteopenia status (aBMD from DXA) of proximal of bilateral femurs		~	Unadjusted value HR for osteopenia= 2.068 (95%CI, 1.202- 3.561), P= 0.009; (Osteopenia high risk-(zBMD < - 1)	Adjusted value= HR= 2.245 (95%CI, 1.201- 4.198), P= 0.011; %	М	S
Ŷ	Yip, et al. [8]	Bone Morphometry Tot Area (cm2), Tb.Ar (cm ²), CrAr/TotAr (%), CtPm, CtTh); at non- dominant distal radius	× × ×	√* √*	T-test: P=0.555 T-test: P=0.920 T-test: P=0.023 T-test: P=0.874 T-test: P=0.015		М	S
Bone Density	Yip, et al. [8]	Volumetric BMD (SD) (Dtot(mgHA/cm3), Dcort (mgHA/cm3), Dtrab (mgHA/cm3)); at non-dominant distal radius	× ×	4	T test, P=0.63, P=0.002, (Dcort < 570 mgHA/cm3 has high risk group), P=0.505);	$\begin{array}{l} (Dcort < 570 \\ mgHA/cm3 \\ high risk group) \\ combined with \\ Cobb angle \geq \\ 24^{\circ} \ lead to \\ SN=43\%, \\ SP=100\%, \\ PPV=100\%, \\ NPV=95 \end{array}$	М	S
	Yip, et al. [8]	Trabecular Bone Micro-architecture (SD) TrN, BV/TV (%), Tb.Th, Tb.Sp); at non- dominant distal radius	× × ×	√*	T test, P=0.178 T test, P=0.503 T test, P= 0.021 T test, P=0.415		М	S

Note that the abbreviations used in table described as: Non-pre: Non-predictor, Pre: Predictor, Progression, Non-pro: Non-progression, HR: Hazard ratio, P: P value, NS: Not significant, NM: Not mentioned, CI: Confidence interval, Log reg: Logistic regression, B: Beta coefficient, SE: Standard error, AUC: Area under curve, OR: Odds ratio, AOR: Adjusted odds ration, SN: Sensitivity, SP: Specificity, -PV: negative predictive value, +PV: Positive predictive value, -LR: Negative likelihood ratio, +LR: Positive likelihood ratio, ACC: Accuracy, Kg= kilogram, cm=centimeter, kg/m² =kilogram/square meter, aBMD= areal bone mineral density, DXA= dual-energy X-ray absorptiometry, BV/TV = bone volume over total volume, CrAr = cortical bone area, CtPm = cortical (periosteal) perimeter, CtTh = cortical thickness, Dcort = volumetric density of cortical bone measured at distal radius, Dtrab = volumetric density of trabecular, bone measured at distal radius, Tb.Ar = trabecular, bone area, Tb.Sp = trabecular spacing, Tb.Th = trabecular thickness, Tot Area = total bone area, TrN = trabecular number

* denotes that predictor is significant from univariate analysis but not from multivariate analysis

parameters included in this study: Internal rotation (R, L hip), External rotation (R, L hip), Total range rotation (R, L hip), Difference in IR: Right minus left, Difference in ER: Right minus left, Absolute difference left-right in IR, Absolute difference left-right in ER, Mid-point of rotation for right hip, Mid-point of rotation for left hip, Static rotational offset of the pelvis

Appendix 3.5 References

1. Goldberg, C.J., et al., *Adolescent idiopathic scoliosis: the effect of brace treatment on the incidence of surgery*. Spine, 2001. **26**(1): p. 42-7.

2. Karol, L.A., et al., *The Effect of the Risser Stage on Bracing Outcome in Adolescent Idiopathic Scoliosis.* Journal of Bone and Joint Surgery-American Volume, 2016. **98**(15): p. 1253-1259.

- Lara, T., et al., The Risk of Curve Progression and Surgery in African Americans With Adolescent Idiopathic Scoliosis. Spine deformity, 2017. 5(4): p. 250-254.
- 4. Lee, C.F., et al., *A new risk classification rule for curve progression in adolescent idiopathic scoliosis*. The spine journal : official journal of the North American Spine Society, 2012. **12**(11): p. 989-95.
- Shi, B., et al., Curve Progression in Adolescent Idiopathic Scoliosis With a Minimum of 2 Years' Follow-up After Completed Brace Weaning With Reference to the SRS Standardized Criteria. Spine deformity, 2016. 4(3): p. 200-205.
- 6. Sun, X., et al., Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis. Orthopaedic surgery, 2010. 2(4): p. 285-90.
- 7. Tan, K.-J., et al., Curve Progression in Idiopathic Scoliosis Follow-up Study to Skeletal Maturity. Spine, 2009. 34(7): p. 697-700.
- Yip, B.H., et al., Prognostic Value of Bone Mineral Density on Curve Progression: A Longitudinal Cohort Study of 513 Girls with Adolescent Idiopathic Scoliosis. Scientific reports, 2016. 6((Yip, Yu, Wang, Hung, Lam, Ng, Zhu, Cheng) Department of Orthopaedics and Traumatology, Faculty of Medicine, Prince of Wales Hospital, The Chinese University of Hong Kong, Hong Kong): p. 39220.
- 9. Davis, L., et al., Nighttime bracing with the Providence thoracolumbosacral orthosis for treatment of adolescent idiopathic scoliosis: A retrospective consecutive clinical series. Prosthetics and orthotics international, 2018(ozz, 7707720): p. 309364618792727.
- 10. Sitoula, P., et al., Prediction of Curve Progression in Idiopathic Scoliosis: Validation of the Sanders Skeletal Maturity Staging System. Spine, 2015. 40(13): p. 1006-13.
- 11. Cheung, J.P.Y., et al., *APSS-ASJ Best Clinical Research Award: Predictability of Curve Progression in Adolescent Idiopathic Scoliosis Using the Distal Radius and Ulna Classification.* Asian spine journal, 2018. **12**(2): p. 202-213.
- 12. Sun, X., et al., Osteopenia Predicts Curve Progression of Adolescent Idiopathic Scoliosis in Girls Treated With Brace Treatment. Journal of Pediatric Orthopaedics, 2013. 33(4): p. 366-371.
- 13. Karol, L.A., Effectiveness of bracing in male patients with idiopathic scoliosis. Spine, 2001. 26(18): p. 2001-5.
- Thompson, R.M., et al., Brace Success Is Related to Curve Type in Patients with Adolescent Idiopathic Scoliosis. The Journal of bone and joint surgery. American volume, 2017. 99(11): p. 923-928.
- Charles, Y.P., F. Canavese, and A. Dimeglio, Curve progression risk in a mixed series of braced and nonbraced patients with idiopathic scoliosis related to skeletal maturity assessment on the olecranon. Journal of Pediatric Orthopaedics-Part B, 2017. 26(3): p. 240-244.
- 16. Goodbody, C.M., et al., *It's not just the big kids: both high and low BMI impact bracing success for adolescent idiopathic scoliosis.* Journal of Childrens Orthopaedics, 2016. **10**(5): p. 395-404.
- 17. O'Neill, P.J., et al., *Decreased orthotic effectiveness in overweight patients with adolescent idiopathic scoliosis.* Journal of Bone and Joint Surgery-American Volume, 2005. **87A**(5): p. 1069-1074.
- 18. Kotwicki, T., A. Walczak, and A. Szulc, *Trunk rotation and hip joint range of rotation in adolescent girls with idiopathic scoliosis: does the "dinner plate" turn asymmetrically* ? Scoliosis, 2008. **3**(101271527): p. 1.
- 19. Karol, L.A., et al., *Progression of the curve in boys who have idiopathic scoliosis.* The Journal of bone and joint surgery. American volume, 1993. **75**(12): p. 1804-10.
- Pasquini, G., et al., The outcome of a modified version of the Cheneau brace in adolescent idiopathic scoliosis (AIS) based on SRS and SOSORT criteria: a retrospective study. European Journal of Physical and Rehabilitation Medicine, 2016. 52(5): p. 618-626

Predictors	Authors	Parameters	Non- pre	Pre	Univariate analysis	ROB	FU intervals
rs)	Ylikoski [1]	Initial age (11-12 vs 13-16)		~	Student's t T-test, P< 0.01 (higher risk for 11-12)	М	S
Age (years)	Yrjonen and Ylikoski [2]	Initial age		~	For age 12-13 had the highest risk of progression	М	S
Menarche Status	Ylikoski [1]	Menarche		v	T- test, P< 0.001 (Premenarche showed higher risk)	М	S
ug	Ylikoski [1]	Risser (0-2 vs 3-5)		~	Student's t T-test, P<0.001, (higher risk for Risser 0-2)	М	S
Risser sign	Yrjonen and Ylikoski [2]	Risser (0-1 vs 2-3)	×		No significant difference between the two groups	М	S
Distal Radius and Ulna Classification	Cheung, et al. [3]	The distal radius (R4- R10) and ulna (U2- U8) (DRU) classification		~	Peak curve progression matched with R7 (0.80±0.89 cm/month) and U5 stages (0.84±0.78 cm/month) (low grades have higher risk)	М	L
State of Maturity	Duval- Beaupere [4]	State of maturity (I-IV)		~	F test, P<0.001. Risk is higher when first examination is between start of puberty and menarche (state II was at higher risk)	Н	S
	Cheung, et al. [3]	Growth rate (body height)		~	Height: Cor Coef=0.26, P<0.001, (higher values are risk factors)	М	L
تع م	Cheung, et al. [3]	Growth rate (arm span)		~	Arm span: Cor Coef=0.26, P<0.001 (higher values are risk factors)	М	L
rowth Factors	Ylikoski [1]	Growth of the T4-L4 segment		~	Cor Coef=0.38, P< 0.001 (higher values are risk factors)	М	S
Growt	Yrjonen and Ylikoski [2]	Growth velocity (standing heights between the last and the first visit were subtracted and divided by the follow-up time)		~	Growth velocity \geq 4 cm/year with curves \geq 25° increased progression velocity, Student's t-test, P< 0.001	М	S
Cobb Angle (°)	Duval- Beaupere [4]	Supine Cobb angle		~	Reg Coef= 0.59, P< 0.001	Н	S

Appendix 3.6. Summary of Articles that Reported Change in Cobb Angle as °/year

	Yrjonen and Ylikoski [2]	Cobb \geq 25 and \leq 25	~	Curves with Cobb ≥ 25 and Growth velocity ≥ 4 cm/year are at higher risk, Student's t-test, P< 0.001	М	S
	Duval- Beaupere [4]	TH, TL, L, double, triple	~	TH, Double, and triple curves showed higher rate of progression	Н	S
Curve Pattern	Ylikoski [1]	TH, TL, L	~	TH: Cor Coef= 0.56, P< 0.001, TL, L: Cor Coef=0.15 ((Correlation between the growth of the T4- L4 segment and the progression of curves in thoracic was more significant than TL and L)	М	S
	Yrjonen and Ylikoski [2]	Right and left TH, TL, L curves	>	Right TH was most progressive, other curve types didn't significantly differ in progression velocity	М	S
Rib Hump	Duval- Beaupere [4]	Rib hump	~	Reg Coef, r= 0.52 with annual increment of progression	Н	S

Note that the abbreviations used in table described as: Non-pre: Non-predictor, Pre: Predictor, Cor Coef: Correlation coefficient, Reg Coef: Regression coefficient, TH: thoracic, TL: Thoracolumbar, L: Lumbar

Appendix 3.6 References

- 1. Ylikoski, M., *Spinal growth and progression of adolescent idiopathic scoliosis*. European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society, 1993. 1(4): p. 236-9.
- Yrjonen, T. and M. Ylikoski, *Effect of growth velocity on the progression of adolescent idiopathic scoliosis in boys*. Journal of Pediatric Orthopaedics-Part B, 2006. 15(5): p. 311-315.
- 3. Cheung, J.P.Y., et al., *Curve Progression in Adolescent Idiopathic Scoliosis Does Not Match Skeletal Growth*. Clinical Orthopaedics and Related Research, 2018. **476**(2): p. 429-436.
- 4. Duval-Beaupere, G., *Rib hump and supine angle as prognostic factors for mild scoliosis.* Spine, 1992. **17**(1): p. 103-7.

Chapter 4: Ultrasound Reflection Coefficient (RC) Index

4.1. Overview

Based on the previous chapter regarding the study parameters suggesting the potential of measurement of bone quality to predict progression, a new ultrasound parameter called reflection coefficient (RC) index is introduced and described in this chapter. This novel index is considered to reflect the bone strength directly from the spine area. Section 4.2 includes a manuscript describing the theory behind the RC index including extraction, calculation, and validation of the RC index. Also, a correlation between the RC and the curve severity was also investigated. The title of the manuscript is "Investigation of Ultrasonic Soft Tissue-Bone Reflection Coefficients Correlating with Curve Severity in Children with Adolescent Idiopathic Scoliosis" and has been submitted to "Journal of Medicine and Engineering". Section 4.3 reports a clinical pilot study with the title of "Reliability of Measurements of a Reflection Coefficient Index to Indicate Spinal Bone Strength on Adolescents with Idiopathic Scoliosis (AIS): A Pilot Study". The manuscript was published in "European Spine Journal" in July 2021. Section 4.4 provides the summary of this chapter.

4.2. The Fundamentals of the Reflection Coefficient (RC) Index

4.2.1. Introduction

Scoliosis is a three-dimensional spinal deformity characterized by a lateral curvature of the spine. Adolescent idiopathic scoliosis (AIS) is the most commonly diagnosed form of scoliosis with unknown etiology [1], affecting 1–4% adolescent population especially children from 10 to 16 years old [2,3]. The curves with severe progression can have a negative impact not only on patients' psychosocial health but also on their physical well-being such as back pain, diminished pulmonary function, and increased mortality rate [4,5]. Understanding the etiopathogenesis of AIS can aid in providing more effective treatment to AIS patients.

Osteopenia or low bone density, quantified by bone mineral density (BMD) which means bone quantity or bone mass, is more commonly observed in children with AIS than normal children [6–9]. BMD measures the amount of calcium and is usually tested by dual-energy X-ray absorptiometry (DXA). Furthermore, children with AIS not only have lower BMD, but also have lower bone quality [10]. Bone quality, which is related to bone strength, describes microarchitecture, mineralization, turnover rate, and micro-fractures [11]. Bone quality can be assessed by radiographic methods [12]. Studies have revealed that the AIS groups have deteriorated bone quality and lower bone mass when compared with the control groups [10, 13–16]. Lee et al. also found that curve severity (Cobb angle) of scoliosis was inversely and independently associated with BMD [8].

On the other hand, quantitative ultrasound (QUS) is a non-invasive and ionizing radiationfree technique to evaluate bone quality. Lam et al. used the ultrasound transmissionthrough technique to measure bone properties at the calcanei in healthy group as well as in both mild and moderate AIS [10, 17]. Two measured QUS parameters, namely speed of sound (SOS) and broadband ultrasound attenuation (BUA), and a third derived stiffness index (SI), which is a combination of BUA and SOS, were found statistically lower in the group with AIS than in the healthy control group. Further, they suggested that SI was an independent and prognostic factor for curve progression and treatment planning [17]. Similar to the study of finding the relationship between BMD and severity of AIS [8], Du et al. attempted to correlate SOS with Cobb angles [18]. They found AIS subjects had lower SOS-values compared to non-scoliotic controls; however, they concluded there was no significant correlation between SOS and Cobb angles. So far, research findings have shown consistently that both bone quantity (BMD) and bone quality (SOS, BUA, and SI) could be found lower in AIS patients as compared to those of normal controls. However, a positive correlation between bone quantity or quality and curve severity has not been assertively established yet. In addition, quantitative US usually uses the transmission method to assess bone quality at peripheral sites other than the spine, namely calcaneus and radius [10, 18]. The use of reflected echoes from spinal scans to evaluate bone quality of the spines is limited.

When ultrasound propagates into the tissue, energy is reflected by any scatter which has a contrast of acoustic impedance with the surrounding tissue, and the strength of the echo mainly depends upon the magnitude of the contrast. Soft tissue–bone interface is an example of strong reflector [19, 20]. Given that the acoustic impedance of bone, which is a product of SOS and density, is related to the bone stiffness, and thus bone quality. Zheng et al. proposed to measure the ultrasound echoes directly from spines to assess bone quality and study if it correlated with the severity of AIS [21]. They introduced a reflection index (RI), which is a ratio of the received echo from the vertebra and a reference echo from a referenced phantom. The results showed that the RI decreased with the increase of curve severity; however, the correlation coefficient was small and the thickness of the soft tissue was ignored. Due to the limited fundamental work and theory in Zheng et al. study, we would like to develop a framework to explain the fundamental phenomenon of ultrasound echo from the spine to assess bone properties.

Therefore, this study aimed to a) develop the theoretical framework to explore the feasibility of determining a new parameter, called the reflection coefficient, from the ultrasound echoes acquired directly from the spine, and b) investigate if there are correlations between the reflection coefficient and the curve severity in children with AIS.

4.2.2. Materials and Methods

4.2.3. Theoretical Formulation

We consider a simple bone model with the cortex overlaid by soft tissue (Figure 4.1a). Both cortex and soft tissue are assumed to be homogeneous and isotropic. The thickness of soft tissue is h while the thickness of cortex is not relevant in this study. The cortical bone and soft tissue are characterized by the velocity (c), density (ρ), and the attenuation coefficient (α), where the subscripts b and s are used to denote bone and soft tissue, respectively. A source signal of amplitude A_0 is generated by the transducer, travels through the soft tissue, and is reflected by the soft tissue–cortex interface. The recorded amplitude of the echo, A, can be described by

$$A = A_0 R_{sb} e^{-2\alpha_s h} \tag{4.1}$$

where R_{sb} is the reflection coefficient of the soft tissue-cortex interface, i.e.,

$$R_{sb} = \frac{Z_b - Z_s}{Z_b + Z_s} \tag{4.2}$$

 $Z = \rho c$ is the acoustic impedance, and the factor $e^{-2\alpha_s h}$ accounts for the attenuation of the signal in the soft-tissue layer. The factor "2" in the exponent refers to the two-way travel within the soft tissue. In Equation 4.1, the unit of α_s is nepers/cm (np/cm) and $\alpha_s h$ is dimensionless.

The extraction of R_{sb} in Equation 4.1 requires the knowledge of A_0 of the source signal and the attenuation coefficient, α_s of the soft tissue. To estimate the two parameters, we utilize the multiple reflections or reverberation within a piece of soft-tissue mimic. Figure 4.1b shows one primary echo and two reverberations within the mimic, which can be respectively expressed by

$$A_{1} = A_{0}R_{ta}e^{-2\alpha_{t}h}$$

$$A_{2} = A_{0}(R_{ta})^{2}R_{tm}e^{-4\alpha_{t}h}$$

$$A_{3} = A_{0}(R_{ta})^{3}(R_{tm})^{2}e^{-6\alpha_{t}h}$$
(4.3)

where the subscript t, a, and m refer to soft-tissue mimic, acrylic, and matching layer of the transducer respectively; α_t , R_{ta} and R_{tm} are respectively the attenuation coefficient of the soft-tissue mimic, the reflection coefficients of the soft-tissue mimic–acrylic, and softtissue mimic–matching layer interfaces.

By defining

$$A_1' = \frac{A_1}{R_{ta}} = A_0 e^{-2\alpha_t h},$$

$$A_{2}' = \frac{A_{2}}{(R_{ta})^{2}R_{tm}} = A_{0}e^{-4\alpha_{t}h},$$

$$A_{3}' = \frac{A_{3}}{(R_{ta})^{3}(R_{tm})^{2}} = A_{0}e^{-6\alpha_{t}h},$$
(4.4)

Equation 4.4 can be generalized as

$$A_n' = A_0 e^{-2\alpha_t(nh)} \quad (n = 1, 2, 3)$$
(4.5)

Taking logarithm of both sides linearizes the equation

$$\ln(A_n') = -2\alpha_t(nh) + \ln(A_0)$$
(4.6)

where $-2\alpha_t$ and $\ln(A_0)$ are the slope and intercept of the best fitting line $(\ln(A_n')$ versus nh) by linear regression.

In this study, the clinical application of the reflection coefficient is to estimate the bone properties of spine. The elastic modulus of cortical bone, E, varies with its density ρ , through $E = 0.09\rho^{7.4}$ [22]. Other forms of the relationship were provided in a review study by Hegason et al [23]. Elasticity is then associated with velocity through $c = \left(\frac{E(1-\nu)}{\rho(1+\nu)(1-2\nu)}\right)^{1/2}$, where ν is the Poisson's ratio [24]. Changes in bone properties such as density and velocity affect the acoustic impedance of the bone Z_b , and higher R_{sb} reflects larger acoustic impedance of the bone Z_b , thus implying better bone quality. From the



Figure 4.1 (a) A bone model. (b) The primary echo and multiple reflections within the tissue mimicking layer used to estimate A_0 and α_t .

literature, the density of cortical bone in human ranges from 1800 to 2000 kg/m³ [25, 26], and correspondingly velocity ranges from 2617 to 3666 m/s [27, 28], which corresponds to the impedance range of 4.7–7.3 Mrayls, and thus to a R_{sb} -range of 0.49–0.64.

4.2.4. Phantom Studies

The SonixTouch Q+ ultrasound system equipped with a 128-element C5-2/60 convex transducer (BK Medical, MA, USA) was employed for the study. The transducer has an adjustable center frequency including: 2, 2.5, 3.3, and 4 MHz. The center frequency was set at 3.3 MHz for the whole study unless otherwise stated.

Determine the Magnitude of the Source of the Ultrasound Transmission Signal (A_0) , the Velocity of Sound (c_t) and the Attenuation Coefficient (α_t) of the Soft-Tissue Mimic (Blue Phantom)

As shown in Figure 4.2a, the transducer was mounted on an aluminum frame and placed in contact with a piece of 2.9-cm thick blue phantom (BP) (CAE Healthcare, FL, USA) overlying an acrylic plate, which was supported by two rubber corks. BP was used because it has ultrasound properties similar to those of soft tissue. Ultrasonic gel (Parker Laboratories, NJ, USA) was applied to all contacting surfaces between the transducer and the BP as well as between the BP and the plate to ensure good coupling. The thickness of BP after being compressed by the transducer was slightly under 2.9 cm, i.e., 2.2 cm. The



Figure 4.2 (a) Experimental setup to measure the ultrasound properties of Blue Phantom and plates, and to estimate the subsequent reflection coefficients. (b) Experimental setup to measure the reflection from a rough curved surface. The surface is also tilted about 3 degrees in a direction (out of the page) perpendicular to the long axis of the transducer array.

received RF ultrasound signal was recorded and used to determine the c_t , α_t and A_0 based on Equation 4.6.

Determine the Reflection Coefficient (R) of Five Different Materials

Based on the previous experimental setup, the acrylic plate was also replaced by different metal plates such as aluminum, brass, copper, and steel to estimate their reflection coefficients with BP. The stiffness of the tested materials ranged from soft to hardest (related to density) while the acoustic impedance of bone should be between that of acrylic and aluminum. To determine the reflection coefficient of the five materials based on measurements, the measured ultrasound signals with Equation 4.1 was used.

The determination of theoretical reflection coefficients is based on Equation 4.2 and requires the predetermined densities, velocities, or acoustic impedance. The density of the BP was measured by the Archimedes' principle while the densities of the five plates were referenced from the literature. The velocities of the used materials, i.e. BP and five plates, were measured by the ultrasonic pulse-echo method using the previous described framework. Regarding the matching layer of the transducer, its acoustic impedance is pooled from references. These values used for the calculation of reflection coefficients are listed in Table 4.1.

Material	Density (kg/m3)	Velocity (m/s)	Impedance (Mrayls)
Water [29]	1000	1480	1.48
Blue Phantom (BP) ^a	900	1485	1.34
Soft tissue [30, 31]	1050	1540	1.62
Cortical bone [32–34]	1930	3250	6.27
Acrylic [35]	1180	2720	3.21
Aluminum [35, 36]	2700	6150	16.61
Brass [36]	8415	4275	35.97
Copper [35, 36]	8900	4515	40.18
Steel [36]	8030	5700	45.77
PZT [37]	-	-	33.00
Matching layer ^b	-	-	7.30°

Table 4.1 Properties of the materials used in the study.

^aBoth density and velocity were experimentally determined.

 ${}^{b}Z_{matching} = (Z_{s}Z_{PZT})^{1/2} [37]$

^cThe value falls within the range given by the literature [38, 39].

Estimate the Effect of Surface Roughness and Inclination upon the Echo-amplitude

Beside the plate study, a 2.4-cm thick acrylic plate with a 19-cm diameter arc on one side (Figure 4.2b) was used to mimic the posterior arch of a vertebra. The surface of the arc was rough, which was caused by the circular-saw cutting. The transducer was placed at 3.5 cm from the rough surface of the phantom, which was similar to a normal distance from an ultrasound transducer to a lamina when the ultrasound was used to scan a scoliotic patient. To investigate the influence of the surface roughness, three scans were performed on: a) a smooth flat surface (SFS), b) rough curved surface (RCS), and c) a smooth curved surface (SCS) which was created by sanding the rough curved surface (RCS). The phantom with the RCS was also tilted at about 3 and 5 degrees in the direction perpendicular to the long axis of the transducer array, named RCIS, to examine the effect of inclination on the recorded echo amplitudes (Figure 4.2b). The experiments were carried out in a water tank with both the transducer and the phantom immersed in water.

Study a Vertebral Phantom

A second phantom study was performed on a cadaveric dry lumbar vertebra. The vertebra phantom was fixed to the bottom of the water tank with LePage[®] Fun-Tak[®] mounting putty (Lepage, Canada). Similar to the acrylic arch phantom experiment, the vertebra was submerged in water with the transducer set at 3.5 cm from the laminae. Again, the reflection coefficient of the cortical bone surface was calculated using Equation 4.1 with the ultrasound measurement signals.

4.2.5. In-vivo Pilot Study

Study Participants

Thirty-seven children (9 M; 28 F), aged 14.0 ± 1.6 years old (ranged between 11 and 17), were recruited from the local scoliosis clinic. Ethics approval was obtained from the University of Alberta Research Ethics Board. All participants signed the written consents prior to participation. The inclusion criteria were participants who 1) were diagnosed with

AIS, 2) had the age ranging from 10 to 18 years old, 3) the Cobb angle was between 10° to 45° (mild to moderate cases) and 4) had no prior surgeries.

Data Acquisition

The same ultrasound system described in the phantom study with the convex transducer was used for this in-vivo study. The scanning parameters used were: -15 dB power, 6-cm imaging depth, and 50% gain with linear time gain compensation (TGC). These parameters were selected based on the previous studies with some minor adjustments to ensure optimal image quality [40]. Data was obtained by an operator with two-years experience in using the US system to scan AIS subjects. The participants were asked to stand in a standard upright posture (Figure 4.3a) within a frame to prevent twisting of the body. The ultrasound gel was applied to their backs prior to scanning. During scanning, the transducer was positioned perpendicular to the sagittal profile of the subjects and moved along the spinal curve. Transverse B-mode images (cross-section image of a vertebra) (Figure 4.3b) were displayed in real-time, this allowed the operator to ensure the transducer was almost perpendicular to the lamina region. In this study, we used the low lumbar region (either L4 or L5) for analyses because these two vertebrae usually had little axial rotation in the spinal axis. Approximately 50 to 100 B-mode images and the corresponding radio-frequency (RF) data were exported for further analysis.



Figure 4.3 (a) The ultrasound scan of children with AIS in standing position. (b) A B-mode image of a vertebra. A cadaver vertebra is shown in the inset.

Selection of B-mode Frames

The lamina has been identified as a strong ultrasound reflector because the lamina area is usually a relatively flat surface [19]. This was similar to the arch phantom and the linear plate studies. The middle of the lumbar vertebra L5 was first identified. If the quality of image on L5 was poor, L4 was then used. Five consecutive B-mode frames around the middle of the vertebra with the most levelled pair of laminae were used.

4.2.6. Signal Processing

The RF data thus acquired was exported by Matlab software (R2019a, MathWorks, MA, USA). Each frame had 256 time series (A-lines). In the selected frames, the series with the largest reflected amplitudes from the laminae were selected for the analysis. Hilbert transform was applied to the series to obtain the envelopes of the signals [41]. The peaks of the envelopes were used as amplitudes to calculate the reflection coefficients. The envelope technique has been shown to be robust in facilitating the detection of the peaks of noisy signals, thereby yielding more consistent and stable results. Three recording with maximum amplitudes from the center of the transducer were used for the measurements in the experiments with plate phantoms. In the vertebral phantom and in-vivo studies about three to five recording of maximum echo-magnitudes corresponding to each lamina were exploited to measure the R. The measured R is the result of averaging the R from both laminae.

4.2.7. Results

4.2.8. Phantom Studies

Determination of A_0 , c_t , and α_t

The primary echo and two reverberations traveling within the BP are shown in Figure 4.4a with the envelope peaks (the A_n) being 15,863, 2,120, and 274 respectively, which were the average of the three time series at locations corresponding to the center of the transducer. By means of Equation 4.4, the A_n are converted to A_n' with R_{tm} and R_{ta} from Table 4.2. The best fitting line goes through the three points (Figure 4.4b) with a r^2 close

to unity, indicating the perfect fit between the data and the line. The source amplitude, A_0 was calculated to be 84,000. The best-fitted α_t is 0.18 np/cm or 1.53 dB/cm, which is compatible to the results obtained by transmission-through measurements with the correction for transmission loss through the water–BP interfaces [42]. The determined α_t thus can be considered the intrinsic absorption coefficient of BP. Further, using the echo and multiples, the velocity of the BP (c_t) was estimated to be 1,485 m/s.

Reflection Coefficient (R) of Five Different Materials

Based on the received reflection signals and Equation 4.1, the measured reflection coefficients, R, of the BP-plate interfaces are shown in Table 4.2. The velocity measurements used for the reference reflection coefficients fall within the ranges provided in literature (Table 4.1). The discrepancies of the measured R versus the calculated R ranged from 1.60% to 6.68%. Figure 4.4c also shows the measured reflection coefficients versus the calculated for the five plates.

Reflection coefficients	Calculated	Measured	ε (%)
BP/Acrylic (R_{ta})	0.41	0.38 ± 0.02	6.68
BP/Aluminum	0.85	0.84 ± 0.06	1.60
BP/Brass	0.93	0.94 ± 0.07	1.73
BP/Copper	0.94	0.98 ± 0.11	4.41
BP/Steel	0.94	0.96 ± 0.02	2.14

Table 4.2 The calculated and measured reflection coefficients of the interfaces involved in this study. ϵ is given by the absolute value of the difference (calculated - measured) × 100% divided by the calculated value.



Figure 4.4: (a) Envelopes of the recorded echo and two reverberations within the BP. The amplitude of the 2nd reverberation is small and the zoomed signal is shown in the inset. (b) The linear regression line of the three data points. (c) Comparison between the measured and predicted reflection coefficients. Error bars denote the standard deviations (see Table 4.2). (d) The simulated amplitude ratio with change of soft-tissue thickness for three α_s -values.

Influence of Surface Roughness and Inclination

Using the signals from the smooth flat surface (SFS) as the reference, the measured R_{wa} from the SFS, RCS, and SCS are 0.37, 0.28, and 0.33, respectively. The signal loss due to surface roughness is observed when the R_{wa} measured from the RCS is 85% of that of the SCS. Comparison between the acquired signals on the SFS and RCIS at 3 degrees is displayed in Figure 4.5a–b.



Figure 4.5 The ultrasonographs (Left) and the corresponding RF data (Right) of the following target: (a) acrylic phantom with the SFS, (b) acrylic phantom with the RCIS, (c) lumbar vertebra phantom, and (d) lumbar vertebra of a subject.

The RCIS signals (Figure 4.5b) show weaker intensity and amplitude in the ultrasound image and the corresponding RF plot than those of the SFS signals (Figure 4.5a). While the R_{wa} of the RCS accounts for about 76% of the reference R_{wa} (SFS), further loss due

to 3-degree inclination reduces the R_{wa} recovered from the RCIS signal (Figure 4.5b) to 0.23, just 62% of the reference. As the inclination increases to 5 degrees, the R_{wa} is reduced to just 0.12 or 32% of that from the SFS. The image of the cadaveric lumbar vertebra is shown in Figure 4.5c1–c2. The predicted R_{wb} is about 0.62 while the recovered coefficient is 0.23. Figure 4.5d1–d2 shows an example of the in-vivo ultrasound image of the lumbar vertebra and the corresponding RF data.

4.2.9. In-vivo Study



Figure 4.6 The correlation between the R_{sb} and Cobb angle.

The decay in echo-amplitude due to the attenuation of ultrasound in the soft tissue was compensated using an α_s -value of 1.65 dB/cm at 3.3 MHz. The measured R_{sb} of the 37 AIS patients ranges from 0.03 to 0.16. For example, the R_{sb} recovered from Figure 4.5d2 is 0.07. Based on the magnitude of the curves (Cobb angle – CA), the data was divided into two groups, mild ($10^\circ < CA \le 24^\circ$) and moderate ($25^\circ \le CA < 45^\circ$) curves. Figure 4.6 plots the measured R_{sb} versus CA; the mild curve group is colored with blue and the latter is shaded with red. The average R_{sb} values for the mild and moderate scoliosis regions are 0.11 ± 0.02 (n = 20) and 0.07 ± 0.03 (n = 17), respectively. Independent T-test analysis was performed to compare the measurements of the two groups (alpha of 0.05). The result shows that the measured R_{sb} values between the two groups are significantly different between the two groups (p-value = 0.0001 < 0.05). There was also a trend observed in the moderate curve region with $r^2 = 0.3$. The best fitting line shows the reflection coefficient declines with the increasing CA, i.e., with the severity of the scoliosis.

4.2.10. Discussion

The theoretical calculation of the reflection coefficients of the BP and five different materials' interfaces (acrylic, aluminum, brass, copper, and steel) was matched to the measured values and maximum discrepancy was 6.7% (Table 4.2). As the stiffness of cortical bone is in between acrylic and aluminum, it is expected that the reflection coefficient of BP-cortical bone interface is between 0.38 to 0.84.

From the surface roughness and inclination studies, the roughness of the surface demonstrated approximately 25% of the signal loss. When the surface was smoothened again, the received signal was back to nearly 90% of the original. The small curve on the surface (SCS) had a small effect when compared to the flat surface (SFS). Furthermore, the effect of the tilt angle was actually more significant. Together with the surface roughness, a 3-degree tilt demonstrated about 40% loss while a 5-degree tilt demonstrated a nearly 70% signal loss. These experiments demonstrated that a significant amount of energy loss was expected due to both the inclination and roughness of the surface.

From the cadaveric vertebra study, the recovered coefficient is 0.23 ± 0.06 , only 33% of the predicted R_{wb} (0.62). Besides, these factors affect the signal loss, the thickness of the human soft tissue may also have effects. Figure 4.4d shows the simulated responses of the amplitude ratio with soft-tissue thickness for three values of α_s . With a constant source amplitude A_0 and a fixed tissue thickness, the amplitude of the echo decreases with increasing α_s -value while for a fixed α_s , the amplitude decreases exponentially with thickness. The attenuation coefficient of soft tissue, α_s , in human ranges from 0.3 to 0.8 dB/cm at 1 MHz with an average of about 0.50 dB/cm [31, 43]. Since the relationship between α_s and frequency is linear [30, 44], α_s was extrapolated to be 1.65 dB/cm at 3.3 MHz, which is similar to α_t obtained by the best fitting regression and transmissionthrough experiments. Ultrasound energy decreases with distance exponentially as it propagates through the absorptive soft tissue. The absorption mechanism dissipates energy within the tissue as heat, reducing the energy and magnitude of the signal. Using the ultrasound images of the 37 AIS subjects, the thickness of soft tissue, measured from the lamina to the skin surface, ranged from 2 to 6 cm with an average of about 3.5 cm. At 3.5 cm soft tissue thickness, the simulated echo-amplitude is only 26% of the source amplitude for $\alpha_s = 1.65$ dB/cm. Therefore, the effect of soft tissue on the echo-amplitude cannot be ignored.

The effect of absorption upon the reflection coefficient was also studied. The first-order approximation of the amplitude acoustic reflection coefficient for two lossy biological tissues is given by [45]

And

$$R_{lossy} = R_{lossless} + i\frac{\Delta}{2} \tag{4.7}$$

$$\Delta = \frac{\alpha_s}{k_{\alpha_s}} - \frac{\alpha_b}{k_{\alpha_b}} \tag{4.8}$$

where $i^2 = -1$, $R_{lossless}$ is the lossless acoustic reflection coefficient, and Δ is the change of α/k (absorption coefficient per wavenumber) across the tissue interface. Assuming cortical bone has an attenuation coefficient of 45 dB/cm at 3.3 MHz [46], the absorption term (imaginary component) accounts for less than 10% or precisely 6.8% of the $|R_{lossy}|$. For the simplicity, we ignore the attenuation effect on the reflection coefficient and only consider the lossless reflection coefficient in this study.

After an approximate compensation of the attenuation in the soft tissue, the measured R_{sb} was in range of 0.03–0.16, lower than the expected values of 0.49–0.64 due to the mentioned loss factors. This loss is challenging to measure to fully recover the R_{sb} . However, by assuming the loss is similar in each AIS subject, meaningful conclusion could be drawn from the measured R_{sb} . The average R_{sb} value for the moderate curve was lower than that of the mild curve region (0.07 vs 0.11). There was also a mild tendency of decreasing R_{sb} when the curve was larger. A moderate correlation between the R_{sb} and CA was found with r^2 of 0.3.

4.2.11. Conclusion

This preliminary study has demonstrated the feasibility of using ultrasonic reflection coefficients for the estimation of material properties and its application in estimating bone properties of AIS patient groups. The total loss in echo amplitude due to the absorptive soft tissue layer, surface roughness and inclination is significant and difficult to quantify. However, although these losses hinder the recovery of true bone properties, the average R_{sb} on the mild curve region was larger than that of the moderate curve region. An inverse relationship between the R_{sb} and CA is found in the AIS group with moderate curves, indicating that bone properties may decrease with the severity of scoliosis. Future investigation of the change of R_{sb} on individual person should be performed to track the risk of progression.

References

[1]J. A. Farady, "Current principles in the nonoperative management of structural adolescent idiopathic scoliosis," *Physical Therapy*, vol. 63, no. 4, pp. 512–523, 1983.

[2] S. L. Weinstein, L. A. Dolan, J. C. Cheng, A. Danielsson, and J. A. Morcuende, "Adolescent idiopathic scoliosis," *The Lancet*, vol. 371, no. 9623, pp. 1527–1537, 2008.

[3] J. C. Cheng, R. M. Castelein, W. C. Chu et al., "Adolescent idiopathic scoliosis," *Nature Reviews Disease Primers*, vol. 1, no. 1, pp. 1–21, 2015.

[4] C. B. Barton and S. L. Weinstein, "Adolescent idiopathic scoliosis: natural history," In *Pathogenesis of Idiopathic Scoliosis*, Springer, Tokyo, pp. 27–50, 2018.

[5] E. Diarbakerli, A. Grauers, A. Danielsson, A. Abbott, and P. Gerdhem, "Quality of life in males and females with idiopathic scoliosis," *Spine*, vol. 44, no. 6, pp. 404–410, 2019.

[6] S. D. Cook, A. F. Harding, E. L. Morgan et al., "Trabecular bone mineral density in idiopathic scoliosis," *Journal of Pediatric Orthopedics*, vol. 7, no. 2, pp. 168–174, 1987.

[7] J. C. Cheng, L. Qin, C. S. Cheung et al, "Generalized low areal and volumetric bone mineral density in adolescent idiopathic scoliosis," *Journal of Bone and Mineral Research,* vol. 15, no. 8, pp. 1587–1595, 2000.

[8] W. T. Lee, C. S. Cheung, K. T. Yee et al., "Association of osteopenia with curve severity in adolescent idiopathic scoliosis: a study of 919 girls," *Osteoporosis International*, vol. 16, no. 12, pp. 1924–1932, 2005.

[9] X. F. Li, H. Li, Z. D. Liu, and L. Y. Dai, "Low bone mineral status in adolescent idiopathic scoliosis," *European Spine Journal*, vol. 17, no. 11, pp. 1431–1440, 2008.

[10] T. P. Lam, V. W. Y. Hung, H. Y. Yeung et al., "Abnormal bone quality in adolescent idiopathic scoliosis: a case-control study on 635 subjects and 269 normal controls with bone densitometry and quantitative ultrasound," *Spine*, vol. 36, no. 15, pp. 1211–1217, 2011.

[11] A. Klibanski, L. Adams-Campbell, T. Bassford et al., "Osteoporosis prevention, diagnosis, and therapy," *Journal of the American Medical Association*, vol. 285, no. 6, pp. 785–795, 2001.

[12] M. Machida, S. L. Weinstein, J. Dubousset, eds. *Pathogenesis of idiopathic scoliosis*, Springer Japan, 2018.

[13] W. S. Yu, K. Y. Chan, F. W. P. Yu et al., "Bone structural and mechanical indices in adolescent idiopathic scoliosis evaluated by high-resolution peripheral quantitative computed tomography (HR-pQCT)," *Bone*, vol. 61, pp. 109–115, 2014.

[14] C. S. K. Cheung, W. T. K. Lee, Y. K. Tse et al., "Generalized osteopenia in adolescent idiopathic scoliosis–association with abnormal pubertal growth, bone turnover, and calcium intake?," *Spine*, vol. 31, no. 3, pp. 330–338, 2006.

[15] K. T. Suh, S. S. Lee, S. H. Hwang, S. J. Kim, and J. S. Lee, "Elevated soluble receptor activator of nuclear factor-κB ligand and reduced bone mineral density in patients with adolescent idiopathic scoliosis," *European Spine Journal*, vol. 16, no. 10, pp. 1563–1569, 2007.

[16] K. Ishida, Y. Aota, N. Mitsugi et al., "Relationship between bone density and bone metabolism in adolescent idiopathic scoliosis," *Scoliosis*, vol. 10, no. 19, 2015.

[17] T. P. Lam, V. W. Hung, H. Y. Yeung et al, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound in Medicine & Biology*, vol. 39, no. 3, pp. 381–387, 2013.

[18] Q. Du, X. Zhou, J. A. Li et al, "Quantitative ultrasound measurements of bone quality in female adolescents with idiopathic scoliosis compared to normal controls," *Journal of Manipulative and Physiological Therapeutics*, vol. 38, no. 6, pp. 434–441, 2015.

[19] W. Chen, L. H. Le, and E. H. Lou, "Ultrasound imaging of spinal vertebrae to study scoliosis," *Open Journal of Acoustics*, vol. 2, no. 3, pp. 95–103, 2012.

[20] H. Li, L. H. Le, M. D. Sacchi, and E. H. Lou, "Ultrasound imaging of long bone fractures and healing with the split-step Fourier imaging method," *Ultrasound in Medicine & Biology*, vol. 39, no. 8, pp. 1482–1490, 2013.

[21] R. Zheng, L. H. Le, D. Hill, and E. Lou, "Estimation of bone quality on scoliotic subjects using ultrasound reflection imaging method-a preliminary study," In *2015 IEEE International Ultrasonics Symposium (IUS)*, IEEE, pp 1–4, 2015.

[22] M. B. Schaffer and D. B. Burr, "Stiffness of compact bone: effects of porosity and density," *Journal of Biomechanics*, vol. 21, no. 1, pp. 13–16, 1988.

[23] B. Helgason, E. Perilli, E. Schileo et al., "Mathematical relationships between bone density and mechanical properties: a literature review," *Clinical Biomechanics*, vol. 23, no. 2, pp. 135–146, 2008.

[24] C. T. Eneh, M. K. Malo, J. P. Karjalainen et al., "Effect of porosity, tissue density, and mechanical properties on radial sound speed in human cortical bone," *Medical Physics*, vol. 43, no. 5, pp. 2030–2039, 2016.

[25] J. C. Wall, S. K. Chatterji, and J. W. Jeffery, "Age-related changes in the density and tensile strength of human femoral cortical bone," *Calcified tissue International*, vol. 27, no. 1, pp. 105–108, 1979.

[26] L. J. Gibson and M. F. Ashby, *Cellular solids: structure and properties*. Cambridge university press, 1999.

[27] W. Abendschein and G. W. Hyatt, "33 Ultrasonics and selected physical properties of bone," *Clinical Orthopaedics and Related Research (1976-2007)*, vol. 69, pp. 294–301, 1970.

[28] J. D. Craven, M. A. Costantini, M. A. Greenfield, and R. Stern, "Measurement of the velocity of ultrasound in human cortical bone and its potential clinical importance: an in vivo preliminary study," *Investigative Radiology*, vol. 8, no. 2, pp. 72–77, 1973.

[29] Q. Ji, L. H. Le, L. J. Filipow, and S. A. Jackson, "Ultrasonic wave propagation in water-saturated aluminum foams," *Ultrasonics*, vol. 36, no. 6, pp. 759–765, 1998.

[30] W. R. Hendee and E. R. Ritenour, *Medical imaging physics*. John Wiley & Sons, 2003.

[31] J. T. Bushberg and J. M. Boone, *The essential physics of medical imaging*. Lippincott Williams & Wilkins, 2011.

[32] M. P. Andre, J. D. Craven, M. A. Greenfield, and R. Stern, "Measurement of the velocity of ultrasound in the human femur in vivo," *Medical Physics*, vol. 7, no. 4, pp. 324–330, 1980.

[33] S. Lees, J. M. Ahern, and M. Leonard, "Parameters influencing the sonic velocity in compact calcified tissues of various species," *The Journal of the Acoustical Society of America*, vol. 74, no. 1, pp. 28–33, 1983.

[34] L. H. Le, Y. J. Gu, Y. Li, and C. Zhang, "Probing long bones with ultrasonic body waves," *Applied Physics Letters*, vol. 96, no. 11, p. 114102, 2010.

[35] Z. Nazarchuk, V. Skalskyi, and O. Serhiyenko, *Acoustic emission: methodology and application*, Springer, 2017.

[36] J. Krautkramer and H. Krautkramer, *Ultrasonic testing of materials. 4th ed.* Springer Science & Business Media, 2013.

[37] D. Callens, C. Bruneel, and J. Assaad, "Matching ultrasonic transducer using two matching layers where one of them is glue," *NDT & E International*, vol. 37, no. 8, pp. 591–596, 2004.

[38] H. Persson and C. Hertz, "Acoustic impedance matching of medical ultrasound transducers," *Ultrasonics*, vol. 23, no. 2, pp. 83–89, 1985.

[39] S. Rhee, T. A. Ritter, K. K. Shung, H. Wang, and W. Cao, "Materials for acoustic matching in ultrasound transducers," In *2001 IEEE Ultrasonics Symposium. Proceedings. An International Symposium (Cat. No. 01CH37263)*, vol. 2, pp. 1051–1055, 2001.

[40] M. Khodaei, D. Hill, R. Zheng, L. H. Le, and E. H. Lou, "Intra-and inter-rater reliability of spinal flexibility measurements using ultrasonic (US) images for non-surgical candidates with adolescent idiopathic scoliosis: a pilot study," *European Spine Journal*, vol. 27, no. 9, pp. 2156–2164, 2018.

[41] L. H. Le, "An investigation of pulse-timing techniques for broadband ultrasonic velocity determination in cancellous bone: a simulation study," *Physics in Medicine & Biology*, vol. 43, no. 8, p. 2295, 1998.

[42] C. Zhang, L. H. Le, R. Zheng, D. Ta, and E. Lou, "Measurements of ultrasonic phase velocities and attenuation of slow waves in cellular aluminum foams as cancellous bone mimicking phantoms," *The Journal of the Acoustical Society of America*, vol. 129, no. 5, pp. 3317–3326, 2011.

[43] T. D. Mast, "Empirical relationships between acoustic parameters in human soft tissues," *Acoustics Research Letters Online*, vol. 1, no. 2, pp. 37–42, 2000.

[44] E. L. Carstensen, "The mechanism of the absorption of ultrasound in biological materials," *IRE Transactions on Medical Electronics*, vol. 3, pp. 158–162, 1960.

[45] L. H. Le and L. J. Filipow, "Approximating the acoustic reflection coefficient of lossy biological tissues," *Physics in Medicine & Biology*, vol. 42, no. 4, p. 757, 1997.

[46] R. Lakes, H. S. Yoon, and J. L. Katz, "Ultrasonic wave propagation and attenuation in wet bone," *Journal of Biomedical Engineering*, vol. 8, no. 2, pp. 143–148, 1986.

4.3. Test-retest and Reliability Analyses of the Reflection Coefficient (RC) Index

4.3.1. Introduction

Adolescent idiopathic scoliosis (AIS) is a lateral curvature of spine coupled with axial vertebral rotation (AVR). The Cobb angle measure on posteroanterior (PA) radiographs is the gold standard to quantify the severity of scoliosis [1]. Children with AIS are usually monitored every 6 to 12 months to check if curve progresses (Cobb change $>5^{\circ}$) [2]. Families and patients are concerned about the ionization radiation exposure and the risk of cancer [3, 4]. A study showed that the cancer rate in children with AIS was 4.3% which was five times higher than the age-matched population [5]. According to the literature, 27% to 38% of children with AIS suffer from osteopenia [6, 7], which means these children have lower bone mineral density (BMD) than normal. Dual x-ray absorptiometry (DXA) is the gold standard method to assess BMD, but it also exposes patients to ionizing radiation [8]. The BMD measured by DXA is also costly and rarely available in scoliosis clinics [8]. Other than the term bone density, bone quality is a better explanation when referring to all the bone properties which include the internal structure, geometry and the mechanical properties [9]. Quantitative Ultrasound (QUS) has been introduced to assess bone quality. The advantages of using ultrasound (US) are no exposure to ionizing radiation, low cost, and portability. More importantly, the elastic nature of the US waves provides an ideal method to evaluate the bone quality [10]. The main QUS parameters associated with the bone quality or bone strength are broadband ultrasound attenuation (BUA), the velocity of sound (VOS) and the stiffness index (SI) [11, 12]. The BUA is a measure of the ultrasound attenuation at different frequencies. It provides information about the cortical bone thickness, elasticity, and micro-architecture of bone. The VOS indicates the transmission velocity of ultrasound passing through bone which can evaluate the bone elasticity and density. Furthermore, the SI is a composite of BUA and VOS parameters and reflects bone elasticity. Langton et al. [13] conducted a study and showed that BUA was a good predictor of elasticity ($R^2=75\%$) and bone strength ($R^2=73\%$). In the 1990s a transmission method

which used US signals passing through the heel area to obtain QUS parameters was introduced. Lam et al. [14] used this transmission method to evaluate the bone quality in children with AIS. They found that the QUS parameters were correlated with the BMD value measured with DXA at the femoral neck, and these parameters were smaller than in a control group without scoliosis. Another study [12] also showed that the SI value measured at the heel area reflected low bone quality in children with AIS. This SI value was a prognostic factor of curve progression for AIS with an odds ratio of 2 (1.0-3.7, 95% CI) while BUA and VOS were not. However, none of these studies measured the bone quality of the spine directly. The transmission technique is not suitable for assessing the spine. Literatures [15-17] also reported that the speed of ultrasound and the reflection coefficient parameters were correlated with the stiffness and mechanical strength of cortical bones. Zheng et al. used an US pulse echo technique to measure the US reflection signals directly from spine [18]. In their preliminary study, they introduced a parameter called the frequency amplitude index (FAI) which reflected the cortical bone stiffness of spine. Their index, however, ignored the soft tissue attenuation effect and only showed a mild correlation with curve severity on children with AIS. In the present study, a modified FAI parameter called the reflection coefficient (RC) index was introduced. This ultrasound RC index accounts for soft tissue effect and reflectsbone stiffness. The objectives of this study were to investigate the test-retest, intra-rater and inter-rater reliabilities of the RC index measured from the lowest lumbar vertebra to reflect bone strength of the spine in children with AIS.

4.3.2. Methods

4.3.3. Clinical Participants

Ethics approval was granted from the local health research ethics board. Fifty-eight children were recruited. Participants signed written consents before data collection. The inclusion criteria were participants who a) were diagnosed with AIS, b) were aged between 10-18 years old, c) had major Cobb angle $<55^{\circ}$ and d) had no prior surgical treatment.

Among these 58 participants, 37 (64%) were under observation and 21 (36%) were under brace treatment.

4.3.4. Equipment

A SonixTouch Q+ ultrasound system equipped with a 128-element C5-2/60 convex transducer and a position sensor (BK Medical, MA, USA) was used. The US parameters were set at: frequency 3.3MHz, gain 50%, power -15dB and the imaging depth of 6 cm. These parameters were verified through *in-vivo* experiments to ensure the quality of the US image and raw data for measurements.

4.3.5. Data Collection

Each participant was scanned in a standardized standing posture (Figure 4.7). Prior to scanning, the spinous processes of C7, L5 and all vertebrae in between were palpated, identified, and marked on the skin. All scans started from the vertebral level C7 and terminated at L5. The US system captured both the spine image and the raw data. The raw data consisted of the amplitude of the US reflection signal at specific locations. Repeated scans were acquired on the first 24 participants during a single session. After the first scan, the operator told the participant to relax and then instructed him/her to hold the same posture as before again. A second scan was then performed.



Figure 4.7 Acquisition of ultrasound data from a participant with AIS in a standardized standing position with feet shoulder-width apart, hands holding the wooden frame, looking straight ahead, standing still and breathing normally.

4.3.6. Data Measurements

Two raters (R1, R2), both with 2+ years of experience measuring US images but novice on RC index measurements were involved. Both raters measured 10 practice images (not included in the analysis) and discussed any disagreement to ensure both raters were consistent and confident to measure the RC. For the test-retest study, rater 1 measured the RC for both scans (24 images) one week apart to minimize memory bias. For the reliability study, both raters measured all 58 images twice one week apart. Both raters were blinded to all participants' information and the other rater's measurements before measurements.

4.3.7. Reflection Coefficient (RC)

Equation (4.9) shows the calculation of the RC based on the US waves propagate through soft tissue with thickness (h), hitting the surface of the cortical bone and reflect back to the transducer. The magnitude of the US source signal (A_0) was determined in an *in-vitro* study prior to this study. A portion of the US transmitted signals was absorbed by the soft tissue due to attenuation. The magnitude of the reflected signal (A) was related to the stiffness and elasticity of the cortical bone [17]. Since the acoustic impedance of bone is significantly higher than that of soft tissue, most of the US signals hitting the soft tissue-bone interface are bounced back to the transducer if the surface is flat. The reflection signal is stronger if the bone is stiffer.

$$RC = \frac{Ae^{-2\alpha_{s}h}}{A_{0}}$$
(4.9)

where RC is the Reflection Coefficient of the soft tissue-bone interface; A is the amplitude of the echo; α_s is the attenuation coefficient of the soft tissue (1.65dB/cm at 3.3 MHz); h is the thickness of soft tissue, and A₀ is the amplitude of the source signal.

4.3.8. Measurement of the Reflection Coefficient (RC)

An in-house software was developed to semi-automatic measure the RC index. Figure 4.8 shows the procedures after the US image was reconstructed. The reconstruction process was described in [18]. Step 1) placing a red box around the L5 vertebra on the US image

(Figure 4.8a). Step 2) selecting the US frames around the centers of laminae of L5 on the transverse view (Figure 4.8b and c). Step 3) measuring the soft tissue thickness (h) from the surface of the skin to the point where the peak signal occurs from the laminae (Figure 4.8d and e). The magnitude of the maximal reflection signal A was extracted from the raw data based on the selected location. Then, the software automatically calculated the RC value using equation (4.9). Four other US frames, 2 above and 2 below were also selected to repeat the RC measurements (Figure 4.8b). Each selected US frame had 5 frames in between to ensure the RC measurements covered the entire region of the laminae.



Figure 4.8a-e The 3 steps process to measure the reflection coefficient (RC) from a reconstructed ultrasound images

4.3.9. Data Analysis

Three different methods with a total of 5 RC measurements were investigated: i) the maximum RC (MRC) on the left and right laminae among the 5 selected frames, ii) the average RC (ARC) values on the left and right laminae among the 5 selected frames, and iii) the combined average RC (CARC) value on both left and right laminae. The soft-tissue thickness h was measured for each method.

For the test-retest, the two-way mixed model and consistency ICC [3,1] was used. For the reliabilities analyses, a two-way random model and absolute agreement ICC [2,1] was calculated. The reliability was interpreted as; poor: ICC< 0.5, moderate: $0.5 \le ICC < 0.75$, good: $0.75 \le ICC \le 0.9$, and excellent: ICC> 0.90 [19]. The mean absolute difference (MAD) and the standard error of mean (SEM) [20] were reported. All statistical analyses were performed using the IBM SPSS v23 software (IBM, Armonk, New York, USA).

4.3.10. Results

Fifty-eight participants (47F, 11M) were included with age 14.0 ± 1.57 (10.3 - 16.9) years old and the major Cobb angle ranging from 12° to 48° ($24.3^{\circ}\pm8.6^{\circ}$) identified on both radiographs and US images without missing any data or measurements.

4.3.11. Test-retest Reliability Analysis

Table 4.3a reports the average RC values from scan 1 versus scan 2 on all 5 measurements. The average RC values was similar between left and right laminae for the same method. However, the overall range of the Left-MRC (0.03-0.34) was smaller than the Right-MRC (0.04-0.45). The difference was not significant (p<0.05). Also, the ARC was not significantly different from the CARC. Table 4.3b shows the ICC [3,1] for the test-retest between scan 1 and 2, and all five measurements show good reliability (range: 0.77 to 0.83). Table 4.3c shows the comparison of the h measurements with all methods. The overall variation was less than 0.54 cm (13% of average thickness) which was approximately 9 US frames (distance between each frame is 0.59 mm).

	Left L	amina	Right L	CARC	
	MRC	ARC	MRC	ARC	CARC
0 1	$0.16{\pm}0.08$	0.08 ± 0.04	0.17 ± 0.11	0.09 ± 0.04	$0.08 {\pm} 0.04$
Scan 1	(0.06-0.34)	(0.04-0.18)	(0.07-0.45)	(0.04-0.22)	(0.04-0.18)
G A	$0.16{\pm}0.07$	0.09 ± 0.04	0.18±0.11	0.09±0.05	0.09±0.03
Scan 2	(0.03-0.3)	(0.04-0.2)	(0.04-0.45)	(0.02-0.22)	(0.04-0.18)

Table 4.3a The average RC values for scan 1 and 2 from rater 1.

Note: MRC = maximum reflection coefficient, ARC = average reflection coefficient, CARC = combined average reflection coefficient,

Table 4.3b The test-retest reliabilit	of RC values between scan 1 and 2 from rater 1.

	Left Lamina		Right Lamina		CARC	
	ICC [3,1] (95% CI)	SEM	ICC [3,1] (95% CI)	SEM	ICC [3,1] (95% CI)	SEM
MRC	0.77 (0.54-0.89)	0.01	0.83 (0.66-0.92)	0.01	N/A	
ARC	0.81 (0.61-0.91)	0.01	0.81 (0.62-0.91)	0.01		
CARC		Ν	//A		0.82 (0.62-0.92)	0.01

Note: MRC = maximum reflection coefficient, ARC = average reflection coefficient, CARC = combined average reflection coefficient, M1 = measurement 1, M2 = measurement 2, R1 = rater 1, R2 = rater 2, CI= confidence interval, ICC= intraclass correlation coefficient, SEM= standard error of measurements.

Table 4.3c The MAD of soft tissue thickness h (cm) measured by the MRC, ARC and CARC methods on the first 24 images from rater 1.

Categories	Method	Left Lamina	Right Lamina	Average of Left & Right	
		MAD (cm)	MAD (cm)	MAD (cm)	
	MRC	0.54±0.43	0.48±0.43	N/A	
Soft tissue thickness (h)	ARC	0.46±0.39	0.43±0.36	1 N / A	
_	CARC	N/A	Ą	0.44±0.37	

Note: MRC = maximum reflection coefficient, ARC = average reflection coefficient, CARC = combined average reflection coefficient, MAD = mean absolute difference between measurements from scan 1 vs scan 2.

4.3.12. Intra-rater Reliabilities

Table 4.4a shows the RC values of both the left and right MRC and ARC for repeated measurements from raters 1 and 2. The right RC measurements were higher than the left. Table 4.4b shows the intra-rater reliabilities of the left and right MRC and ARC. Rater 2
showed an excellent reliability, but rater 1 only showed moderate and good reliability with left and right measurements, respectively. Between the ARC and MRC, ARC showed better reliability. Also, the right measurement was more reliable for both raters. Table 4.4c shows the intra-rater reliability for CARC with ICC [2, 1] \geq 0.84 and the SEM was \leq 0.01 for both raters. Among the 3 methods, the CARC showed the best reliability.

Table 4.4a The average RC measurements on the first and second measurements (M1 and M2) on the same 58 images using MRC and ARC methods from rater 1 (R1) and rater 2 (R2).

		Left Lamina		Right I	Lamina
		M1	M2	M1	M2
	D1	$0.14{\pm}0.07$	$0.14{\pm}0.08$	0.17±0.12	0.16±0.12
MDG	R1	(0.04-0.32)	(0.05-0.41)	(0.03-0.78)	(0.03-0.86)
MRC	DA	$0.12{\pm}0.06$	0.12±0.06	0.13±0.10	0.14 ± 0.11
	R2	(0.05-0.31)	(0.04-0.32)	(0.03-0.67)	(0.03-0.72)
	D1	$0.07{\pm}0.03$	0.08 ± 0.03	$0.09{\pm}0.05$	0.08 ± 0.05
	R1	(0.03-0.17)	(0.03-0.18)	(0.02-0.38)	(0.02-0.36)
ARC	DA	0.07 ± 0.03	0.07 ± 0.03	0.07 ± 0.04	0.08 ± 0.04
	R2	(0.03-0.14)	(0.02-0.14)	(0.02-0.27)	(0.02-0.28)

Table 4.4b The intra-rater reliability on RC measurements from R1 and R2 on the same 58 images using the MRC and ARC methods.

		I	Left Lamina	Rig	ght Lamina
		SEM	SEM ICC [2,1] (95% CI range)		ICC [2,1] (95% CI range)
	R1	0.04	0.53 (0.31-0.69)	0.03	0.75 (0.62-0.84)
MRC	R2	< 0.01	0.93 (0.88-0.95)	< 0.01	0.96 (0.93-0.97)
	R1	0.01	0.71 (0.55-0.81)	0.01	0.86 (0.78-0.91)
ARC	R2	< 0.01	0.97 (0.95-0.97)	< 0.01	0.98 (0.97-0.99)

		M1	M2	SEM	ICC [2,1] (95% CI range)	
CARC –	D1	0.08 ± 0.04	0.08±0.03	0.01	0.04 (0.72, 0.00)	
	R1	(range 0.03-0.22)	(range 0.03-0.20)	0.01	0.84 (0.73-0.90)	
	R2	0.070±0.03	0.07±0.03	<0.01	0.02 (0.07 0.00)	
		(range 0.03-0.17)	(range 0.03-0.17)	< 0.01	0.98 (0.97-0.99)	

Table 4.4c The mean±SD and intra-rater reliability for RC measurements from R1 and R2 on the same 58 images using the CARC method.

4.3.13. Inter-rater Reliabilities

Table 4.5a shows the inter-rater reliability of the RC measurements for MRC and ARC. Again, the ARC was more reliable than MRC, and the first measurements from both raters had better agreement. Table 4.5b shows the inter-rater reliability for CARC with ICC [2, 1] \geq 0.85 for both measurements and an SEM of 0.00. The MAD of the thickness h for the CARC method between raters 1 and 2 was almost the same for both measurements. In terms of the frame selection variation, there was maximum 10 frames difference between raters 1 and 2.

Table 4.5a The inter-rater reliability for RC measurements from R1 and R2 on the same 58 images using the MRC and ARC methods.

		Left Lamina		Right Lamina	
	Measurements	SEM	ICC [2,1] (95% CI)	SEM	ICC [2,1] (95% CI)
	M1	0.02	0.70 (0.50-0.82)	0.02	0.82 (0.60-0.91)
MRC	M2	0.04	0.54 (0.32-0.70)	0.02	0.81 (0.70-0.88)
	M1	0.01	0.83 (0.69-0.90)	0.01	0.86 (0.66-0.93)
ARC	M2	0.01	0.79 (0.66-0.87)	0.01	0.85 (0.76-0.91)

	Measurements	MAD±SD	SEM	ICC [2,1] (95% CI range)
CARC	M1	0.01 ± 0.01	< 0.01	0.86 (0.56-0.94)
	M2	0.01±0.01	< 0.01	0.85 (0.73-0.91)
Soft tissue thickness (h) (cm) difference	M1	$0.20{\pm}0.07$		
	M2	0.21±0.08		N/A
Frame selected difference	M1	7.47±7.29		
	M2	9.18±11.46		

Table 4.5b The inter-rater reliability for RC measurements on the same 58 images using the CARC method, the soft tissue thickness (h) measurement difference and the frame number selection differences between R1 and R2.

4.3.14. Discussion

This is the first study to investigate the reliabilities of the RC index. According to US theory, when the sound wave hits a tilted surface, the echo will reflect away from the transducer. The lamina at L5 is a relatively flat area with minimum rotation and should reflect most of the signals. In this study, the vertebral rotation on L5 was $2.8^{\circ} \pm 1.8^{\circ}$ (n=58), which was in a small magnitude range.

The test-retest study showed that the RC measurements are repeatable for each method. However, the RC value on the right lamina was slightly higher than left, this indicated that the right side of the vertebral bone was slightly stronger than the left. This phenomenon might be related to the curve direction. Among the 58 participants, 31 had major curves on right side. Fourteen of these had major curves > 25°. On the other hand, 27 subjects had left major curves but only 9 of their curves were > 25°. From a biomechanical point of view, the lumbar vertebra supports most of the body weight [23]. The Wolff's law stated that when the loading on a particular bone increase, that bone remodels itself over time and becomes stiffer [23, 24]. The soft-tissue thickness measured on the left lamina was also slightly bigger than that on the right. This might relate to more L5 rotation toward the right. The overall AVR for the 24 participants involved in the test-retest study was $2.7^{\circ}\pm 1.7^{\circ}$. The average AVR toward the right was $3.0^{\circ}\pm 2.0^{\circ}$ (ranged 0.6 to 8.4) while toward the left was $-2.1^{\circ}\pm 1.0^{\circ}$ (ranged -3.3° to -0.6°). For the intra-rater, the reliability on the right side

was better than on the left side. This was due to the RC value being smaller on the left side, hence it was more difficult to identify the peak value from the image. As shown in Figure 4.8c, the image is darker on the right which means signals are stronger. Also, R2 seems to measure the RC index more reliably. R1 had more deviations on cases when AVR was > 5°. There were five outliers on R1-MRC measurements. If these 5 cases were excluded, the intra-rater on the left RC improved from 0.53 to 0.70. The CARC presented better interrater reliability than both the MRC and ARC. Again, by excluding those five R1 outliers, the MRC inter-rater for left lamina improved from 0.54 to 0.66 for the second measurement. Also, the maximum average inter-rater difference for the soft tissue thickness h and for the number of frames were 2.1 mm and 9, respectively. Since the spacing between each frame was 0.59 mm, the 9 frames difference was equivalent to 5.4mm, which was smaller than the height of a single vertebra. In addition, the CARC showed the best reliability among the methods while the MRC had the least. This may be due to a single extreme value in MRC might affect the result significantly. To highlight the clinical significance, a preliminary association of the RC value and curve progression was investigated. Figure 4.9 shows the distribution of the CARC on all 58 participants. Eighteen out of 58 cases showed curve progression and their RC values were captured within 7 months of the RC data collection visit. The overall average of the CARC for all participants was 0.07 ± 0.03 . If we use the threshold of 0.07 (dashed line), 19 participants were above the average. Among those, 13/19 cases (68%) showed no progression. If we use 0.08 as the threshold, 12/15 cases (80%) showed no progression. Furthermore, there are 7 cases with values above 0.10 (mean + 1 SD) and all of those showed no progression. Hence, the stronger the bone, the less likely the curve is to progress.



Figure 4.9 Distribution of the participants with and without progression based on the CARC measurements (n=58)

4.3.15. Conclusions

The reflection coefficient (RC) index can be measured repeatably and reliably. The CARC method provided the best reliability (ICC [2, 1] \ge 0.77 and SEM \le 0.01). A RC value above the average already indicate a lower chance of curve progression. Future studies can consider using the RC index and other risk factors to predict curve progression.

4.4. Summary

This chapter reviewed a new ultrasound index called reflection coefficient (RC) which showed that this index is linearly related with the stiffness of the study materials. Based on the experimental results, the stiffer the bone, the higher the RC value. A moderate Correlation between RC and the curve severity was observed ($R^2=0.3$). The factors that affected the RC value and could cause signal loss were the absorption by soft tissue layer, surface roughness, and the inclination which are difficult to quantify. A pilot study was also conducted to test the repeatability and reliability of this index. In summary, the CARC

which was the combined average RC value on both the left and right laminae was the most repeatable and reliable to measure the RC index. The reliability for CARC method was good (ICC [2, 1] \ge 0.82 and SEM \le 0.01). Clinical application of this index was discussed, and our findings showed that 68% children who had the RC value above the average (RC = 0.07) showed a lower chance of curve progression.

References

- Kotwicki, T (2008) Evaluation of scoliosis today: examination, X-rays and beyond. Disability and rehabilitation, 30(10), 742-751.
- Negrini, S., Hresko, T. M., O'brien, J. P., Price, N., & SRS Non-Operative Committee (2015) Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee. Scoliosis, 10(1), 8.
- Levy, A. R., Goldberg, M. S., Mayo, N. E., Hanley, J. A., & Poitras, B (1996). Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis. Spine, 21(13), 1540-1547.
- Luo, T. D., Stans, A. A., Schueler, B. A., & Larson, A. N. (2015). Cumulative radiation exposure with EOS imaging compared with standard spine radiographs. Spine deformity, 3(2), 144-150.
- Simony, A., Hansen, E. J., Christensen, S. B., Carreon, L. Y., & Andersen, M. O. (2016). Incidence of cancer in adolescent idiopathic scoliosis patients treated 25 years previously. European Spine Journal, 25(10), 3366-3370.
- 6. Cheng, J. C., Guo, X., & Sher, A. H. (1999). Persistent osteopenia in adolescent idiopathic scoliosis: a longitudinal follow-up study. Spine, 24(12), 1218-1222.
- 7. Cheng, J. C., & Guo, X. (1997). Osteopenia in adolescent idiopathic scoliosis: a primary problem or secondary to the spinal deformity?. Spine, 22(15), 1716-1721.

- Deng, H. W., Xu, F. H., Davies, K. M., Heaney, R., & Recker, R. R. (2002). Differences in bone mineral density, bone mineral content, and bone areal size in fracturing and non-fracturing women, and their interrelationships at the spine and hip. Journal of bone and mineral metabolism, 20(6), 358-366.
- Glüer, C. C., Wu, C. Y., & Genant, H. K. (1993). Broadband ultrasound attenuation signals depend on trabecular orientation: an in vitro study. Osteoporosis International, 3(4), 185-191.
- 10. Laugier, P (2006) *Quantitative ultrasound of bone: looking ahead. Joint Bone Spine* 73(2):125-128.
- 11. Lam TP, Hung VWY, Yeung HY, et al. (2010) Quantitative ultrasound for detection of deranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS) - A prospective cohort study till skeletal maturity. Osteoporosis Int 21(SUPPL. 5):S686-S687.
- 12. Lam, T. P., Hung, V. W. Y., Yeung, H. Y., et al. (2013). Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity. Ultrasound in medicine & biology, 39(3), 381-387.
- Langton, C. M., Njeh, C. F., Hodgskinson, R., & Currey, J. D. (1996). Prediction of mechanical properties of the human calcaneus by broadband ultrasonic attenuation. Bone,18(6), 495-503.
- 14. Lam, T. P., Hung, V. W. Y., Yeung, H. Y. (2011) Abnormal bone quality in adolescent idiopathic scoliosis: a case-control study on 635 subjects and 269 normal controls with bone densitometry and quantitative ultrasound. Spine, 36(15), 1211-1217.
- Mathieu V, Chappard C, Vayron R, Michel A, Haiat G (2013) Radial anatomic variation of ultrasonic velocity in human cortical bone. Ultrasound Med Bio 39(11):2185-2193. https://doi.org/10.1016/j.ultrasmedbio.2013.06.005.

- Mesquita AQ, Barbieri G, Barbien CH (2016) Correlation between ultrasound velocity and densitometry in fresh and demineralized cortical bone. Clinics (Sao Paulo) 71(11):657-663. https://doi.org/ 10.3031/clinics/2016(11)07.
- Matusin DP, Fontes-Pereira AJ, Barboza T, Augusto S, Antonio M, Pereira WC (2018) Exploring Cortical Bone Density through the Ultrasound Integrated Reflection Coefficient. Acta Orthopedica Brasileira 26(4) Sao Paulo, https://doi.org/10.1590/1413785220182604177202.
- Zheng, R., Le, L. H., Hill, D., & Lou, E. (2015) Estimation of bone quality on scoliotic subjects using ultrasound reflection imaging method-a preliminary study. In 2015 IEEE International Ultrasonics Symposium (IUS) (pp. 1-4). IEEE.
- Bushberg, J.T. and J.M. Boone (2011) The essential physics of medical imaging: Lippincott Williams & Wilkins.
- 20. Khodaei, M., Pacheco-Pereira, C., Trac, S., et al (2018) Radiographic methods to estimate surgical outcomes based on spinal flexibility assessment in patients who have adolescent idiopathic scoliosis: a systematic review. The Spine Journal, 18(11), 2128-2139.
- Koo, T. K., & Li, M. Y. (2016). A guideline of selecting and reporting intraclass correlation coefficients for reliability research. Journal of chiropractic medicine, 15(2), 155-163.
- Weir, J. P. (2005). Quantifying test-retest reliability using the intraclass correlation coefficient and the SEM. The Journal of Strength & Conditioning Research, 19(1), 231-240.
- 23. Wolff J. (2012) The law of bone remodelling. Springer Science & Business Media.
- 24. Frost HM. Wolff's Law and bone's structural adaptations to mechanical usage: an overview for clinicians. The Angle Orthodontist. 1994;64(3):175-188.

Chapter 5: Ultrasound Kyphotic Angle (KA)

5.1. Overview

This chapter describes a new US measurement method of the kyphotic angle (KA) on sagittal US images. The materials of this chapter are based on two studies which have been submitted to refereed journals. In section 5.2, a pilot study with the title "Intra- and Interrater Reliabilities and Differences of Kyphotic Angle Measurements on Ultrasound Images versus Radiographs for Children with Adolescent Idiopathic Scoliosis – A Preliminary Study" is presented. This study has been submitted to "Journal of Spine Deformity". The section 5.3 presents a research study with the title" Factors Influencing Kyphotic Angle Measurements on Ultrasound Spinal Images in Children with Adolescent Idiopathic Scoliosis (AIS)" which has been submitted to "European Spine Journal". Section 5.4 provides the summary of this chapter.

5.2. Reliabilities and Differences of Kyphotic Angle Measurements on Ultrasound Images versus Radiographs

5.2.1. Introduction

Adolescent idiopathic scoliosis (AIS) is a three-dimensional (3D) spinal condition characterized by lateral curvature and vertebral rotation. It has no known cause and primarily affects girls aged 10 to 18 years old. The current gold standard to diagnose and monitor AIS is to measure the Cobb angle on standing posteroanterior (PA) radiographs [1]. Currently, the clinically accepted error for the Cobb angle measurement is 5° and the definition of curve progression is indicated by an increase of 6° or more [2] of the Cobb angle. According to the Scoliosis Research Society, the recommended treatments are observation, orthotic treatment (bracing), and surgery [2]–[4], and most scoliosis centers use the value of the Cobb angle to inform treatment recommendations. However, treatment decision based solely on the Cobb angle measured on PA radiographs may fail to account for the 3D nature of AIS, an issue which is only recently being addressed according to the

International Scientific Society on Scoliosis Orthopaedic and Rehabilitation Treatment (SOSORT) guidelines.

Sagittal parameters such as the kyphotic angle (KA), lumbar lordosis, and pelvic incidence have been identified as important factors according to newer SOSORT guidelines on the treatment of AIS as a 3D spinal disorder [4]. Thoracic hypokyphosis in particular is prevalent in patients with AIS [5]–[9]. Mac-Thiong et al. found a correlation between the thoracic curve type and the degree of the KA, and additionally suggested that inadequate sagittal biomechanics could alter the loading of the spine, affecting the progression of AIS [6]. Furthermore, Van Loon et al. have demonstrated that the correction of a flattening sagittal spine via the introduction of a wedge at the thoracolumbar junction in the supine position worked to reduce the Cobb angle on PA radiographs. This approach suggested a role for the correction of thoracic hypokyphosis in the conservative treatment of AIS [10].

Scoliotic and kyphotic angles are generally measured from standing radiographs. There are only few studies have addressed the reliability and repeatability of KA measurements. This is due to the inconsistencies in vertebral selection, making comparisons between measurements difficult. Among all the KA measurement approaches, there were studies using the Cobb method to measure the kyphosis between superior endplate of T1 and inferior endplate of T12 [5], [11]–[16]. A study by Carman et al. measuring KA found that an 11° difference in KA was accepted to rule out measurement error with 95% confidence [13]. Ohrt-Nissen et al. found an intra-rater reproducibility of 9° and inter-rater reliability of 13° [16].

However, the frequency of taking lateral (LAT) radiographs exposes children with AIS to higher ionizing radiation than PA radiographs, which increases lifetime cancer risk [17]. Even though the EOS Imaging machine is recently available in many scoliosis centers and the ionizing radiation dosage is reduced, the clarity of identifying the endplates of vertebrae at the upper thoracic region is still a challenge. Ultrasound (US) imaging is a radiation-free alternative that has been used to successfully measure the Cobb angle and the axial vertebral rotation (AVR) on standing PA and transverse views [18], [19], respectively. As

a single US scan of the spine can provide information on all 3 planes: PA (Cobb), transverse (AVR) and lateral (KA), it may help to measure the KA more reliably. Recently, Lee et al. also used US to measure the sagittal curvature. However, their method only used the sagittal information. Similar to the US PA view, the endplates of vertebrae are invisible in the lateral view, and so an alternate method to measure sagittal angles must be developed. The objectives of this study were to 1) present a new method to measure KA based on 3D US images, 2) determine the intra-rater and inter-rater reliabilities of the KA measurement, and 3) determine the inter-method accuracy in comparison to the Cobb method measured from radiographs.

5.2.2. Methods

5.2.3. Clinical Participants

Twenty subjects (17F, 3M, aged 13.7±2.2 years old, range 10-17 years old) were recruited in this study from a local scoliosis clinic. The inclusion criteria were subjects who were 1) diagnosed with AIS, 2) Cobb angle $\leq 50^{\circ}$, 3) required out of brace PA and sagittal radiographs, and 4) had no prior surgical treatment. The Cobb angle of $\leq 50^{\circ}$ was chosen, which can eliminate patients who have large axial vertebral rotation, to make sure laminae are visible on the ultrasound images. Ethics approval was obtained from the local health research ethics board and written consents were obtained from all subjects prior to data acquisition.

5.2.4. Data Acquisition

Standing out of brace PA and LAT radiographs and an US scan were obtained from each subject on the same day, with the US scan following the radiographs within 1 hour. The PA and LAT radiographs were acquired simultaneously using the EOS system (EOS Imaging, Paris, France) with subjects standing in a standard posture: subjects looked forward without tilting their heads and their hands were placed on the front chamber wall of the EOS system. US images were acquired using the Sonix TABLET medical US system equipped with a 128-element C5-2/60 GPS convex transducer (Analogic Ultrasound – BK

Medical, Peabody, Massachusetts, USA). This system recorded the orientation and location of the transducer to capture the 3D information of the spine. Each US scan started at the C7 vertebra and terminated at L5, with the transducer positioned perpendicular to the subject's back and moved along the path of the curve. During the scan, each subject was positioned in a standing frame in a position similar to that used in the EOS chamber with the hands placed at chest height against the opposite wall of the standing frame. The hips and the shoulders were also positioned in the same plane in a neutral standing position to minimize movement of the subject during the scan. Each US scan lasted for less than 1 minute. The US operator had 3 years of experience to acquire good US spinal images. An in-house developed program, called the Medical Imaging Analysis System (MIAS), was used to compile and process the acquired US data to produce coronal, transverse and sagittal plane images of the spine. This software allows users to zoom in and out the image, adjust the contrast and brightness.

5.2.5. Raters

Two raters (R1 and R2) performed ultrasound KA measurements from US images. R1 had 2 years of experience measuring KA on both radiographs and US LAT images and R2 had 4 years of experience measuring KA on LAT radiographs, but less than 1-year experience on US images. R3 who had over 20 years of experience on radiographs measurement measured the KA on the corresponding radiographs. Prior to the study, both R1 and R2 measured an extra 10 LAT US images from previous patients' records as a training set to obtain a mutual measurement agreement. These 10 measurements were not used for analysis.

5.2.6. Data Measurements

The KA on both radiographs and US images were measured using the MIAS program. All raters were blinded to the clinical information and R3 measured the KA from the radiographs once while R1 and R2 measured the KA from the US images twice. The radiographs and US images were randomly coded with numbers. The second US

measurement (M2) was measured one-week apart from the first measurement (M1) to minimize recall bias. On LAT radiographs, the Cobb method was applied. During the measurements, all raters were permitted to adjust contrast, brightness and magnification. For the radiography measurement, two lines were then drawn parallel to the top endplate of T1 and bottom endplate of T12, and the MIAS program displayed the value automatically.

On US images, the centers of lamina (COL) of the T1-T3 and T10–T12 vertebrae are first identified on the PA view (Figure 5.1a). The tips of the 6 spinous processes (SP) were then identified in the transverse views (Figure 5.1b) which was more precise than the PA view. The SP would automatically display in the sagittal view, where its location could be fine-tuned (Figure 5.1c). The identification of T3 and T10 were only used to confirm the trend of the spinal curve. The MIAS program also automatically displayed the KA calculated using the intersection of a line connecting the SP of T1 and T2 and another line connecting the SP of T11 and T12. The program calculated the acute angle formed by the intersection of these two lines, which was taken to represent the proxy KA.



Figure 5.1 (a) An ultrasound image showing a coronal PA view of spine with T1-T3 and T10-T12 identified on the image, (b) an US transverse view of T1 with the laminas identified and joined using a line and the midpoint of the line as the spinous process, (c) US sagittal view of spine and the kyphotic angle measured 27 ° using the slope of a line joining spinous process of T1 and T2 at the top to T11 and T12 at the bottom, d) the sagittal view X-ray of the spine correspond to the same patient and the kyphotic angle measured using the Cobb method as 27° .

5.2.7. Statistical Analysis

Statistical analysis was performed using IBM SPSS Statistics v23 software (IBM, Armonk, New York, USA). The mean and standard deviation (SD) of the KA measurements on both radiographs and US images from both raters were reported. The intra- and inter-rater reliabilities of the US KA measurements were calculated using the intraclass correlation coefficient [ICC (2,1)] with a 2-way random model and absolute agreement with a confidence interval of 95%. The inter-rater reliability was determined by comparing the measurements made by both raters (R1 and R2) with 2 different measurements (M1, M2). The accuracy of the US measurements was analyzed by comparing the radiographic measurements with US measurements. The ICC value was considered excellent (\geq 0.90), good (0.75-0.90), moderate (0.5-0.75), or poor (<0.5) reliability based on Koo's report [20]. The mean absolute difference (MAD) and standard error of the mean (SEM) [21]were determined for the intra-rater, inter-rater and inter-method analyses. The inter-method of inter-rater comparison was calculated to report the measurement differences. Bland-Altman analyses were also performed. A regression equation between the US KA and the radiographic KA value was generated based on all rater 1 and 2 US measurements.

5.2.8. Results

Among the 20 subjects, 10 subjects were under observation and they were in their initial visits. Another 10 subjects were under brace treatment and they were in those clinics that their braces were first prescribed. The major Cobb angle of the subjects ranged from 13° to 41° ($25^{\circ}\pm8^{\circ}$) and KA from radiography ranged from 16° to 56° ($38^{\circ}\pm10^{\circ}$). A total of 20 KAs were measured on both radiographs and US images. Table 5.1 summarizes the mean, standard deviation, and the range of the KA measurements by each rater.

	M1 (°)	M2 (°)
R1 (US)	40±10 (22-56)	37±12 (11-58)
R2 (US)	41±11 (18-59)	39±13 (17-60)
R3 (Radiograph)	38±10 (16-56)	N/A

Table 5.1 KA measurement means, standard deviations, and range on radiographs and US images (n = 20)

R -Rater, M - Measurements

Table 5.2 shows the MAD±SD, SEM and the intra-rater reliability (ICC[2,1]) of the US KA measurements of R1 and R2. Both raters have excellent reliability.

	5		1
	MAD±SD (°)	SEM (°)	ICC [2,1]
R1	3.1±1.7	0.4	0.94
R2	3.2±2.3	0.5	0.95

Table 5.2 The intra-rater reliability of the US KA measurements (n = 20)

R - Rater; MAD - mean absolute difference; SD - standard deviation; SEM - standard error of the mean; ICC - intraclass correlation coefficient

Table 5.3 compares the inter-rater reliability on both US KA measurements. The ICC values showed good reliability (>0.85), and there was no statistically significant difference on the measurements (P < 0.05).

Table 5.3 The inter-rater reliability for KA measurements on both US and radiographs (n = 20)

Image Modality	MAD±SD (°)	SEM (°)	ICC [2,1]
M1 US	4.2±4.6	1.8	0.85
M2 US	4.8±4.2	1.6	0.86

Table 5.4 shows the inter-method comparison of R1 and R2. The results from both raters on KA measurements from radiographic measurements, US-M1 and US-M2 showed similar as all the SEM values were under 1.7° and the ICC[2,1] values were >0.84.

Variable	MAD±SD (°)	SEM (°)	ICC [2,1]
R1 (M1 vs X-ray)	4.2±3.0	1.1	0.87
R1 (M2 vs X-ray)	3.9±2.9	0.9	0.90
R2 (M1 vs X-ray)	5.0±4.1	1.7	0.84
R2 (M2 vs X-ray)	4.1±3.9	1.3	0.88

Table 5.4 The inter-method comparison and reliability of the KA measurements from both raters (n=20)

Figure 5.2 shows a comparison between the US M2 vs. the radiographic measurements for both R1 and R2. Both R1 and R2 showed the US KA always overestimated. The variability was smaller when the KA values were larger. The US measurements of both raters were averaged and plotted against the average radiographic KA measurements by both raters. A linear equation of the radiographic KA = 0.82° x US KA + 5.60° was generated, which can be used to convert the US KA to its radiographic equivalent. When using this equation to assess the accuracy of the US measurements, the overall MAD between US and radiographic KA was $2.9\pm1.6^{\circ}$.



Figure 5.2 Inter-method comparison between the radiographic measurements and US measurements from both raters (R1 and R2) and the average of both raters with R2 indicating the percent of variance explained by the regression.

Figure 5.3 shows the Bland-Altman analyses for R1 x-ray versus the second US measurement and the same repeated for R2. R1 demonstrates less bias than R2 between the two types of measurements, with a scatter that indicates moderate overestimation but all points within 1.96 standard deviations about the mean. R2 showed a systematic overestimation of larger values of KA than R1, confirming the results found in Figure 5.2.



Figure 5.3 The Bland-Altman plot of mean and ± 1.96 standard deviations of the mean of radiographic and US measurements versus the differences between the radiographic and US measurements for (a) Rater 1 and (b) Rater 2.

5.2.9. Discussion

Scoliosis is a three-dimensional problem which may include abnormal spinal curvature in all 3 planes. Besides the Cobb angle on the coronal view and the axial vertebral rotation on the transverse view, the KA on the sagittal view is also an important parameter to be measured. However, due to the increase of cancer risk in taking an extra radiograph, most orthopedic surgeons decide not to take the LAT radiograph in the follow-up clinic unless patients complain about back pain. Although the low dose x-ray system (EOS) has become common, clinicians and parents are still concerned about the accumulated ionizing radiation. The US method has been introduced and it has been demonstrated that it can measure the Cobb angle and vertebral rotation reliably, but it has not been developed for kyphotic measurement using the available 3D information. From the literature, Carman et al. reported that, without fixing the end vertebrae to measure the KA, the measurement difference could be as large as an 11° difference in between consecutive measurements of the same subject [13]. Also, there is no standard clinically accepted error in KA measurements. The KA measurement differences between the inter-rater and the intermethod of both raters (Tables 5.3 and 5.4) ranged from 3.9° to 4.8°, which showed very little difference, indicating that kyphotic angle measurements from both radiographs and US images are similar.

A limitation in assessing the relevance of the data acquired in this study is the wide variety of methodology used in determining thoracic kyphosis (TK), including but not limited to a range of radiographic parameters (T1-T12, T2-T12, T5-T12, and nonfixed endplate methods), computer assisted methods, and surface topography [5], [12], [16], [22], [23]. While the T1-T12 method is the most commonly reported, for both ease of comparison of the degree of kyphosis between subjects, and also for its relevance in capturing the entirety of the global TK. Of course, it has some radiograph-specific drawbacks such as the difficulty in visualizing the T1 vertebra due to the overlap of the shoulder girdle in some subjects [5], [11]–[14], [16]. This limitation does not exist in US images because the vertebral selection and the COL identification on both coronal and transverse planes are

shown clearly. This makes T1 clearly visible in every US image, and this highlights a key strength of using US images for kyphotic measurements. Additionally, a study [24] has compared different methods based on plumbline measures, using correlation of plumbline distance, or based on video rasterstereography methods to measure kyphosis. These studies only provided fair to good results on reliability, but no direct measurements were reported.

In addition to the lack of consistency in terms of measurement method for TK, there are only two studies reported the standard intra-rater and inter-rater reliabilities of the KA measurement based on T1-T12 [15], [16]. Among the two studies, Ilharreborde et al. [15] reported an average intra-operator difference of three experienced raters was 6° and an inter-operator reproducibility of 7°. The current study demonstrates similar results. Furthermore, Ilharreborde et al. already used 3D EOS system to capture the TK, which may have more reliable images. Ohrt-Nissen et al. [16] reported the intra-rater and interrater reliabilities of 0.87 and 0.82, respectively. Our results reported similar or better results on both intra-rater and inter-rater reliabilities while using the US images.

Limitations of our study include a) the US image quality, which may depend on the skill of the US operator, and b) the identification of the lamina, which requires experience to identify COL on US images. However, the advantage of the US images is the ability to view the spinal images in all three-dimensional views (coronal, transverse, and sagittal). This eliminates the difficulty of only observing on the sagittal view. Furthermore, as this is the only known study to measure KA directly from US images, the reduction of exposing children to extra ionizing radiation can be made.

5.2.10. Conclusions

The intra-rater and inter-rater reliabilities of the KA measurements on US images can be performed reliably and the measurement differences are within the clinical accepted range. Also, the average differences of the KA measurements between the US images and radiographs is 4°, which shows no significant difference on repeat KA measurements from radiographs. However, since the number of study cases is small, a larger clinical trial is needed to validate the US method can be applied to scoliosis clinics.

References

- J. R. Cobb, "Outline for the study of scoliosis," *Instr Course Lect AAOS*, vol. 5, pp. 261–275, 1948.
- B. P. Wills *et al.*, "Comparison of Cobb angle measurement of scoliosis radiographs with preselected end vertebrae: traditional versus digital acquisition," *Spine*, vol. 32, no. 1, pp. 98–105, 2007.
- J. A. Janicki and B. Alman, Scoliosis: Review of diagnosis and treatment," *Paediatr. Child Health*, vol. 12, no. 9, pp. 771–776, 2007.
- [4] S. Negrini, et al. "2016 SOSORT guidelines: orthopaedic and rehabilitation treatment of idiopathic scoliosis during growth," *Scoliosis* 13, 3 (2018). https://doi.org/10.1186/s13013-017-0145-8.
- [5] J. Legaye, G. Duval-Beaupère, J. Hecquet, and C. Marty, "Pelvic incidence: a fundamental pelvic parameter for three-dimensional regulation of spinal sagittal curves," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 7, no. 2, pp. 99–103, 1998, doi: 10.1007/s005860050038.
- [6] J.-M. Mac-Thiong, H. Labelle, M. Charlebois, M.-P. Huot, and J. A. de Guise, "Sagittal plane analysis of the spine and pelvis in adolescent idiopathic scoliosis according to the coronal curve type," *Spine*, vol. 28, no. 13, pp. 1404–1409, 2003.
- [7] S. L. Propst-Proctor and E. E. Bleck, "Radiographic determination of lordosis and kyphosis in normal and scoliotic children.," *J. Pediatr. Orthop.*, vol. 3, no. 3, pp. 344–346, 1983.
- [8] V. V. Upasani *et al.*, "Analysis of sagittal alignment in thoracic and thoracolumbar curves in adolescent idiopathic scoliosis: how do these two curve types differ?," *Spine*, vol. 32, no. 12, pp. 1355–1359, 2007.

- [9] Q. Yong *et al.*, "Comparison of sagittal spinopelvic alignment in Chinese adolescents with and without idiopathic thoracic scoliosis," *Spine*, vol. 37, no. 12, pp. E714–E720, 2012.
- [10] P. J. M. van Loon, B. A. G. Kühbauch, and F. B. Thunnissen, "Forced lordosis on the thoracolumbar junction can correct coronal plane deformity in adolescents with double major curve pattern idiopathic scoliosis," *Spine*, vol. 33, no. 7, pp. 797–801, Apr. 2008, doi: 10.1097/BRS.0b013e3181694ff5.
- [11] A. M. Briggs, T. V. Wrigley, E. A. Tully, P. E. Adams, A. M. Greig, and K. L. Bennell, "Radiographic measures of thoracic kyphosis in osteoporosis: Cobb and vertebral centroid angles," *Skeletal Radiol.*, vol. 36, no. 8, pp. 761–767, 2007.
- [12] D. Harrison, R. Cailliet, D. Harrison, T. Janik, and B. Holland, "Reliability of Centroid, Cobb, and Harrison Posterior Tangent Methods: Which to Choose for Analysis of Thoracic Kyphosis," *Spine*, vol. 26, pp. E227-34, Jun. 2001, doi: 10.1097/00007632-200106010-00002.
- [13] D. L. Carman, R. H. Browne, and J. G. Birch, "Measurement of scoliosis and kyphosis radiographs. Intraobserver and interobserver variation.," *J. Bone Joint Surg. Am.*, vol. 72, no. 3, pp. 328–333, 1990.
- [14] R. P. Jackson and A. C. McManus, "Radiographic analysis of sagittal plane alignment and balance in standing volunteers and patients with low back pain matched for age, sex, and size. A prospective controlled clinical study," *Spine*, vol. 19, no. 14, pp. 1611–1618, Jul. 1994, doi: 10.1097/00007632-199407001-00010.
- [15] B. Ilharreborde *et al.*, "Angle measurement reproducibility using EOSthreedimensional reconstructions in adolescent idiopathic scoliosis treated by posterior instrumentation," *Spine*, vol. 36, no. 20, pp. E1306–E1313, 2011.
- [16] S. Ohrt-Nissen *et al.*, "Reproducibility of thoracic kyphosis measurements in patients with adolescent idiopathic scoliosis," *Scoliosis Spinal Disord.*, vol. 12, no. 1, pp. 1–8, 2017.

- [17] A. Simony, E. J. Hansen, S. B. Christensen, L. Y. Carreon, and M. O. Andersen, "Incidence of cancer in adolescent idiopathic scoliosis patients treated 25 years previously," *Eur. Spine J.*, vol. 25, no. 10, pp. 3366–3370, 2016.
- [18] W. Chen, E. H. M. Lou, P. Q. Zhang, L. H. Le, and D. Hill, "Reliability of assessing the coronal curvature of children with scoliosis by using ultrasound images," *J. Child. Orthop.*, vol. 7, no. 6, pp. 521–529, Dec. 2013, doi: 10.1007/s11832-013-0539-y.
- [19] M. Young, D. L. Hill, R. Zheng, and E. Lou, "Reliability and accuracy of ultrasound measurements with and without the aid of previous radiographs in adolescent idiopathic scoliosis (AIS)," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 24, no. 7, pp. 1427–1433, Jul. 2015, doi: 10.1007/s00586-015-3855-8.
- [20] T. K. Koo and M. Y. Li, "A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research," *J. Chiropr. Med.*, vol. 15, no. 2, pp. 155–163, Jun. 2016, doi: 10.1016/j.jcm.2016.02.012.
- [21] P. W. Stratford and C. H. Goldsmith, "Use of the standard error as a reliability index of interest: an applied example using elbow flexor strength data," *Phys. Ther.*, vol. 77, no. 7, pp. 745–750, 1997.
- [22] J. S. Lewis and R. E. Valentine, "Clinical measurement of the thoracic kyphosis. A study of the intra-rater reliability in subjects with and without shoulder pain," *BMC Musculoskelet. Disord.*, vol. 11, no. 1, p. 39, Mar. 2010, doi: 10.1186/1471-2474-11-39.
- [23] D. J. Kolessar, G. T. Stollsteimer, and R. R. Betz, "The value of the measurement from T5 to T12 as a screening tool in detecting abnormal kyphosis," *J. Spinal Disord.*, vol. 9, no. 3, pp. 220–222, Jun. 1996.

[24] F. Zaina, S. Donzelli, M. Lusini, and S. Negrini, "How to measure kyphosis in everyday clinical practice: a reliability study on different methods," *Stud. Health Technol. Inform.*, vol. 176, pp. 264–267, 2012.

5.3. Factors Influencing Kyphotic Angle Measurements on Ultrasound images

5.3.1. Introduction

Adolescent idiopathic scoliosis (AIS) is a 3D structural spinal disorder which includes lateral curvatures in the coronal plane, vertebral rotation in axial plane, and may have abnormal curvatures in the sagittal plane. The current imaging standard to measure the severity of scoliosis is radiography. Children with AIS usually take an X-ray every 6 months to monitor curve progression and review treatment decision [1, 2]. Traditionally, a posterioanterior (PA) and a lateral (LAT) radiograph are taken at the initial visit. Then, for children under observation, LAT radiographs are only requested when patients have abnormal sagittal curvature or complain of back pain. For brace candidates, LAT radiographs are only obtained at the in-brace follow-up. Since the EOS X-ray system (EOS Imaging, Paris, France), a low dose ionizing radiation system, has become more common, more clinicians prescribe PA and LAT radiographs at every follow-up clinic. Surgeons can thus review the 3D alignment of the spine. In addition to the Cobb angle measured on the LAT radiograph.

The KA value is especially important for children with AIS undergoing brace treatment or planning surgery. Orthotists and surgeons need to consider the sagittal profile while they design the brace or plan to operate the spine to ensure that the sagittal balance and alignment are optimized. Inappropriate treatment planning may reduce the treatment effectiveness or lead to undesirable changes in sagittal profile such as hypokyphosis or flattening of the thoracic kyphosis. Flattening of the spine increases the risk of death related to scoliosis because it may cause a reduction of the lung function [3]. In addition, the KA has also been considered as a parameter associated with curve progression. Studies reported that children who had curve progression showed a significantly lower KA versus the cases without progression [4, 5].

The KA can be measured using different top vertebral levels along the curve. Researchers have reported using T1-T12 [6, 7], T3-T12 [8], T4-T12 [9], T5-T12 [10] for KA measurements. The most common method to measure KA is the Cobb method and the angle is formed between the superior endplate of T1 and inferior endplate of T12 [11]. Carmen et al. used the Cobb method from T1 to T12 on 20 radiographs and the measurement error reported was 11 degrees [6]. Ohrt-Nissen et al. reported kyphotic angle measurements from different levels and showed a variability in intra- and inter-rater results from 8° to 13°[9] with the best intra- and inter-rater reproducibility for T4–T12 and T5–T12 levels.

Taking the LAT radiographs in addition to PA radiographs adds ionizing radiation to children with AIS. This is a worrisome issue since children with AIS have to undergo multiple visits and this excessive radiation may increase the incidence of cancer [12, 13]. Even though the EOS system uses lower ionizing radiation, the cumulative radiation is still a concern to children's families [12, 13]. Ultrasound (US) is a non-ionizing radiation imaging modality. An US imaging method had been introduced which could be used to measure the Cobb angle and the axial vertebral rotation reliably and accurately [14-16]. Recently, our group conducted a pilot study on 20 subjects to evaluate the intra-rater and inter-rater reliabilities for measuring the KA from US spinal images and the results showed good reliability ((ICC) [2, 1] \geq 0.84). However, this pilot study did not explore which factors can affect the accuracy of KA measurements. Therefore, this study aimed to further validate the accuracy and reliability of the US KA measurements based on a larger data set and to investigate how curve severity, X-ray imaging comparisons and curve type could influence the accuracy and reliability of KA measurements on US images in children with AIS.

5.3.2. Methodology

5.3.3. Clinical Participants

Sixty participants (53 F, 7 M, aged 13.4±1.6 years old) were consecutively recruited and scanned using our US scanner. The inclusion criteria were participants who a) were

diagnosed with AIS, b) had a major Cobb angle <55°, c) had no prior surgical treatment, d) had an out of brace radiograph at the recruitment clinic, and e) required PA and LAT radiographs. Among the 60 cases, 32/60 (53%) were undergoing brace treatment and 28/60 (47%) were under observation. The ethics approval was obtained from the local health research ethics board and all participants consented prior to participation.

5.3.4. Data Acquisition

There were 34/60 participants where the EOS X-ray system was used and 26/60 where a conventional X-ray system was used. All X-ray and US images were acquired in a standard standing position. During EOS acquisition, the PA and LAT radiographs were taken simultaneously, but for the conventional X-ray system, the LAT radiograph was taken independently after repositioning the patient. During the LAT radiographs, the participants were asked to stand in an upright position, without tilting the head and looking forward, hands placed at chest height against the opposite wall and without leaning forward or backward. US images were acquired using the Sonix TABLET which is equipped with a 128-element C5-2/60 GPS convex transducer (Analogic Ultrasound – BK Medical, Peabody, Massachusetts, USA). During the US data acquisition, participants stood in a frame devised to stabilize them in their natural position. They stood with feet shoulderwidth apart, looking straight and without tilting the head, with both hands held on the poles of the frame at around chest height to ensure the arms were not overlapping the spine. The operator then identified and marked the C7 and L5 vertebrae as the starting and end points of the scan. Both US and X-ray were performed in the same day, but two different rooms and the US images were acquired blind to the X-ray acquisition. One operator with 4+ years of experience performed the US scanning. Among the 60 participants, 40 had a second US acquisition at their subsequent follow-up visits within 7.0±3.1 months from their first visit. An in-house custom software was used to reconstruct the US images in coronal, axial and sagittal planes and measure the KA.

5.3.5. Data Measurement

Two raters, R1 and R2, were involved in this study. R1 had 20+ years of experience doing radiographic measurement and measured the KA on the LAT radiographs. R2 had 2+ years of experience measuring the KA on US images. Both raters measured all images twice one week apart to reduce memory bias. R1 measured the KA using the Cobb method using the T1/T12 and T4/T12 approaches. The procedures for using 3D US information to measure the KA measurements on US images are shown in Figure 5.4. The rater first identified the center of laminae (COL) for the vertebral levels from T1 to T5 and T10 to T12 on the PA images (Figure 5.4a). When the vertebra was selected on the PA image, the corresponding axial view was displayed. The operator then labelled the tip of the spinous process at that level (Figure 5.4b). Once all the center of laminae and spinous processes were labelled, the sagittal view with all labelled points were displayed automatically (Figure 5.4c). Labelling the T3 and T10 vertebrae helps confirm the curve trend of the spine on the sagittal view. The rater then drew a line manually to join the spinous processes of T1 and T2 and joining the spinous processes of T11 and T12 to measure the KA between these segments for the T1/T12 level measurement method (Figure 5.4c). The same procedure was repeated to measure the KA angle from T4/T12.



Figure 5.4 The process of measuring KA on US images. a) The center of the laminae of T1 to T5 and of T10 to T12 are first identified on the coronal US image. b) Then, the two centers of laminae and the tips of spinous process are identified in the axial view of the 3D US image at the level passing horizontally by both laminae. c) The spinous processes locations are then displayed in the sagittal view and used to measure the KA. The T1/T12 KA is obtained by measuring the angle between segments connecting T1T2 and T11/T12. The T4/T12 KA is obtained by measuring the angle between segments connecting T4T5 and T11/T12.

5.3.6. Data Analysis

The mean and standard deviation (SD) for both X-ray and ultrasound KA measurements at different occasions, measurement 1 and 2 (M1 and M2) for reanalysis of baseline images were reported. The intra-rater (M1 vs M2) and inter-method (X-ray vs US for both levels of measurements) reliabilities were calculated using the intraclass correlation coefficient ICC [2, 1] with a 2-way random model and absolute agreement with 95% confidence interval. The mean absolute difference (MAD) and standard error of measurements (SEM) between first and second measurements (M1 and M2) on each the US and the X-ray imaging modalities were reported. The Pearson correlation coefficient between US and X-ray was plotted using the Bland-Altman method. The independent Student t-test was

used to compare the X-ray and US measurements in terms of the KA severity. Also, the Chi square analysis was used to investigate the association of MAD thresholds (5° and 7°) with the X-ray acquisition systems and the curve type including number of scoliotic curves and apex location at upper thoracic (UT), main thoracic (MT), thoracolumbar (TL) and lumbar (L) regions. For the Chi square analysis, the X-ray acquisition systems coded as conventional (0) and EOS (1) and the number of coronal curves was coded to single (0) and double or multiple as (1). The number of curves and apex location were extracted from our scoliosis database which were measured by the clinical staff with 10+ years of experience. The alpha level < 0.05 was used to determine statistical significance.

5.3.7. Results

Ten of the 60 cases were excluded after comparison. Even though we tried to standardize the standing posture, the excluded 10 cases had adopted a standing posture that was significantly different between X-ray and US images because during x-ray acquisition the standardized instructions were not followed. Figure 5.5 shows an example in which the subject placed the hands above the head during the X-ray acquisition. Hence, fifty participants (43 F, 7 M, aged 13.4 \pm 1.9 years old) were analyzed. There were 83 coronal scoliosis curves, and the average of all curves was 23.1° \pm 8.2° (10° to 44°).



Figure 5.5 Comparison between the KA measurements on US and X-ray images for the same subject. The orange line shows the sagittal curve for a subject who has an arm flexion of around 60° during US data acquisition and >90° during X-ray data acquisition. a) The US KA are 48° and 38° when using T1/T12 and T4/T12 measurement, respectively. b) The corresponding X-ray KA values are 26° and 25°, respectively.

5.3.8. Reliability and Agreement Analysis

Table 5.5 shows the mean, standard deviation, MAD, SEM and the intra-rater reliability (ICC [2,1]) of KA measurements of the first and second intra-rater measurements when using the two different levels measurement method for the US and for the x-ray images, respectively.

The intra-rater MAD of X-ray KA was slightly bigger than the MAD of US KA, but there was no difference between level measurement method for a given imaging system. When comparing the different levels measurement method, T1/T12 showed a better intra-rater reliability. Overall, the intra-rater reliability results of KA measurements on both X-ray image types combined and on US images were excellent with ICC [2,1] >0.9, MAD \leq 3.1° and SEM \leq 0.79° when using the guidelines from Koo et al. [17]. (Table 5.5)

	Measureme nt Levels		D (range) Baseline	Intra-rater Reliability Results			
		M1 (°)	M2 (°)	MAD±SD (°)	SEM (°)	ICC [2,1] (95%CI, range)	
X-ray (combining	R1 (T1/T12)	32.6±11.0 (9 to 59)	32.7±10.0 (8 to 57)	2.5±2.2	0.52	0.95 (0.91 to 0.97)	
both systems)	R1 (T4/T12)	26.4±10.8 (2 to 51)	26.4±10.1 (2 to 50)	3.1±2.7	0.79	0.92 (0.86 to 0.95)	
Ultrasound	R2 (T1/T12)	35.9±9.6 (13 to 54)	35.5±9.6 (8 to 58)	2.2±1.9	0.43	0.95 (0.92 to 0.97)	
	R2 (T4/T12)	25.7±10.3 (1 to 51)	25.9±9.5 (4 to 46)	2.8±2.5	0.67	0.93 (0.88 to 0.96)	

Table 5.5 The average, standard deviation, SEM and intra-rater reliability of the KA measurements for both vertebral levels measurement methods on the X-ray and US images.

Table 5.6 shows the inter-method reliability in which measurements for both set of vertebral levels show good reliability (ICC [2,1] \ge 0.82). The maximum MAD±SD and the SEM of the differences between X-ray KA and US KA were 5.0°±3.8° and 1.6°, respectively.

Table 5.6 The comparison of inter-method differences and reliabilities at different occasions (M1, M2) and for different levels measurement methods as T1/T12 and T4/T12.

	Maagunamant Lavala		SEM (0)	ICC [2,1]
	Measurement Levels	MAD±SD (°)	SEM (°)	(95%CI, range)
T1/T12	US (M1) vs X-ray (M1)	4.8±3.2	1.30	0.84 (0.67 to 0.92)
	US (M2) vs X-ray (M2)	4.6±3.1	1.23	0.84 (0.69 to 0.91)
T4/T12	US (M1) vs X-ray (M1)	5.0±3.8	1.24	0.82 (0.71 to 0.90)
	US (M2) vs X-ray (M2)	4.6±3.7	1.56	0.82 (0.70 to 0.89)

Overall, the US KA had a linear correlation with X-ray KA for both different levels measurement method (T1/T12 and T4/T12). (Figure 5.6) The T1T12 X-ray KA from the EOS X-ray predicted the corresponding US KA ($R^2 = 0.79$) better than the conventional X-ray ($R^2 = 0.67$). For the EOS system, when the X-ray T1/T12 KA values were small (<15°), the variation was bigger.



(a)

Scatter plot of the T1T12 and T4T12 KA on US images versus EOS



Figure 5.6 Scatter plots of the US KA versus the X-ray KA from the second measurement occasion (M2) for both different levels measurement method (T1/T12 and T4/T12) when using data from comparisons to (a) the conventional X-ray system and (b) the EOS X-ray system.

Scatter plot of the T1T12 and T4T12 KA on US images versus conventional X-ray

Figure 5.7 shows the Bland-Altman plot of the inter-method measurement difference in KA versus the average of the US and X-ray measurements at the second occasion (M2) using the T1/T12 level measurement method. On average, there was a -2.4° bias, which means US measurements were larger than X-ray and there was no correlation between the severity of the KA and the differences in measurements. The 95% limits of agreement were from 12.4 to 7.4 degrees suggesting that X-ray measurements corresponding to US measurement can be expected to be 12.4 degrees smaller and up to 7.4 degrees bigger with 95% confidence.



Figure 5.7 Bland-Altman plot of the inter-method differences between the X-ray T1/T12 KA – US T1/T12 KA versus the average KA.

5.3.9. Factors Influencing the Accuracy and Reliability

Table 5.7 shows the association of measurement difference (MAD) with the X-ray KA severity, the X-ray systems and curve type for the T1/T12 level measurement method since it showed a better reliability. Two thresholds of MADs were tested and compared. The 5-degree threshold was selected based on the maximum MAD observed from inter-method

analysis (Table 5.7a). Also, the 7-degree threshold was selected based on the minimum threshold reported on inter-rater reliability analysis on KA from literature [18] (Table 5.7b). Table 5.8 shows the distribution of the curve type based on apex location for both thresholds.

Table 5.7 Test of association between the magnitude of the absolute measurement difference and the KA severity, the X-ray system and the curve type for the T1/T12 measurements using a) a 5° MAD threshold, and b) for a 7° MAD threshold.

T1/T12 Level of Measurements		MAD (°) (between X-ray (M2) and of US KA (M2) ≤5.0 >5.0 (n=27) (n=23)		Statistic value	P value	
X-ray	KA Severity	Mean±SD	33.7±8.9	31.4±11.2	t=0.80	0.42
X-ray	Conventional (n=23)	Number	13 (57%)	10 (43%)	Chi square=0.10	0.74
System	EOS (n=27)	n (%)	14 (63%)	13 (37%)		
Curve type —		Single (n=19)	10 (53%)	9 (47%)	Chi	0.07
		Others (n=31)	17 (55%)	14 (45%)	square=0.02	0.87

(a)

T1/T12 Level of Measurements			MAD (°) (X-ray KA - US KA) (M2)		Statistic	Desta
			≤7.0 (n=40)	>7.0 (n=10)	value	P value
X-ray KA		Mean±SD	33.9±9.5	27.8±11.1	t=1.74	0.08
X-ray	Conventional (n=23)	Number n (%)	19 (83%)	4 (17%)	Chi	0.67
System	EOS (n=27)		21 (78%)	6 (22%)	square=0.18	
Curve type		Single (n=19)	15 (79%)	4 (21%)	Chi	0.88
		Others (n=31)	25 (81%)	6 (45%)	square=0.02	

(b)

MAD= mean absolute difference, KA= kyphotic angle, M2= measurement 2, US= ultrasound

Kyphotic Angle Severity

The average X-ray KA value did not differ significantly between patients with large intermethod differences or not for both thresholds. (Table 5.7, p = 0.42 vs. p = 0.08.)

X-ray System

The number of cases with a MAD \leq 5° when comparing the US KA to the conventional or the EOS X-ray were 57% vs 63%. However, the number of cases with a MAD \leq 7° when comparing the US KA to the conventional or the EOS X-ray were 83% vs 78%, respectively. There was no statically significant difference in frequency of error above and below either large MAD threshold based on the X-ray acquisition system used in the comparison with US KA (Table 5.7).

Curve Type

There also was no significant difference between the number with single or more than single curve among those with larger errors based on MAD thresholds of 5° and 7° when comparing US KA to Xray KA, with 53% vs 55% and 79% vs 81%, respectively. (Table 5.8) Among 83 scoliotic curves in total, 50/83 (60%) were within 5° threshold, but 68/83 (82%) were within 7° threshold. For MT curve, accuracy increased significantly when comparing the 5° to 7° thresholds while the number of cases increased from 22/33 (67%) to 30/33 (90%). Similar trend occurred on the TL/L, it increased from 24/43 (56%) to 35/43 (81%).

Curve type (Total n = 83)	$MAD \le 5.0^{\circ}$ $(n = 50)$	MAD > 5.0° (n = 33)	$MAD \le 7.0^{\circ}$ $(n = 68)$	MAD > 7.0° (n = 15)
UT (n = 7)	4 (8%)	3 (9%)	3 (4%)	4 (27%)
MT (n = 33)	22 (44%)	11 (33%)	30 (44%)	3 (20%)
TL/L (n = 43)	24 (48%)	19 (58%)	35 (51%)	8 (53%)

Table 5.8 The association between the magnitude of the absolute errors of measurements $(5^{\circ} \text{ and } 7^{\circ} \text{ thresholds})$ and the apex location of the scoliotic curves.

Abbreviations: UT= Upper Thoracic, MT= Main Thoracic, TL= Thoracolumbar, L= Lumbar

5.3.10. Discussion

The current gold standard to measure the KA is to use the Cobb method on LAT radiograph. Apart from the ionizing radiation issue, the X-ray modality is a projection

method which means when axial vertebral rotation exists, it may introduce error in KA measurements. Also, if the pelvis rotates in the axial plane while standing it may add errors on the projection. Therefore, the true KA measurement may not be obtained. In contrast, during the US imaging acquisition, the operator captures the information along the curve. The sagittal US images can capture the vertebral rotation aspect information. Hence, the X-ray KA and US KA may have more discrepancy when vertebral rotation is significant.

The X-ray KA and US KA showed higher correlation when using measurements from the EOS X-ray than the conventional X-ray. This result may be explained by better image quality from the EOS system than the conventional X-ray. The conventional X-ray system uses divergent X-ray beam for exposure in which the X-ray photons need to travel longer distance to expose the upper level of vertebra. This may lead to lower penetration energy on the upper level, which affects the image quality [19]. In addition, T1/T12 level measurement method may have higher correlation because it has less chance to have vertebral rotation on the T1 level than T4.

In this study, 10 cases we excluded due to posture variations. The MAD between X-ray KA and US KA measurements ranged from 11 to 21 degrees in these 10 cases. The major reason was the arm flexion. Using an ImageJ (NIH, USA) angle measurement tool, the measured arm flexion on LAT radiographs ranged from 20° to 162° for these participants. Normally, if the subject follows the X-ray technician's instructions, the arm flexion should be approximately 60° . Of these 10 cases, 8 had arm flexion $>60^{\circ}$, one $<40^{\circ}$ and one within 40 to 60 degrees. In Figure 5.5, the arm flexion for that subject is over 90° during the X-ray acquisition and was 60° during the US acquisition, respectively. This patient presented a 22° difference on the KA measurement. Based on the literature, the arm position could affect the sagittal alignment [20, 21]. In addition, among these 10 cases, differences in head position as well as pelvic tilt were noticed. The hyper extension or flexion of the head may affect the KA especially when using the T1/T12 level measurement method. Among the 10 excluded cases, 4 also showed hyper extension of the head. Also, since there is no support to stabilize the hips and pelvis while acquiring X-ray images, children may lean their abdomen forward which make their lower body closer to the opposite wall. This could
affect the bottom endplates of T11 or T12; hence affecting the KA measurements. This was also observed in 2/10 cases. Therefore, standardization of the posture during data acquisition is important. Overall, the US data acquisition seems more consistent while all participants stood in a frame that supports the shoulder and hips and while offering support for the arms flexed at an angle between 40° to 60°. The same operator scanned all the participants, and she ensured all patients stood properly. However, X-ray data acquisitions were performed by a number of technicians. An unrestricted technician might allow participants with slightly more variable positioning.

Among the 50 cases analysed, 40 had a second US data at the consecutive follow-up clinic . Figure 5.8 shows a comparison of US KA measurements (T1/T12 method) with 23 under brace treatment and 17 under observation for both measurements. The participants with bracing have lower mean value of the KA versus the ones who were under observation; however, the difference is not statistically significant (1st visit - brace vs observation: $33.2^{\circ}\pm9.9^{\circ}$ vs 38.0 ± 10.0 ; 2^{nd} visit $33.5^{\circ}\pm11.5^{\circ}$ vs $37.5^{\circ}\pm11.4^{\circ}$). There is no significant change on the KA values between the first and the follow-up visits. A longer follow-up period may require.



Figure 5.8 Comparison of the US KA (M2) between the observation and bracing group (n=40) as well as the changes of the KA between the first and second US visits for both groups

The intra-rater reliability and measurement error of the X-ray KA (ICC $[2,1] \ge 0.92 \le 3.1^{\circ}\pm 2.7^{\circ}$) were better in this study than others [6,18]. The rater experience is important. This is the only study to compare the US KA with X-ray KA on a large data set. The ICC [2,1] of the US KA and measurement error was slightly lower than the X-ray method, but

the US rater had only 2+ year of experience. However, the result is still compatible to other X-ray measurement studies in which the intra-rater measurement error ranged between 6° to 11° [6, 18] and inter-rater errors ranged from 7° to 13° [9, 18]. Approximately 80% of the US KA measurements were within 7° differences with the X-ray KA.

In term of limitation, the US scan is operator dependent. A good image quality must be acquired to ensure an acceptable sagittal reconstruction of the spine. Moreover, US data acquisition usually takes 30 seconds to 1 minute to scan a spine, which is longer than the X-ray acquisition (6 seconds). This may increase chances of body movements and may affect the body alignment. Also, a test-retest analysis could be performed to investigate the repeatability of the KA measurements.

5.3.11. Conclusions

US imaging is a reliable and safe method to capture KA measurements from a sagittal image. The US KA measurement is reliable and accurate enough and can be used in every clinic. The standing posture must be standardized and enforced while acquiring images no matter which imaging modality is used.

5.4. Summary

The sagittal X-ray images are not often acquired due to concerns related with radiation. This chapter reported a new US method to measure kyphotic angle on US sagittal images. The results showed that KA can be measured reliably on US images and the maximum measurement differences was 4.0° when compared with X-ray KA. Furthermore, the intrarater and inter-method reliability coefficients were high (ICC > 0.84) when using the T1/T12 method to measure the KA, and 80% of KA measurements were within the clinical acceptance error (7°). None of the 3 investigated factors showed a significant influence on the measurement error. However, the standing posture during image acquisition was the only factor which might influence the measurement difference of the KA measurements. Since KA can be measured reliably, this parameter will be considered as a potential prognostic factor for prediction of curve progression.

References

- Negrini, S., Hresko, T. M., O'brien, J. P., et al. (2015) Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee. *Scoliosis*, 10(1), 8.
- Kotwicki, T (2008) Evaluation of scoliosis today: examination, X-rays and beyond. *Disability and rehabilitation*, 30(10), 742-751.
- Winter R, Lovell WW, Moe J. (1975). Excessive thoracic lordosis and loss of pulmonary function in patients with idiopathic scoliosis. *J Bone Joint Surg Am*. 57:972–977.
- Nault, M. L., Mac-Thiong, J. M., Roy-Beaudry, et al. (2013). Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis. *Journal of Pediatric Orthopaedics*, 33(6), 618-623.
- Nault, M. L., Mac-Thiong, J. M., Roy-Beaudry, M., et al. (2014). Threedimensional spinal morphology can differentiate between progressive and nonprogressive patients with adolescent idiopathic scoliosis at the initial presentation: a prospective study. *Spine*, 39(10), E601.
- D. L. Carman, R. H. Browne, and J. G. Birch (1990) "Measurement of scoliosis and kyphosis radiographs. Intraobserver and interobserver variation.," *J. Bone Joint Surg. Am.*, vol. 72, no. 3, pp. 328–333.
- R. P. Jackson, T. Kanemura, N. Kawakami, and C. Hales (2000) "Lumbopelvic lordosis and pelvic balance on repeated standing lateral radiographs of adult volunteers and untreated patients with constant low back pain," *Spine*, vol. 25, no. 5, pp. 575–586.
- 8. K. P. Singer, S. J. Edmondston, R. E. Day, and W. H. Breidahl (1994) "Computerassisted curvature assessment and Cobb angle determination of the thoracic

kyphosis. An in vivo and in vitro comparison," *Spine*, vol. 19, no. 12, pp. 1381–1384, Jun. 1994, doi: 10.1097/00007632-199406000-00012.

- Ohrt-Nissen, S., Cheung, J. P. Y., Hallager, D. W.(2017) "Reproducibility of thoracic kyphosis measurements in patients with adolescent idiopathic scoliosis," *Scoliosis Spinal Disord.*, vol. 12, no. 1, pp. 1–8.
- D. J. Kolessar, G. T. Stollsteimer, and R. R. Betz (1996) "The value of the measurement from T5 to T12 as a screening tool in detecting abnormal kyphosis," *J. Spinal Disord.*, vol. 9, no. 3, pp. 220–222, Jun.
- Tran, T. H., Wing, D., Davis, A., Bergstrom, J., Schousboe, J. T., Nichols, J. F., et al. (2016). Correlations among four measures of thoracic kyphosis in older adults. *Osteoporosis International*, 27(3), 1255-1259.
- T. D. Luo, A. A. Stans, B. A. Schueler, and A. N. Larson (2015) "Cumulative Radiation Exposure With EOS Imaging Compared With Standard Spine Radiographs," *Spine Deform.*, vol. 3, no. 2, pp. 144–150, Mar. doi: 10.1016/j.jspd.2014.09.049.
- A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras (1996) "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine*, vol. 21, no. 13, pp. 1540–1547; discussion 1548, Jul. doi: 10.1097/00007632-199607010-00011.
- 14. Chen, W., Lou, E. H., Zhang, P. Q., et al. (2013). Reliability of assessing the coronal curvature of children with scoliosis by using ultrasound images. *Journal of children's orthopaedics*, 7(6), 521-529.
- 15. Zheng, Y. P., Lee, T. T. Y., Lai, K. K. L., et al (2016). A reliability and validity study for Scolioscan: a radiation-free scoliosis assessment system using 3D ultrasound imaging. *Scoliosis and spinal disorders*, 11(1), 1-15.

- Chen, W., Le, L. H., & Lou, E. H. (2016). Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study. *European Spine Journal*, 25(10), 3265-3273.
- Koo, T. K., & Li, M. Y. (2016). A guideline of selecting and reporting intraclass correlation coefficients for reliability research. Journal of chiropractic medicine, 15(2), 155-163.
- Ilharreborde B, Steffen JS, Nectoux E, et al. (2011) Angle measurement reproducibility using EOS three-dimensional reconstructions in adolescent idiopathic scoliosis treated by posterior instrumentation. *Spine* 36(20):E1306-1313. https://doi.org/10.1097/BRS.0b013e3182293548.
- Melhem, E., Assi, A., El Rachkidi, R., & Ghanem, I. (2016). EOS® biplanar X-ray imaging: concept, developments, benefits, and limitations. *Journal of children's orthopaedics*, 10(1), 1-14.
- 20. Kaneko, K., Aota, Y., Sekiya, T., Yamada, K., & Saito, T. (2016). Validation study of arm positions for evaluation of global spinal balance in EOS imaging. *European Journal of Orthopaedic Surgery & Traumatology*, 26(7), 725-733.
- Horton, W. C., Brown, C. W., Bridwell, K. H., Glassman, S. D., Suk, S. I., & Cha,
 C. W. (2005). Is there an optimal patient stance for obtaining a lateral 36" radiograph?: a critical comparison of three techniques. *Spine*, *30*(4), 427-433.

Chapter 6: Reliability and Factors Influencing the Measurement Variation of Ultrasound Axial Vertebral Rotation (AVR)

6.1. Overview

This chapter reports a study to investigate the intra-rater, inter-method reliability coefficients and factors influencing the measurement variation of ultrasound (US) axial vertebral rotation (AVR) when compares with Stokes' method on radiographs in AIS. Section 6.2 is the manuscript entitled "Investigation of Factors Influencing the Reliability and Measurement Variation of Ultrasound Axial Vertebral Rotation on Adolescents with Idiopathic Scoliosis" which has been submitted to "The Spine Journal". Section 6.6 provides a summary of this chapter.

6.2. Introduction

Idiopathic scoliosis is the most common type of scoliosis among adolescents [1]. The standard of care for scoliosis is taking radiographs every 6 months to evaluate this spine disorder. As per Scoliosis Research Society (SRS) guidelines, the three treatment options available for the adolescent idiopathic scoliosis (AIS) are observation, bracing and surgical interventions. The treatment option is selected based on the severity of curvature, the maturity status of the patients, the patients' self-image and the risk of curve progression [2], [3]. However, to recommend an appropriate management, the spinal curvature in coronal, axial and sagittal planes must be reviewed, as AIS is a three-dimensional (3D) spinal disorder.

Besides the Cobb angle in the coronal plane, axial vertebral rotation (AVR) is another important curve characteristics on AIS which is defined as the rotation of the vertebra along the longitudinal axis and projected in the axial plane [4]. A spinal curvature with large AVR could cause the thoracic rib hump which is an asymmetry due to prominence of one side of rib cage [5]. Also, AVR is considered as a potential predictor that could be used to predict curve progression [6].

A few 2D and 3D methods have been suggested to measure the AVR [7]–[13]. The challenge about 2D methods is that the AVR measurements are not directly measured on axial plane which may affect the accuracy of measurements. Of the 3D methods, computed tomography (CT) [11], magnetic resonance imaging (MRI) [14], and the EOS 3D reconstruction [15] modalities have also been suggested to measure the AVR. The AVR measurements are performed in axial plane for CT and MRI; however, there are some limitations about these two methods. Both of imaging methods require a long data acquisition time and are relatively costly when comparing with radiography. They acquire the images in supine position which may affect the curvature characteristics, hence influencing the rotation severity. Lastly, the CT method, uses even higher level of radiation versus conventional X-ray which is not desirable [16]. Another 3D method is using the 3D EOS system which applies low level of radiation and capture a posteroanterior (PA) and lateral (LAT) spinal radiographs simultaneously. However, the EOS 3D reconstruction of spine is a timely process which could last up to an hour for severe idiopathic scoliosis cases [17]. Also, for EOS 3D reconstruction, two acquisitions including PA and LAT are required. The accumulated radiation with both PA and LAT radiographs may increase the health risk to these children. Therefore, most of the orthopedic surgeons use the landmark features to estimate the AVR without actual measurements.

Another option that can measure AVR directly and fast is to acquire spinal ultrasonography. Suzuki et al. [18] was the first group using an ultrasound system combined with an inclinometer to measure the AVR in a prone posture. However, no reliability analysis of the AVR was reported from their study. Burwell *et al.* was another group [19] who applied ultrasound to measure rib rotation and lamina rotation in the prone position. The reduction of curve severity due to the prone posture may happen. Recently, Vo et al. used a freehand 3D ultrasound system to scan children with AIS in upright position. A single scan of ultrasound from top to bottom along the lateral spinal curvature on the coronal view allows a 3D reconstruction of spine. An *in-vitro* study was conducted to investigate the reliability of the 3D ultrasound method for AVR measurements using laminae. The results showed an excellent intra- and inter-rater reliabilities. However, no

clinical participants and no comparison with X-ray measurements were included for this study [20]. Chen et al reported on the center of lamina (COL) method to measure AVR in both in-vitro and in-vivo studies [21], the results showed high intra-rater and inter-rater reliability coefficients for both *in-vitro* and *in-vivo* studies (ICCs > 0.91, MAD < 1.4°) [21]. They also compared their measurements with Stokes' method and showed a good intermethod reliability for the *in-vitro* study (ICC = 0.84-0.85) with the mean absolute difference (MAD) within 4.5°-5.0°; while the agreement for the *in-vivo* study was poor to moderate (ICC = 0.49-0.54, MAD = $2.7^{\circ}-3.5^{\circ}$). Another study compared and validated the AVR measurements using 3D US measurements COL method and an MRI method. Even though their US measurements were reliable, both MRI and US scans were acquired in supine position [22]. These two studies [21, 22] only focused on the apical AVR measurements and did not investigate the US measurement reliability on the entire spine. Also, no investigations were reported on which factors might influence the US AVR measurements. Therefore, this study aimed to determine the reliability and variation of US AVR measurements at vertebrae levels T4 to L4 and what factors might affect the measurement.

6.3. Methods

6.3.1. Clinical Participants

Twenty-four participants (20 F, 4 M, aged 13.6 ± 1.6 years old) were randomly selected from our US database. The ethics approval was obtained from the local health research ethics board and all participants consented prior to participation. The inclusion criteria were participants who were diagnosed with AIS, had a Cobb angle <55 degree, and had no prior surgical treatments.

6.3.2. Data Acquisition

All participants were scanned by an US imager in addition to the X-ray acquisition as the standard of care. The US images were acquired using the Sonix TABLET medical US

system equipped with a 128-element C5-2/60 GPS convex transducer (Analogic Ultrasound – BK Medical, Peabody, Massachusetts, USA). The X-ray data acquisition was performed while participants were positioned in an upright standing position. For the conventional X-ray, the chest was against the metal plate of X-ray machine and the hands grabbed on the side bars at a height around chest level. For the EOS system, the patients placed their hands at chest level and touching in front of them. During the US data acquisition, participants were asked to stand in an upright position, look straight and no tilt head in a frame devised to stabilize them. Both hands grabbed on the frame poles around chest height. One operator with 4+ years of experience performed the scanning. An inhouse software was used to reconstruct the US images. The coronal and sagittal views looked like PA and LAT radiographs. The axial view showed a single US B-mode image of a slice of vertebra on transverse view.

6.3.3. Raters

Rater 1 who had over 20+ years of scoliosis experience measured the AVR on the radiographs from T4 to L4 vertebrae using Stokes' method [5] (X-ray AVR). Rater 2 who had around 5 years of US measurement experience measured the AVR using COL [21] at the same level on the US images (US AVR). Rater 1 only measured the AVR once since he had demonstrated good reliability and repeatability for AVR measurements in the study [21]. Rater 2 measured the US images twice in two measurement occasions (M1 and M2) with one week apart to reduce the memory bias.

6.3.4. AVR Measurements

Both measurements methods were performed using a custom semi-automatic software. On a coronal radiograph, R1 first identified and outlined the pedicles on a selected vertebra. The software then automatically estimated the centers of the pedicles. R1 then placed a line across the minimum distance of the vertebral body and selected the vertebral level. The program then calculated the AVR automatically based on Stokes' method [23]. For the US measurement, R2 first identified the center of laminae at each vertebral level on the coronal view as shown in Figure 6.1a. The rater then selected the vertebral level to perform a fine adjustment of the COL locations in the corresponding transverse view. Once selected, the program would display the axial view (Figure 6.1b). Rater 2 could make COL adjustment in the axial view; afterwards, the software would then display the AVR for each level.



Figure 6.1 The process of AVR measurement starting from coronal to axial US images. a) The center of laminae are identified on the coronal view, and at each of the vertebra levels (T4 to L4) a line is automatically drawn and joined the COL. b) On the axial view, the selected vertebra (T6), the center of the laminae can be fine-tuned, and the AVR angle forms between the line which joins the COL and a horizontal reference line (dashed line).

6.3.5. Data Analysis

The mean and standard deviation (SD) for each set of measurement including X-ray AVR and US AVR (M1 and M2) were reported. The intra-rater reliability of the US AVR and the inter-method reliability coefficients were analyzed using the intraclass correlation coefficient ICC [2,1] with a 2-way random model and absolute agreement with 95% confidence interval. The mean absolute difference (MAD) and standard error of

measurements (SEM) between X-ray AVR and US AVR measurements were reported. The agreement between two methods measurements was also plotted using the Bland-Altman's method. An independent Student t-test was performed to compare the difference between the AVR measurements from different subgroups defined by factors of interest: a) the apical region (inferior apical, apical and superior apical vertebral levels) versus the non-apical region, b) the curve severity (Cobb angle between 10° to 25° (mild), and Cobb angle between 25° to 45°(moderate)), and c) curve type either upper thoracic (UT)/main thoracic (MT) and thoracolumbar (TL)/lumbar (L). Also, a one-way ANOVA test was performed to compare the effect of AVR severity: mild (0° to 5°), moderate (6° to 10°) and severe (>10°) on differences of measurements between the US and x-ray imaging methods, followed by post-hoc Bonferroni test to investigate the differences between these three categories. An alpha level <0.05 was considered as a statically significant difference.

6.3.6. Results

Forty-two curves were identified in these 24 radiographs, and the average of the Cobb angle was $24.4^{\circ}\pm9.0^{\circ}$ (10° to 43°). In total, 312 (24 spine x 13) AVR were measured in which 125 were within the apical regions and 187 were outside. For one case, the apex level was at L4 therefore, only two vertebrae including the superior and apical vertebral levels were available.

Table 6.1 shows the X-ray and the US AVR measurements including the mean±SD from both raters, the MAD, SEM and intra-rater reliability between M1 and M2. The intra-rater reliability on US measurements was excellent with an ICC [2,1] = 0.93, the MAD was $1.7^{\circ}\pm1.7^{\circ}$ and the SEM was 0.46° .

Table 6.1 The average, standard deviation, SEM and intra-rater results of the AVR measurements on both X-ray and US images

Measurements	M1 (°)	M2 (°)	MAD±SD (°)	SEM (°)	ICC [2,1] (95%CI, range)	
X-ray	-0.4±4.3	N/A				
Measurements	(-17.0 to 12.1)					
US	1.1±6.5	1.1±6.0	1.7±1.7	0.46	0.93	
Measurements	(-14.8 to 20.5)	(-14.8 to 17.3)	1./±1./	0.40	(0.91 to 0.94)	

MAD: Mean absolute difference, SD: Standard deviation, SEM: Standard error of measurement, ICC: Intra-class correlation coefficient, US: Ultrasound

Table 6.2 shows the MAD, SEM and inter-method reliability results of the AVR measurements between X-ray and US methods. The MADs were calculated based on differences between X-ray and US measurements and the results on both measurement occasions were consistent (< 4.6°). The SEM was less than 2.4° and the ICC [2,1] was poor to moderate (ICC [2,1] \ge 0.49) as per interpretation guidelines provided by Koo et al [24].

Table 6.2 The MAD, SEM and inter-method reliability results of the AVR measurements

 between X-ray and US methods

Measurements	MAD±SD (°)	SEM (°)	ICC [2,1] (95%CI, range)
X-ray vs US (M1)	4.6±3.3	2.4	0.49 (0.38 to 0.57)
X-ray vs US (M2)	4.3±3.2	2.2	0.50 (0.39 to 0.59)

Figure 6.2 shows the Bland-Altman plot of the AVR difference of the inter-method measurement versus the average of US (M2) and X-ray measurements. Majority of the measurements 278/312 (89%) were within the lower and upper boundaries of 95% confidence interval. A slight positive correlation (R^2 =0.13) was observed between AVR severity and the differences of the measurement indicating larger AVR severity, larger variations of the measurements.



Average of X-ray and US AVR measurements

Figure 6.2 Bland-Altman plot of the differences between the X-ray and US AVR versus the average of the AVR from the X-ray and US(M2) measurements

6.3.1. Factors Influencing the AVR Measurement Difference

Apical vs Non-Apical Regions

The average of AVR measurement difference between US (M2) and radiographic methods at both apical and non-apical regions were compared, and the results were statistically significant (p<0.05). The mean of AVR measurement difference at the apical region was larger than non-apical region ($4.0^{\circ}\pm3.3^{\circ}$ vs $2.9^{\circ}\pm2.2^{\circ}$) (Table 6.3).

Curve Severity

Table 6.3 also shows the MAD \pm SD of inter-method measurements was not statically significant different between mild and moderate groups (p=.97). The average MAD in both curve severity groups was approximately 4°.

Curve Type

There was no statically significant difference (p=.29) between the inter-method measurement difference between UT/MT and TL/L curve types (Table 6.3). In general, the inter-method measurement difference TL/L was slightly smaller 3.7° vs 4.3° when compared to the UT/MT curves.

Table 6.3 Comparison of the AVR measurement difference (MAD) in terms of region	of
measurement, curve severity and curve type	

Average of difference between X-ray and US (M2)	Apical Regions (N=125)	Non-apical Regions (N=187)	t test	P value
(MAD±SD)	4.0±3.3	2.9±2.2	-3.4	.001
MAD±SD at apical region	Mild curve (n=78)	Moderate curve (n=47)	0.02	.97
in id=52 at apical region	4.1±3.2	4.0±3.2	0.02	
MAD±SD at apical region	UT/MT (n=60)	TL/L (n=65)	1.0	.29
MAD±SD at apical region	4.3±3.7	3.7±3.0	1.0	

MAD: Mean absolute difference, SD: Standard deviation, US: Ultrasound, M2: Measurement 2, UT: Upper thoracic, MT: Main thoracic, TL: Thoracolumbar, L: Lumbar

AVR Severity

Table 6.4 shows the average of MAD among the three AVR severity categories for the apical region. As the results show, there is a significant difference among the three categories. The largest mean of measurement difference between X-ray and US was observed for the largest severity category (X-ray AVR >10°). Further analysis showed that mean of MAD±SD for the largest severity category was significantly larger versus the second category (p<0.05), while it was not significant when compared with first category.

AVR Severity (°)	MAD±SD (°)	F value	P value	Comparisons	P value
Mild: (0 to 5) (n=88)	4.1±3.0			1 vs 2	.35
Moderate: (6 to 10) (n=27)	3.0±3.2	4.03	.02	1 vs 3	0.16
Severe: >10 (n=10)	6.3±4.2			2 vs 3	.017

 Table 6.4 Comparison between the average of the inter-method difference in AVR severity

AVR: Axial vertebral rotation

6.4. Discussion

The intra-rater and the inter-method reliability coefficient ICC [2,1] of the US AVR measurement in the current study was similar to the results from the previous study [21]. The US AVR can be reliably measured (ICC [2,1] > 0.9), but it is in the poor to moderate range (ICC[2,1] = 0.49 to 0.5) when compared with Stokes' method. This means that the US method cannot directly compare with the X-ray estimation method. As Stokes' method uses the pedicle shadows projected on the 2D X-ray image, the AVR value is always underestimated using Stokes' method. The overall US AVR is 4.45° higher than Stokes' method.

The average of AVR measurement difference between X-ray and US method at the apical region was statistically larger than non-apical region. However, there were 88/125 (70%) of the AVR differences are within 5°, the majority is still within the measurement error. When the X-ray AVR is larger, the more difference occurs. This phenomenon may either due to the underestimation of the X-ray AVR from Stokes' method or the poor US image quality which makes the US AVR inaccurate. When a large AVR exists, it causes more

uneven back surface which makes the US probe more difficult to contact the surface of skin properly. Since the majority of the X-ray AVR is within 5° at the apical region, the overall inter-method reliability (ICC [2,1]) for apical and non-apical region groups was 0.6 and 0.41, respectively.

There was no correlation between the curve severity and the difference of the X-ray AVR and US AVR measurements (Figure 6.3). It may be due to the mild curve is more dominant.



Figure 6.3 Scatter plot of the inter-method measurement difference and curve severity

Furthermore, the MAD of inter-method measurements was not significantly different between UT/MT and TL/L curve types, the error of measurements at UT/MT was almost the same as TL/L region (0.6° different). Although the COL is more difficult to identify on the lumbar region in the coronal view [25], using the axial view can ensure the rater can determine the COL more precisely. Hence, the advantage of using the US method is the 3D information. Also, another advantage of using US was the time required to obtain the AVR measurements using the COL method versus Stokes' method. The COL method was significantly faster than Stokes' method (5 seconds versus 45 seconds per vertebra).

In terms of limitations, the US data acquisition could take more time particularly for the cases with large AVR. To scan cases with large AVR, the operator needs to make sure there is an appropriate contact for imaging of the laminae. Longer scan could increase chances of body movements which may affect the curvature characteristics. Also, the standing posture instructions is very important during data acquisition particularly for X-ray imaging. For x-ray data acquisition, if the patient's body tilts, it may cause other vertebra features such as spinous process to overlap pedicles and project over them which could make identifications of the pedicles difficult.

To further verify the accuracy of the US COL method for the AVR measurements, the AVR acquired from the EOS 3D reconstruction measurements in a standing position should be conducted.

6.5. Conclusion

The results of the study showed an excellent US intra-rater reliability with the average of difference 1.7° and a moderate inter-method reliability of ICC [2,1] \geq 0.49. Overall, 89% of the measurements were within the lower and upper boundaries of 95% confidence interval. The error measurement was larger when AVR severity was larger. The curve severity and curve type were not factors influencing the difference between the X-ray AVR and US AVR measurements. The apical region and AVR severity were the influencing factors.

6.6. Summary

This chapter shows a high intra-rater reliability of the AVR measurements on US images. However, the inter-method reported a moderate reliability. The discrepancy between the two methods could be related to AVR measurement on a 2D X-ray using Stokes' method versus COL method. The US AVR is measured directly in the axial plane whereas the Xray AVR is measured on a projected 2D radiographic image. Also, the inter-method AVR measurement difference at the apical region was also significantly larger than non-apical. The inter-method measurement difference was significantly larger when AVR severity was >10° versus 6° to 10° at the apical region. As a result, the US AVR can be investigated as a potential prognostic factor of progression for model development.

References

- M. N. Choudhry, Z. Ahmad, and R. Verma, "Adolescent idiopathic scoliosis," *The open orthopaedics journal*, vol. 10, p. 143, 2016.
- [2] J. A. Janicki and B. Alman, "Scoliosis: Review of diagnosis and treatment," *Paediatrics & child health*, vol. 12, no. 9, pp. 771–776, 2007.
- S. L. Weinstein, "The natural history of adolescent idiopathic scoliosis," *Journal of Pediatric Orthopaedics*, vol. 39, pp. S44–S46, 2019.
- [4] T. Vrtovec, F. Pernuš, and B. Likar, "A review of methods for quantitative evaluation of axial vertebral rotation," *European Spine Journal*, vol. 18, no. 8, pp. 1079–1090, 2009.
- [5] I. A. Stokes, "Axial rotation component of thoracic scoliosis," *Journal of orthopaedic research*, vol. 7, no. 5, pp. 702–708, 1989.
- [6] M.-L. Nault *et al.*, "Three-Dimensional Spinal Morphology Can Differentiate Between Progressive and Nonprogressive Patients With Adolescent Idiopathic Scoliosis at the Initial Presentation," *Spine (Phila Pa 1976)*, vol. 39, no. 10, pp. E601–E606, May 2014, doi: 10.1097/BRS.00000000000284.
- [7] J. R. Cobb, "Outline for the study of scoliosis," *Instr Course Lect AAOS*, vol. 5, pp. 261–275, 1948.
- [8] R. Perdriolle and J. Vidal, "Thoracic idiopathic scoliosis curve evolution and prognosis," *Spine (Phila Pa 1976)*, vol. 10, no. 9, pp. 785–791, Nov. 1985, doi: 10.1097/00007632-198511000-00001.
- [9] I. A. Stokes, L. C. Bigalow, and M. S. Moreland, "Measurement of axial rotation of vertebrae in scoliosis.," *Spine*, vol. 11, no. 3, pp. 213–218, 1986.

- [10] C. L. Nash and J. H. Moe, "A study of vertebral rotation," J Bone Joint Surg Am, vol. 51, no. 2, pp. 223–229, 1969.
- [11] E. K. Ho, S. S. Upadhyay, F. L. Chan, L. C. Hsu, and J. C. Leong, "New methods of measuring vertebral rotation from computed tomographic scans. An intraobserver and interobserver study on girls with scoliosis," *Spine (Phila Pa 1976)*, vol. 18, no. 9, pp. 1173–1177, Jul. 1993, doi: 10.1097/00007632-199307000-00008.
- [12] D. Birchall, D. G. Hughes, J. Hindle, L. Robinson, and J. B. Williamson, "Measurement of vertebral rotation in adolescent idiopathic scoliosis using threedimensional magnetic resonance imaging," *Spine (Phila Pa 1976)*, vol. 22, no. 20, pp. 2403–2407, Oct. 1997, doi: 10.1097/00007632-199710150-00016.
- [13] L. Boyer, J. Shen, S. Parent, S. Kadoury, and C.-E. Aubin, "Accuracy and precision of seven radiography-based measurement methods of vertebral axial rotation in adolescent idiopathic scoliosis," *Spine deformity*, vol. 6, no. 4, pp. 351–357, 2018.
- [14] D. Birchall, D. Hughes, B. Gregson, and B. Williamson, "Demonstration of vertebral and disc mechanical torsion in adolescent idiopathic scoliosis using threedimensional MR imaging," *European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society*, vol. 14, pp. 123–9, Apr. 2005, doi: 10.1007/s00586-004-0705-5.
- [15] T. Illés and S. Somoskeöy, "Comparison of scoliosis measurements based on threedimensional vertebra vectors and conventional two-dimensional measurements: advantages in evaluation of prognosis and surgical results," *European spine journal*, vol. 22, no. 6, pp. 1255–1263, 2013.
- [16] E. Melhem, A. Assi, R. El Rachkidi, and I. Ghanem, "EOS® biplanar X-ray imaging: concept, developments, benefits, and limitations," *Journal of children's orthopaedics*, vol. 10, no. 1, pp. 1–14, 2016.

- [17] E. Amzallag-Bellenger *et al.*, "Idiopathic scoliosis in children and adolescents: assessment with a biplanar X-ray device," *Insights into imaging*, vol. 5, no. 5, pp. 571–583, 2014.
- S. Suzuki, T. Yamamuro, J. Shikata, K. Shimizu, and H. Iida, "Ultrasound measurement of vertebral rotation in idiopathic scoliosis," *J Bone Joint Surg Br*, vol. 71, no. 2, pp. 252–255, Mar. 1989, doi: 10.1302/0301-620X.71B2.2647754.
- [19] R. Burwell *et al.*, "Spine-rib rotation differences at the apex in preoperative patients with adolescent idiopathic scoliosis (AIS): Evaluation of a three-level ultrasound method," *Studies in health technology and informatics*, vol. 91, pp. 246–50, Feb. 2002, doi: 10.3233/978-1-60750-935-6-246.
- [20] Q. N. Vo, E. H. M. Lou, and L. H. Le, "Measurement of vertebral rotation using a three-dimensional ultrasound image," *Scoliosis*, vol. 10, no. 1, p. O40, Jan. 2015, doi: 10.1186/1748-7161-10-S1-O40.
- [21] W. Chen, L. H. Le, and E. H. M. Lou, "Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study," *Eur Spine J*, vol. 25, no. 10, pp. 3265–3273, Oct. 2016, doi: 10.1007/s00586-016-4492-6.
- [22] Q. Wang *et al.*, "Validity Study of Vertebral Rotation Measurement Using 3-D Ultrasound in Adolescent Idiopathic Scoliosis," *Ultrasound Med Biol*, vol. 42, no. 7, pp. 1473–1481, Jul. 2016, doi: 10.1016/j.ultrasmedbio.2016.02.011.
- [23] J. Zhang et al., "Computer-aided assessment of scoliosis on posteroanterior radiographs," Med Biol Eng Comput, vol. 48, no. 2, pp. 185–195, Feb. 2010, doi: 10.1007/s11517-009-0556-7.
- [24] T. K. Koo and M. Y. Li, "A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research," *J Chiropr Med*, vol. 15, no. 2, pp. 155–163, Jun. 2016, doi: 10.1016/j.jcm.2016.02.012.

[25] R. Zheng *et al.*, "Factors influencing spinal curvature measurements on ultrasound images for children with adolescent idiopathic scoliosis (AIS)," *PLoS One*, vol. 13, no. 6, p. e0198792, Jun. 2018, doi: 10.1371/journal.pone.0198792.

Chapter 7: Development of Detection Models for Curve Progression

7.1. Overview

This chapter overviews two studies focusing on model development for curve progression. The first study is a pilot study and the second one is a large clinical study. Section 7.2 covers a brief description of the pilot study. The result of that pilot study has been reported in the manuscript entitled "Development and Validation of a Model to Predict X-ray Progression at a Follow-up Visit Based on Ultrasound and Clinical Parameters for Adolescent Idiopathic Scoliosis (AIS)" which has been submitted to the "Journal of Spine Deformity". It also has been presented in 56th Annual Meeting of Scoliosis Research Society (SRS) in 2021. The section 7.3 presents the large clinical study that is used to investigate the association of baseline, radiographic and US parameters with curve progression. The materials of the section 7.3 are taken from the manuscript entitled "Identifying Prognostic Parameters to Develop a Detection Model of Curve Progression with Clinical Validation on Adolescents with Idiopathic Scoliosis" which has been submitted to the "European Spine Journal". Section 7.4 provides a summary of this chapter.

7.2. A Preliminary Analysis on Development and Validation of a Model to Detect Progression

Prior to conduct a large study on development of detection model, multiple models were developed and tested. One of the preliminary study models based on four baseline parameters including a) the X-ray Cobb, b) a novel US parameter called reflection coefficient (RC) as an index of bone strength, c) chronological age, and d) the menarche status plus an extra parameter of the US Cobb change measured from ultrasonographs at the baseline and follow-up visits are reported in this section. In this study, seventy-five female participants with AIS aged 13.9 ± 1.5 years old were recruited and consented from a single center. All participants were followed 8.5 ± 5.3 months away from the baseline visit

and had the US scans during the baseline and follow-up visits plus the standard of care. Among the 75 participants, 56 (75%) (age 13.6 \pm 1.6yr) were randomly selected for model development and 19 (25%) (age 13.5 \pm 2.1yr) were used for model validation. The curve progression cases for the model development and validation groups with the consecutive Cobb angle increase >5° were 25 and 7, respectively. The results showed that only US Cobb change and RC index were the final predictors, and the model achieved the sensitivity=85.7%, specificity=91.7% and accuracy=89.5%. However, a larger study was required to investigate more prognostic factors, use those potential parameters for model development and validate the model which is explained in detail in section 7.3.

7.3. Identifying Prognostic Parameters to Develop a Detection Model of Curve Progression with Clinical Validation

7.3.1. Introduction

Adolescents with idiopathic scoliosis (AIS) mainly present lateral deviations of spine which typically also include rotations, and twists in their spine. Children with AIS need to be monitored by consecutive posterioanterior (PA) and sometimes lateral (LAT) radiographs every six months to measure the curve severity using the Cobb method [1]. Ronckers et al. reported that on average 22.9 radiographs per person were acquired during treatment and follow-up of scoliosis after following 5,513 females [2]. On the other hands, approximately 15% of children with AIS showed progression defined as a Cobb angle change over 5°at the 6 months follow-up visit [3].

Exposing children to ionizing radiation may increase the risk of cancer incidence [4]–[6]. Also, the long-term negative effects because of cumulative radiation may appear years after for the exposed person or even in the next generations [4]–[6]. One recommendation would be using a safe non-ionizing imaging modality, ultrasound (US), to image spine. From the literature, few parameters measured from US images have shown reliable and accurate results with both intra- and inter-rater and inter-method reliabilities were good (ICCs>0.82) when compared from radiographic measurements [7]–[11]. The measured parameters

included the Cobb angle [7], axial vertebra rotation (AVR) [9], Cobb angle at the plane of maximum curvature (PMC) [10], and kyphotic angle (KA) [11]. Recently, a new US parameter called the reflection coefficient index (RC), which was proposed to provide spinal stiffness information. This parameter can be extracted and measured reliably from US images [12]. However, US cannot replace X-ray because radiography is needed for the diagnosis of scoliosis. However, it is possible to reduce the follow-up X-rays for children without progression if we could predict risk and detect curve progression.

From the literature, a list of baseline, clinical, radiographic and physiologic parameters have been associated with curve progression [13]–[18]. The most cited parameters that were linked with curve progression were a) the severity of the Cobb angle, b) AVR, c) curve type, d) Risser sign, and growth-related factors such as e) gender, f) age, and g) menarche status [13]-[17]. In 2005, Hung et al. put forward a new parameter, osteopenia, as a new potential predictor. Their study showed that younger age at the time of diagnosis, premenarchal status in females, low Risser sign (0-1) and presence of osteopenia could be used to predict curve progression [13]. They used dual X-ray absorptiometry (DXA) to measure the bone mineral density (BMD) of the hip which exposed these children to extra ionizing radiation. Their prediction model had the sensitivity and specificity of 76% and 70%, respectively. Another study by Lam et al. showed that a lower ultrasound stiffness index - a bone quality assessment index acquired at calcaneus, younger age, pre-menarche status, and larger Cobb angle were associated with a higher risk of curve progression with a sensitivity of 85% and a specificity of 66% [19]. Also, progressive, and non-progressive groups were compared in terms of the sagittal parameters including kyphotic and lordotic angles and the results showed that the progressive group had a significantly lower kyphotic angle. However, no prediction model was developed in that study [20].

Based on a systematic review that we conducted recently, the level of evidence is limited or conflicting for most of the parameters reported in the literature. To develop a prediction model, clinicians are always looking for parameters which can be captured from their regular scoliosis clinic. Therefore, a larger and rigorous study is needed to further confirm the prognostic parameters to predict curve progression. This study aimed to identify different baseline, radiographic and US parameters to develop a detection model on curve progression and to validate the model based on clinical data for children with AIS.

7.3.2. Materials and Methods

7.3.3. Clinical Participants

One hundred and sixty-two female participants with AIS were consecutively recruited and signed informed consent. Ethics approval was obtained from the local research ethics board. The inclusion criteria were participants who a) were diagnosed with AIS, b) were aged between 10-18 years old, c) had major Cobb angle $<55^{\circ}$ and d) had no surgical treatment during or prior to the study. Among the 162, 94 were under conservative treatment (bracing and Schroth's exercises), and 68 were under observation. All participants were followed-up 8.0 ± 4.1 months away from their baseline visit. All participants had the US scans during their baseline and follow-up visits in addition to the standard care. One-hundred participants ($13.6 \pm 1.6 \text{ yr}$) were randomly selected for development of detection model and the rest of 62 ($13.4 \pm 1.7 \text{ yr}$) were used for model validation. Among the model development and validation, 25 and 11 cases had curve progression (Cobb angle change >5 degrees) as per the Scoliosis Research Society guidelines [21], respectively.

7.3.4. US Data Acquisition

The participants were scanned by an US system (Sonix TABLET with SonixGPS system, BK Medical, MA, USA), in a standard standing position as described in our previous studies [22] for both visits. The US system is equipped with a 128-element C5-2/60 GPS convex transducer. The parameters that were set for imaging were: 2.5 MHz center frequency, focus on the imaging depth at 6 cm, and the signal gain of 15%. During data acquisition, the B mode images that included the US reflection signals as well as position

and orientation information were acquired. Then, a custom software was used to reconstruct the US images in 3D and display on coronal, axial and sagittal planes.

7.3.5. Investigated Parameters & Data Measurements

a) Demographic Parameters

• Chronological Age, Body Mass Index (BMI), Menarche Status

The demographic information of each participant including the age, BMI and menarche status were extracted from our local clinic database.

b) Radiographic Parameters

• X-ray Cobb Angle, Number of Curves, Risser Sign at Baseline

The X-ray Cobb angle, number of curves (NOC) and Risser sign which were measured and recorded by the clinical staff who had at least 10+ years of experience. This radiographic data was extracted from the database as well.

c) Ultrasound Parameters

Raters Involved in US Parameters Measurements

The US parameters were measured by three raters. R1 and R2 had over 2 years of experience measuring the US parameters. Rater 3 was a novice. For the training R3 first practiced on 10 practice files apart from study files and compared her measurements with R1 and discussed the discrepancies. Once she was confident, she measured study cases. All the US parameters were measured using a custom software. Fig 7.1 shows the measurements of the Cobb angle, AVR, KA and RC index on US images.

• Ultrasound Cobb Change

The US Cobb was measured at both the baseline and follow-up visits using the center of lamina (COL) method [7] on the US coronal view image (Fig 7.1a). Then, the difference of the US Cobb angle between the two visits was calculated for each participant.

• Maximal Axial Vertebral Rotation (AVR)

The COL method was used to measure the AVR on transverse view US images (Fig 7.1b). The maximum AVR among the apical vertebra, the superior and inferior of the apical level was used.

• Plane of Maximum Curvature (PMC) Curve Angle

The Curve angle in the PMC usually occurs in the maximum axial rotation plane. Similar to the study in [10], to calculate the maximum Curve angle in the plane of PMC, the Curve angles in the maximum AVR at the apical region with five AVR offsets values -4° , -2° , 0° , $+2^\circ$, $+4^\circ$ were recorded.

• Kyphotic angle (KA)

The centers of the laminae for T1 and T2 and T11 and T12 were determined on US coronal view (Fig 7.1a). Then, the tips of spinous process for each of those vertebrae were identified in the transverse view at the skin level (Fig 7.1b). The angle between the line joining the spinous process of T1 and T2 and the line joining the spinous process of T11 and T12 in the sagittal plane was measured as US KA angle (Fig 7.1c).

• US Reflection Coefficient (RC) Index

To measure the RC index, the L5 vertebra was identified on coronal US image (Fig 7.1a). The B-mode US image around the center of the laminae was selected on the coronal view. The transverse view image was then displayed, and the operator marked the laminae on the transverse image. The distances from the surface of the skin to the center of left and right laminae which visualized as the maximum US reflection signals in Fig 7.1d were determined. A custom developed software was then used to calculate the RC index. Four more B-mode US images, 2 above and 2 below of the center B-mode image were used to calculate the corresponding RC values at these levels. The average of the 5 left and 5 right RC values called as combined average reflection coefficient (CARC) method, which was described in [12], was used as the investigated parameter. The method of measurement was selected because it showed a good reliability.



Figure 7.1 The measurements of US parameters including. a) US Cobb angle, b) AVR, c) KA and d) RC index on US images. a) US Cobb angle measured 28° on a coronal US image using the center of laminae method, b) the US AVR is measured on the transverse view US image using the center of laminae, c) to measure the KA, the angle between the two line that joins the spinous process of T1 and T2 and T11 and T12 is measured d) the RC index is measured based on the magnitude of the US refection from the laminae at L5 level by accounting the thickness of the soft tissue from the skin to laminae.

7.3.6. Statistical Analysis

The univariate analysis including Student's t test was conducted to investigate the difference between cases with and without progression with an alpha level of p<0.05 for our continuous variables. The association of the categorical variable i.e., menarche status, NOC and Risser sign with curve progression were tested using Chi square analysis with an alpha level of p<0.05. The dependent variable *curve progression* was coded into "1" for participants "with progression" and "0" for participants "without progression". To develop the detection model, logistic regression analysis was used. The *forward selection* method was chosen for analysis. Three classification cut-offs set 0.4, 0.5 and 0.6 were tested. The model statistics including sensitivity, specificity, accuracy, positive predictive value (PPV) and negative predictive value (NPV) were reported. The receiver operating characteristic curve (ROC) was also performed to compare different threshold values for significant

continuous parameters and area under the curve (AUC) was reported. All statistical analyses were performed using the IBM SPSS v23 software (IBM, Armonk, New York, USA).

7.3.7. Results

Univariate Analysis

Table 7.1 presents the demographic information of the 100 participants used for model development into two categories: with progression and without progression. The average values of the BMI, X-ray Cobb angle, max AVR, PMC Cobb, KA and RC index were smaller in cases with progression versus those without progression. The US Cobb change was higher among the cases with progression versus without progression. Age was about the same among the two groups. Among these parameters only the US Cobb change and RC index were significantly different between the two groups (p<0.05).

Table 7.1 The mean±SD for age, BMI, X-ray Cobb and US parameters for the cases with and without progression and the statistical t-test values on the described parameters between the two groups.

Variables	With progression (n=25)	Without progression (n=75)	t-test (p value)
Age (yr)	13.7±1.5	13.5±1.7	-0.44 (0.66)
Body Mass Index (BMI) (kg/m ²)	18.9±2.7	19.5±3.8	0.73 (0.46)
X-ray Cobb (°)	22.6±9.7	25.7±9.8	1.38 (0.17)
US Cobb change (°)	$6.6{\pm}5.0$	-2.4±6.5	-6.39 (0.00*)
Max AVR (°)	3.4±9.9	5.2±9.4	0.82 (0.41)
PMC (°)	24.5±10.5	27.1±10.4	1.09 (0.27)
Kyphotic Angle (KA) (°)	32.0±11.2	35.1±11.6	1.18 (0.24)
Reflection Coefficient (RC)	0.05 ± 0.02	$0.07{\pm}0.02$	3.95 (<0.01*)

US= Ultrasound, PMC=Plane of maximum curvature, AVR=Axial Vertebra Rotation, the asteroid indicates significant difference (p<0.05)

Table 7.2 indicates the association of the categorical parameters with curve progression. The majority of the participants already had menarche (70% vs 30%). Also, 56% of the participants had Risser sign >1. In addition, only 46% of the participants had a single curve. Of these 3 parameters, only the NOC was statistically associated with curve progression (p<0.05).

Parameters	Parameters	Coding	With Progression	Without	Chi square	
rarameters	division		(n=25)	Progression (n=75)	(p value)	
Menarche	Pre-menarche	1	7 (28%)	23 (31%)	0.062 (0.80)	
Status	Menarche	0	18 (72%)	52 (69%)	0.063 (0.80)	
Risser sign -	≤1	1	12 (48%)	32 (43%)	0.21 (0.64)	
	>1	0	13 (52%)	43 (57%)		
NOC -	Single	0	6 (24%)	40 (53%)	((0 (0 0114)	
	Double or more	1	19 (76%)	35 (47%)	6.49 (0.011*)	

Table 7.2 The association between categorical parameters with curve progression

NOC= Number of curves, the asterisk indicates significant difference (p < 0.05)

Multivariate analysis

Table 7.3 shows the developed model specifications and statistics. The curve progression detection model was: Logit (p) = Log (p/1-p) =-1.40+0.28(US Cobb change)-39.45(RC)+1.36 (NOC). The participants who had higher values of US Cobb Change and NOC, and lower values of RC were at a higher probability to have curve progression using 0.4 cut-off threshold. The statistical values of the model development data were sensitivity=68.0% (95% CI, 46.5 to 85.0), specificity= 89% (95%CI, 80.6 to 95.2) and accuracy=84% (95%CI, 75.3 to 90.5).

	p			er mit mererepen	
Predictors	В	Standard error	Level of significance	EXP (B) 95% CI	Logistic regression model
NOC	1.36	0.68	0.04	3.90 (1.02 to 14.84)	
RC	-39.45	17.27	0.02	0.00 (0.000 to 0.004)	Log (p/1-p) = -1.40+0.28(US) Cobb change)-
US Cobb Change	0.28	0.07	< 0.01	1.33 (1.14 to 1.55)	39.45(RC)+1.36 (NOC)
Constant	-1.40	1.62	0.38	0.24	-
Model	Sensiti	ivity= 68%	Specif	icity= 89%	Accuracy=84%
Characteristics	(95%CI,	46.5 to 85.0)	(95%CI,	80.6 to 95.2)	(95%CI, 75.3 to 90.5)

Table 7.3 The final predictors and the statistics of the developed model

RC=reflection coefficient; NOC= Number of curve, B= Beta coefficient, EXP (B): Odds ratios for the predictors which are exponentiation of coefficients; CI=Confidence interval; P= Probability of progression

Model Validation

Of the three tested probability cut-off thresholds, 0.4 classification results were the best and presented in Table 7.4. The accuracy of the developed model was tested in 62 new participants and the probability of curve progression was calculated for each participant. The following accuracy statistics achieved: the sensitivity=90.9%, specificity=90% and accuracy=90% (Table 7.4).

N = 62	Actual progression	Actual non- progression	
Predicted progression	10	5	Positive Predictive value = $10/15 =$ 67.0%
Predicted non-progression	1	46	Negative Predictive value = 46/47 = 97.8%
	Sensitivity= 10/11 = 90.9%	Specificity= 46/51 = 90.1%	Accuracy = 56/62 = 90.3%

Table 7.4 The validation results from the test data set

Thresholds for US Cobb change and RC

• US Cobb change

The ROC analysis on US Cobb change resulted in the AUC= 0.89 which was significant. Figure 7.2 compares three different threshold values of US Cobb change as 4° , 5° and 6° versus the percentage of sensitivity and specificity. As the results show, the 4° threshold provides the best sensitivity (78%), but the lowest specificity. On the other hand, the 6° threshold provides the reverse result, the lowest sensitivity and the highest specificity. Using the 5° threshold provides the middle values for both sensitivity and specificity.



Figure 7.2 Compares three different thresholds of US Cobb change as 4° , 5° and 6° in terms of their sensitivity and specificity

• Reflection Coefficient (RC) Index

The AUC for the RC index was 0.66 which was also significant (p<0.05). Fig 7.3 shows the comparison of 3 different threshold values for the RC index from 0.05 - 0.07 versus the sensitivity and specificity. The best sensitivity was achieved when RC was equal to 0.07 (83%); however, the specificity was the lowest (40%). The 0.05 threshold shows the highest specificity (73%); however, its sensitivity is lower for the 0.06 threshold (56% vs 69%). Using the 0.06 threshold provides a middle value on both sensitivity and specificity.



Figure 7.3 The comparison of different threshold values for the RC index to discriminate the cases with and without progression

7.3.8. Discussion

Based on the literature, for majority of the prognostic factor of curve progression the evidence is insufficient in quantity or quality to formulate conclusive statements. In this study, we conducted a comprehensive step-by step analysis to identify the prognostic factors which associate with curve progression on children with AIS. Using our detection model 90% of accuracy was achieved. Our model was able to avoid unnecessary radiographs for 74% of similar cases. Even though by using our model there was only one missing progressive case as false negative (NPV= 98%), the positive predictive value was 67%. This shows that there were 5 false positive cases that were wrongly classified as progression cases. It should be noted in clinical practice, all children, even those without progression would need to take radiographs if requested. Even though there is a 6/62 (10%)

chance of misclassification, using our model, it is possible to recommend skipping x-rays for the majority of children predicted to be without progression. Among the 6 misclassification cases, 1 was from the progression group, but 5 from the non-progression group. For the missed progression case, the participant had the RC index of 0.11 which is above the average of the RC measured for the progression group (0.05 ± 0.02) . Also, this case was already mature with a Risser sign of 4 and one year had passed her menarche at the time of data acquisition which may explain high RC value for her.

Table 7.5 shows the information on the predictors (US Cobb change, RC and NOC) as well as the predicted probability of progression for the 5 cases misclassified as progression cases. Even though these cases were misclassified as progression by using our model, their probability is not significantly larger than the 0.4 threshold used to identify as progression. This means that most of the measurements for the US Cobb change (5° to 8°) and particularly RC (0.04 to 0.06) were close to marginal values that may overlap between progression and non-progression group. More accurate measurements could reduce the risk of misclassifications.

Cases	Ultrasound (US) Cobb change	Reflection coefficient (RC) index	Number of curve (NOC)	Predicted probability of progression
1	5	0.04	0	0.48
2	6	0.04	0	0.52
3	5	0.06	1	0.59
4	8	0.05	0	0.58
5	5	0.04	0	0.46

Table 7.5 Predictors values (US Cobb change, RC and NOC) as well as the predicted probability of progression for the 5 cases misclassified as progression cases.

Note that the predicted probability >0.4 was considered as threshold to identify those with progression

Figure 7.4 shows the participants with and without progression in terms of their US Cobb change and RC index. The region outlined with the red box indicates the high-risk area for curve progression. Of the cases in this region, 28/36=77% had an US Cobb change $\geq 5^{\circ}$, 25/36=69% had RC ≤ 0.06 and 26/36=72% had multiple curvatures. Within this region, there are 9 cases without progression as well. These cases showed RC ≤ 0.06 and US Cobb

change $\geq 5^{\circ}$ while 7/9 of them had only a single curvature. Nevertheless, they still meet the progression criteria for two of our predictors and are still at risk of showing curve progression. When these cases were tracked in our data base, we found that 3/9 showed curve progression within a year or 2 from their study scan visit.



Figure 7.4 Illustration of the RC index versus US cobb changes for 162 cases with and without progression

In our study we focused on female participants to maintain the homogeneity of the data. However, we also tested our model on 29 boys with AIS. Of these 29 cases 6 showed curve progression and 23 no progression. The validity results showed a sensitivity=68%, specificity=87% and accuracy=83%. Since the number of male cases was small, no conclusive statement could be made.

To compare the results of our model with existing models [18] the sensitivity reported from literature is within 60% [23] to 85% [24] and specificity from 48% [24] to 81% [25]. The most cited parameters were low age, low Risser sign (0-1), pre-menarche status, large Cobb angle and presence of osteopenia [18]. Our results using US Cobb angle change in addition

to these promising candidate variables showed better accuracy results to detect curve progression at follow-up without use of x-ray radiation. Also, in our study we introduced new parameters extracted from US which is a safe method. There are some limitations regarding our study. Our sample size does not include large group of progression cases. Larger study may help to improve the validation results. Another issue is about heterogeneity of the baseline parameters. Our participants were recruited at different timeline in their growth which may generate some heterogeneities. It is suggested to follow participants prospectively from the initial visit after diagnosis to control for the effect of confounders and check one variable at a time. Lastly, our sample included participants under observation and conservative treatment which could affect the results. It is suggested to examine the predictability of parameters for each group separately. Nevertheless, this study showed promising results for the detection of progression, and improvemed predictability could be achieved by including a larger sample and a homogenous dataset.

7.3.9. Conclusion

The final model of curve progression was associated with US Cobb change, RC index and NOC. A large US Cobb change, small RC value and presenting multiple curves increases the risk of presenting curve progression at the follow-up visit. The developed model was validated with sensitivity, specificity, and accuracy all over 90%. Future study should include a greater number of participants to provide better validation of the model.

7.4. Summary

This chapter described two studies to develop a detection model for curve progression and validated with clinical data. The first study was a pilot study that investigated development of detection model using some baseline, clinical, radiographic data and US parameters from both the baseline and follow-up visit. The results showed that the probability of presenting curve progression at time of follow-up for girls with AIS was higher when the US Cobb change was bigger, and the RC value was lower (weaker bone). The 2nd study included more participants for model development and validation. Also, more parameters,

particularly US parameters extracted from coronal, axial and sagittal planes, were investigated. The model achieved sensitivity, specificity, and accuracy of 90%. The results indicated that participants with lower RC (weaker bone, RC ≤ 0.06), larger US Cobb change ($\geq 5^{\circ}$) and multiple curvatures (NOC>1) were at higher risk of curve progression. Using the predicted model, it may possible to reduce radiographs at that follow-up visit for 74% of children which were predicted to not present progression.

References

- T. Kotwicki, "Evaluation of scoliosis today: examination, X-rays and beyond," Disability and rehabilitation, vol. 30, no. 10, pp. 742–751, 2008.
- [2] C. M. Ronckers, C. E. Land, J. S. Miller, M. Stovall, J. E. Lonstein, and M. M. Doody, "Cancer mortality among women frequently exposed to radiographic examinations for spinal disorders," *Radiation research*, vol. 174, no. 1, pp. 83–90, 2010.
- P. N. Soucacos, K. Zacharis, J. Gelalis, *et al.*, "Assessment of curve progression in idiopathic scoliosis," *Eur Spine J*, vol. 7, no. 4, pp. 270–277, 1998, doi: 10.1007/s005860050074.
- [4] L. H. Thompson, "Recognition, signaling, and repair of DNA double-strand breaks produced by ionizing radiation in mammalian cells: the molecular choreography," *Mutation Research/Reviews in Mutation Research*, vol. 751, no. 2, pp. 158–246, 2012.
- [5] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine (Phila Pa 1976)*, vol. 21, no. 13, pp. 1540–1547; discussion 1548, Jul. 1996, doi: 10.1097/00007632-199607010-00011.
- [6] A. R. Levy, M. S. Goldberg, J. A. Hanley, N. E. Mayo, and B. Poitras, "Projecting the lifetime risk of cancer from exposure to diagnostic ionizing radiation for adolescent idiopathic scoliosis.," *Health physics*, vol. 66, no. 6, pp. 621–633, 1994.

- [7] R. Zheng, A.C. Chan, W. Chen, *et al.*, "Intra- and Inter-rater Reliability of Coronal Curvature Measurement for Adolescent Idiopathic Scoliosis Using Ultrasonic Imaging Method-A Pilot Study," *Spine Deform*, vol. 3, no. 2, pp. 151–158, Mar. 2015, doi: 10.1016/j.jspd.2014.08.008.
- [8] Y.-P. Zheng, T.T.Y. Lee, K.K. L. Lai, *et al.*, "A reliability and validity study for Scolioscan: a radiation-free scoliosis assessment system using 3D ultrasound imaging," *Scoliosis and Spinal Disorders*, vol. 11, no. 1, p. 13, May 2016, doi: 10.1186/s13013-016-0074-y.
- [9] W. Chen, L. H. Le, and E. H. M. Lou, "Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study," *Eur Spine J*, vol. 25, no. 10, pp. 3265–3273, Oct. 2016, doi: 10.1007/s00586-016-4492-6.
- [10] S. Trac, R. Zheng, D. L. Hill, and E. Lou, "Intra- and Interrater Reliability of Cobb Angle Measurements on the Plane of Maximum Curvature Using Ultrasound Imaging Method," *Spine Deform*, vol. 7, no. 1, pp. 18–26, Jan. 2019, doi: 10.1016/j.jspd.2018.06.015.
- [11] Sayed T, Khodaei M, Lou E, "Intra- and inter-rater reliabilities and accuracy of kyphotic angle measurements on ultrasound images for children with adolescent idiopathic scoliosis - a pilot study," presented at the WCHRI Research Day 2019.
- [12] M. Khodaei, T. Sayed, D. Hill, *et al.*, "Reliability of measurements of a reflection coefficient index to indicate spinal bone strength on adolescents with idiopathic scoliosis (AIS): a pilot study," *European Spine Journal*, pp. 1–8, 2021.
- [13] V. W. Y. Hung, L. Qin, C. S. K. Cheung, *et al.*, "Osteopenia: a new prognostic factor of curve progression in adolescent idiopathic scoliosis," *J Bone Joint Surg Am*, vol. 87, no. 12, pp. 2709–2716, Dec. 2005, doi: 10.2106/JBJS.D.02782.
- [14] X. Sun, B. Wang, Y. Qiu, *et al.*, "Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis," *Orthopaedic surgery*, vol. 2, no. 4, pp. 285–290, 2010.
- [15] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for detection of dranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS)-a prospective cohort study till skeletal maturity," *Osteoprosis International*, 2010, vol. 21, pp. S686-S687.
- [16] K.-J. Tan, M. M. Moe, R. Vaithinathan, and H.-K. Wong, "Curve progression in idiopathic scoliosis: follow-up study to skeletal maturity," *Spine*, vol. 34, no. 7, pp. 697–700, 2009.
- [17] S. S. Upadhyay, I. W. Nelson, E. K. Ho, L. C. Hsu, and J. C. Leong, "New prognostic factors to predict the final outcome of brace treatment in adolescent idiopathic scoliosis," *Spine (Phila Pa 1976)*, vol. 20, no. 5, pp. 537–545, Mar. 1995, doi: 10.1097/00007632-199503010-00006.
- [18] A. Noshchenko, L. Hoffecker, E. M. Lindley, *et al.*, "Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with metaanalysis," *World journal of orthopedics*, vol. 6, no. 7, p. 537, 2015.
- [19] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound Med Biol*, vol. 39, no. 3, pp. 381–387, Mar. 2013, doi: 10.1016/j.ultrasmedbio.2012.09.012.
- [20] M.-L. Nault, J.-M. Mac-Thiong, M. Roy-Beaudry, J. deGuise, H. Labelle, and S. Parent, "Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis," *J Pediatr Orthop*, vol. 33, no. 6, pp. 618–623, Sep. 2013, doi: 10.1097/BPO.0b013e318292462a.

- [21] S. Negrini, T. M. Hresko, J. P. O'Brien, N. Price, SOSORT Boards, and SRS Non-Operative Committee, "Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee," *Scoliosis*, vol. 10, p. 8, 2015, doi: 10.1186/s13013-014-0025-4.
- [22] M. Khodaei, D. Hill, R. Zheng, L. H. Le, and E. H. M. Lou, "Intra- and inter-rater reliability of spinal flexibility measurements using ultrasonic (US) images for nonsurgical candidates with adolescent idiopathic scoliosis: a pilot study," *Eur Spine J*, vol. 27, no. 9, pp. 2156–2164, Sep. 2018, doi: 10.1007/s00586-018-5546-8.
- [23] A. J. Danielsson, R. Hasserius, A. Ohlin, and A. L. Nachemson, "A prospective study of brace treatment versus observation alone in adolescent idiopathic scoliosis: a follow-up mean of 16 years after maturity," *Spine (Phila Pa 1976)*, vol. 32, no. 20, pp. 2198–2207, Sep. 2007, doi: 10.1097/BRS.0b013e31814b851f.
- [24] X. Sun, T. Wu, Z. Liu, *et al.*, "Osteopenia predicts curve progression of adolescent idiopathic scoliosis in girls treated with brace treatment," *J Pediatr Orthop*, vol. 33, no. 4, pp. 366–371, Jun. 2013, doi: 10.1097/BPO.0b013e31827b7b5f.
- [25] L. E. Peterson and A. L. Nachemson, "Prediction of progression of the curve in girls who have adolescent idiopathic scoliosis of moderate severity. Logistic regression analysis based on data from The Brace Study of the Scoliosis Research Society," J Bone Joint Surg Am, vol. 77, no. 6, pp. 823–827, Jun. 1995, doi: 10.2106/00004623-199506000-00002.

Chapter 8: Conclusions and Recommendations

8.1. Overview

This chapter provides a summary of this PhD work and reports major contributions, the clinical significance, limitations, and future recommendations. Section 8.2 summarizes the final findings from each study leading to the answers to the objectives of my thesis. Section 8.3 outlines the major contributions of this thesis. In section 8.4, the clinical significance of this thesis work is presented. Limitations and future recommendations are described in Section 8.5.

8.2. Summary of Thesis Work

This thesis has outlined the process of developing a detection model for curve progression for children with AIS. To begin, a comprehensive systematic review was conducted to identify potential prognostic predictors of curve progression. The results showed that only a large Cobb angle and pre-menarche status were strongly supported by the literature to predict curve progression $>5^{\circ}$ during a short-term follow-up. However, conflicting results and limited evidence were reported for most parameters. Further investigations were required for many of these parameters. The potential baseline parameters that were selected based on that systematic review were age, body mass index (BMI), menarche status, X-ray Cobb angle, number of curve (NOC), Risser sign. Also, the ultrasound (US) reflection coefficient (RC) which is related to bone quality was selected.

A new US parameter called US RC index was derived based on the fundamentals of ultrasound imaging physics. To learn about the theory, extract the RC index, and calculate its value, both *in-vitro* and *in-vivo* experiments were conducted. From the *in-vitro* experiments, the RC index was found to be linearly correlated with the stiffness of the scanning materials which also applied to the bone in the *in-vivo* study. Also, in a pilot clinical study, the RC index was moderately correlated with the curve severity of 37 participants ($R^2=0.3$) suggesting this parameter may have value to predict curve progression.

Then, a study was conducted to evaluate the repeatability and reliability of RC. Different methods of measurements were tested throughout this study and the most repeatable and reliable was the average of the RC index measured on each lamina for five consecutive frames at L5 with intra-rater and inter-rater reliabilities of ICC $[2,1] \ge 0.84$ and SEM ≤ 0.01 . The results showed 68% of children who had a larger RC value (stronger bone) than the average of the study participants (0.07) showed no curve progression. However, it was a pilot study and the ability of the RC index to predict curve progression required further evaluation.

The US parameters that were considered for investigation as predictors of progression consisted of US Cobb change, plus the US parameters at baseline including US maximum axial vertebral rotations (AVR), Curve angle in the plane of maximum curvature (PMC), US kyphotic angle (KA), and US RC index.

Two studies were conducted to evaluate the reliability and validity of the KA on US images. The first pilot study (n=20) showed US KA was a reliable measurement with an ICC [2, 1] \geq 0.85 for the intra-rater, inter-rater, and inter-method analyses. The average difference between US and X-ray measurements was within 5°. Then, a large clinical study (n=50) was conducted to further evaluate the intra-rater and inter-method reliabilities of the KA as well as to investigate factors that might affect the accuracy of the measurements. The results showed good intra-rater and inter-method reliability ICC [2, 1] \geq 0.84 using the T1/T12 level measurement method. In addition, 80% of the measurements were within the clinically accepted error (7°). The position of the arms during standing posture was the only factor that might affect the measurements.

The factors influencing the reliability and variation measurement of the US AVR were investigated. The ultrasound AVR measurement using the center of laminae (COL) method was compared with Stokes' method on radiographs. The intra-rater result was excellent (ICC [2,1] =0.9); however, the inter-method reliability was poor to moderate (ICC [2,1] \geq 0.49) and the maximum average difference in AVR measurement between X-ray and US

was 4.6°. The factors that showed a significant influence on the differences in the measurements were AVR measurements at the apical region and AVR severity.

Next, a pilot study with 56 cases was conducted to develop a detection model of curve progression using the following parameters age, menarche status, X-ray Cobb angle, RC index that all acquired at the baseline visit except the US Cobb change. Of these, only the RC index and then US Cobb change were the final predictors of curve progression. Then, the model was validated in 19 new participants including 8 progression cases and the results showed sensitivity, specificity, and accuracy of 86%, 92%, and 90% during validation, respectively. This preliminary study showed promising results for developing a model to predict curve progression using US parameters. However, since the number of cases used to develop the model and validate it was small, a larger clinical study was conducted.

Finally, a large clinical study with 162 girls with AIS was conducted. Among those, 100 participants including 25 progression cases were used for model development. The parameters were demographic extracted at baseline: age, body mass index (BMI), and menarche status; radiographic extracted at baseline: X-ray Cobb angle, NOC, and Risser sign; and ultrasonic: US Cobb change, and baseline US parameters including max AVR, PMC Cobb angle, KA, and RC index. After a step-by-step analysis, the significant parameters were US Cobb change, RC index, and NOC. The generated model was Log (p/1-p) = -1.40+0.28(US Cobb change)-39.45(RC)+1.36 (NOC). Then, the model was validated on 62 cases. The sensitivity, specificity, and accuracy results were over 90% which is promising. Using our model at the follow-up visit, we were able to reduce radiographs for 74% of non-progressive cases.

8.3. Major Contributions

The main contributions of this PhD thesis are:

• Introduction of a new US parameter called RC index, *in-vitro* validation of the index and demonstration of the reliable measurements from the developed method

- Introduction of a new method to measure KA on US images, presenting the reliability of measurements, validation of the measurement in a large study, and reporting on factors that may affect the accuracy of the US KA measurements
- Development of a new curve progression detection model based on my proposed parameters, and accurate up to 90%.

8.4. Clinical Significance

The developed model focuses on the detection of curve progression as measured on radiograph at a follow-up visit for children with AIS (Figure 8.1). At the initial visit, PA and LAT radiographs are requested as well as an ultrasound scan. After the patient is diagnosed with AIS, the number of curves (NOC) is determined based on an X-ray image. From the US image, the RC index as well as the first-visit US Curve angle are measured and recorded. Then, for the next follow-up visit, an ultrasonography is acquired. The US Cobb change is calculated. The parameters including the US Cobb change, the baseline RC index, and the baseline NOC are used to calculate the probability of the curve progression on the radiograph based on the equation: Log (p/1-p) =-1.40+0.28(US Cobb change)-39.45(RC)+1.36 (NOC). If the value is > 0.4, the case is considered as high-risk of showing progression. A radiograph is then requested to confirm the Cobb angle change and for planning the course of treatment. Following this approach, it is possible to reduce the number of radiographs at the second visit for the cases that are considered to have a low risk of progression.



Figure 8.1 Clinical application of the detection model to detect cases with progression in clinical practice

8.5. Limitations and Future Recommendations

The main limitation of this study is the number of participants used to develop and validate the model. The total number is smaller than the sample size required to provide 80% power, 500 participants including 72 progression cases. The lack of a larger sample size is due to the COVID-19 pandemic which affected the recruitment and follow-up testing in clinical study.

Another point is the heterogeneities that exist in our data set. The children were recruited at different timelines in their care path; more specifically, they were also at different growth stages. Even though we tried to focus on only female cases and reduce the heterogeneities to some level with our selection criteria excluding larger curves and those treated with surgeries, the baseline information varied between participants.

The next limitation is regarding the treatment status of the participants included in our study. The children were either under observation or conservative treatment during data acquisition. The interventions could have affected the outcome of progression. Also, since the number of cases under each category was not large, it was difficult to run the analysis for each category during the model development.

The following recommendations have been provided for consideration in future investigations:

- Continue recruiting subjects for the study so that a larger data sample and more homogenous data can be used to develop and validate the model.
- Use all participants' data at their first visit to the scoliosis clinic.
- Apply different machine learning algorithms to develop the prediction models and to help choose the most accurate model for clinics.
- In terms of treatment status, develop the model for observation and treatment groups separately in order to investigate whether treatment has any effect on the results.
- Consider other progression criteria, as well as a short-term or long-term follow-up interval for developing the prediction model rather than solely focusing on prediction of curve progression at the follow-up visit.
- Develop a program to automatically measure all the US parameters instead of manual measurement and extraction to minimize measurement errors.

References

Chapters 1 and 2

- M. N. Choudhry, Z. Ahmad, and R. Verma, "Adolescent idiopathic scoliosis," *The open orthopaedics journal*, vol. 10, p. 143, 2016.
- [2] M. M. Doody *et al.*, "Breast cancer mortality after diagnostic radiography: findings from the US Scoliosis Cohort Study," *Spine*, vol. 25, no. 16, pp. 2052–2063, 2000.
- [3] A. R. Levy, M. S. Goldberg, J. A. Hanley, N. E. Mayo, and B. Poitras, "Projecting the lifetime risk of cancer from exposure to diagnostic ionizing radiation for adolescent idiopathic scoliosis.," *Health physics*, vol. 66, no. 6, pp. 621–633, 1994.
- P. N. Soucacos *et al.*, "Assessment of curve progression in idiopathic scoliosis," *Eur Spine J*, vol. 7, no. 4, pp. 270–277, 1998, doi: 10.1007/s005860050074.
- [5] A. Noshchenko *et al.*, "Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis," *World journal of orthopedics*, vol. 6, no. 7, p. 537, 2015.
- [6] W. Chen, E. H. M. Lou, P. Q. Zhang, L. H. Le, and D. Hill, "Reliability of assessing the coronal curvature of children with scoliosis by using ultrasound images," *J Child Orthop*, vol. 7, no. 6, pp. 521–529, Dec. 2013, doi: 10.1007/s11832-013-0539-y.
- [7] R. Zheng *et al.*, "Improvement on the Accuracy and Reliability of Ultrasound Coronal Curvature Measurement on Adolescent Idiopathic Scoliosis With the Aid of Previous Radiographs," *Spine (Phila Pa 1976)*, vol. 41, no. 5, pp. 404–411, Mar. 2016, doi: 10.1097/BRS.000000000001244.
- [8] J. C. Cheng and X. Guo, "Osteopenia in adolescent idiopathic scoliosis. A primary problem or secondary to the spinal deformity?," *Spine (Phila Pa 1976)*, vol. 22, no. 15, pp. 1716–1721, Aug. 1997, doi: 10.1097/00007632-199708010-00006.
- [9] G. Kayalioglu, "The vertebral column and spinal meninges," in *The Spinal Cord*, Elsevier, 2009, pp. 17–36.

- [10] V. Mahadevan, "Anatomy of the vertebral column," *Surgery (Oxford)*, vol. 36, no. 7, pp. 327–332, 2018.
- [11] D. Bijendra, X. Wu, Z. Jiang, L. Zhu, M. Promish, and S. Ratish, "Adjacent Level Vertebral Fractures in Patients Operated with Percutaneous Vertebroplasty," *Open Journal of Orthopedics*, vol. 8, no. 3, pp. 116–126, 2018.
- [12] S. Donzelli *et al.*, "State of the art of current 3-D scoliosis classifications: a systematic review from a clinical perspective," *Journal of neuroengineering and rehabilitation*, vol. 12, no. 1, pp. 1–11, 2015.
- [13] D. Hedequist and J. Emans, "Congenital scoliosis," JAAOS-Journal of the American Academy of Orthopaedic Surgeons, vol. 12, no. 4, pp. 266–275, 2004.
- [14] R. Vialle, C. Thévenin-Lemoine, and P. Mary, "Neuromuscular scoliosis," Orthopaedics & Traumatology: Surgery & Research, vol. 99, no. 1, pp. S124–S139, 2013.
- [15] J. A. Janicki and B. Alman, "Scoliosis: Review of diagnosis and treatment," *Paediatrics & child health*, vol. 12, no. 9, pp. 771–776, 2007.
- [16] S. L. Weinstein, L. A. Dolan, J. C. Cheng, A. Danielsson, and J. A. Morcuende, "Adolescent idiopathic scoliosis," *The lancet*, vol. 371, no. 9623, pp. 1527–1537, 2008.
- [17] J. A. Janicki and B. Alman, "Scoliosis: Review of diagnosis and treatment," *Paediatrics & child health*, vol. 12, no. 9, pp. 771–776, 2007.
- [18] S. L. Weinstein, "The natural history of adolescent idiopathic scoliosis," *Journal of Pediatric Orthopaedics*, vol. 39, pp. S44–S46, 2019.
- [19] T. Kotwicki, "Evaluation of scoliosis today: examination, X-rays and beyond," *Disability and rehabilitation*, vol. 30, no. 10, pp. 742–751, 2008.
- [20] A. Jada *et al.*, "Evaluation and management of adolescent idiopathic scoliosis: a review," *Neurosurgical focus*, vol. 43, no. 4, p. E2, 2017.

- [21] S. Negrini, T. M. Hresko, J. P. O'brien, and N. Price, "Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee," *Scoliosis*, vol. 10, no. 1, pp. 1–12, 2015.
- [22] S. L. Weinstein and I. V. Ponseti, "Curve progression in idiopathic scoliosis.," *The Journal of bone and joint surgery. American volume*, vol. 65, no. 4, pp. 447–455, 1983.
- [23] F. P. Castro Jr, "Adolescent idiopathic scoliosis, bracing, and the Hueter-Volkmann principle," *The Spine Journal*, vol. 3, no. 3, pp. 180–185, 2003.
- [24] I. A. Stokes, R. G. Burwell, and P. H. Dangerfield, "Biomechanical spinal growth modulation and progressive adolescent scoliosis-a test of the'vicious cycle'pathogenetic hypothesis: Summary of an electronic focus group debate of the IBSE," *Scoliosis*, vol. 1, no. 1, pp. 1–21, 2006.
- [25] C. Van Rooyen *et al.*, "The effectiveness of Schroth exercises in adolescents with idiopathic scoliosis: A systematic review and meta-analysis," *South African Journal of Physiotherapy*, vol. 75, no. 1, pp. 1–9, 2019.
- [26] S. L. Weinstein, L. A. Dolan, J. G. Wright, and M. B. Dobbs, "Effects of bracing in adolescents with idiopathic scoliosis," *New England Journal of Medicine*, vol. 369, no. 16, pp. 1512–1521, 2013.
- [27] F. Canavese and A. Kaelin, "Adolescent idiopathic scoliosis: Indications and efficacy of nonoperative treatment," *Indian journal of orthopaedics*, vol. 45, pp. 7– 14, 2011.
- [28] M. Romano et al., "Exercises for adolescent idiopathic scoliosis," Cochrane Database of Systematic Reviews, no. 8, 2012.
- [29] S. Schreiber *et al.*, "Schroth physiotherapeutic scoliosis-specific exercises added to the standard of care lead to better Cobb angle outcomes in adolescents with

idiopathic scoliosis-an assessor and statistician blinded randomized controlled trial," *PloS one*, vol. 11, no. 12, p. e0168746, 2016.

- [30] S. Otman, N. Kose, and Y. Yakut, "The efficacy of Schroth's 3-dimensional exercise therapy in the treatment of adolescent idiopathic scoliosis in Turkey," *Neurosciences* (*Riyadh*), vol. 10, no. 4, pp. 277–83, 2005.
- [31] C. Fusco, F. Zaina, S. Atanasio, M. Romano, A. Negrini, and S. Negrini, "Physical exercises in the treatment of adolescent idiopathic scoliosis: an updated systematic review," *Physiotherapy theory and practice*, vol. 27, no. 1, pp. 80–114, 2011.
- [32] S. Schreiber *et al.*, "The effect of Schroth exercises added to the standard of care on the quality of life and muscle endurance in adolescents with idiopathic scoliosis-an assessor and statistician blinded randomized controlled trial: 'SOSORT 2015 Award Winner," *Scoliosis*, vol. 10, p. 24, 2015, doi: 10.1186/s13013-015-0048-5.
- [33] K. H. Bridwell, "Surgical treatment of idiopathic adolescent scoliosis," *Spine*, vol. 24, no. 24, p. 2607, 1999.
- [34] M. de Kleuver *et al.*, "Optimal surgical care for adolescent idiopathic scoliosis: an international consensus," *European Spine Journal*, vol. 23, no. 12, pp. 2603–2618, 2014.
- [35] T. Maruyama and K. Takeshita, "Surgical treatment of scoliosis: a review of techniques currently applied," *Scoliosis*, vol. 3, no. 1, pp. 1–6, 2008.
- [36] W. C. Warner, J. R. Sawyer, and D. M. Kelly, "Scoliosis and kyphosis," *Campbell's operative orthopaedics*. 12th ed. Philadelphia: Elsevier-Mosby, pp. 1691–895, 2013.
- [37] I. Helenius *et al.*, "Harrington and Cotrel-Dubousset instrumentation in adolescent idiopathic scoliosis. Long-term functional and radiographic outcomes," *J Bone Joint Surg Am*, vol. 85, no. 12, pp. 2303–2309, Dec. 2003, doi: 10.2106/00004623-200312000-00006.

- [38] G. J. Bansal, "Digital radiography. A comparison with modern conventional imaging," *Postgrad Med J*, vol. 82, no. 969, pp. 425–428, Jul. 2006, doi: 10.1136/pgmj.2005.038448.
- [39] G. Compagnone, M. C. Baleni, L. Pagan, F. L. Calzolaio, L. Barozzi, and C. Bergamini, "Comparison of radiation doses to patients undergoing standard radiographic examinations with conventional screen-film radiography, computed radiography and direct digital radiography," *Br J Radiol*, vol. 79, no. 947, pp. 899–904, Nov. 2006, doi: 10.1259/bjr/57138583.
- [40] T. Kluba, J. Schäfer, T. Hahnfeldt, and T. Niemeyer, "Prospective randomized comparison of radiation exposure from full spine radiographs obtained in three different techniques," *Eur Spine J*, vol. 15, no. 6, pp. 752–756, Jun. 2006, doi: 10.1007/s00586-005-1005-4.
- [41] H. Geijer, B. Verdonck, K.-W. Beckman, T. Andersson, and J. Persliden, "Digital radiography of scoliosis with a scanning method: radiation dose optimization," *Eur Radiol*, vol. 13, no. 3, pp. 543–551, Mar. 2003, doi: 10.1007/s00330-002-1476-1.
- [42] S. C. N. Hui *et al.*, "Radiation dose of digital radiography (DR) versus micro-dose x-ray (EOS) on patients with adolescent idiopathic scoliosis: 2016 SOSORT- IRSSD 'John Sevastic Award' Winner in Imaging Research," *Scoliosis Spinal Disord*, vol. 11, p. 46, 2016, doi: 10.1186/s13013-016-0106-7.
- [43] J. Dubousset *et al.*, "A new 2D and 3D imaging approach to musculoskeletal physiology and pathology with low-dose radiation and the standing position: the EOS system," *Bulletin de l'Academie nationale de medecine*, vol. 189, no. 2, pp. 287–97, 2005.
- [44] E. Amzallag-Bellenger *et al.*, "Idiopathic scoliosis in children and adolescents: assessment with a biplanar X-ray device," *Insights into imaging*, vol. 5, no. 5, pp. 571–583, 2014.

- [45] J. W. Raczkowski, B. Daniszewska, and K. Zolynski, "Functional scoliosis caused by leg length discrepancy," *Archives of medical science: AMS*, vol. 6, no. 3, p. 393, 2010.
- [46] K. Kaneko, Y. Aota, T. Sekiya, K. Yamada, and T. Saito, "Validation study of arm positions for evaluation of global spinal balance in EOS imaging," *European Journal of Orthopaedic Surgery & Traumatology*, vol. 26, Oct. 2016, doi: 10.1007/s00590-016-1813-8.
- [47] S. Deschênes *et al.*, "Diagnostic imaging of spinal deformities: reducing patients radiation dose with a new slot-scanning X-ray imager," *Spine*, vol. 35, no. 9, pp. 989–994, 2010.
- [48] T. Illés and S. Somoskeöy, "The EOS[™] imaging system and its uses in daily orthopaedic practice," *Int Orthop*, vol. 36, no. 7, pp. 1325–1331, Jul. 2012, doi: 10.1007/s00264-012-1512-y.
- [49] M. Burkus, Á. T. Schlégl, I. O'Sullivan, I. Márkus, C. Vermes, and M. Tunyogi-Csapó, "Sagittal plane assessment of spino-pelvic complex in a Central European population with adolescent idiopathic scoliosis: a case control study," *Scoliosis and spinal disorders*, vol. 13, no. 1, pp. 1–8, 2018.
- [50] D. H. Heo, Y. J. Cho, S. H. Sheen, M. S. Hong, S. M. Cho, and S. H. Park, "3D reconstructions of spinal segmental arteries using CT angiography: applications in minimally invasive spinal procedures," *American journal of neuroradiology*, vol. 31, no. 9, pp. 1635–1639, 2010.
- [51] D. A. Glaser, J. Doan, and P. O. Newton, "Comparison of 3-dimensional spinal reconstruction accuracy: biplanar radiographs with EOS versus computed tomography," *Spine (Phila Pa 1976)*, vol. 37, no. 16, pp. 1391–1397, Jul. 2012, doi: 10.1097/BRS.0b013e3182518a15.

- [52] S. Kuraishi et al., "Pedicle Morphology Using Computed Tomography-based Navigation System in Adolescent Idiopathic Scoliosis," *Journal of spinal disorders* & techniques, vol. 26, Sep. 2011, doi: 10.1097/BSD.0b013e31823162ef.
- [53] S. Langensiepen *et al.*, "Measuring procedures to determine the Cobb angle in idiopathic scoliosis: a systematic review," *European Spine Journal*, vol. 22, no. 11, pp. 2360–2371, 2013.
- [54] T. R. Kuklo, B. K. Potter, D. W. Polly Jr, M. F. O'Brien, T. M. Schroeder, and L. G. Lenke, "Reliability analysis for manual adolescent idiopathic scoliosis measurements," *Spine*, vol. 30, no. 4, pp. 444–454, 2005.
- [55] T. R. Kuklo *et al.*, "Reliability analysis for digital adolescent idiopathic scoliosis measurements," *Clinical Spine Surgery*, vol. 18, no. 2, pp. 152–159, 2005.
- [56] T. R. Kuklo, B. K. Potter, T. M. Schroeder, and M. F. O'Brien, "Comparison of Manual and Digital Measurements in Adolescent Idiopathic Scoliosis," *Spine*, vol. 31, no. 11, pp. 1240–1246, May 2006, doi: 10.1097/01.brs.0000217774.13433.a7.
- [57] N. Chung *et al.*, "Spinal phantom comparability study of Cobb angle measurement of scoliosis using digital radiographic imaging," *Journal of orthopaedic translation*, vol. 15, pp. 81–90, 2018.
- [58] M.-E. Lamarre *et al.*, "Assessment of spinal flexibility in adolescent idiopathic scoliosis: suspension versus side-bending radiography," *Spine*, vol. 34, no. 6, pp. 591–597, 2009.
- [59] S. Ohrt-Nissen, D. W. Hallager, M. Gehrchen, and B. Dahl, "Supine Lateral Bending Radiographs Predict the Initial In-brace Correction of the Providence Brace in Patients With Adolescent Idiopathic Scoliosis," *Spine (Phila Pa 1976)*, vol. 41, no. 9, pp. 798–802, May 2016, doi: 10.1097/BRS.00000000001519.
- [60] C. He *et al.*, "An effective assessment method of spinal flexibility to predict the initial in-orthosis correction on the patients with adolescent idiopathic scoliosis (AIS)," *PLoS One*, vol. 12, no. 12, Dec. 2017, doi: 10.1371/journal.pone.0190141.

- [61] K. Watanabe *et al.*, "Traction versus supine side-bending radiographs in determining flexibility: what factors influence these techniques?," *Spine (Phila Pa 1976)*, vol. 32, no. 23, pp. 2604–2609, Nov. 2007, doi: 10.1097/BRS.0b013e318158cbcb.
- [62] R. W. Liu, A. L. Teng, D. G. Armstrong, C. Poe-Kochert, J. P. Son-Hing, and G. H. Thompson, "Comparison of supine bending, push-prone, and traction under general anesthesia radiographs in predicting curve flexibility and postoperative correction in adolescent idiopathic scoliosis," *Spine*, vol. 35, no. 4, pp. 416–422, 2010.
- [63] M. Khodaei, C. Pachêco-Pereira, S. Trac, A. Chan, L. H. Le, and E. Lou, "Radiographic methods to estimate surgical outcomes based on spinal flexibility assessment in patients who have adolescent idiopathic scoliosis: A systematic review," *The Spine Journal*, vol. 18, no. 11, pp. 2128–2139, Nov. 2018, doi: 10.1016/j.spinee.2018.06.344.
- [64] K. M. C. Cheung and K. D. K. Luk, "Prediction of Correction of Scoliosis with Use of the Fulcrum Bending Radiograph*," *JBJS*, vol. 79, no. 8, pp. 1144–50, Aug. 1997.
- [65] C. Hirsch, B. Ilharreborde, and K. Mazda, "EOS suspension test for the assessment of spinal flexibility in adolescent idiopathic scoliosis," *Eur Spine J*, vol. 24, no. 7, pp. 1408–1414, Jul. 2015, doi: 10.1007/s00586-015-3771-y.
- [66] E. Melhem, A. Assi, R. El Rachkidi, and I. Ghanem, "EOS® biplanar X-ray imaging: concept, developments, benefits, and limitations," *Journal of children's orthopaedics*, vol. 10, no. 1, pp. 1–14, 2016.
- [67] T. Vrtovec, F. Pernuš, and B. Likar, "A review of methods for quantitative evaluation of axial vertebral rotation," *European Spine Journal*, vol. 18, no. 8, pp. 1079–1090, 2009.
- [68] J. R. Cobb, "Outline for the study of scoliosis," *Instr Course Lect AAOS*, vol. 5, pp. 261–275, 1948.
- [69] C. L. Nash and J. H. Moe, "A study of vertebral rotation," J Bone Joint Surg Am, vol. 51, no. 2, pp. 223–229, 1969.

- [70] R. Perdriolle and J. Vidal, "Thoracic idiopathic scoliosis curve evolution and prognosis," *Spine (Phila Pa 1976)*, vol. 10, no. 9, pp. 785–791, Nov. 1985, doi: 10.1097/00007632-198511000-00001.
- [71] C. M. Barsanti, A. deBari, and B. M. Covino, "The torsion meter: a critical review," *J Pediatr Orthop*, vol. 10, no. 4, pp. 527–531, Aug. 1990.
- [72] I. A. Stokes, L. C. Bigalow, and M. S. Moreland, "Measurement of axial rotation of vertebrae in scoliosis.," *Spine*, vol. 11, no. 3, pp. 213–218, 1986.
- [73] G. C. Lam, D. L. Hill, L. H. Le, J. V. Raso, and E. H. Lou, "Vertebral rotation measurement: a summary and comparison of common radiographic and CT methods," *Scoliosis*, vol. 3, p. 16, Nov. 2008, doi: 10.1186/1748-7161-3-16.
- [74] P. Cerny, I. Marik, and I. Pallova, "The radiographic method for evaluation of axial vertebral rotation-presentation of the new method," *Scoliosis*, vol. 9, no. 1, pp. 1–9, 2014.
- [75] E. K. Ho, S. S. Upadhyay, F. L. Chan, L. C. Hsu, and J. C. Leong, "New methods of measuring vertebral rotation from computed tomographic scans. An intraobserver and interobserver study on girls with scoliosis," *Spine (Phila Pa 1976)*, vol. 18, no. 9, pp. 1173–1177, Jul. 1993, doi: 10.1097/00007632-199307000-00008.
- [76] M. Yazici, E. R. Acaroglu, A. Alanay, V. Deviren, A. Cila, and A. Surat, "Measurement of vertebral rotation in standing versus supine position in adolescent idiopathic scoliosis," *J Pediatr Orthop*, vol. 21, no. 2, pp. 252–256, Apr. 2001.
- [77] G. Torell, A. Nachemson, K. Haderspeck-Grib, and A. Schultz, "Standing and Supine Cobb Measures in Girls with Idiopathic Scoliosis:," *Spine*, vol. 10, no. 5, pp. 425–427, Jun. 1985, doi: 10.1097/00007632-198506000-00004.
- [78] T. Illés and S. Somoskeöy, "Comparison of scoliosis measurements based on threedimensional vertebra vectors and conventional two-dimensional measurements: advantages in evaluation of prognosis and surgical results," *European spine journal*, vol. 22, no. 6, pp. 1255–1263, 2013.

- [79] A. M. Agha *et al.*, "Contribution of Screw Position to Correction of Axial Rotation in Adolescent Idiopathic Scoliosis".
- [80] D. Harrison, R. Cailliet, D. Harrison, T. Janik, and B. Holland, "Reliability of Centroid, Cobb, and Harrison Posterior Tangent Methods: Which to Choose for Analysis of Thoracic Kyphosis," *Spine*, vol. 26, pp. E227-34, Jun. 2001, doi: 10.1097/00007632-200106010-00002.
- [81] S. Goh, R. I. Price, P. J. Leedman, and K. P. Singer, "A comparison of three methods for measuring thoracic kyphosis: implications for clinical studies," *Rheumatology*, vol. 39, no. 3, pp. 310–315, Mar. 2000, doi: 10.1093/rheumatology/39.3.310.
- [82] S. Ohrt-Nissen *et al.*, "Reproducibility of thoracic kyphosis measurements in patients with adolescent idiopathic scoliosis," *Scoliosis and spinal disorders*, vol. 12, no. 1, pp. 1–8, 2017.
- [83] T. H. Tran *et al.*, "Correlations among four measures of thoracic kyphosis in older adults," *Osteoporos Int*, vol. 27, no. 3, pp. 1255–1259, Mar. 2016, doi: 10.1007/s00198-015-3368-7.
- [84] D. L. Carman, R. H. Browne, and J. G. Birch, "Measurement of scoliosis and kyphosis radiographs. Intraobserver and interobserver variation.," *The Journal of bone and joint surgery. American volume*, vol. 72, no. 3, pp. 328–333, 1990.
- [85] R. P. Jackson, T. Kanemura, N. Kawakami, and C. Hales, "Lumbopelvic lordosis and pelvic balance on repeated standing lateral radiographs of adult volunteers and untreated patients with constant low back pain," *Spine*, vol. 25, no. 5, pp. 575–586, 2000.
- [86] K. P. Singer, S. J. Edmondston, R. E. Day, and W. H. Breidahl, "Computer-assisted curvature assessment and Cobb angle determination of the thoracic kyphosis. An in vivo and in vitro comparison," *Spine (Phila Pa 1976)*, vol. 19, no. 12, pp. 1381– 1384, Jun. 1994, doi: 10.1097/00007632-199406000-00012.

- [87] D. J. Kolessar, G. T. Stollsteimer, and R. R. Betz, "The value of the measurement from T5 to T12 as a screening tool in detecting abnormal kyphosis," *J Spinal Disord*, vol. 9, no. 3, pp. 220–222, Jun. 1996.
- [88] R. P. Jackson and A. C. McManus, "Radiographic analysis of sagittal plane alignment and balance in standing volunteers and patients with low back pain matched for age, sex, and size. A prospective controlled clinical study," *Spine (Phila Pa 1976)*, vol. 19, no. 14, pp. 1611–1618, Jul. 1994, doi: 10.1097/00007632-199407001-00010.
- [89] A. Moskowitz and S. Trommanhauser, "Surgical and clinical results of scoliosis surgery using Zielke instrumentation," *Spine (Phila Pa 1976)*, vol. 18, no. 16, pp. 2444–2451, Dec. 1993, doi: 10.1097/00007632-199312000-00013.
- [90] M. Bernhardt and K. H. Bridwell, "Segmental analysis of the sagittal plane alignment of the normal thoracic and lumbar spines and thoracolumbar junction," *Spine (Phila Pa 1976)*, vol. 14, no. 7, pp. 717–721, Jul. 1989, doi: 10.1097/00007632-198907000-00012.
- [91] R. Fernand and D. E. Fox, "Evaluation of lumbar lordosis. A prospective and retrospective study," *Spine (Phila Pa 1976)*, vol. 10, no. 9, pp. 799–803, Nov. 1985, doi: 10.1097/00007632-198511000-00003.
- [92] T. R. Yochum and L. J. Rowe, "Essentials of skeletal radiology," 1987.
- [93] Polly DW Jr, Kilkelly FX, McHale KA, Asplund LM, Mulligan M, Chang AS. Measurement of lumbar lordosis. Evaluation of intraobserver, interobserver, and technique variability. *Spine (Phila Pa 1976)*. 1996 Jul 1;21(13):1530-5; discussion 1535-6. doi: 10.1097/00007632-199607010-00008. PMID: 8817780.
- [94] P. Knott et al., "SOSORT 2012 consensus paper: reducing x-ray exposure in pediatric patients with scoliosis," Scoliosis, vol. 9, no. 1, p. 4, Apr. 2014, doi: 10.1186/1748-7161-9-4.

- [95] L. H. Thompson, "Recognition, signaling, and repair of DNA double-strand breaks produced by ionizing radiation in mammalian cells: the molecular choreography," *Mutation Research/Reviews in Mutation Research*, vol. 751, no. 2, pp. 158–246, 2012.
- [96] D. J. Shah, R. K. Sachs, and D. J. Wilson, "Radiation-induced cancer: a modern view," *The British journal of radiology*, vol. 85, no. 1020, pp. e1166–e1173, 2012.
- [97] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine*, vol. 21, no. 13, pp. 1540–1547, 1996.
- [98] C. M. Ronckers, C. E. Land, J. S. Miller, M. Stovall, J. E. Lonstein, and M. M. Doody, "Cancer mortality among women frequently exposed to radiographic examinations for spinal disorders," *Radiation research*, vol. 174, no. 1, pp. 83–90, 2010.
- [99] T. D. Luo, A. A. Stans, B. A. Schueler, and A. N. Larson, "Cumulative Radiation Exposure With EOS Imaging Compared With Standard Spine Radiographs," *Spine Deform*, vol. 3, no. 2, pp. 144–150, Mar. 2015, doi: 10.1016/j.jspd.2014.09.049.
- [100] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine (Phila Pa 1976)*, vol. 21, no. 13, pp. 1540–1547; discussion 1548, Jul. 1996, doi: 10.1097/00007632-199607010-00011.
- [101] G. Kalifa *et al.*, "Evaluation of a new low-dose digital x-ray device: first dosimetric and clinical results in children," *Pediatr Radiol*, vol. 28, no. 7, pp. 557–561, Jul. 1998, doi: 10.1007/s002470050413.
- [102] M. Chung et al., Emerging MRI Technologies for Imaging Musculoskeletal Disorders Under Loading Stress. Rockville (MD): Agency for Healthcare Research and Quality (US), 2011. Accessed: Jun. 09, 2021. [Online]. Available: http://www.ncbi.nlm.nih.gov/books/NBK82287/

- [103] D. Birchall, D. Hughes, B. Gregson, and B. Williamson, "Demonstration of vertebral and disc mechanical torsion in adolescent idiopathic scoliosis using threedimensional MR imaging," *European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society*, vol. 14, pp. 123–9, Apr. 2005, doi: 10.1007/s00586-004-0705-5.
- [104] D. Birchall, D. G. Hughes, J. Hindle, L. Robinson, and J. B. Williamson, "Measurement of vertebral rotation in adolescent idiopathic scoliosis using threedimensional magnetic resonance imaging," *Spine (Phila Pa 1976)*, vol. 22, no. 20, pp. 2403–2407, Oct. 1997, doi: 10.1097/00007632-199710150-00016.
- [105] W. W. Lovell, R. B. Winter, R. T. Morrissy, and S. L. Weinstein, Lovell and Winter's pediatric orthopaedics. Philadelphia: Lippincott Williams & Wilkins, 2006.
- [106] M. Thomsen and R. Abel, "Imaging in scoliosis from the orthopaedic surgeon's point of view," *Eur J Radiol*, vol. 58, no. 1, pp. 41–47, Apr. 2006, doi: 10.1016/j.ejrad.2005.12.003.
- [107] M. Dewey, T. Schink, and C. F. Dewey, "Claustrophobia during magnetic resonance imaging: cohort study in over 55,000 patients," *J Magn Reson Imaging*, vol. 26, no. 5, pp. 1322–1327, Nov. 2007, doi: 10.1002/jmri.21147.
- [108] J. T. Bushberg, The Essential Physics of Medical Imaging. Lippincott Williams & Wilkins, 2002.
- [109] S. Suzuki, T. Yamamuro, J. Shikata, K. Shimizu, and H. Iida, "Ultrasound measurement of vertebral rotation in idiopathic scoliosis," *J Bone Joint Surg Br*, vol. 71, no. 2, pp. 252–255, Mar. 1989, doi: 10.1302/0301-620X.71B2.2647754.
- [110] Y.-P. Zheng *et al.*, "A reliability and validity study for Scolioscan: a radiation-free scoliosis assessment system using 3D ultrasound imaging," *Scoliosis and Spinal Disorders*, vol. 11, no. 1, p. 13, May 2016, doi: 10.1186/s13013-016-0074-y.

- [111] M. Young, D. L. Hill, R. Zheng, and E. Lou, "Reliability and accuracy of ultrasound measurements with and without the aid of previous radiographs in adolescent idiopathic scoliosis (AIS)," *Eur Spine J*, vol. 24, no. 7, pp. 1427–1433, Jul. 2015, doi: 10.1007/s00586-015-3855-8.
- [112] W. Chen, L. H. Le, and E. H. M. Lou, "Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study," *Eur Spine J*, vol. 25, no. 10, pp. 3265–3273, Oct. 2016, doi: 10.1007/s00586-016-4492-6.
- [113] R. Zheng et al., "Intra- and Inter-rater Reliability of Coronal Curvature Measurement for Adolescent Idiopathic Scoliosis Using Ultrasonic Imaging Method-A Pilot Study," Spine Deform, vol. 3, no. 2, pp. 151–158, Mar. 2015, doi: 10.1016/j.jspd.2014.08.008.
- [114] R. C. Brink *et al.*, "A reliability and validity study for different coronal angles using ultrasound imaging in adolescent idiopathic scoliosis," *The Spine Journal*, vol. 18, no. 6, pp. 979–985, Jun. 2018, doi: 10.1016/j.spinee.2017.10.012.
- [115] C.-W. J. Cheung, G.-Q. Zhou, S.-Y. Law, K.-L. Lai, W.-W. Jiang, and Y.-P. Zheng, "Freehand three-dimensional ultrasound system for assessment of scoliosis," J Orthop Translat, vol. 3, no. 3, pp. 123–133, Jun. 2015, doi: 10.1016/j.jot.2015.06.001.
- [116] C.-W. J. Cheung, G.-Q. Zhou, S.-Y. Law, T.-M. Mak, K.-L. Lai, and Y.-P. Zheng, "Ultrasound Volume Projection Imaging for Assessment of Scoliosis," *IEEE Trans Med Imaging*, vol. 34, no. 8, pp. 1760–1768, Aug. 2015, doi: 10.1109/TMI.2015.2390233.
- [117] H.-D. Wu, W. Liu, and M.-S. Wong, "Reliability and validity of lateral curvature assessments using clinical ultrasound for the patients with scoliosis: a systematic review," *Eur Spine J*, vol. 29, no. 4, pp. 717–725, Apr. 2020, doi: 10.1007/s00586-019-06280-y.

- [118] S. Trac, R. Zheng, D. L. Hill, and E. Lou, "Intra- and Interrater Reliability of Cobb Angle Measurements on the Plane of Maximum Curvature Using Ultrasound Imaging Method," *Spine Deform*, vol. 7, no. 1, pp. 18–26, Jan. 2019, doi: 10.1016/j.jspd.2018.06.015.
- [119] J. C. Cheng, X. Guo, and A. H. Sher, "Persistent osteopenia in adolescent idiopathic scoliosis. A longitudinal follow up study," *Spine (Phila Pa 1976)*, vol. 24, no. 12, pp. 1218–1222, Jun. 1999, doi: 10.1097/00007632-199906150-00008.
- [120] H.-W. Deng, F.-H. Xu, K. M. Davies, R. Heaney, and R. R. Recker, "Differences in bone mineral density, bone mineral content, and bone areal size in fracturing and non-fracturing women, and their interrelationships at the spine and hip," *J Bone Miner Metab*, vol. 20, no. 6, pp. 358–366, 2002, doi: 10.1007/s007740200052.
- [121] A. Licata, "Bone density vs bone quality: what's a clinician to do?," *Cleve Clin J Med*, vol. 76, no. 6, pp. 331–336, Jun. 2009, doi: 10.3949/ccjm.76a.08041.
- [122] J. C. Cheng *et al.*, "Generalized low areal and volumetric bone mineral density in adolescent idiopathic scoliosis," *J Bone Miner Res*, vol. 15, no. 8, pp. 1587–1595, Aug. 2000, doi: 10.1359/jbmr.2000.15.8.1587.
- [123] V. W. Y. Hung *et al.*, "Osteopenia: a new prognostic factor of curve progression in adolescent idiopathic scoliosis," *J Bone Joint Surg Am*, vol. 87, no. 12, pp. 2709–2716, Dec. 2005, doi: 10.2106/JBJS.D.02782.
- [124] Y. K. Jeon *et al.*, "Effect of increased axial rotation angle on bone mineral density measurements of the lumbar spine," *Spine J*, vol. 14, no. 9, pp. 2150–2154, Sep. 2014, doi: 10.1016/j.spinee.2014.01.052.
- [125] J. Cheng, H. Sher, X. Guo, V. Hung, and A. Cheung, "The effect of vertebral rotation of the lumbar spine on dual energy X-ray absorptiometry measurements: observational study," p. 5.

- [126] C.-C. Glüer, C. Y. Wu, and H. K. Genant, "Broadband ultrasound attenuation signals depend on trabecular orientation: An in vitro study," *Osteoporosis Int*, vol. 3, no. 4, pp. 185–191, Jul. 1993, doi: 10.1007/BF01623674.
- [127] G. I. Baroncelli, "Quantitative Ultrasound Methods to Assess Bone Mineral Status in Children: Technical Characteristics, Performance, and Clinical Application," *Pediatr Res*, vol. 63, no. 3, Art. no. 3, Mar. 2008, doi: 10.1203/PDR.0b013e318163a286.
- [128] P. Laugier, "Quantitative ultrasound of bone: looking ahead," *Joint Bone Spine*, vol. 73, no. 2, pp. 125–128, Mar. 2006, doi: 10.1016/j.jbspin.2005.10.012.
- [129] T. P. Lam *et al.*, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound Med Biol*, vol. 39, no. 3, pp. 381–387, Mar. 2013, doi: 10.1016/j.ultrasmedbio.2012.09.012.
- [130] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for detection of dranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS)-a prospective cohort study till skeletal maturity," *Osteoprosis International*, 2010, vol. 21, pp. S686-S687.
- [131] T. P. Lam *et al.*, "Abnormal bone quality in adolescent idiopathic scoliosis: a casecontrol study on 635 subjects and 269 normal controls with bone densitometry and quantitative ultrasound," *Spine (Phila Pa 1976)*, vol. 36, no. 15, pp. 1211–1217, Jul. 2011, doi: 10.1097/BRS.0b013e3181ebab39.
- [132] R. Zheng, L. Le, D. Hill, and E. Lou, "Estimation of bone quality on scoliotic subjects using ultrasound reflection imaging method - a preliminary study," Oct. 2015, pp. 1–4. doi: 10.1109/ULTSYM.2015.0218.
- [133] J. E. Lonstein and J. M. Carlson, "The prediction of curve progression in untreated idiopathic scoliosis during growth," *J Bone Joint Surg Am*, vol. 66, no. 7, pp. 1061– 1071, Sep. 1984.

- [134] K.-J. Tan, M. M. Moe, R. Vaithinathan, and H.-K. Wong, "Curve progression in idiopathic scoliosis: follow-up study to skeletal maturity," *Spine*, vol. 34, no. 7, pp. 697–700, 2009.
- [135] W. T. Lee *et al.*, "Association of osteopenia with curve severity in adolescent idiopathic scoliosis: a study of 919 girls," *Osteoporosis international*, vol. 16, no. 12, pp. 1924–1932, 2005.
- [136] D. Zhao, G. Qiu, Y. Wang, J. Zhang, J. Shen, and Z. Wu, "Association between adolescent idiopathic scoliosis with double curve and polymorphisms of calmodulin1 gene/estrogen receptor-α gene," *Orthop Surg*, vol. 1, no. 3, pp. 222– 230, Jul. 2009, doi: 10.1111/j.1757-7861.2009.00038.x.
- [137] H. Y. Yeung *et al.*, "Genetic association study of insulin-like growth factor-I (IGF-I) gene with curve severity and osteopenia in adolescent idiopathic scoliosis," *Stud Health Technol Inform*, vol. 123, pp. 18–24, 2006.
- [138] D. Steinberg, "CART: Classification and Regression Trees," in *The Top Ten Algorithms in Data Mining*, Chapman and Hall/CRC, 2009.
- [139] Y. Qi, "Random Forest for Bioinformatics," in *Ensemble Machine Learning: Methods and Applications*, C. Zhang and Y. Ma, Eds. Boston, MA: Springer US, 2012, pp. 307–323. doi: 10.1007/978-1-4419-9326-7_11.
- [140] X. Wu *et al.*, "Top 10 algorithms in data mining," *Knowl Inf Syst*, vol. 14, no. 1, pp. 1–37, Jan. 2008, doi: 10.1007/s10115-007-0114-2.
- [141] A. P. Engelbrecht, Computational Intelligence: An Introduction. John Wiley & Sons, 2007.
- [142] S. Halabi and K. Owzar, "The Importance of Identifying and Validating Prognostic Factors in Oncology," *Semin Oncol*, vol. 37, no. 2, pp. e9-18, Apr. 2010, doi: 10.1053/j.seminoncol.2010.04.001.

Chapter 3

- [1] S. Negrini, T. M. Hresko, J. P. O'Brien, N. Price, SOSORT Boards, and SRS Non-Operative Committee, "Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee," *Scoliosis*, vol. 10, p. 8, 2015, doi: 10.1186/s13013-014-0025-4.
- [2] P. N. Soucacos *et al.*, "Assessment of curve progression in idiopathic scoliosis," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 7, no. 4, pp. 270–277, 1998, doi: 10.1007/s005860050074.
- [3] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine*, vol. 21, no. 13, pp. 1540–1547, 1996.
- [4] V. W. Y. Hung *et al.*, "Osteopenia: a new prognostic factor of curve progression in adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 87, no. 12, pp. 2709–2716, Dec. 2005, doi: 10.2106/JBJS.D.02782.
- [5] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for detection of dranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS)-a prospective cohort study till skeletal maturity," *Osteoprosis International*, 2010, vol. 21, pp. S686-S687.
- [6] J. E. Lonstein and J. M. Carlson, "The prediction of curve progression in untreated idiopathic scoliosis during growth," *J. Bone Joint Surg. Am.*, vol. 66, no. 7, pp. 1061–1071, Sep. 1984.
- M.-L. Nault, J.-M. Mac-Thiong, M. Roy-Beaudry, J. deGuise, H. Labelle, and S. Parent, "Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis," *J. Pediatr. Orthop.*, vol. 33, no. 6, pp. 618–623, Sep. 2013, doi: 10.1097/BPO.0b013e318292462a.

- [8] A. Noshchenko *et al.*, "Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with meta-analysis," *World J. Orthop.*, vol. 6, no. 7, p. 537, 2015.
- [9] K.-J. Tan, M. M. Moe, R. Vaithinathan, and H.-K. Wong, "Curve progression in idiopathic scoliosis: follow-up study to skeletal maturity," *Spine*, vol. 34, no. 7, pp. 697–700, 2009.
- [10] L. E. Peterson and A. L. Nachemson, "Prediction of progression of the curve in girls who have adolescent idiopathic scoliosis of moderate severity. Logistic regression analysis based on data from The Brace Study of the Scoliosis Research Society," *J. Bone Joint Surg. Am.*, vol. 77, no. 6, pp. 823–827, Jun. 1995, doi: 10.2106/00004623-199506000-00002.
- [11] D. Moher, A. Liberati, J. Tetzlaff, and D. G. Altman, "Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement," *Ann. Intern. Med.*, vol. 151, no. 4, pp. 264–269, Aug. 2009, doi: 10.7326/0003-4819-151-4-200908180-00135.
- [12] T. S. Zotero, "George Mason University," *Fairfax VA USA Cent. Hist. New Media*, 2016.
- [13] J. A. Hayden, D. A. van der Windt, J. L. Cartwright, P. Côté, and C. Bombardier, "Assessing bias in studies of prognostic factors," *Ann. Intern. Med.*, vol. 158, no. 4, pp. 280–286, Feb. 2013, doi: 10.7326/0003-4819-158-4-201302190-00009.
- [14] L. R. Cornelius, J. J. L. van der Klink, J. W. Groothoff, and S. Brouwer, "Prognostic factors of long term disability due to mental disorders: a systematic review," J. Occup. Rehabil., vol. 21, no. 2, pp. 259–274, Jun. 2011, doi: 10.1007/s10926-010-9261-5.
- [15] M. L. McHugh, "Interrater reliability: the kappa statistic," *Biochem. Medica*, vol. 22, no. 3, pp. 276–282, 2012.

- [16] J. Cheung *et al.*, "The relation between electromyography and growth velocity of the spine in the evaluation of curve progression in idiopathic scoliosis," *Spine*, vol. 29, no. 9, pp. 1011–1016, May 2004, doi: 10.1097/00007632-200405010-00012.
- [17] L. Davis, J. S. Murphy, K. A. Shaw, K. Cash, D. P. Devito, and M. L. Schmitz, "Nighttime bracing with the Providence thoracolumbosacral orthosis for treatment of adolescent idiopathic scoliosis: A retrospective consecutive clinical series," *Prosthet. Orthot. Int.*, vol. 43, no. 2, pp. 158–162, Apr. 2019, doi: 10.1177/0309364618792727.
- [18] T. Lara *et al.*, "The Risk of Curve Progression and Surgery in African Americans With Adolescent Idiopathic Scoliosis," *Spine Deform.*, vol. 5, no. 4, pp. 250–254, Jul. 2017, doi: 10.1016/j.jspd.2017.01.013.
- [19] S. Mao *et al.*, "Initial Cobb angle reduction velocity following bracing as a new predictor for curve progression in adolescent idiopathic scoliosis," *Eur. Spine J.*, vol. 25, no. 2, pp. 500–505, Feb. 2016, doi: 10.1007/s00586-015-3937-7.
- [20] P. J. O'Neill *et al.*, "Decreased orthotic effectiveness in overweight patients with adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 87, no. 5, pp. 1069– 1074, May 2005, doi: 10.2106/JBJS.C.01707.
- [21] S. Ohrt-Nissen, D. W. Hallager, M. Gehrchen, and B. Dahl, "Flexibility Predicts Curve Progression in Providence Nighttime Bracing of Patients With Adolescent Idiopathic Scoliosis," *Spine*, vol. 41, no. 22, pp. 1724–1730, Nov. 2016, doi: 10.1097/BRS.00000000001634.
- [22] P. Sitoula *et al.*, "Prediction of Curve Progression in Idiopathic Scoliosis: Validation of the Sanders Skeletal Maturity Staging System," *Spine*, vol. 40, no. 13, pp. 1006– 1013, Jul. 2015, doi: 10.1097/BRS.00000000000952.
- [23] X. Sun *et al.*, "Osteopenia predicts curve progression of adolescent idiopathic scoliosis in girls treated with brace treatment," *J. Pediatr. Orthop.*, vol. 33, no. 4, pp. 366–371, Jun. 2013, doi: 10.1097/BPO.0b013e31827b7b5f.

- [24] F. Zaina, S. Donzelli, and S. Negrini, "Overweight is not predictive of bracing failure in adolescent idiopathic scoliosis: results from a retrospective cohort study," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 26, no. 6, pp. 1670–1675, Jun. 2017, doi: 10.1007/s00586-017-4985y.
- [25] G. Duval-Beaupère, "Rib hump and supine angle as prognostic factors for mild scoliosis," *Spine*, vol. 17, no. 1, pp. 103–107, Jan. 1992, doi: 10.1097/00007632-199201000-00015.
- [26] M. R. Konieczny, P. Hieronymus, and R. Krauspe, "Time in brace: where are the limits and how can we improve compliance and reduce negative psychosocial impact in patients with scoliosis? A retrospective analysis," *Spine J. Off. J. North Am. Spine Soc.*, vol. 17, no. 11, pp. 1658–1664, Nov. 2017, doi: 10.1016/j.spinee.2017.05.010.
- [27] T. Kotwicki, A. Walczak, and A. Szulc, "Trunk rotation and hip joint range of rotation in adolescent girls with idiopathic scoliosis: does the" dinner plate" turn asymmetrically?," *Scoliosis*, vol. 3, no. 1, pp. 1–11, 2008.
- [28] A. G. Aulisa, V. Guzzanti, M. Galli, C. Perisano, F. Falciglia, and L. Aulisa, "Treatment of thoraco-lumbar curves in adolescent females affected by idiopathic scoliosis with a progressive action short brace (PASB): assessment of results according to the SRS committee on bracing and nonoperative management standardization criteria," *Scoliosis*, vol. 4, no. 1, pp. 1–8, 2009.
- [29] A. G. Aulisa, M. Giordano, F. Falciglia, E. Marzetti, A. Poscia, and V. Guzzanti, "Correlation between compliance and brace treatment in juvenile and adolescent idiopathic scoliosis: SOSORT 2014 award winner," *scoliosis*, vol. 9, no. 1, pp. 1–9, 2014.
- [30] D. D. Bohl, C. J. Telles, N. S. Golinvaux, B. A. Basques, P. A. DeLuca, and J. N. Grauer, "Effectiveness of Providence nighttime bracing in patients with adolescent idiopathic scoliosis," *Orthopedics*, vol. 37, no. 12, pp. e1085–e1090, 2014.

- [31] W. P. Bunnell, "The natural history of idiopathic scoliosis before skeletal maturity.," *Spine*, vol. 11, no. 8, pp. 773–776, 1986.
- [32] Y. P. Charles, F. Canavese, and A. Diméglio, "Curve progression risk in a mixed series of braced and nonbraced patients with idiopathic scoliosis related to skeletal maturity assessment on the olecranon," *J. Pediatr. Orthop. B*, vol. 26, no. 3, pp. 240– 244, 2017.
- [33] W. H. P. Cheung, K. D. K. Luk, D. Samartzis, and J. P. Y. Cheung, "Curve progression matching with skeletal growth in adolescent idiopathic scoliosis using the distal radius and ulna (DRU) classification," 2018.
- [34] J. P. Y. Cheung, P. W. H. Cheung, D. Samartzis, and K. D.-K. Luk, "APSS-ASJ Best Clinical Research Award: Predictability of Curve Progression in Adolescent Idiopathic Scoliosis Using the Distal Radius and Ulna Classification," *Asian Spine J.*, vol. 12, no. 2, pp. 202–213, Apr. 2018, doi: 10.4184/asj.2018.12.2.202.
- [35] C. R. D'Amato, S. Griggs, and B. McCoy, "Nighttime bracing with the Providence brace in adolescent girls with idiopathic scoliosis," *Spine*, vol. 26, no. 18, pp. 2006– 2012, Sep. 2001, doi: 10.1097/00007632-200109150-00014.
- [36] A. J. Danielsson, R. Hasserius, A. Ohlin, and A. L. Nachemson, "A prospective study of brace treatment versus observation alone in adolescent idiopathic scoliosis: a follow-up mean of 16 years after maturity," *Spine*, vol. 32, no. 20, pp. 2198–2207, Sep. 2007, doi: 10.1097/BRS.0b013e31814b851f.
- [37] M.-Q. Fang, C. Wang, G.-H. Xiang, C. Lou, N.-F. Tian, and H.-Z. Xu, "Long-term effects of the Chêneau brace on coronal and sagittal alignment in adolescent idiopathic scoliosis," *J. Neurosurg. Spine*, vol. 23, no. 4, pp. 505–509, Oct. 2015, doi: 10.3171/2015.2.SPINE14970.
- [38] R. Gepstein *et al.*, "Effectiveness of the Charleston bending brace in the treatment of single-curve idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 22, no. 1, pp. 84–87, Feb. 2002.

- [39] C. J. Goldberg, D. P. Moore, E. E. Fogarty, and F. E. Dowling, "Adolescent idiopathic scoliosis: the effect of brace treatment on the incidence of surgery," *Spine*, vol. 26, no. 1, pp. 42–47, Jan. 2001, doi: 10.1097/00007632-200101010-00009.
- [40] C. M. Goodbody, I. B. Asztalos, W. N. Sankar, and J. M. Flynn, "It's not just the big kids: both high and low BMI impact bracing success for adolescent idiopathic scoliosis," *J. Child. Orthop.*, vol. 10, no. 5, pp. 395–404, Oct. 2016, doi: 10.1007/s11832-016-0763-3.
- [41] J. Guo *et al.*, "Pelvic tilt and trunk inclination: new predictive factors in curve progression during the Milwaukee bracing for adolescent idiopathic scoliosis," *Eur. Spine J.*, vol. 21, no. 10, pp. 2050–2058, Oct. 2012, doi: 10.1007/s00586-012-2409-6.
- [42] J. Guo et al., "A prospective randomized controlled study on the treatment outcome of SpineCor brace versus rigid brace for adolescent idiopathic scoliosis with followup according to the SRS standardized criteria," Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc., vol. 23, no. 12, pp. 2650–2657, Dec. 2014, doi: 10.1007/s00586-013-3146-1.
- [43] L. A. Karol, C. E. Johnston, R. H. Browne, and M. Madison, "Progression of the curve in boys who have idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 75, no. 12, pp. 1804–1810, Dec. 1993, doi: 10.2106/00004623-199312000-00010.
- [44] L. A. Karol, D. Virostek, K. Felton, C. Jo, and L. Butler, "The Effect of the Risser Stage on Bracing Outcome in Adolescent Idiopathic Scoliosis," *J. Bone Joint Surg. Am.*, vol. 98, no. 15, pp. 1253–1259, Aug. 2016, doi: 10.2106/JBJS.15.01313.
- [45] L. A. Karol, "Effectiveness of bracing in male patients with idiopathic scoliosis," *Spine*, vol. 26, no. 18, pp. 2001–2005, Sep. 2001, doi: 10.1097/00007632-200109150-00013.

- [46] D. E. Katz and A. A. Durrani, "Factors that influence outcome in bracing large curves in patients with adolescent idiopathic scoliosis," *Spine*, vol. 26, no. 21, pp. 2354–2361, Nov. 2001, doi: 10.1097/00007632-200111010-00012.
- [47] D. E. Katz, J. A. Herring, R. H. Browne, D. M. Kelly, and J. G. Birch, "Brace wear control of curve progression in adolescent idiopathic scoliosis," *J. Bone Joint Surg. Am.*, vol. 92, no. 6, pp. 1343–1352, Jun. 2010, doi: 10.2106/JBJS.I.01142.
- [48] H. Kuroki, N. Inomata, H. Hamanaka, K. Higa, E. Chosa, and N. Tajima, "Efficacy of the Osaka Medical College (OMC) brace in the treatment of adolescent idiopathic scoliosis following Scoliosis Research Society brace studies criteria," *Scoliosis*, vol. 10, p. 12, Apr. 2015, doi: 10.1186/s13013-015-0036-9.
- [49] T. P. Lam *et al.*, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound Med. Biol.*, vol. 39, no. 3, pp. 381–387, Mar. 2013, doi: 10.1016/j.ultrasmedbio.2012.09.012.
- [50] R. LeBlanc, H. Labelle, F. Forest, and B. Poitras, "Morphologic discrimination among healthy subjects and patients with progressive and nonprogressive adolescent idiopathic scoliosis," *Spine*, vol. 23, no. 10, pp. 1109–1115, 1998.
- [51] C. S. Lee *et al.*, "Effectiveness of the Charleston night-time bending brace in the treatment of adolescent idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 32, no. 4, pp. 368–372, 2012.
- [52] M.-L. Nault *et al.*, "Three-Dimensional Spinal Morphology Can Differentiate Between Progressive and Nonprogressive Patients With Adolescent Idiopathic Scoliosis at the Initial Presentation," *Spine*, vol. 39, no. 10, pp. E601–E606, May 2014, doi: 10.1097/BRS.00000000000284.
- [53] G. Pasquini *et al.*, "The outcome of a modified version of the Cheneau brace in adolescent idiopathic scoliosis (AIS) based on SRS and SOSORT criteria: a retrospective study," *Eur J Phys Rehabil Med*, vol. 52, no. 5, pp. 618–629, 2016.

- [54] A. Ploumis, V. Trivedi, J.-H. Shin, K. B. Wood, and B. E. Grottkau, "Progression of idiopathic thoracic or thoracolumbar scoliosis and pelvic obliquity in adolescent patients with and without limb length discrepancy," *Scoliosis Spinal Disord.*, vol. 13, no. 1, pp. 1–7, 2018.
- [55] P. M. Ryan, E. G. Puttler, W. M. Stotler, and R. L. Ferguson, "Role of the triradiate cartilage in predicting curve progression in adolescent idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 27, no. 6, pp. 671–676, 2007.
- [56] B. Shi *et al.*, "Curve progression in adolescent idiopathic scoliosis with a minimum of 2 years' follow-up after completed brace weaning with reference to the SRS standardized criteria," *Spine Deform.*, vol. 4, no. 3, pp. 200–205, 2016.
- [57] X. Sun *et al.*, "Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis," *Orthop. Surg.*, vol. 2, no. 4, pp. 285–290, 2010.
- [58] R. M. Thompson, E. W. Hubbard, C.-H. Jo, D. Virostek, and L. A. Karol, "Brace Success Is Related to Curve Type in Patients with Adolescent Idiopathic Scoliosis," *JBJS*, vol. 99, no. 11, pp. 923–928, Jun. 2017, doi: 10.2106/JBJS.16.01050.
- [59] J. M. Trivedi and J. D. Thomson, "Results of Charleston bracing in skeletally immature patients with idiopathic scoliosis," *J. Pediatr. Orthop.*, vol. 21, no. 3, pp. 277–280, Jun. 2001.
- [60] V. Vijvermans, G. Fabry, and J. Nijs, "Factors determining the final outcome of treatment of idiopathic scoliosis with the Boston brace: a longitudinal study," J. Pediatr. Orthop. Part B, vol. 13, no. 3, pp. 143–149, May 2004, doi: 10.1097/00009957-200405000-00001.
- [61] H. R. Weiss, "The Progression of Idiopathic Scoliosis under the Influence of a Physiotherapy Rehabilitation Programme," *Physiotherapy*, vol. 78, no. 11, pp. 815– 821, Nov. 1992, doi: 10.1016/S0031-9406(10)60452-X.
- [62] H. Wu, J. L. Ronsky, F. Cheriet, J. Harder, J. C. Küpper, and R. F. Zernicke, "Time series spinal radiographs as prognostic factors for scoliosis and progression of spinal

deformities," *Eur. Spine J.*, vol. 20, no. 1, pp. 112–117, Jan. 2011, doi: 10.1007/s00586-010-1512-9.

- [63] Y. Yamauchi, T. Yamaguchi, and Y. Asaka, "Prediction of curve progression in idiopathic scoliosis based on initial roentgenograms. A proposal of an equation," *Spine*, vol. 13, no. 11, pp. 1258–1261, Nov. 1988, doi: 10.1097/00007632-198811000-00009.
- [64] B. H. K. Yip *et al.*, "Prognostic Value of Bone Mineral Density on Curve Progression: A Longitudinal Cohort Study of 513 Girls with Adolescent Idiopathic Scoliosis," *Sci. Rep.*, vol. 6, p. 39220, Dec. 2016, doi: 10.1038/srep39220.
- [65] M. Ylikoski, "Spinal growth and progression of adolescent idiopathic scoliosis," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 1, no. 4, pp. 236–239, Mar. 1993, doi: 10.1007/BF00298366.
- [66] T. Yrjönen and M. Ylikoski, "Effect of growth velocity on the progression of adolescent idiopathic scoliosis in boys," *J. Pediatr. Orthop. Part B*, vol. 15, no. 5, pp. 311–315, Sep. 2006, doi: 10.1097/01202412-200609000-00001.
- [67] T. Yrjönen, M. Ylikoski, D. Schlenzka, R. Kinnunen, and M. Poussa, "Effectiveness of the Providence nighttime bracing in adolescent idiopathic scoliosis: a comparative study of 36 female patients," *Eur. Spine J.*, vol. 15, no. 7, pp. 1139–1143, Jul. 2006, doi: 10.1007/s00586-005-0049-9.
- [68] Y. Zhang *et al.*, "Factors relating to curve progression in female patients with adolescent idiopathic scoliosis treated with a brace," *Eur. Spine J.*, vol. 24, no. 2, pp. 244–248, Feb. 2015, doi: 10.1007/s00586-014-3674-3.
- [69] Z. Zhu et al., "Is Brace Treatment Appropriate for Adolescent Idiopathic Scoliosis Patients Refusing Surgery With Cobb Angle Between 40 and 50 Degrees," Clin. Spine Surg., vol. 30, no. 2, pp. 85–89, Mar. 2017, doi: 10.1097/BSD.0b013e3182a1de29.

Chapter 4 (Section 4.2)

[1] J. A. Farady, "Current principles in the nonoperative management of structural adolescent idiopathic scoliosis," *Physical Therapy*, vol. 63, no. 4, pp. 512–523, 1983.

[2] S. L. Weinstein, L. A. Dolan, J. C. Cheng, A. Danielsson, and J. A. Morcuende, "Adolescent idiopathic scoliosis," *The Lancet*, vol. 371, no. 9623, pp. 1527–1537, 2008.

[3] J. C. Cheng, R. M. Castelein, W. C. Chu et al., "Adolescent idiopathic scoliosis," *Nature Reviews Disease Primers*, vol. 1, no. 1, pp. 1–21, 2015.

[4] C. B. Barton and S. L. Weinstein, "Adolescent idiopathic scoliosis: natural history," In *Pathogenesis of Idiopathic Scoliosis*, Springer, Tokyo, pp. 27–50, 2018.

[5] E. Diarbakerli, A. Grauers, A. Danielsson, A. Abbott, and P. Gerdhem, "Quality of life in males and females with idiopathic scoliosis," *Spine*, vol. 44, no. 6, pp. 404–410, 2019.

[6] S. D. Cook, A. F. Harding, E. L. Morgan et al., "Trabecular bone mineral density in idiopathic scoliosis," *Journal of Pediatric Orthopedics*, vol. 7, no. 2, pp. 168–174, 1987.

[7] J. C. Cheng, L. Qin, C. S. Cheung et al, "Generalized low areal and volumetric bone mineral density in adolescent idiopathic scoliosis," *Journal of Bone and Mineral Research*, vol. 15, no. 8, pp. 1587–1595, 2000.

[8] W. T. Lee, C. S. Cheung, K. T. Yee et al., "Association of osteopenia with curve severity in adolescent idiopathic scoliosis: a study of 919 girls," *Osteoporosis International*, vol. 16, no. 12, pp. 1924–1932, 2005.

[9] X. F. Li, H. Li, Z. D. Liu, and L. Y. Dai, "Low bone mineral status in adolescent idiopathic scoliosis," *European Spine Journal*, vol. 17, no. 11, pp. 1431–1440, 2008.

[10] T. P. Lam, V. W. Y. Hung, H. Y. Yeung et al., "Abnormal bone quality in adolescent idiopathic scoliosis: a case-control study on 635 subjects and 269 normal controls with bone densitometry and quantitative ultrasound," *Spine*, vol. 36, no. 15, pp. 1211–1217, 2011.

[11] A. Klibanski, L. Adams-Campbell, T. Bassford et al., "Osteoporosis prevention, diagnosis, and therapy," *Journal of the American Medical Association*, vol. 285, no. 6, pp. 785–795, 2001.

[12] M. Machida, S. L. Weinstein, J. Dubousset, eds. *Pathogenesis of idiopathic scoliosis*, Springer Japan, 2018.

[13] W. S. Yu, K. Y. Chan, F. W. P. Yu et al., "Bone structural and mechanical indices in adolescent idiopathic scoliosis evaluated by high-resolution peripheral quantitative computed tomography (HR-pQCT)," *Bone*, vol. 61, pp. 109–115, 2014.

[14] C. S. K. Cheung, W. T. K. Lee, Y. K. Tse et al., "Generalized osteopenia in adolescent idiopathic scoliosis–association with abnormal pubertal growth, bone turnover, and calcium intake?," *Spine*, vol. 31, no. 3, pp. 330–338, 2006.

[15] K. T. Suh, S. S. Lee, S. H. Hwang, S. J. Kim, and J. S. Lee, "Elevated soluble receptor activator of nuclear factor-κB ligand and reduced bone mineral density in patients with adolescent idiopathic scoliosis," *European Spine Journal*, vol. 16, no. 10, pp. 1563–1569, 2007.

[16] K. Ishida, Y. Aota, N. Mitsugi et al., "Relationship between bone density and bone metabolism in adolescent idiopathic scoliosis," *Scoliosis*, vol. 10, no. 19, 2015.

[17] T. P. Lam, V. W. Hung, H. Y. Yeung et al, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound in Medicine & Biology*, vol. 39, no. 3, pp. 381–387, 2013.

[18] Q. Du, X. Zhou, J. A. Li et al, "Quantitative ultrasound measurements of bone quality in female adolescents with idiopathic scoliosis compared to normal controls," *Journal of Manipulative and Physiological Therapeutics*, vol. 38, no. 6, pp. 434–441, 2015.

[19] W. Chen, L. H. Le, and E. H. Lou, "Ultrasound imaging of spinal vertebrae to study scoliosis," *Open Journal of Acoustics*, vol. 2, no. 3, pp. 95–103, 2012.
[20] H. Li, L. H. Le, M. D. Sacchi, and E. H. Lou, "Ultrasound imaging of long bone fractures and healing with the split-step Fourier imaging method," *Ultrasound in Medicine & Biology*, vol. 39, no. 8, pp. 1482–1490, 2013.

[21] R. Zheng, L. H. Le, D. Hill, and E. Lou, "Estimation of bone quality on scoliotic subjects using ultrasound reflection imaging method-a preliminary study," In *2015 IEEE International Ultrasonics Symposium (IUS)*, IEEE, pp 1–4, 2015.

[22] M. B. Schaffer and D. B. Burr, "Stiffness of compact bone: effects of porosity and density," *Journal of Biomechanics*, vol. 21, no. 1, pp. 13–16, 1988.

[23] B. Helgason, E. Perilli, E. Schileo et al., "Mathematical relationships between bone density and mechanical properties: a literature review," *Clinical Biomechanics*, vol. 23, no. 2, pp. 135–146, 2008.

[24] C. T. Eneh, M. K. Malo, J. P. Karjalainen et al., "Effect of porosity, tissue density, and mechanical properties on radial sound speed in human cortical bone," *Medical Physics*, vol. 43, no. 5, pp. 2030–2039, 2016.

[25] J. C. Wall, S. K. Chatterji, and J. W. Jeffery, "Age-related changes in the density and tensile strength of human femoral cortical bone," *Calcified tissue International*, vol. 27, no. 1, pp. 105–108, 1979.

[26] L. J. Gibson and M. F. Ashby, *Cellular solids: structure and properties*. Cambridge university press, 1999.

[27] W. Abendschein and G. W. Hyatt, "33 Ultrasonics and selected physical properties of bone," *Clinical Orthopaedics and Related Research (1976-2007)*, vol. 69, pp. 294–301, 1970.

[28] J. D. Craven, M. A. Costantini, M. A. Greenfield, and R. Stern, "Measurement of the velocity of ultrasound in human cortical bone and its potential clinical importance: an in vivo preliminary study," *Investigative Radiology*, vol. 8, no. 2, pp. 72–77, 1973.

[29] Q. Ji, L. H. Le, L. J. Filipow, and S. A. Jackson, "Ultrasonic wave propagation in water-saturated aluminum foams," *Ultrasonics*, vol. 36, no. 6, pp. 759–765, 1998.

[30] W. R. Hendee and E. R. Ritenour, *Medical imaging physics*. John Wiley & Sons, 2003.

[31] J. T. Bushberg and J. M. Boone, *The essential physics of medical imaging*. Lippincott Williams & Wilkins, 2011.

[32] M. P. Andre, J. D. Craven, M. A. Greenfield, and R. Stern, "Measurement of the velocity of ultrasound in the human femur in vivo," *Medical Physics*, vol. 7, no. 4, pp. 324–330, 1980.

[33] S. Lees, J. M. Ahern, and M. Leonard, "Parameters influencing the sonic velocity in compact calcified tissues of various species," *The Journal of the Acoustical Society of America*, vol. 74, no. 1, pp. 28–33, 1983.

[34] L. H. Le, Y. J. Gu, Y. Li, and C. Zhang, "Probing long bones with ultrasonic body waves," *Applied Physics Letters*, vol. 96, no. 11, p. 114102, 2010.

[35] Z. Nazarchuk, V. Skalskyi, and O. Serhiyenko, *Acoustic emission: methodology and application*, Springer, 2017.

[36] J. Krautkramer and H. Krautkramer, *Ultrasonic testing of materials. 4th ed.* Springer Science & Business Media, 2013.

[37] D. Callens, C. Bruneel, and J. Assaad, "Matching ultrasonic transducer using two matching layers where one of them is glue," *NDT & E International*, vol. 37, no. 8, pp. 591–596, 2004.

[38] H. Persson and C. Hertz, "Acoustic impedance matching of medical ultrasound transducers," *Ultrasonics*, vol. 23, no. 2, pp. 83–89, 1985.

[39] S. Rhee, T. A. Ritter, K. K. Shung, H. Wang, and W. Cao, "Materials for acoustic matching in ultrasound transducers," In *2001 IEEE Ultrasonics Symposium. Proceedings. An International Symposium (Cat. No. 01CH37263)*, vol. 2, pp. 1051–1055, 2001.

[40] M. Khodaei, D. Hill, R. Zheng, L. H. Le, and E. H. Lou, "Intra-and inter-rater reliability of spinal flexibility measurements using ultrasonic (US) images for non-surgical

candidates with adolescent idiopathic scoliosis: a pilot study," *European Spine Journal*, vol. 27, no. 9, pp. 2156–2164, 2018.

[41] L. H. Le, "An investigation of pulse-timing techniques for broadband ultrasonic velocity determination in cancellous bone: a simulation study," *Physics in Medicine & Biology*, vol. 43, no. 8, p. 2295, 1998.

[42] C. Zhang, L. H. Le, R. Zheng, D. Ta, and E. Lou, "Measurements of ultrasonic phase velocities and attenuation of slow waves in cellular aluminum foams as cancellous bone mimicking phantoms," *The Journal of the Acoustical Society of America*, vol. 129, no. 5, pp. 3317–3326, 2011.

[43] T. D. Mast, "Empirical relationships between acoustic parameters in human soft tissues," *Acoustics Research Letters Online*, vol. 1, no. 2, pp. 37–42, 2000.

[44] E. L. Carstensen, "The mechanism of the absorption of ultrasound in biological materials," *IRE Transactions on Medical Electronics*, vol. 3, pp. 158–162, 1960.

[45] L. H. Le and L. J. Filipow, "Approximating the acoustic reflection coefficient of lossy biological tissues," *Physics in Medicine & Biology*, vol. 42, no. 4, p. 757, 1997.

[46] R. Lakes, H. S. Yoon, and J. L. Katz, "Ultrasonic wave propagation and attenuation in wet bone," *Journal of Biomedical Engineering*, vol. 8, no. 2, pp. 143–148, 1986.

Chapter 4 (Section 4.3)

- Kotwicki, T (2008) Evaluation of scoliosis today: examination, X-rays and beyond. Disability and rehabilitation, 30(10), 742-751.
- Negrini, S., Hresko, T. M., O'brien, J. P., Price, N., & SRS Non-Operative Committee (2015) Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee. Scoliosis, 10(1), 8.

- Levy, A. R., Goldberg, M. S., Mayo, N. E., Hanley, J. A., & Poitras, B (1996). Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis. Spine, 21(13), 1540-1547.
- Luo, T. D., Stans, A. A., Schueler, B. A., & Larson, A. N. (2015). Cumulative radiation exposure with EOS imaging compared with standard spine radiographs. Spine deformity, 3(2), 144-150.
- Simony, A., Hansen, E. J., Christensen, S. B., Carreon, L. Y., & Andersen, M. O. (2016). Incidence of cancer in adolescent idiopathic scoliosis patients treated 25 years previously. European Spine Journal, 25(10), 3366-3370.
- 6. Cheng, J. C., Guo, X., & Sher, A. H. (1999). Persistent osteopenia in adolescent idiopathic scoliosis: a longitudinal follow-up study. Spine, 24(12), 1218-1222.
- 7. Cheng, J. C., & Guo, X. (1997). Osteopenia in adolescent idiopathic scoliosis: a primary problem or secondary to the spinal deformity?. Spine, 22(15), 1716-1721.
- Deng, H. W., Xu, F. H., Davies, K. M., Heaney, R., & Recker, R. R. (2002). Differences in bone mineral density, bone mineral content, and bone areal size in fracturing and non-fracturing women, and their interrelationships at the spine and hip. Journal of bone and mineral metabolism, 20(6), 358-366.
- Glüer, C. C., Wu, C. Y., & Genant, H. K. (1993). Broadband ultrasound attenuation signals depend on trabecular orientation: an in vitro study. Osteoporosis International, 3(4), 185-191.
- 10. Laugier, P (2006) *Quantitative ultrasound of bone: looking ahead. Joint Bone Spine* 73(2):125-128.
- 11. Lam TP, Hung VWY, Yeung HY, et al. (2010) Quantitative ultrasound for detection of deranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS) - A prospective cohort study till skeletal maturity. Osteoporosis Int 21(SUPPL. 5):S686-S687.

- Lam, T. P., Hung, V. W. Y., Yeung, H. Y., et al. (2013). Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity. Ultrasound in medicine & biology, 39(3), 381-387.
- Langton, C. M., Njeh, C. F., Hodgskinson, R., & Currey, J. D. (1996). Prediction of mechanical properties of the human calcaneus by broadband ultrasonic attenuation. Bone,18(6), 495-503.
- 14. Lam, T. P., Hung, V. W. Y., Yeung, H. Y. (2011) Abnormal bone quality in adolescent idiopathic scoliosis: a case-control study on 635 subjects and 269 normal controls with bone densitometry and quantitative ultrasound. Spine, 36(15), 1211-1217.
- Mathieu V, Chappard C, Vayron R, Michel A, Haiat G (2013) Radial anatomic variation of ultrasonic velocity in human cortical bone. Ultrasound Med Bio 39(11):2185-2193. https://doi.org/10.1016/j.ultrasmedbio.2013.06.005.
- Mesquita AQ, Barbieri G, Barbien CH (2016) Correlation between ultrasound velocity and densitometry in fresh and demineralized cortical bone. Clinics (Sao Paulo) 71(11):657-663. https://doi.org/ 10.3031/clinics/2016(11)07.
- Matusin DP, Fontes-Pereira AJ, Barboza T, Augusto S, Antonio M, Pereira WC (2018) Exploring Cortical Bone Density through the Ultrasound Integrated Reflection Coefficient. Acta Orthopedica Brasileira 26(4) Sao Paulo, https://doi.org/10.1590/1413785220182604177202.
- Zheng, R., Le, L. H., Hill, D., & Lou, E. (2015) Estimation of bone quality on scoliotic subjects using ultrasound reflection imaging method-a preliminary study. In 2015 IEEE International Ultrasonics Symposium (IUS) (pp. 1-4). IEEE.
- Bushberg, J.T. and J.M. Boone (2011) The essential physics of medical imaging: Lippincott Williams & Wilkins.
- 20. Khodaei, M., Pacheco-Pereira, C., Trac, S., et al (2018) Radiographic methods to estimate surgical outcomes based on spinal flexibility assessment in patients who have

adolescent idiopathic scoliosis: a systematic review. The Spine Journal, 18(11), 2128-2139.

- 21. Koo, T. K., & Li, M. Y. (2016). A guideline of selecting and reporting intraclass correlation coefficients for reliability research. Journal of chiropractic medicine, 15(2), 155-163.
- Weir, J. P. (2005). Quantifying test-retest reliability using the intraclass correlation coefficient and the SEM. The Journal of Strength & Conditioning Research, 19(1), 231-240.
- 23. Wolff J. (2012) The law of bone remodelling. Springer Science & Business Media.
- 24. Frost HM. Wolff's Law and bone's structural adaptations to mechanical usage: an overview for clinicians. The Angle Orthodontist. 1994;64(3):175-188.

Chapter 5 (Section 5.2)

- J. R. Cobb, "Outline for the study of scoliosis," *Instr Course Lect AAOS*, vol. 5, pp. 261–275, 1948.
- B. P. Wills *et al.*, "Comparison of Cobb angle measurement of scoliosis radiographs with preselected end vertebrae: traditional versus digital acquisition," *Spine*, vol. 32, no. 1, pp. 98–105, 2007.
- J. A. Janicki and B. Alman, Scoliosis: Review of diagnosis and treatment," *Paediatr. Child Health*, vol. 12, no. 9, pp. 771–776, 2007.
- [4] S. Negrini, et al. "2016 SOSORT guidelines: orthopaedic and rehabilitation treatment of idiopathic scoliosis during growth," *Scoliosis* 13, 3 (2018). https://doi.org/10.1186/s13013-017-0145-8.
- [5] J. Legaye, G. Duval-Beaupère, J. Hecquet, and C. Marty, "Pelvic incidence: a fundamental pelvic parameter for three-dimensional regulation of spinal sagittal curves," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 7, no. 2, pp. 99–103, 1998, doi: 10.1007/s005860050038.

- [6] J.-M. Mac-Thiong, H. Labelle, M. Charlebois, M.-P. Huot, and J. A. de Guise, "Sagittal plane analysis of the spine and pelvis in adolescent idiopathic scoliosis according to the coronal curve type," *Spine*, vol. 28, no. 13, pp. 1404–1409, 2003.
- [7] S. L. Propst-Proctor and E. E. Bleck, "Radiographic determination of lordosis and kyphosis in normal and scoliotic children.," *J. Pediatr. Orthop.*, vol. 3, no. 3, pp. 344–346, 1983.
- [8] V. V. Upasani *et al.*, "Analysis of sagittal alignment in thoracic and thoracolumbar curves in adolescent idiopathic scoliosis: how do these two curve types differ?," *Spine*, vol. 32, no. 12, pp. 1355–1359, 2007.
- [9] Q. Yong *et al.*, "Comparison of sagittal spinopelvic alignment in Chinese adolescents with and without idiopathic thoracic scoliosis," *Spine*, vol. 37, no. 12, pp. E714–E720, 2012.
- [10] P. J. M. van Loon, B. A. G. Kühbauch, and F. B. Thunnissen, "Forced lordosis on the thoracolumbar junction can correct coronal plane deformity in adolescents with double major curve pattern idiopathic scoliosis," *Spine*, vol. 33, no. 7, pp. 797–801, Apr. 2008, doi: 10.1097/BRS.0b013e3181694ff5.
- [11] A. M. Briggs, T. V. Wrigley, E. A. Tully, P. E. Adams, A. M. Greig, and K. L. Bennell, "Radiographic measures of thoracic kyphosis in osteoporosis: Cobb and vertebral centroid angles," *Skeletal Radiol.*, vol. 36, no. 8, pp. 761–767, 2007.
- [12] D. Harrison, R. Cailliet, D. Harrison, T. Janik, and B. Holland, "Reliability of Centroid, Cobb, and Harrison Posterior Tangent Methods: Which to Choose for Analysis of Thoracic Kyphosis," *Spine*, vol. 26, pp. E227-34, Jun. 2001, doi: 10.1097/00007632-200106010-00002.
- [13] D. L. Carman, R. H. Browne, and J. G. Birch, "Measurement of scoliosis and kyphosis radiographs. Intraobserver and interobserver variation.," *J. Bone Joint Surg. Am.*, vol. 72, no. 3, pp. 328–333, 1990.

- [14] R. P. Jackson and A. C. McManus, "Radiographic analysis of sagittal plane alignment and balance in standing volunteers and patients with low back pain matched for age, sex, and size. A prospective controlled clinical study," *Spine*, vol. 19, no. 14, pp. 1611–1618, Jul. 1994, doi: 10.1097/00007632-199407001-00010.
- [15] B. Ilharreborde *et al.*, "Angle measurement reproducibility using EOSthreedimensional reconstructions in adolescent idiopathic scoliosis treated by posterior instrumentation," *Spine*, vol. 36, no. 20, pp. E1306–E1313, 2011.
- [16] S. Ohrt-Nissen *et al.*, "Reproducibility of thoracic kyphosis measurements in patients with adolescent idiopathic scoliosis," *Scoliosis Spinal Disord.*, vol. 12, no. 1, pp. 1–8, 2017.
- [17] A. Simony, E. J. Hansen, S. B. Christensen, L. Y. Carreon, and M. O. Andersen, "Incidence of cancer in adolescent idiopathic scoliosis patients treated 25 years previously," *Eur. Spine J.*, vol. 25, no. 10, pp. 3366–3370, 2016.
- [18] W. Chen, E. H. M. Lou, P. Q. Zhang, L. H. Le, and D. Hill, "Reliability of assessing the coronal curvature of children with scoliosis by using ultrasound images," *J. Child. Orthop.*, vol. 7, no. 6, pp. 521–529, Dec. 2013, doi: 10.1007/s11832-013-0539-y.
- [19] M. Young, D. L. Hill, R. Zheng, and E. Lou, "Reliability and accuracy of ultrasound measurements with and without the aid of previous radiographs in adolescent idiopathic scoliosis (AIS)," *Eur. Spine J. Off. Publ. Eur. Spine Soc. Eur. Spinal Deform. Soc. Eur. Sect. Cerv. Spine Res. Soc.*, vol. 24, no. 7, pp. 1427–1433, Jul. 2015, doi: 10.1007/s00586-015-3855-8.
- [20] T. K. Koo and M. Y. Li, "A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research," *J. Chiropr. Med.*, vol. 15, no. 2, pp. 155–163, Jun. 2016, doi: 10.1016/j.jcm.2016.02.012.

- [21] P. W. Stratford and C. H. Goldsmith, "Use of the standard error as a reliability index of interest: an applied example using elbow flexor strength data," *Phys. Ther.*, vol. 77, no. 7, pp. 745–750, 1997.
- [22] J. S. Lewis and R. E. Valentine, "Clinical measurement of the thoracic kyphosis. A study of the intra-rater reliability in subjects with and without shoulder pain," *BMC Musculoskelet. Disord.*, vol. 11, no. 1, p. 39, Mar. 2010, doi: 10.1186/1471-2474-11-39.
- [23] D. J. Kolessar, G. T. Stollsteimer, and R. R. Betz, "The value of the measurement from T5 to T12 as a screening tool in detecting abnormal kyphosis," *J. Spinal Disord.*, vol. 9, no. 3, pp. 220–222, Jun. 1996.

[24] F. Zaina, S. Donzelli, M. Lusini, and S. Negrini, "How to measure kyphosis in everyday clinical practice: a reliability study on different methods," *Stud. Health Technol. Inform.*, vol. 176, pp. 264–267, 2012.

Chapter 5 (Section 5.3)

- Negrini, S., Hresko, T. M., O'brien, J. P., et al. (2015) Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee. *Scoliosis*, 10(1), 8.
- Kotwicki, T (2008) Evaluation of scoliosis today: examination, X-rays and beyond. *Disability and rehabilitation*, 30(10), 742-751.
- Winter R, Lovell WW, Moe J. (1975). Excessive thoracic lordosis and loss of pulmonary function in patients with idiopathic scoliosis. *J Bone Joint Surg Am*. 57:972–977.
- Nault, M. L., Mac-Thiong, J. M., Roy-Beaudry, et al. (2013). Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis. Journal of Pediatric Orthopaedics, 33(6), 618-623.

- Nault, M. L., Mac-Thiong, J. M., Roy-Beaudry, M., et al. (2014). Threedimensional spinal morphology can differentiate between progressive and nonprogressive patients with adolescent idiopathic scoliosis at the initial presentation: a prospective study. *Spine*, 39(10), E601.
- D. L. Carman, R. H. Browne, and J. G. Birch (1990) "Measurement of scoliosis and kyphosis radiographs. Intraobserver and interobserver variation.," *J. Bone Joint Surg. Am.*, vol. 72, no. 3, pp. 328–333.
- R. P. Jackson, T. Kanemura, N. Kawakami, and C. Hales (2000) "Lumbopelvic lordosis and pelvic balance on repeated standing lateral radiographs of adult volunteers and untreated patients with constant low back pain," *Spine*, vol. 25, no. 5, pp. 575–586.
- K. P. Singer, S. J. Edmondston, R. E. Day, and W. H. Breidahl (1994) "Computerassisted curvature assessment and Cobb angle determination of the thoracic kyphosis. An in vivo and in vitro comparison," *Spine*, vol. 19, no. 12, pp. 1381– 1384, Jun. 1994, doi: 10.1097/00007632-199406000-00012.
- Ohrt-Nissen, S., Cheung, J. P. Y., Hallager, D. W.(2017) "Reproducibility of thoracic kyphosis measurements in patients with adolescent idiopathic scoliosis," *Scoliosis Spinal Disord.*, vol. 12, no. 1, pp. 1–8.
- D. J. Kolessar, G. T. Stollsteimer, and R. R. Betz (1996) "The value of the measurement from T5 to T12 as a screening tool in detecting abnormal kyphosis," *J. Spinal Disord.*, vol. 9, no. 3, pp. 220–222, Jun.
- Tran, T. H., Wing, D., Davis, A., Bergstrom, J., Schousboe, J. T., Nichols, J. F., et al. (2016). Correlations among four measures of thoracic kyphosis in older adults. *Osteoporosis International*, 27(3), 1255-1259.
- 12. T. D. Luo, A. A. Stans, B. A. Schueler, and A. N. Larson (2015) "Cumulative Radiation Exposure With EOS Imaging Compared With Standard Spine

Radiographs," *Spine Deform.*, vol. 3, no. 2, pp. 144–150, Mar. doi: 10.1016/j.jspd.2014.09.049.

- A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras (1996) "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine*, vol. 21, no. 13, pp. 1540–1547; discussion 1548, Jul. doi: 10.1097/00007632-199607010-00011.
- Chen, W., Lou, E. H., Zhang, P. Q., et al. (2013). Reliability of assessing the coronal curvature of children with scoliosis by using ultrasound images. *Journal of children's orthopaedics*, 7(6), 521-529.
- 15. Zheng, Y. P., Lee, T. T. Y., Lai, K. K. L., et al (2016). A reliability and validity study for Scolioscan: a radiation-free scoliosis assessment system using 3D ultrasound imaging. *Scoliosis and spinal disorders*, *11*(1), 1-15.
- 16. Chen, W., Le, L. H., & Lou, E. H. (2016). Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study. *European Spine Journal*, 25(10), 3265-3273.
- Koo, T. K., & Li, M. Y. (2016). A guideline of selecting and reporting intraclass correlation coefficients for reliability research. Journal of chiropractic medicine, 15(2), 155-163.
- Ilharreborde B, Steffen JS, Nectoux E, et al. (2011) Angle measurement reproducibility using EOS three-dimensional reconstructions in adolescent idiopathic scoliosis treated by posterior instrumentation. *Spine* 36(20):E1306-1313. https://doi.org/10.1097/BRS.0b013e3182293548.
- Melhem, E., Assi, A., El Rachkidi, R., & Ghanem, I. (2016). EOS® biplanar X-ray imaging: concept, developments, benefits, and limitations. *Journal of children's orthopaedics*, 10(1), 1-14.

- 20. Kaneko, K., Aota, Y., Sekiya, T., Yamada, K., & Saito, T. (2016). Validation study of arm positions for evaluation of global spinal balance in EOS imaging. *European Journal of Orthopaedic Surgery & Traumatology*, 26(7), 725-733.
- 21. Horton, W. C., Brown, C. W., Bridwell, K. H., Glassman, S. D., Suk, S. I., & Cha,
 C. W. (2005). Is there an optimal patient stance for obtaining a lateral 36" radiograph?: a critical comparison of three techniques. *Spine*, *30*(4), 427-433.

Chapter 6

- M. N. Choudhry, Z. Ahmad, and R. Verma, "Adolescent idiopathic scoliosis," *The open orthopaedics journal*, vol. 10, p. 143, 2016.
- [2] J. A. Janicki and B. Alman, "Scoliosis: Review of diagnosis and treatment," *Paediatrics & child health*, vol. 12, no. 9, pp. 771–776, 2007.
- S. L. Weinstein, "The natural history of adolescent idiopathic scoliosis," *Journal of Pediatric Orthopaedics*, vol. 39, pp. S44–S46, 2019.
- [4] T. Vrtovec, F. Pernuš, and B. Likar, "A review of methods for quantitative evaluation of axial vertebral rotation," *European Spine Journal*, vol. 18, no. 8, pp. 1079–1090, 2009.
- [5] I. A. Stokes, "Axial rotation component of thoracic scoliosis," *Journal of orthopaedic research*, vol. 7, no. 5, pp. 702–708, 1989.
- [6] M.-L. Nault *et al.*, "Three-Dimensional Spinal Morphology Can Differentiate Between Progressive and Nonprogressive Patients With Adolescent Idiopathic Scoliosis at the Initial Presentation," *Spine (Phila Pa 1976)*, vol. 39, no. 10, pp. E601–E606, May 2014, doi: 10.1097/BRS.00000000000284.
- [7] J. R. Cobb, "Outline for the study of scoliosis," *Instr Course Lect AAOS*, vol. 5, pp. 261–275, 1948.

- [8] R. Perdriolle and J. Vidal, "Thoracic idiopathic scoliosis curve evolution and prognosis," *Spine (Phila Pa 1976)*, vol. 10, no. 9, pp. 785–791, Nov. 1985, doi: 10.1097/00007632-198511000-00001.
- [9] I. A. Stokes, L. C. Bigalow, and M. S. Moreland, "Measurement of axial rotation of vertebrae in scoliosis.," *Spine*, vol. 11, no. 3, pp. 213–218, 1986.
- [10] C. L. Nash and J. H. Moe, "A study of vertebral rotation," J Bone Joint Surg Am, vol. 51, no. 2, pp. 223–229, 1969.
- [11] E. K. Ho, S. S. Upadhyay, F. L. Chan, L. C. Hsu, and J. C. Leong, "New methods of measuring vertebral rotation from computed tomographic scans. An intraobserver and interobserver study on girls with scoliosis," *Spine (Phila Pa 1976)*, vol. 18, no. 9, pp. 1173–1177, Jul. 1993, doi: 10.1097/00007632-199307000-00008.
- [12] D. Birchall, D. G. Hughes, J. Hindle, L. Robinson, and J. B. Williamson, "Measurement of vertebral rotation in adolescent idiopathic scoliosis using threedimensional magnetic resonance imaging," *Spine (Phila Pa 1976)*, vol. 22, no. 20, pp. 2403–2407, Oct. 1997, doi: 10.1097/00007632-199710150-00016.
- [13] L. Boyer, J. Shen, S. Parent, S. Kadoury, and C.-E. Aubin, "Accuracy and precision of seven radiography-based measurement methods of vertebral axial rotation in adolescent idiopathic scoliosis," *Spine deformity*, vol. 6, no. 4, pp. 351–357, 2018.
- [14] D. Birchall, D. Hughes, B. Gregson, and B. Williamson, "Demonstration of vertebral and disc mechanical torsion in adolescent idiopathic scoliosis using threedimensional MR imaging," *European spine journal : official publication of the European Spine Society, the European Spinal Deformity Society, and the European Section of the Cervical Spine Research Society*, vol. 14, pp. 123–9, Apr. 2005, doi: 10.1007/s00586-004-0705-5.
- [15] T. Illés and S. Somoskeöy, "Comparison of scoliosis measurements based on threedimensional vertebra vectors and conventional two-dimensional measurements:

advantages in evaluation of prognosis and surgical results," *European spine journal*, vol. 22, no. 6, pp. 1255–1263, 2013.

- [16] E. Melhem, A. Assi, R. El Rachkidi, and I. Ghanem, "EOS® biplanar X-ray imaging: concept, developments, benefits, and limitations," *Journal of children's orthopaedics*, vol. 10, no. 1, pp. 1–14, 2016.
- [17] E. Amzallag-Bellenger *et al.*, "Idiopathic scoliosis in children and adolescents: assessment with a biplanar X-ray device," *Insights into imaging*, vol. 5, no. 5, pp. 571–583, 2014.
- S. Suzuki, T. Yamamuro, J. Shikata, K. Shimizu, and H. Iida, "Ultrasound measurement of vertebral rotation in idiopathic scoliosis," *J Bone Joint Surg Br*, vol. 71, no. 2, pp. 252–255, Mar. 1989, doi: 10.1302/0301-620X.71B2.2647754.
- [19] R. Burwell *et al.*, "Spine-rib rotation differences at the apex in preoperative patients with adolescent idiopathic scoliosis (AIS): Evaluation of a three-level ultrasound method," *Studies in health technology and informatics*, vol. 91, pp. 246–50, Feb. 2002, doi: 10.3233/978-1-60750-935-6-246.
- [20] Q. N. Vo, E. H. M. Lou, and L. H. Le, "Measurement of vertebral rotation using a three-dimensional ultrasound image," *Scoliosis*, vol. 10, no. 1, p. O40, Jan. 2015, doi: 10.1186/1748-7161-10-S1-O40.
- [21] W. Chen, L. H. Le, and E. H. M. Lou, "Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study," *Eur Spine J*, vol. 25, no. 10, pp. 3265–3273, Oct. 2016, doi: 10.1007/s00586-016-4492-6.
- [22] Q. Wang *et al.*, "Validity Study of Vertebral Rotation Measurement Using 3-D Ultrasound in Adolescent Idiopathic Scoliosis," *Ultrasound Med Biol*, vol. 42, no. 7, pp. 1473–1481, Jul. 2016, doi: 10.1016/j.ultrasmedbio.2016.02.011.

- [23] J. Zhang et al., "Computer-aided assessment of scoliosis on posteroanterior radiographs," Med Biol Eng Comput, vol. 48, no. 2, pp. 185–195, Feb. 2010, doi: 10.1007/s11517-009-0556-7.
- [24] T. K. Koo and M. Y. Li, "A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research," *J Chiropr Med*, vol. 15, no. 2, pp. 155–163, Jun. 2016, doi: 10.1016/j.jcm.2016.02.012.
- [25] R. Zheng *et al.*, "Factors influencing spinal curvature measurements on ultrasound images for children with adolescent idiopathic scoliosis (AIS)," *PLoS One*, vol. 13, no. 6, p. e0198792, Jun. 2018, doi: 10.1371/journal.pone.0198792.

Chapter 7

- T. Kotwicki, "Evaluation of scoliosis today: examination, X-rays and beyond," Disability and rehabilitation, vol. 30, no. 10, pp. 742–751, 2008.
- [2] C. M. Ronckers, C. E. Land, J. S. Miller, M. Stovall, J. E. Lonstein, and M. M. Doody, "Cancer mortality among women frequently exposed to radiographic examinations for spinal disorders," *Radiation research*, vol. 174, no. 1, pp. 83–90, 2010.
- [3] P. N. Soucacos, K. Zacharis, J. Gelalis, *et al.*, "Assessment of curve progression in idiopathic scoliosis," *Eur Spine J*, vol. 7, no. 4, pp. 270–277, 1998, doi: 10.1007/s005860050074.
- [4] L. H. Thompson, "Recognition, signaling, and repair of DNA double-strand breaks produced by ionizing radiation in mammalian cells: the molecular choreography," *Mutation Research/Reviews in Mutation Research*, vol. 751, no. 2, pp. 158–246, 2012.
- [5] A. R. Levy, M. S. Goldberg, N. E. Mayo, J. A. Hanley, and B. Poitras, "Reducing the lifetime risk of cancer from spinal radiographs among people with adolescent idiopathic scoliosis," *Spine (Phila Pa 1976)*, vol. 21, no. 13, pp. 1540–1547; discussion 1548, Jul. 1996, doi: 10.1097/00007632-199607010-00011.

- [6] A. R. Levy, M. S. Goldberg, J. A. Hanley, N. E. Mayo, and B. Poitras, "Projecting the lifetime risk of cancer from exposure to diagnostic ionizing radiation for adolescent idiopathic scoliosis.," *Health physics*, vol. 66, no. 6, pp. 621–633, 1994.
- [7] R. Zheng, A.C. Chan, W. Chen, *et al.*, "Intra- and Inter-rater Reliability of Coronal Curvature Measurement for Adolescent Idiopathic Scoliosis Using Ultrasonic Imaging Method-A Pilot Study," *Spine Deform*, vol. 3, no. 2, pp. 151–158, Mar. 2015, doi: 10.1016/j.jspd.2014.08.008.
- [8] Y.-P. Zheng, T.T.Y. Lee, K.K. L. Lai, *et al.*, "A reliability and validity study for Scolioscan: a radiation-free scoliosis assessment system using 3D ultrasound imaging," *Scoliosis and Spinal Disorders*, vol. 11, no. 1, p. 13, May 2016, doi: 10.1186/s13013-016-0074-y.
- [9] W. Chen, L. H. Le, and E. H. M. Lou, "Reliability of the axial vertebral rotation measurements of adolescent idiopathic scoliosis using the center of lamina method on ultrasound images: in vitro and in vivo study," *Eur Spine J*, vol. 25, no. 10, pp. 3265–3273, Oct. 2016, doi: 10.1007/s00586-016-4492-6.
- [10] S. Trac, R. Zheng, D. L. Hill, and E. Lou, "Intra- and Interrater Reliability of Cobb Angle Measurements on the Plane of Maximum Curvature Using Ultrasound Imaging Method," *Spine Deform*, vol. 7, no. 1, pp. 18–26, Jan. 2019, doi: 10.1016/j.jspd.2018.06.015.
- [11] Sayed T, Khodaei M, Lou E, "Intra- and inter-rater reliabilities and accuracy of kyphotic angle measurements on ultrasound images for children with adolescent idiopathic scoliosis - a pilot study," presented at the WCHRI Research Day 2019.
- [12] M. Khodaei, T. Sayed, D. Hill, *et al.*, "Reliability of measurements of a reflection coefficient index to indicate spinal bone strength on adolescents with idiopathic scoliosis (AIS): a pilot study," *European Spine Journal*, pp. 1–8, 2021.

- [13] V. W. Y. Hung, L. Qin, C. S. K. Cheung, *et al.*, "Osteopenia: a new prognostic factor of curve progression in adolescent idiopathic scoliosis," *J Bone Joint Surg Am*, vol. 87, no. 12, pp. 2709–2716, Dec. 2005, doi: 10.2106/JBJS.D.02782.
- [14] X. Sun, B. Wang, Y. Qiu, *et al.*, "Outcomes and predictors of brace treatment for girls with adolescent idiopathic scoliosis," *Orthopaedic surgery*, vol. 2, no. 4, pp. 285–290, 2010.
- [15] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for detection of dranged bone quality and bone density and prognostication of curve progression in adolescent idiopathic scoliosis (AIS)-a prospective cohort study till skeletal maturity," *Osteoprosis International*, 2010, vol. 21, pp. S686-S687.
- [16] K.-J. Tan, M. M. Moe, R. Vaithinathan, and H.-K. Wong, "Curve progression in idiopathic scoliosis: follow-up study to skeletal maturity," *Spine*, vol. 34, no. 7, pp. 697–700, 2009.
- [17] S. S. Upadhyay, I. W. Nelson, E. K. Ho, L. C. Hsu, and J. C. Leong, "New prognostic factors to predict the final outcome of brace treatment in adolescent idiopathic scoliosis," *Spine (Phila Pa 1976)*, vol. 20, no. 5, pp. 537–545, Mar. 1995, doi: 10.1097/00007632-199503010-00006.
- [18] A. Noshchenko, L. Hoffecker, E. M. Lindley, *et al.*, "Predictors of spine deformity progression in adolescent idiopathic scoliosis: A systematic review with metaanalysis," *World journal of orthopedics*, vol. 6, no. 7, p. 537, 2015.
- [19] T. P. Lam, V. W. Y. Hung, H. Y. Yeung, *et al.*, "Quantitative ultrasound for predicting curve progression in adolescent idiopathic scoliosis: a prospective cohort study of 294 cases followed-up beyond skeletal maturity," *Ultrasound Med Biol*, vol. 39, no. 3, pp. 381–387, Mar. 2013, doi: 10.1016/j.ultrasmedbio.2012.09.012.
- [20] M.-L. Nault, J.-M. Mac-Thiong, M. Roy-Beaudry, J. deGuise, H. Labelle, and S. Parent, "Three-dimensional spine parameters can differentiate between progressive and nonprogressive patients with AIS at the initial visit: a retrospective analysis," J

Pediatr Orthop, vol. 33, no. 6, pp. 618–623, Sep. 2013, doi: 10.1097/BPO.0b013e318292462a.

- [21] S. Negrini, T. M. Hresko, J. P. O'Brien, N. Price, SOSORT Boards, and SRS Non-Operative Committee, "Recommendations for research studies on treatment of idiopathic scoliosis: Consensus 2014 between SOSORT and SRS non-operative management committee," *Scoliosis*, vol. 10, p. 8, 2015, doi: 10.1186/s13013-014-0025-4.
- [22] M. Khodaei, D. Hill, R. Zheng, L. H. Le, and E. H. M. Lou, "Intra- and inter-rater reliability of spinal flexibility measurements using ultrasonic (US) images for nonsurgical candidates with adolescent idiopathic scoliosis: a pilot study," *Eur Spine J*, vol. 27, no. 9, pp. 2156–2164, Sep. 2018, doi: 10.1007/s00586-018-5546-8.
- [23] A. J. Danielsson, R. Hasserius, A. Ohlin, and A. L. Nachemson, "A prospective study of brace treatment versus observation alone in adolescent idiopathic scoliosis: a follow-up mean of 16 years after maturity," *Spine (Phila Pa 1976)*, vol. 32, no. 20, pp. 2198–2207, Sep. 2007, doi: 10.1097/BRS.0b013e31814b851f.
- [24] X. Sun, T. Wu, Z. Liu, *et al.*, "Osteopenia predicts curve progression of adolescent idiopathic scoliosis in girls treated with brace treatment," *J Pediatr Orthop*, vol. 33, no. 4, pp. 366–371, Jun. 2013, doi: 10.1097/BPO.0b013e31827b7b5f.
- [25] L. E. Peterson and A. L. Nachemson, "Prediction of progression of the curve in girls who have adolescent idiopathic scoliosis of moderate severity. Logistic regression analysis based on data from The Brace Study of the Scoliosis Research Society," J Bone Joint Surg Am, vol. 77, no. 6, pp. 823–827, Jun. 1995, doi: 10.2106/00004623-199506000-00002.